CENTER FOR DRUG EVALUATION AND RESEARCH

APPROVAL PACKAGE FOR:

APPLICATION NUMBER 20-825

Medical Review(s)

REVIEW AND EVALUATION OF CLINICAL DATA

NDA 20-825 Sponsor: Pfizer Drug: (ziprasidone) Material Submitted: Response to Nonapprovable Letter (3/10/00), Responses to Request for Information (4/6/00, 5/19/00, 5/22/00), Safety Update #3 (6/2/00)

I. Background

Summary

Ziprasidone is a new antipsychotic which has serotonin (5-HT_{2A}) and dopamine (D₂) antagonist properties and received a non-approvable on 6/17/98 for the original NDA submission. The nonapprovable letter delineated concerns regarding ziprasidone's safety data which demonstrated a dose dependent QTc prolongation observed in the short-term placebo controlled studies, and that this represented a risk of "potentially fatal ventricular arrhythmias that is not outweighed by a demonstrated and sufficient advantage of ziprasidone over already marketed drug products." In response to this nonapprovable letter, the sponsor designed and conducted Study 054 which would characterize the QTc effects of currently marketed neuroleptic agents (risperidone, olanzapine, haloperidol, thioridazine, and quetiapine) in an effort to compare ziprasidone's ability to prolong the QTc with these other neuroleptic agents.

The sponsor's response to the nonapprovable letter was submitted on March 10, 2000 and included the study report of Study 054; the submission included a small amount of summary safety information up to February 5, 1999. In subsequent submissions, at FDA's request, the sponsor submitted narratives of all serious adverse events, cases of overdose, updated mortality tables and selected discontinuations occurring from May 15, 1997 to February 5, 2000, and all deaths in the entire NDA data base. The cut-off date for the safety_data base reviewed in this document is February 5, 2000.

Administrative History

The original consultations obtained by FDA Cardiorenal Division written by Charles Ganley, M.D. (1/13/98 &11/24/97) stated that "if ziprasidone does prolong the QT interval, as the short term, fixed-dose, placebo controlled trials suggest, then some patients will be at risk for the development of torsades de pointes based on the experiences with other QT prolonging drugs [1/13/98]." Dr. Ganley recommended that, "unless efficacy data suggests superior benefit over currently available drugs, ziprasidone should be considered for second line therapy with adequate warnings of risk associated with drugs that prolong the QT interval [11/24/97]."

In his memo of May 14, 1998, Thomas Laughren, M.D., Team Leader for Psychiatric Drug Products, Division of Neuropharmacological Drug Products expressed concern for the approval of ziprasidone at the recommended doses because of the issues of QTc prolongation, and recommended approval of ziprasidone at a lower dosing range (20-40 mg bid). He also recommended that ziprasidone "be made available only as a second line drug with very strong labeling." Dr. Laughren recommended that labeling include a contraindicated use with other drugs that prolong the QTc interval, and "a bolded and boxed warning regarding QTc prolongation...and... [recommendations] for screening and monitoring (ECGs; serum potassium; and Holter monitoring for symptomatic patients). "

In a memo of June 1, 1998, Paul Leber, M.D., then Director of the Division of Neuropharmacological Drug Products, stated that he may have considered recommending

approval if ziprasidone had demonstrated a "unique benefit or advantage not provided by already marketed antipsychotics...however, the only evidence...that we have makes a point to state that ziprasidone's comparative performance (Study 115) supports a conclusion that it is less efficacious than haloperidol, a long marketed antipsychotic drug." Dr. Leber recommended a nonapprovable action.

The nonapprovable letter of June 17, 1998 asserted that a sufficient advantage over currently marketed antipsychotics had not been demonstrated that could outweigh the risk of potentially fatal arrhythmias because of the demonstrated QTc prolongation. The letter went onto say that "… we would find QTc prolongation at maximum blood levels in the 5-10 msec range, with adequate assurances that there are very few outliers and that there are no factors that lead to substantially greater values in individuals (such as drug-drug interactions), sufficiently reassuring…to support approval of a new antipsychotic such as ziprasidone."

Advisory Committee Meeting

A Psychopharmacological Drug Advisory Committee meeting was held on July 19, 2000 to discuss the NDA 20-825 for Zeldox (ziprasidone hydrochloride capsules) sponsored by Pfizer. The following summary is based on the Flash minutes distributed July 20, 2000 (full transcript to be available after August 20,2000 at < <u>http://www.fda.gov/ohrms/dockets/ac/acmenu.htm</u> >).The majority of the committee voted for the approval of ziprasidone based on the safety information presented. There were concerns expressed regarding the lack of dose escalation data; it is unknown what would occur when ziprasidone might be used as a concomitant medications in which the dose levels increased significantly, and whether or not there could be an additive effect of QTc when administered with another drug which prolonged the QTc. There was consensus that there needed to be strong warnings regarding the potential for fatal arrhythmias, but no clear consensus of how that should be communicated to practitioners.

II. Updated Clinical Data

The original NDA clinical review discussed the sponsor's primary integrated safety data base which included 2588 patients participating in Phase II/III studies as of a cut off of May 15, 1997. This original safety data base incorporated data from both the original NDA submission (cut-off 10/31/96) and a four month safety update (cut-off 5/15/97).

Updated Primary Integrated Database

This review will cover the sponsor's primary integrated safety data base with a new cut-off date of February 5, 2000. It is also noted that the sponsor has included trials conducted in Japan in this new integrated safety data base; whereas in the original NDA submission, Japanese studies were not included in the integrated safety data base.

The following table shows the number of patients and patient-years exposure of the primary integrated data base for Phase II/III oral ziprasidone trials as of the cut-off date of February 5, 2000:

TREATMENT GROUP	N	Patient-Years
Ziprasidone (cut-off: 2/2000)	4571	1732.6
Placebo	605	91.8
Haloperidol	1071	298.6
Risperidone	426	196.4

Since the original NDA review cut-off date of May 15, 1997 to the most recent cut-off of February 5, 2000, ziprasidone has had an additional 961 patient years exposure; the additional patient years exposure for the other treatment groups are as follows: 1) placebo: 39, 2) haloperidol:167, and 3) risperidone:91.

Of the 4571 patients exposed to oral ziprasidone, 603 patients participated in the oral portion/extension to an IM study, 991 patients received ziprasidone for longer than 6 months, and 605 patients were exposed to ziprasidone for at least one year.

<u>Deaths</u>

As of the sponsor's cut off date of February 5, 2000, there have been a total of 50 deaths. Please refer to Appendix I for a listing of all deaths known to occur in patients exposed to ziprasidone. Since the original NDA review of 4/30/98 (cut-off of 5/15/97), there have been an additional eleven deaths which occurred within 30 days or less of the patients' discontinuation of ziprasidone, making a total of twenty-eight deaths occurring within 30 days.

The following table updates the mortality rates for patients in the Phase II/III trials of the integrated safety data base who have died during the study or within thirty days of discontinuing treatment with ziprasidone; as in the previous review, the placebo group appears to demonstrate the highest mortality.

DRUGS	Number of Patients ¹	Patient-years exposure ¹	Total # deaths	# deaths ≤ 30 days	Crude mortality rate ²	Mortality per 100 patient- years ²
Ziprasidone (cut-off: 2/2000)	4571	1732.6	50	28	0.006	1.62
Placebo	605	91.8	10	5	0.008	5.45
Haloperidol	1071	298.6	3	3	0.003	1.00
Risperidone	426	196.4	2	1	0.002	0.51

Mortality rate for Phase II/III clinical programs in ziprasidone NDA 20-825

Includes integrated safety data base (which has been redefined to include Japanese studies), Study 105 (IM: ziprasidone n=11; placebo n=12) and Study 120 (dementia: ziprasidone n=12)

²Based on # of deaths \leq 30 days

The tables below updates the sudden unexpected death rate. With the increased patient years exposure since the last NDA, it appears that there have been a decrease from 9.1 SUD per 1000 (n=772) to 5.8 SUD per 1000 for the ziprasidone treatment groups.

DRUGS	Number of Subjects ¹	Subject- years exposure	# Sudden Deaths	SUD per 1000 subject years
Ziprasidone	4571	1732.6	11"	6.3
Placebo	605	91.8	0	0
Haloperidol	1071	298.6	0	0
Risperidone	426	196.4	1	5.1

Rate of Sudden Unexpected Death* (SUD) in Ziprasidone NDA 20-825

*Sudden Unexpected Death (SUD) refers to subjects found dead or who died within 24 hours of symptoms Refer to Appendix 1 for listing of deaths considered to be SUD.

¹Does not include subject 115-6940394; please refer to the text of Section 8.1.1 of NDA review of 4/98. ¹Includes integrated safety data base, Study 105 (IM: ziprasidone n=11; placebo n=12) and Study 120 (dementia: ziprasidone n=12)

However, when updating the SUD comparison rate of ziprasidone and the most recently submitted antipsychotic NDAs (table below), it appears that both ziprasidone and sertindole continue to surpass the SUD rate of olanzapine, risperidone, and quetiapine:

DRUGS	Subject- years exposure	# Sudden Deaths	SUD per 1000 subject years
Ziprasidone (cut-off: 2/2000)	1733	11	6.3
Ziprasidone (orig. cut-off 6/98)	772	7	9.1
Sertindole	476	5	10.5
Olanzapine	1122.2	4	3.5
Risperidone	508	1•	1.9*
Quetiapine	865.3	1	1.1

Rate of SUD in most recently submitted antipsychotic NDA data bases *

*Sources are the current NDA 20-852 and Review of Clinical Data: General Characteristics of the Deaths in the NDAs for Olanzapine, Risperidone, Quetiapine and Sertindole by Greg Burkhart, M.D. (HFD-120: 3/3/98) *Correction from NDA Review of 4/30/98 to exclude a SUD during drowning episode.

There may be limitations to interpreting a comparison of the sudden deaths across data bases. The system of assigning sudden deaths was not tested for consistency with blinded readers, and the definition of sudden death may not have been consistently applied for each NDA data base, not to mention the inherent difficulty in being able to accurately assign the true cause of death in many cases.

Overdose Experience

The sponsor did not define what they considered to be an overdose. The table below includes cases in which the overdose was thought to include ziprasidone. Please see Appendix II for cases in which the patient was currently being treated with ziprasidone, but the overdose was believed to be with a different medication.

Summary of overdoses with ziprasidone

Patient #	Age/ Sex	Overdose Mg	Concomitant medications	Comments
116B5290019	49/M	240 mg	Valproic acid	Intentional O/D. Hospitalized because of command hallucinations. Unclear what associated effects of O/D were. CT of head for chronic headaches and dizziness was WNL.
127E5810001	43/F	640 mg		Patient reported O/D. Hospitalized for moderate sedation.
301E1950995	21/F	24 day supply dose unclear		Admitted to hospital after repeated vomiting. Received gastric lavage.
NY970020018	50/M	3240 mg		Initially reported as coma due to O/D; sponsor revised report to state that patient was drowsy and minimally sedated. QTc readings (central reading): Baseline QTc=454; Day of event: maxQTc=478
116B5950006	64/F	360 mg		Narrative not located. Table states that patient accidentally took ziprasidone. No QTc prolongation evident one day after incident.
116B5950014	26/M	Day 1: 400 mg Day 2: 480 mg		Narrative not located. Sponsor's table states that patient accidentally took ziprasidone with no QTc prolongation one day after incident.
NY970020105	29/F	Unknown		Patient reported to take O/D of ziprasidone (amt. unknown). Taken to hospital and released that day. No details submitted.

It appears that nausea, vomiting and sedation were the most prominent effects of overdose. There were no apparent sequela from overdoses with ziprasidone. There were also an additional 6 patients who were in blinded treatment listed as taking overdoses; the narratives did not report any remarkable events or treatments.

There were no deaths reported associated with an overdose of ziprasidone.

It is difficult to make definitive safety conclusions from the above overdose data. In most cases the serum level was not provided, and it was unclear if any of these overdoses were witnessed. At best, the above data speaks to concerns about an single elevated dose rather than allowing for conclusions to a longer exposure to high levels of ziprasidone.

Adverse Events

Since the original NDA review, additional data covered in this review has been collected from open label studies and three placebo controlled studies (Study 307:a 52 week flexible dose placebo controlled trial and Studies 601 & 602: placebo controlled studies being conducted in patients with mania). Data from these studies has been included in the serious adverse events and the discontinuation, but because these three placebo controlled studies are not yet completed and analyzed, there is currently no new information to contribute to the common adverse events profile. The remainder of this section will focus on significant adverse events, serious adverse events, and some adverse events which lead to premature discontinuations.

Upon request, the sponsor provided updates for the adverse events of syncope, rash and seizure as follows: 1) updated incidence rate for episodes of rash is 4.5% or 173 of 3834 patients, 2) for syncope, the rate is 0.57% (22/3824), and 3) for seizure, the rate is 0.39% or 15 of 3834 patients treated with ziprasidone.

Please refer to Appendix III which is a modification of the sponsor's table of all serious adverse events as submitted on May 22, 2000 covering the period of May 15, 1997 to February 5, 2000. All narratives submitted for each patient were read, and, for the most part, the sponsor's listing of the event accurately reflected the narrative.

Most of the serious adverse events observed in this report period were also observed in the original NDA submission, and not unexpected. Of note was a 25 y.o. patient (Subject #128-601E-189-0077 or 128-601E-0540-0077) who had an episode of neutropenia with a WBC =2.3 (NL range: 4.1-12.3) in the first month of treatment with ziprasidone; this neutropenia resolved after hospitalization (treatment unclear), and the patient was reported to have continued taking ziprasidone throughout and after this episode.

The following patient with a serious adverse event deserves mention:

Subject #302E-057-0456: 46 y.o. male with schizophrenia experienced weakness and chest pain after eight months of ziprasidone treatment. After being treated with isosorbide dinitrate in the hospital, he experienced a syncopal event with bradycardia (26 bpm). The centrally read ECGs three days prior to and one week after the event did not demonstrate any QTc prolongation; however, there were no ECGs submitted during the episodes of chest pain or bradycardia/syncope, and none located in the case report form requested from the sponsor (submitted 7/14/00).

Discontinuations Due to Adverse Events

A review of the discontinuations revealed one notable case of an event not previously observed in the original NDA data:

Subject 601-0520-0027: 38 y.o. male experienced **priapism** after twelve days of ziprasidone treatment. The event resolved on the same day that the study drug was discontinued suggesting a temporal relationship of this event to treatment with ziprasidone.

Other reasons for discontinuations during this safety period were also observed in the original NDA submission and not unexpected (please see Appendix IV for a list reasons for discontinuations for this review period).

QTc Outliers Reported by the Sponsor

١.

In the reporting period covered by this review (5/15/97 to 2/5/00), the sponsor reported the following additional patients who had a prolonged QTc, but the readings of whether there was a QTc \geq 500 msec depended on which central reader interpreted the ECG. The sponsor employed two central ECG readers:

Subject 121-590-0362: 48 y.o female with hypertension, had a QTc of 504 msec on Day 4, the first day of oral dosing after treatment with IM ziprasidone: QTc readings are in the following sponsor summary table:

Study Day	<u>Date</u>	Dose (Time)	ECG Time	<u>QTc (GDXI)</u>	QTc (PRW)
Baseline	20 January 1997			420 msec	424 msec
Day 3	23 January 1997	5 mg IM (15:30)		•-	
Day 4	24 January 1997	20 mg PO (9:00)	9:38	504 msec	442 msec
Day 7	27 January 1997	20 mg PO (9:00)	10:31	462 msec	438 msec

Subject 126-701-0079 /127E-701-0003

53 y.o. male with baseline ECG reading of left bundle branch block had the following ECG readings while being treated with ziprasidone:

<u>Study Day</u> Baseline	<u>Date</u> 25 June 1997	Dose (Time)	ECG Time	<u>QTc (GDXI)</u> 426 msec	<u>QTc (PRW)</u> 520 msec
Day 2	26 June 1997	20 mg IM (3:00)	12:53	423 msec	522 msec
Day 2	26 June 1997	80 mg PO (?)	••	••	
Day 6	30 June 1997	120 mg PO (?)	••		••
Day 7	1 July 1997	40 mg PO (?)	8:45	490 msec	468 msec

 Subject 97-R-585-3027-3227: 37 y.o. female diagnosed with schizophrenia with an ECG showing a QTc=513 (later re-read by PRW as a QTc=434) after 3 months of treatment with ziprasidone. She continued treatment with ziprasidone 80 mg bid, started divalproex sodium as a mood stabilizer, and one month later reported experiencing two syncopal events while a new ECG showed nonspecific T wave changes and a QTc=387 msec.

As with the case above, there were numerous ECGs in the NDA data base, which, when locally read had a QTc \ge 500 msec, but when re-read centrally (presumably blinded), the QTc intervals did not exceed 500. The sponsor listed the following as previously reported cases:

Subject 117-648-0167: 39 y.o. male discontinued treatment for a QTc of 503 msec on day 7 of ziprasidone treatment (80 mg/day). Baseline QTc=466 msec.

Subject 301-311-0977 28 y.o. female diagnosed with schizophrenia whose death occurred two days after discontinuing ziprasidone. Upon discharge from the study, ECG changes were consistent with subendocardial ischemia with substernal pinching sensation. ECG after last AM dose of 60 mg ziprasidone showed Qtc=391 msec. After receiving 200 mg thioridazine that afternoon two ECGs showed QTc= 518 &593 msec (timing unclear). Patient died the next day with cause of death reported to be myocarditis.

III. Pfizer's Response to the Nonapprovable Letter.

Study 054

As a response to the nonapprovable letter of June 17, 1998, the sponsor conducted Study 054, an open label, six arm study designed to assess the effects of ziprasidone on the QTc interval compared to currently marketed antipsychotics (risperidone, olanzapine, haloperidol, thioridazine, and quetiapine) at the maximum recommended dosage. This design allowed for assessment of ECGs at the time of maximum concentration (tmax) for each antipsychotic in the absence and presence of an appropriately chosen CYP450 inhibitor. The study was conducted in 185 patients (approximately 30 per treatment group) aged 18-59 y.o. diagnosed with a psychotic disorders (with no acute exacerbation within 3 months).

This study's results were reviewed in depth by Maryann Gordon, M.D. from the Division of CardioRenal Drug Products (Consult: 6/14/00). The following tables (based on tables from Dr. Gordon's review and the sponsor's table 5.2.2.1.1) summarize the results of Study 054:

	Ziprasidone	Risperidone	Olanzapine	Quetiapine	Thioridazine	Haloperidol
QTc*	20.6 msec	10.0	6.4	14.5	35.8	4.7
Heart Rate	4.6 bpm	6.4	6.5	11.2	5.7	-2.9
QT	7.0 msec	-11.8	-9.3	-12.2	19.7	12.5

Mean change from baseline in the absence of a Metabolic Inhibitor

*Using Bazett's formula

Mean change from baseline in the presence of a Metabolic Inhibitor

	Ziprasidone	Risperidone	Olanzapine	Quetiapine	Thioridazine	Haloperidol
QTc*	20.4	3.2	5.3	19.7	28.0	8.9
Heart Rate	3.6	0.5	3.0	15.1	-2.1	-5.7
QT	9.9	1.1	-1.8	-15.8	33.3	22.5

*Using Bazett's formula

From the above tables, it can be seen that ziprasidone demonstrates a QTc mean change from baseline that is higher than the recently marketed atypical antipsychotics and haloperidol, and that thioridazine demonstrates the greatest QTc mean change from baseline. Quetiapine may appear to have a relatively high QTc change from baseline; however, it has been proposed that Bazett's formula may not be accurately applied to drugs which have an appreciable increase in heart rate. In a preliminary review of Study 054, Greg Burkhart, M.D., M.S. (1/11/00), points out that the "for drugs that cause an increase in heart rate, one would expect the QT to decrease, as occurred with risperidone, olanzapine, and quetiapine. However, for thioridazine and ziprasidone, both of which increase in heart rate, the QT increased. (Haloperidol also had an increase in QT, but the heart rate decreased from baseline in this group so that the increase in QT would be expected.)" It can also be seen from the above tables that there is no appreciable change of QTc for ziprasidone in the presence of a metabolic inhibitor. (Please see Dr. Gordon's and Dr. Burkhart's reviews for more result details).

Considering the results of this study and the findings of a dose dependent increase in the QTc from the placebo controlled studies, Dr. Gordon stated that ziprasidone increased the QTc from baseline on an average of 10-20 msec, compared to thioridazine's change of approximately 36 msec, and the 21 msec QTc prolongation observed with sertidole (an antipsychotic withdrawn prior to marketing in the U.S. because of concerns regarding sudden deaths observed in the UK's post-marketing reports). The slight increase in blood level changes observed for ziprasidone in the presence of a metabolic inhibitor were negligible enough to not present an additional concern.

Dr. Gordon concluded that Study 054 demonstrated that ziprasidone and thioridazine adversely affect cardiac repolarization as seen by their ability to prolong the QTc and QT intervals in a concentration-related manner, and that some patients would be at an increased risk of potentially fatal arrhythmias when exposed to either of these drugs. Considering the characteristics of QTc prolongation as an added risk. Dr. Gordon recommended that a drug with this profile either not be marketed or be used only as second line therapy.

In an effort to establish some benefit for the use of ziprasidone as an antipsychotic, the sponsor also provided data showing that patients had a decrease in total cholesterol and triglycerides in the ziprasidone group compared to the other antipsychotic treatment groups in this open label study 054. However, it is noted that in the short term placebo controlled trials, the ziprasidone groups were shown to have statistically significant increases in cholesterol and triglyceride levels when compared to placebo with respect to numbers of patients exceeding threshold values.

Also of interest in this study is that the mean serum concentration of a dose of ziprasidone 80 mg bid was 49 ng/ml on Day 2, and increased to 171 ng/ml at steady state (Day 8). In the presence of the 3A4 metabolic inhibitor ketoconazole, the mean ziprasidone concentration increased to 224 ng/ml.

Data Regarding Weight Changes

The sponsor claims that there is a beneficial weight gain profile to ziprasidone compared to other antipsychotics. However, in the placebo controlled studies, there was an increase of \geq 7 % weight gain observed, which was statistically significant compared to placebo. The only head-tohead comparison study which the sponsor describes is Study 054 (an open label study) in which 2 patients (5.9%) in the ziprasidone group were observed to have a weight gain \geq 7% while 1 patient (3.1%) in the haloperidol group, 6 patients (23.1%) in the olanzapine group, 5 (18.5%) in the risperidone group, 3 patients (10.3%) in the quetiapine group, and 3 patients (9.7%) in the thioridazine group showed a weight gain ≥ 7 %; these results are difficult to interpret as there was no placebo control group, and it was a short term study (less than 28 days) in a relatively small sample (25-35 patients in each group). In the 52 week placebo-controlled study 303, two of the three ziprasidone treatment groups showed an higher percentage of patients with weight gain when compared to placebo (ziprasidone 20 mg bid: 11% gained \geq 7%, in 40 mg bid group: 4.3 %, in 80 mg bid group: 8.6 %, and placebo: 4.3 % patients gained \geq 7 %); it is noted that 46% of the patients in the ziprasidone group and 20% of patients in the placebo group completed this study. Other studies sited by the sponsor were open label studies, and not located in the NDA submissions.

Labeling

If approved, the following are recommended revisions to the sponsor's proposed labeling (3/10/00):

- The sponsor's proposed statements regarding a pharmacokinetic profile in the pediatric population based on a small study (n=25) of pediatric patients with Tourette's Syndrome (under Special Populations). These findings are preliminary and efficacy for schizophrenia (the labeled indication) has not been tested in children or adolescents, and it could be misleading to include this data. It is also recommended that the labeling not include outcomes of this pilot study of Tourette's Syndrome in children/adolescents until the sponsor has proven safety and efficacy in this population for the proposed indication of schizophrenia (under Clinical Trials: Pediatric Studies section of sponsor's proposed labeling).
- 2. Under Contraindications, it may be beneficial to add several other drugs by name which also prolong the QTc that should not be used concomitantly with ziprasidone such as quinidine, pimozide, thioridazine, sotalol, moxifloxicin, and sparfloxacin. It would be prudent to also contraindicate this medication in patients with congenital long QT syndrome, history of cardiac arrhythmias, uncompensated heart failure, and acute myocardial infarction, as the sponsor has proposed.
- 3. Under the Warnings Section, it is recommended that the labeling resemble the new proposed labeling for Mellaril, with a bolded, black box delineating concerns regarding ziprasidone's effect on the QTc interval, and making the language strong enough that this drug would be used as a second line with emphasis that efficacy has not been established in the treatment resistant schizophrenic population.

4. Under the Precautions Section, it is recommended that the following subsections be added to the sponsor's proposed labeling: rash, orthostatic hypotension, potential for cognitive and motor impairment (somnolence), and dysphagia to reflect both the integrated safety data base of ziprasidone and standard language in the labeling of antipsychotic medications.

Because of one case of priapism observed in this data base, it is recommended that priapism be added to the precautions section of the labeling.

- 5. In the Information to Patients, it is important that patients be alerted to the risks involved with syncopal events and their need to seek medication attention if an episode occurs. Other information should include the current understanding of drugs which prolong the QTc interval, as ziprasidone does, to cause potential fatal arrhythmias, and sudden death, in addition to syncope. It may also be of some aid to have a patient insert with every prescription to maximize the efforts to educate patients and families. Ideally, there would be a mechanism to insure informed consent.
- 6. Under Laboratory Tests, it may be prudent to recommend patients obtain screening tests of an ECG and electrolytes to rule out circumstances which may leave patients more vulnerable to the cardiac adverse events associated with drugs which prolong the QTc interval. It might also be prudent to recommend routine ECGs to rule out any new onset ECG changes as a result of ziprasidone exposure.
- 7. Under Drug Interactions, it should be emphasized that there is no data regarding ziprasidone's effect when co-administered with another drug which prolongs the QTc, and that this combination should be avoided at the current time.
- 8. Because the NDA for has been withdrawn at this point, all references to the for he sponsor's proposed labeling.

IV. Financial Disclosure Information

The sponsor submitted a certification of Financial Interests and Arrangements of Clinical Investigators in Section 19.3 entitled Financial Disclosure Information.

There were a total of 5 sites in the study including four U.S. sites (794, 602,782, 529) and one site in South Africa (529). Site 794 had 2 principle investigators and 13 subinvestigators; site 602 had 1 principle investigator and 9 subinvestigators; site 782 had one principle investigator and 13 subinvestigators; site 529 had one principle investigator and 6 subinvestigators, and site 5006 had 1 principle investigator and 6 subinvestigators.

Otherwise, there were no other specific financial disclosures made by other investigators. No disclosures were able to be collected from six individual subinvestigators who were no longer working at the study sites and either had no forwarding address or did not respond to forms sent to their forwarding address. The sponsor's Director of Medical Finance signed the Form 3454 certifying that there was no financial arrangement made with investigators that could affect the

ł

outcome of the study as defined in 21 CFR 54.2(a), and that no listed investigator was the recipient of significant payments of other sorts as defined in 21 CFR 54.2(f).

Because of the E disclosed by 1 , a cursory review of the ECG data of Site compared to other sites was performed. From this informal brief review, results from Site appeared to be consistent with results from the other sites. In a telecon of 7/12/00, Dr. Charles Ritrovato of Pfizer stated that he was unable to provide information at this time as to how many patients Dr. had enrolled in this study site, but it was noted that Dr. . was one of 13 subinvestigators at this site. Based on the limited information available at this time, there does not appear to be a noticeable difference in the findings of Site _____ compared to other sites. It cannot be definitively determined if Dr. potential financial conflict was problematic or not; however, the total results from the study site in which Dr. was a subinvestigator did not appear to distort the final outcome of Study 054 in any obvious manner.

Efforts on the sponsor's part to minimize bias of Study 054 included randomization of subjects through the use of a tele-randomization system operated in the UK. Also, the sponsor utilized a blinding process in which ECG tracings from each site were transmitted electronically to under they were blinded and then forwarded to the central reader at for interval determinations.

V. Foreign Marketing

Ziprasidone is not marketed anywhere in the world at this time.

The following countries have approved the marketing of ziprasidone: Brazil (February 5, 1998), Sweden (as of June 10, 1998), Venezuela (November, 25, 1998), Czech Republic (March 15, 2000), New Zealand (April-20, 2000).

The sponsor listed the following countries where the application for ziprasidone is currently under review: Canada, Turkey, Malaysia, Hungary, South Africa, Switzerland, Egypt, Colombia, Slovakia, Poland, Croatia, Indonesia, Morocco, Slovenia, Mexico, and Bulgaria.

VI. Conclusions/Recommendations

This safety update includes ECG data from Study 054, which indicate that ziprasidone produces a QTc prolongation that is substantially greater than the other tested atypical antipsychotics and haloperidol, but less than thioridazine. The magnitude of the mean change from baseline of 20 msec is well above that suggested as the threshold for concern (5-10 msec) stated in our non-approvable letter of June 17, 1998.

The cardiac risks associated with QTc prolongation in antipsychotic drugs have been discussed in two open public discussions (PDAC: for sertindole and ziprasidone). Aside from the formal votes, there was an acknowledgement of the association of QTc prolongation with the events of syncope, ventricular arrhythmia, and sudden unexplained death; however, there has been a lack of sufficient evidence to directly address the safety risk associated with this degree of QTc prolongation. Hence, there was no data presented to suggest that our previous level of concern regarding the cardiac safety of ziprasidone, the sole determinant of the non-approvable action in 1998, was unwarranted. The sponsor states that ziprasidone has advantages over other marketed antipsychotics such as less weight gain and a less adverse effect on lipid profiles. However, it is questionable if these advantages outweigh the cardiac risks of this drug, such as potentially fatal arrhythmias. Weight gain and lipid abnormalities can be detected early, monitored, and managed, unlike ventricular arrhythmias, which cannot be predicted and may have irreversible consequences.

After consideration of this safety update and the discussion of the ziprasidone PDAC, I must conclude that the extent of QTc prolongation associated with ziprasidone represent a signal of cardiac risks which do not outweigh the benefits of treatment. If the sponsor were able to demonstrate ziprasidone's ability to effectively treat patients with treatment refractory schizophrenia, then, perhaps for some individuals, this would offer an overriding benefit. Until there is evidence to support that advantage, it is recommended that this application not be approved.

- 7/28/00

Roberta L. Glass, M.D. Medical Officer, Division of Neuropharmacological Drug Products

NDA 20-825 Div File HFD-120:Katz/Laughren/Hardeman/Glass

8-9-00 I drange with the above muchusing presmundation I put that yoprovidan can be approved. I to baris for this view is provided in a memo to the file on this date. 151 . MD Tran Leader, 7313 Energy

APPENDIX I

Deaths occurring during or after trial treatment : Cut-off date: 2/5/00

Ziprasidone subjects who died \leq 30 days after treatment

SUBJECT #	AGE	LAST	DAYS	CAUSE OF DEATH/COMMENTS
	1	DOSE	OF	
	SEX	(MG/D)	TREAT-	
			MENT	
108E-7160157**	49/M	160	466	Found unresponsive on bathroom floor immediately after a thump was heard. Was reported to have been revived for several minutes after brother administered CPR, but ECG showed asystole in emergency room Lab tests showed CPK=851 IU(24-195 IU) with MB fraction of 6.0 ng/ml (0-5.0 ng/ml), potassium=6mEq/L(3.3-5.1 mEq/L), bicarbonate=10 mEq/L(24-32 mEq/L). Ziprasidone level (at death)=14 ng/ml. Investigator thought cause of death was acute myocardial infarction. No autopsy performed. No coroner's report available. ECGs during the study: Screening: QTc=432 Baseline: QTc=396 Week 6: QTc=396 Week 40: QTc=415 Was on ziprasidone at time of death.
Yale-9990040** ; ;	34/M	100	319	Found dead. Had complained of chest pain on the day of his death Had complaint of brief episodes of"heart pounding" and "skipping heartbeats" during the study. An autopsy report stated that the cause ofdeath was occlusive coronary atherosclerosis. ECGs during the study:Screening: QTc=377 msecWk1: QTc=413Wk8: QTc=378Wk24: QTc=391Wk40:QTc=429Wk2: QTc=390Wk12:QTc=425Wk28: QTc=390Wk16:QTc=366Wk32: QTc=409Wk16:QTc=366Wk4: QTc=397Wk20:QTc=407Was on ziprasidone at the time of death.
*Included in Sudden U *Reported after 8/29/9	nexpected	Death (SUD) rate calculati	on

.

				Appendix I: Table of Deaths (con't)
NY- 97031E0054**	51/M	40	151	 Found dead. Two days prior to death, patient c/o not feeling well. Five days prior to death was seen by general practitioner with bp=140/100; no neurological findings noted. ECG at Day 7 showed flat T waves. ECGs during study: Screening: QT/QTc=358/428msec; HR=86bpm Wk1: " =351/425 ; " =88 Wk6: " =383/393 ; " =63 Wk13: " =405/416 ; " =63 (patient died Wk21) According to sponsor, a discussion with medical examiner suggested left meningioma without evidence of brain injury was found on autopsy; however, no report available.
JP-96-602- 450081**	50/M	40	Unclear	Unknown cause of death. Baseline values: heart rate=46bpm; QT/QTc= 140/359 msec. One day after starting ziprasidone, patient presented at hospital with c/o palpitations and nausea; bp=150/100 mmHg, hr=119, QT/QTc=282/398. Two days after starting ziprasidone: ECG showed hr=47, QT/QTc=384/349; WBC=13130 cells/UL (NL: 3000-9000 cells/UL), neutrophils 81.4% (NL:40-74%), CPK=297 IU/I (NL:26-200 IU/I); diagnosed with URI and treated with antibiotic, flomoxef. Police contacted patient's physician to report his death; police ruled out suicide and homocide. Death occurred within five days after starting ziprasidone (length of treatment unclear).
105-5340021*	70/F	2	5	Patient had sudden onset of shallow respirations and diaphoresis. Death certificate stated acute <u>cardiopulmonary arrest due to arteriosclerotic cardiovascular disease</u> . Subject with history of right bundle branch block, otherwise ECG was normal. Was taking ziprasidone just prior to death.
108-6070305*	46/M	80	61	Found dead (in heat of 100°F). Autopsy report stated cause of death as <u>acute and chromic asthmatic</u> bronchitis and granulomatous myocarditis. ECG: Screening: QTc =366 msec Baseline: QTc =393 Week 6: QTc=395 Was on ziprasidone at time of death.
108-5920750*	39/F	120	8	Found dead one day after her estimated date of death of unknown cause. Patient's face was burned and it was thought that she had fallen against a hot water pipe. The investigator's postmortem diagnosis was alcohol abuse/diabetic ketoacidosis, but there is no evidence for this. No coroner's report located in the CRF. Was on ziprasidone at time of death.
*Included in Sudden I *Reported after 8/29/		Death (SUD)) rate calculat	ion

i

1

1

J

÷

÷

1

48/M 52/M	120 120 80	71	 Found dead in his hospital bed. Autopsy showed generalized atherosclerosis, coronary artery disease, cerebral artery disease, visceral congestion (liver, spleen, and lung), COPD, and cardiac hypertrophy. ECGs during the study: Screening: QTc=391 msec baseline: QTc=383 week 2: QTc=367 week 6:QTc=391 Patient had complaint of chest pain once during the study, but ECG was normal and diagnosed as anxiety. Was on ziprasidone at time of death. Found dead. CRF showed hypertension and tachycardia on last day of study with hypertension as adverse event during study. Narrative states that subject had history of polydipsia and seizure disorder. Details regarding the death are unclear. Died one day after discontinuing ziprasidone.
			Found dead. CRF showed hypertension and tachycardia on last day of study with hypertension as adverse event during study. Narrative states that subject had history of polydipsia and seizure disorder. Details
52/M	80	221	
		, 221 :	Found dead while taking a nap. No autopsy performed and exact cause of death is <u>unknown</u> . ECG during the study as shown in the safety update: QTc at: Screening=374.7 msec Week 12=415.69 Week 28=413.12 with flat T wave in lead AVL; no evidence of ischemic changes. The CRF had minimal information and the patient profile in the safety update had different ECG QTc values than the original submission. Was on ziprasidone at time of death.
28/F	120	57	Patient reported be cachectic and had ECG changes consistent with subendocardial ischemia with substernal pinching sensation. Patient was d/ced from ziprasidone and treated with thioridazine, nitrazepam and patient died two days later. Cause of death reported to be myocarditis. Death occurred 2 days after d/c from ziprasidone.
63/M	80	485	Sudden collapse and died. Coroner's report stated that cause was a ruptured abdominal aortic aneurysm and atherosclerosis. Was on ziprasidone at the time of death.
43/M	40	16	Found dead. Coroner's cause of death listed as <u>asphyxiation due to aspiration of vomit</u> . Was on risperidone, clonazepam and lorazepam at time of death. Patient had difficulty breathing three days before death, and complained of dyspnea on morning of death. Died 29 days after discontinuing ziprasidone.
4	3/M 3/M	3/M 80 3/M 40	8/F 120 57 3/M 80 485

1

÷

i

				Appendix I: Table of Deaths (con't)
116B-6590001	44/F	120	47	Patient had a UTI upon d/c and was diagnosed with gastritis with Helicobacter pylori 13 days later. She was seen in ER with diagnosis of panic attack 21 days after d/c (three days prior to death). Sponsor reports that the autopsy was not available due to legal issues in medical examiners, but Subject's attending physician reportedly got information from the medical examiners that subject had a benign cardiac neoplasm (myxoma). ECG: screening: QTc=444 msec week 2: QTc=440 week 6: QTc=407Week 1: QTc=433 week 1: QTc=433 week 2: QTc=440 day after starting ziprasidone: cardiology w/u was normal, but had elevated transaminases. Episodes of tachycardia and hypertension during the study: day 6:102 bpm day 20:120/100; 104 bpm day 42:164/98It is unclear what medications she was on as the patient summary and the CRF do not list the same
				medications. Death occurred 24 days after d/c from ziprasidone.
303-1970299	79/F	80	30	Cardiac arrest. No autopsy was performed. Patient had new diagnosis of atrial fibrillation and ischemic heart disease 27 days after d/c. At time of death was taking perphenazine, deparkin, digoxin, verapamil, and enalapril. Death occurred 30 days after d/c from ziprasidone.
Suicides and accid	ents	·		
108-6090381	21/F	160	54	Suicide by gunshot while on ziprasidone.
116B-6940004	24/M	160	146	Suicide by hanging while on ziprasidone. Subject had been complaining of increasing depressed mood; treatment included an increase in ziprasidone.
117-6870317	51/M	120	205	Death by defenestration. According to study profile, patient did not appear suicidal prior to death.
117-7060529	40/M	160	54	Patient stopped ziprasidone on his own and four days later he drove his car off a cliff. Subject was driving his car after a sleep deprived EEG against medical advice. Autopsy listed <u>asphyxiation due to drowning</u> and was classified as a probable traffic accident.
302-2600156	46/M	120	7	Patient's body found drowned in local river after being missing from the hospital for five days.
302E-1590029	22/M	120	179	Suicide by falling under a train. Was being treated with ziprasidone at time of suicide with plans to be admitted to the hospital that same day.
ЛР-95-6011622	53/M	53	20	Suicide seventeen days after discontinuing ziprasidone (Japanese studies: not part of the integrated safety data base.)
NY-97-002-0022*	34	160	9	Probable suicide by drowning. Was on ziprasidone at time of death. Postmortem finding reported pulmonary congestion and pulmonary edema consistent with drowning episode as cause of death.
NY-97-002-0077*	37	60	13	Died of carbon monoxide poisoning and cardio-respiratory arrest secondary to accidental fire, according to coroner's report. Patient found unconscious. Was on ziprasidone at time of death.
NY-97-033-053 *	23	160	53	Suicide by gunshot while on ziprasidone
*Included in Sudden U *Reported after 8/29/9		Death (SUD) rate calcula	ation

1

.

	Appendix I: Table of Deaths (con't)						
NY-97-033-266*	20	160	8	Drowned while swimming in the sea. Patient had c/o palpitations for 4 months prior to death; also c/o dystonia 4 days prior to death, treated with trihexyphenidyl and benztropine, and was on propranolol. On ziprasidone at time of death.			
601E-0187-0118*	41	80	46	Probable suicide; autopsy report states cause of death: exsanguination from multiple deep incised wounds. On ziprasidone at time of death.			
R0553012097- R0267*	47	80	19	Suicide by hanging. Was on ziprasidone at time of death.			

Ziprasidone subjects who died \geq 30 days after treatment

ı.

dden death; cause unknown. Occurred 71/2 months after discontinuation from ziprasidone.
known cause of death but possible seizure and aspiration of vomit. Was taking risperidone at time
death. Death occurred 41/2 months after stopping ziprasidone .
cidental drowning. Died 11/2 months after stopping ziprasidone .
icide by gun shot one year after d/c from ziprasidone .
ed of lobar pneumonia 2 months after stopping ziprasidone. IM:po extension study.
ed of unknown causes over 2 months after stopping ziprasidone.
ed of complication due to pancreatitis 9 months after stopping ziprasidone
icide by hanging approximately 3 months after stopping ziprasidone.
ed of bronchopneumonia with bronchial adenocarcinoma and metastasis. Death occurred 4 months er stopping ziprasidone.
ed of cranial trauma 2° to fall. Ziprasidone was stopped 45 days prior to death.
ed of heart failure more than 2 years after stopping ziprasidone.
dden death due to acute purulent leptomeningitis. Death occurred 4 1/2 months after stopping prasidone.
onchopneumonia. Stopped ziprasidone 112 days before diagnosis.
dden death due to acute cerebral edema. death occurred 60 days after stopping ziprasidone.
ed of post operative cerebral edema after tumor removal. Occurred two months after stopping rasidone

				Appendix I: Table of Deaths (con't)
304-2040222	55/F	160	29	Sudden death with proposed cause of acute heart failure due to pulmonary disease. Death occurred approximately 2 months after stopping ziprasidone.
307-2650034*	47/M	80	365	Death from trauma of fall from window over one year after discontinuing ziprasidone.
307-2690047	49/F	100	196	Died of hepatic coma, cholestatic jaundice and malignant neoplasm 95 days after stopping ziprasidon Discontinued ziprasidone because of jaundice and elevated AST (244 U/L) and ALT (375 U/L).
NY-97-001-355*	25/M	160	41	Suicide by hanging 2 months after stopping ziprasidone. IM/po extension study.
NY-98-035- 0572*	23	160	71	Suicide by hanging 6 weeks after stopping ziprasidone. IM/po extension study.
NY-98-035-586*	30/M	160	60	Died of pneumonia 33 days after stopping ziprasidone.
JP-94-601 0014*	51/F	20	37	Died of myocardial infarction. Symptoms began 6 days after d/c from ziprasidone when patient fell, diagnosed with cyanosis requiring oxygen therapy. Fifty-five days, later patient died. During study ECGs were read as abnormal as follows: Baseline: ST-T AbnormalFollow Up: ST-T Abnormal Right Atrial HypertrophyWk4:ST-T AbnormalRight Atrial Hypertrophy Sinus TachycardiaFollow Up: ST-T Abnormal Right Atrial Hypertrophy

*Included in Sudden Unexpected Death (SUD) rate calculation *Reported after 8/29/97

÷

3083200008	36/F	Unknown		Randomized to <u>ziprasidone</u> group. Patient reported to have ingested alcohol and taken O/D
				of risperidone. Had postural hypotension and lengthening of QTc (centrally read):
				Baseline QTc=413; On day of event: QTc=458.
1085230870	41/M	Unknown		Patient reporting taking O/D of diazepam and
				lorazepam to self treat insomnia.
1085820420	35/M	Unknown		Patient found comatose and thought to have
		0		overdosed. Responded to naloxone with
			!	improved respiration. Drug screen positive for
			}	propylene glycol, cocaine & opiate. Lithium
				level=0.5 mEq/L; ziprasidone serum concentration
				< 1ng/ml (below limit of quantization). Treated
				with gastric lavage and charcoal. Many
				complications during course of treatment (see
				4/6/00 submission for details).
116B5510005	38/M	Unknown	Aspirin	Patient reported O/D of aspirin. ECG was WNL;
	·			blood gas showed respiratory alkalosis and
				metabolic acidosis. Recovered in 24 hours.
116B5553002	62/F	Unknown	Thioridizine	Patient admitted to hospital for sedation and
				discharged 12 days later. Narrative suggests
				possible O/D of thioridizine, but this is unclear.
116B5810014	35/F	Unknown	Lorazepam	Patient reported O/D of lorazepam. Gastric
				lavage performed.
116B5950018	33/F	Unknown	Lorazepam	Staff reported O/D of lorazepam and flurazepam
			flurazepam	
116B6690025	41/F	Unknown	Acetominophen	Patient reported O/D of acetominophen and
				drinking alcohol. No symptoms reported.
1175080352	54/M	Unknown	Chloral hydrate	Found on floor. Thought to be intentional
			Lorazepam	overdose of chloral hydrate and lorazepam. L
				Treated with activated charcoal; rhabdomyolysis
3011110384	28/F	Unknown		in leg. Patient reported O/D on cyamemazine with
3011110304	20/F	Unknown		recovery after gastric lavage.
3011320771	34/M	Unknown		Reported as suicide attempt by poisoning. No
				details located.
3011380814	27/M	Unknown	Temazepam	Report to have ingested zopiclone; hospitalized
			, on acopain	and recovered after gastric lavage.
3021500046	18/M	Unknown	+	Reported to have ingested haloperidol.
				Hospitalized and received gastric lavage.
3040390343	27/F	Unknown	<u> </u>	Reported to have ingested O/D of aspirin.
3041720304	24/M	Unknown	Parcetamol	O/D of lorazepam.
			lorazepam	· ·
601E1920034	40/F	Unknown	Lorazepam	O/D of lorazepam to treat insomnia. Patient
		1	1	hospitalized for confusion, disorganized thoughts
		1	1	and bizarre behavior.
R5550007	19/M	Unknown	1	Excessive alcohol intake reported. Event
-	[1		resolved 4 days later.
NY980350474	?/F	Unknown		Reported to have taken O/D of zopicione.
980350338	33/M	Unknown	1	Reported to have taken O/D of zolpidem &
	1		1	chlordiazepoxide.

APPENDIX II Other Overdose Cases During Treatment with Ziprasidone

r

•

·····	
	SERIOUS ADVERSE EVENTS FOR ZIPRASIDONE GROUPS
	Period of May 15, 1997 to February 5, 2000
	(table revised from sponsor's submission of 5/22/00)
PID	Event
Cardiovascular	
302E-057-0456	Syncope and bradycardia
127E-595-0013	Bradycardia (2 episodes)
127E-701-0003	Cardiomegaly; Possible Congestive Heart Failure; Pneumonia
R-0554-0070	Congestive Heart Failure; Chronic Obstructive Pulmonary Disease
303-269-0207	Hypertension
308-320-0008	Prolonged QT Interval; Worsening Schizophrenia
NY-97-014 252	Convulsion; Prolonged QT Interval
R-0554-0129	Arterial Occlusion; Leg Amputation
JP-96-602	Cerebral Infarction
460001-1	1
Seizure	· · · · · · · · · · · · · · · · · · ·
123-1001-0061	Recurrence of Tonic-Clonic Seizure
127E-795-0002	Seizure
	Generalized Tonic-Clonic Seizure
601E-0624-0123	Seizure
602-0651-0105	Tonic-Clonic Seizure
(also listed as	
602-216-0105)	
R-0554-0126	Grand Mal Seizure
NY-97-014 252	Convulsion; Prolonged QT Interval
Movement Disord	lers
116B-0581-0017	Exacerbation of Tardive Dyskinesia
108E-0523-0148	Left Leg Dystonia; Gait Disturbance
R-0554-012	Restlessness; Abnormal Movements of Lower Extremities; Tightness in Chest;
	Anxiety
R-0553-0045	Acute Dystonia
Possible NMS	
601E-0756-0085	Medication of induced movement disorder (NMS Like); Bipolar disorder, recurrent
(aka	depression;
69202190080)	respiratory distress; aspiration pneumonia
JP-95-601 57-2	Decreased Level of Consciousness; Hyponatremia; Increased Blood Pressure;
	Increased
-	Heart Rate; Increased Creatine Phosphokinase
Metabolic/Hemate	
	Elevated Blood Sugar
1	, view of the second seco
R-0553-0201	Hyperglycemia
601E-0540-0077	Neutropenia
NY-97-014 052	Dizziness; Unstable Balance; Falls; Shallow Respirations; Decreased Hemoglobin;
	Increased Platelets; Unsteadiness in Feet; Oculogyric Crisis; Extrasystoles;
	Postural Hypotension
Skin	
108-0681-0618	Cellulitis, Facial
116B-0590-0003	Cellulitis
108E-0509-0168	Ulcerated Basal Cell Carcinoma
NY-97-031 0041	Cellutitis
307-0265-0035	Sunburn
301-0203-0033	

.

١

_

Pulmonary	
108E-0594-0094	Aspiration of Food
R-0553-0083	Pulmonary Emboli
R-0553-0160	Pneumonia; Worsening of Schizoaffective Disorder
127E-595-0016	Exacerbation of Asthma; Exacerbation of Schizophrenia
127E-669-0008	Traumatic Pneumonthorax,
NY-98-035 0585	Left Lower Lobe Pneumonia; Pneumothorax
	Right Lower Lobe Pneumonia
Gastrointestinal	
NY-98-035 0246	Nausea; Serious Diarrhea
116B-0596-0006	
1100-0090-0000	Worsening of Gastroesphogeal Reflux Disorder; Barrett's Esophagitis; Laparoscopy with
	partial Fundoplication
116B-0556-0001	Worsening Diverculosis
	Monoping Contrip English Monoping Durdenities English to the Contribution
601-0540-0217	Worsening Gastric Erosion; Worsening Duodenitis; Exacerbation of Bipolar
	Disorder, Mania
Miscellaneous	
602E-0651-0084	Excessive Sedation
(Previously listed	
as	
602E-0246-0084)	
108E-0620-0041	Papilloma, left Ureter; Transitional Cell Carcinoma in Situ, Left Ureter
R-0553-0191	Breast Cancer - Female
601E-0756-0086	Carcinoma of the Bladder, Deep Vein Thrombosis
NY-98-035 0576	Tonsillitis
Fractures	
1160-603-0002	Open Reduction Internal Fixation of Right Humerus
116B-0523-0002	Fractured Femur, Accidental
1165-0653-0003	Ankle Fracture; Accidental
168E-0881-0079	Surgical Site Infection, Right Foot
127E-719-0005	Ankle Fracture, Accidental; Staphylococcus Infection; Increased Depression
JP-96-602	Accidental Bone Fracture
270050-4	
R-0554-0024	Fractured Ankle
	sychiatric Illness and Suicidal Gestures
	Exacerbation of Schizophrenia; Recurrent Depression; Suicidal Gesture
116S-0650-0001	Exacerbation of Chronic Paranoid Schizophrenia
108E-0523-0203	Exacerbation of Psychotic Symptoms; Suicidal Ideation; Increased Insomnia
116B-0581-0014	Suicide Attempt; Intentional Drug Overdose; Persistent Depression
116B-717-0003	Exacerbation of Schizophrenia
R-0554-0050	Exacerbation of Psychosis; Attempted Suicide (Laceration of Wrist)
JP-95-601 131-2	Exacerbation of Schizophrenia; Suicide Attempt
301E-0102-0231	Alcohol Intoxication; Alprazolam Intoxication
NY-97-031E 0016	Suicidal Ideation
NY-97-032 020	Relapse of Schizophrenia; Manic Seizure
NY-97-033 288	Exacerbation of Psychosis
NY-98-035 0025	Suicide Attempt
NY-98-035 0043	Exacerbation of Schizophrenic Symptoms; Manic Episodes; Zona (Herpes Zester)
NY-98-035 0058	Psychotic Relapse
NY-98-035	Increased Irritability; Anxiety, Lesions on Hands; Possible Hallucinations;
0079	Exacerbation

· -.

NY-98-035 0093	Exacerbation of Schizophrenia
NY-98-035 0518	Worsening Psychosis
NY-98-035 218	Suicide Attempt (Laceration of Wrists)
NY-98-035 0255	Suicide Attempt; Worsening Hallucinations
NY-98-035-0923	Suicide Attempt By Jumping from window; fractured Clavicle; Brain Contusion
601-0615-0120	Exacerbation of Bipolar Disorder, Mania; Exacerbation of Bipolar Disorder, Depression; Suicidal Gesture
601E-0669-0021	Suicidal gesture; Exacerbation of bipolar disorder, depression
601E-0529-0176	Exacerbation of Bipolar Disorder, Depression; Suicidal Ideation
601E-0646-0059 (Previously listed as 601 E-0201-0059)	Anxiety
601E-0279-0319	Recurrent Suicide Attempt
602E-0719-0123	Exacerbation of Mania
602E*0662-0051 (Previously listed as 602E-0244-0051)	Exacerbation of Mania
602E-0662-0125	Exacerbation of Bipolar Disorder, Mania
602E-0764-0062 (Previously listed as 602E-0255-0062)	Bipolar Disorder, Recurrent Depression; Suicidal Ideation
R-0554-0061	Worsening Psychosis; Suicide Attempt
R-0555-6039	Suicide Attempt
R-0585-6061	Exacerbation of Psychosis

APPENDIX IV

Reasons for Discontinuations For Period of May 15, 1997 to February 5, 2000

Nausea, vomiting, chest pain, headache, dizziness, hypertension, tardive dyskinesia, cerebrovascular accident, dystonias, tic disorder, EPS, akathisia, tardive dyskinesia, excessive weight loss, increased liver function test (alkaline phosphatase, SGPT, SGOT), somnolence, bradycardia, impotence, convulsion, ataxia, multiple sclerosis, insomnia, leg cramps, pancreatic cancer, tuberculosis, hepatic metastases, orthostasis, bronchospasm, meningitis, billiary tract disorder, catatonia, mania, sweating, anxiety, gynocomastia, and anemia.

-

REVIEW AND EVALUATION OF CLINICAL DATA

Application Information

NDA 20-825 Sponsor: Pfizer Inc. Original User Fee Due Date: March 17, 1998 Extended User Fee Due Date: June 17, 1998

Drug Name

J

Generic Name: Ziprasidone hydrochloride Trade Name:

Drug Characterization

Pharmacologic Category: Serotonin and Dopamine Antagonist Proposed Indication: Management of the Manifestations of Psychotic Disorders NDA Classification: 1S Dosage Forms: Oral tablets; 20, 40, 60, 80 mg capsules

Reviewer Information

Ctinical Reviewer: Roberta L. Glass, M.D. Review Completion Date: April 30, 1998 H |

Table of Contents for Clinical Review of NDA 20-825

1.0 M	aterial Utilized in Review			
1.1	Materials from NDA/IND	1		
1.2	Related Reviews, Consults for the NDA	1		
1.3	Other Resources	1		
2.0 Ba	ackground			
2.1	Indication	1		
2.2	Important Information from Related INDs and NDAs and from Pharmacologically Related Agents	2		
2.3	Administrative History	2		
2.4	Proposed Labeling	3		
2.5	Foreign Marketing	3		
3.0 Chemistry, Manufacturing, and Controls				
4.0 A	nimal Pharmacology and Toxicology	3		
5.0 De	escription of Clinical Data Sources			
5.1	Primary Source Data (Development Program)	4		
5.1.1	Study Type and Design/Patient Enumeration	4		
5.1.2	Demographics	5		
5.1.3	Extent of Exposure (dose/duration)	5		
5.2	Secondary Source Data	6		
5.2.1	Other Studies	6		
5.2.2	Postmarketing Experience	6		
5.2.3	Literature	6		
6.0 H	uman Pharmacokinetic Considerations	7		

· _

7.0 Review of Efficacy

7.	1	Background	8
7.	2	Review of individual studies	8
7.	2.1	Study 106	9
7.	2.2	Study 114	12
7.	2.3	Study 115	15
7.	2.4	Study 303	19
7.:	2.5	Study 104	22
7.:	2.6	Other <u>St</u> udies	25
7.	2.7	Pediatric Studies	25
7.	3	Summary of Data Pertinent to Important Clinical Issues	25
7.	3.1	Predictors of Response	25
7.	3.2	Choice of Dose	25
7.	4	Conclusions Regarding Efficacy Data	26
٠			
8.	0 Int	egrated Review of Safety	
8. _ 8.		egrated Review of Safety Methods and Finding for Safety Review	26
_ 8.			26 27
_ 8. 8.	1	Methods and Finding for Safety Review	
_ 8. 8. 8.	1 1.1 1.2	Methods and Finding for Safety Review Deaths	27
_ 8. 8. 8.	1 1.1 1.2 1.2.1	Methods and Finding for Safety Review Deaths Other Serious Adverse Events	27 29
_ 8. 8. 8. 8. 8.	1 1.1 1.2 1.2.1 1.2.2	Methods and Finding for Safety Review Deaths Other Serious Adverse Events Syncope/Hypotension	27 29 29
8. 8. 8. 8. 8. 8.	1 1.1 1.2 1.2.1 1.2.2 1.2.3	Methods and Finding for Safety Review Deaths Other Serious Adverse Events Syncope/Hypotension Rash	27 29 29 30
_ 8. 8. 8. 8. 8. 8. 8.	1 1.1 1.2 1.2.1 1.2.2 1.2.3 1.2.4	Methods and Finding for Safety Review Deaths Other Serious Adverse Events Syncope/Hypotension Rash Elevated Transaminase	27 29 29 30 31
_ 8. 8. 8. 8. 8. 8. 8. 8.	1 1.1 1.2 1.2.1 1.2.2 1.2.3 1.2.4 1.2.5	Methods and Finding for Safety Review Deaths Other Serious Adverse Events Syncope/Hypotension Rash Elevated Transaminase Neuroleptic Malignant Syndrome	27 29 29 30 31 32

8.1.3 Dropouts and "Other Significant Adverse Events"	35
8.1.3.1 Overall Profile of Dropouts	35
8.1.3.2 Adverse Events Associated with Dropout	36
8.1.4 Other Search Strategies	37
8.1.5 Common Adverse Events	37
8.1.5.1 Approach to Eliciting Adverse Events in the Development Program	37
8.1.5.2 Establishing Appropriateness of Adverse Event Categorization and Preferred Terms	37
8.1.5.3 Selecting the Best Adverse Event Tables for Characterizing the Adverse Event Profile	38
8.1.5.4 Identifying Common and Drug-Related Adverse Events	39
8.1.5.5 Additional Analyses and Explorations	39
8.1.6 Laboratory Findings	40
8.1.6.1 Extent of Laboratory Testing in the Development Program	40
8.1.6.2 Selection of Studies and Analyses for Overall Drug-Control Comparisons	40
8.1.6.3 Standard Analyses and Explorations of Laboratory Data	40
8.1.6.3.1 Analyses Focused on Measures of Central Tendency	40
8.1.6.3.2 Analyses Focus on Outliers	40
8.1.6.3.3 Dropouts for Laboratory Abnormalities	41
8.1.6.4 Additional Analyses and Explorations	42
8.1.7 Vital Signs	43
 8.1.7.1 Extent of Vital Sign Testing in the Development Program 8.1.7.2 Selection of Studies and Analyses for Overall Drug-Control Comparisons 	43 43
8.1.7.3 Standard Analyses and Explorations of Vital Sign Data	43
8.1.7.3.1 Analyses focused on Measures of Central Tendency	43

5

8.1.7.3.2 Analyses focused on Outliers	. 43
8.1.7.3.3 Dropouts for Vital Sign Abnormalities	43
8.1.7.4 Additional Analyses and Explorations	44
8.1.8 Electrocardiograms (ECGs)	45
8.1.8.1 Extent of ECG testing in the development program	45
8.1.8.2 Selection of Studies and Analyses for Overall Drug-Control Comparisons	45
8.1.8.3 Standard Analyses and Explorations of ECG Data	45
8.1.8.3.1 Analyses focused on Measures of Central Tendency	45
8.1.8.3.2 Analyses focused on Outliers	46
8.1.8.3.3 Dropouts for ECG Abnormalities	46
8.1.8.4 Additional Analyses and Explorations	46
8.1.9 Special Studies	47
8.1.9.1 Ophthalmology	47
8.1.9.2 Extrapyramidal Symptoms	48
8.1.10 Withdrawal Phenomena/Abuse Potential	48
8.1.11 Human Reproduction Studies	48
8.1.12 Overdose Experience	. 49
8.1.13 Pediatric Studies	. 49
8.2 Adequacy of Patient Exposure and Safety Assessments	50
8.2.1 Adequacy of Clinical Experience	50
8.2.2 Adequacy of Animal and/or In Vitro Testing	50
8.2.3 Adequacy of Routine Clinical Testing	51 -
8.2.4 Adequacy of Metabolic Workup	51

.

.... ··

8.3.3 8.3.4 8.3.5	Orthostatic hypotension/syncope Rash Seizure	53 54 54
8.3 <i>.</i> 6	Cholesterol/triglyceride elevation	54
8.3.7	Hyperprolactinemia	55
8.3.8	Transaminases elevation	55
8.3.9	Weight gain	55
8.3.10	Extrapyramidal Symptoms	56
8.3.11	Neuroleptic Malignant Syndrome	56
8.3.12	Somnolence	56
8.3.13	Tardive Dyskinesia	56
	Aspiration Pneumonia	56
8.3.14		
8.3.14 9.0	Labeling Review	57
	Labeling Review	57 57
9.0		

Med - Page 29 of 223

1.0 Material Utilized in Review

1.1 Material from NDA/IND

This NDA submission was presented in a combination of hard copy and electronic format. Case report forms were submitted in electronic format only. Addendum submissions clarifying issues of the original submission (3/18/97) were in hard copy only except for the safety update (8/29/97) which was submitted in both hard copy electronic format. There were no electronic datasets provided for this review.

The documents most frequently referred to for the purposes of this review were the following:

Integrated summary of efficacy Integrated summary of safety Study reports for trials 104, 106, 114, 115, and 303 Safety update report of 8/29/97 Literature summary

 for agitated

Case report forms were examined for the following subjects: all reported deaths, 116B-5510007,303-1970265, 303-0070098, 109-5720027, 117-6200029, 108-5740080, 114-6560036, 116B-551-0008, 303-2120105, 303-1970265, 102-5130005, 108-5740080, 109-5650041, 303-2710228, 116B-05230001, 304-----1890367, 108-6170817, 117-7060373, 117-7060380, 115-06560036.

1.2 Related Reviews and Consults for the NDA

The Division of Cardio-Renal Drug Products was consulted for issues concerning ziprasidone's prolongation of the QTc interval on ECG recordings by Charles J. Ganley(HFD-110: 11/18/98 and 1/6/98). Also referred to were the following reports: 1) A Review of UK Post-Marketing Surveillance Experience with Sertindole, Olanzapine and Risperidone by Greg Burkhart, M.D., M.S. (HFD-120: 12/12/97), 2) Clinical Pharmacology and Biopharmaceutics Review by Sayed Al-Habet, Ph.D. (HFD-860: 3/3/98), 3) Review of Clinical Data: General Characteristics of the Deaths in the NDAs for Olanzapine, Risperidone, Quetiapine and Sertindole by Greg Burkhart, M.D. (HFD-120: 3/3/98), 4) Review of Data Quality, Coding, All Cause Mortality and Sudden Deaths by Gerard Boehm, M.D., M.P.H. & James F. Knudsen, M.D., Ph.D. (HFD-120: 2/3/98), 5) Review of Ziprasidone ECG Data by Gerard Boehm, M.D., M.P.H. (HFD-120: 1/23/98), and 6) Statistical Review and Evaluation by Sue-Jane Wang, Ph.D. (HFD-710: 11/24/98).

1.3 Other Resources

Dr. Andrew Mosholder provided excellent mentoring in the preparation of this document and throughout ... the review process.

2.0 Background

2.1 Indication

The majority of the fifteen medications currently labeled for the indication of psychosis are considered to be traditional dopamine antagonist agents. These traditional agents have been associated with a high incidence of extrapyramidal symptoms (EPS) and long term risks of tardive dyskinesia. The more recently marketed 'atypical' antipsychotic agents (clozapine, risperidone, olanzapine, and quetiapine) possess serotonin type 2 (5-HT₂) receptor blocking activity in addition to their dopamine antagonist properties. It has been suggested that these 'atypical' agents may reduce the incidence of EPS, result in less risk of the

development of tardive dyskinesia, and be more effective in treating the negative symptoms of schizophrenia. In light of the risks of agranulocytosis associated with clozapine, it is indicated only for refractory patients. Of these 'atypical' antipsychotic drugs, only clozapine has shown superior efficacy in refractory patients thus far.

The sponsor of ziprasidone has characterized this drug as an 'atypical' antipsychotic demonstrating dopamine and serotonin receptor antagonist activity. The sponsor proposes that this medication minimizes EPS and also treats both positive and negative symptoms of schizophrenia.

The NDA for another 'atypical' neuroleptic, sertindole, was recently withdrawn because of concerns about QT interval prolongation and sudden deaths.

2.2 Important Information from Related INDs and NDAs and from Pharmacologically Related Agents

Pfizer has submitted IND

This sponsor also submitted IND(to study ()The review for these other submissions are still in progress and will not be discussed in this review except for relevant safety data.

According to a teleconference of September 4, 1997 with Dr. Ritrovato from Pfizer, all clinical studies of ziprasidone have been done under Pfizer's sponsorship.

In a recently reviewed NDA for the 'atypical' antipsychotic sertindole, concerns arose regarding the high incidence of sudden death and prolongation of the QTc interval within this NDA safety data base. In the literature there have been observations that QTc prolongation may be correlated with the development of ventricular arrhythmia, syncope, and sudden death (Morganroth, 1993). This safety concern was intensified when post marketing data from the U.K. were reviewed by Greg Burkhart, M.D., MPH (12/12/97), and it was found that sertindole demonstrated a higher sudden unexplained death reporting rate than olanzapine and risperidone. The sponsor of sertindole withdrew this NDA.

2.3 Administrative History

The original commercial IND for oral ziprasidone was filed on April 3, 1990. FDA allowed women of child bearing potential to be included in clinical trials in October, 1991; the sponsor began to include women of child bearing potential in August, 1993.

An End of Phase II meeting between FDA and the sponsor was held in March 1994. At that time, the FDA requested that the sponsor repeat the Segment II study in rabbits because of the low survival rate of fetuses and the lack of skeletal and visceral examination in all fetuses. FDA also requested that if the sponsor was interested in correlating neoplastic changes in pre-clinical carcinogenicity studies to an increase in serum prolactin, they would need to document that serum prolactin levels were simultaneously elevated in these animal studies. Characterizing the pharmacokinetic characteristics of ziprasidone and its metabolites was also emphasized.

According to FDA records, a pre-NDA meeting was held in May, 1996. During this meeting, the pharmacology group requested that the sponsor submit separate tables describing neoplastic and non-neoplastic findings from rat carcinogenicity studies; it was again requested that the sponsor submit preclinical data characterizing ziprasidone's effect on serum prolactin concentration and its relationship to neoplastic changes. The sponsor was also encouraged to identify the metabolic pathway of ziprasidone and cytochrome characterization to develop information regarding drug interactions. The format for the NDA submission was discussed during the pre-NDA meeting.

In July 1996, the sponsor met with the FDA to discuss and demonstrate the electronic submission of the NDA. In September, 1996, Pfizer was granted a waiver for submitting the case report forms in hard copy.

1

In November 1996, the sponsor met with the FDA chemistry group to discuss issues regarding chemistry, manufacturing and controls.

The original NDA was submitted to FDA on March 18, 1997 with a cut-off date listed as October 31, 1996. A safety update was submitted by the sponsor on August 29. 1997 with a cut-off date recorded at May 15, 1997.

In October 1997, the Division notified the sponsor by letter that there were concerns regarding the finding of QT interval prolongation with the use of ziprasidone.

On December 16, 1997, the sponsor notified DNDP that they intended to submit a major amendment to the NDA which would subsequently entitle the sponsor to have a 3 month user fee extension; this amendment (submitted 1/23/98) included one pre-clinical study and data from two pediatric studies (one completed study with n=18 and one interim report) in children with Tourette's Syndrome.

2.4 Proposed Labeling

The dosing instructions in the draft labeling recommended an initial dose of 40 mg bid with food, and, if needed, followed by dose titration up to 80 mg bid at intervals of 2 days or more. The labeling states that doses above 80 mg bid were not shown to be more efficacious than 80 mg bid, and that clinical assessment is recommended if a dose greater than 80 mg bid is to be undertaken. It is also mentioned that the safety profile of doses above 100 mg bid have not been assessed.

For maintenance therapy, the draft labeling recommends that the dose be 40 mg bid with allowance for individual patient differences. There is no maximum time period stated for use. It also states that no dose adjustments are necessary for age, gender, race, renal or hepatic impairment.

2.5 Foreign Marketing

Ziprasidone is not marketed anywhere in the world.

3.0 Chemistry, Manufacturing and Controls

The chemical structure for ziprasidone is:



4.0 Animal Pharmacology and Toxicology

In vitro studies have shown that ziprasidone is a serotonin $(5-HT_{2A})$ and dopamine type 2 (D₂) antagonist. Other receptor effects of ziprasidone include histamine (H₁), α_1 adrenergic, $5-HT_{1A}$ (agonist) $5HT_{1D}$ (antagonist), and $5HT_{2C}$ (antagonist) receptor affinity. There is evidence that ziprasidone blocks neuronal reuptake of serotonin and norepinephrine, and inhibits the contractile effect of norepinephrine on guinea pig aortic strips. In animal models, ziprasidone demonstrated potential for antipsychotic properties. The suggestion that it would have less severe motor side effects than traditional antipsychotics was supported by its weak ability to induce catalepsy in animals; its potential to treat negative symptoms was theoretically demonstrated in its ability to increase the release of dopamine in rat prefrontal cortex.

Reproductive studies in rabbits resulted in decreased birth weight, decreased birth survival, decreased maternal weight gain, decrease in the number of viable litters, and abnormal fetal heart development. Ziprasidone has been shown to exhibit placental transfer in rats and rabbits. In chronic toxicology studies, sedation and reduced body weight gain were observed in rats and dogs. Also observed in dogs were motor side effects, and intrahepatic cholestasis (correlated with increases in ALT and alkaline phosphatase).

The mouse carcinogenicity studies showed dose related increases in the incidence of pituitary adenomas and mammary gland adenocarcinomas.

In the opinion of FDA consultants, the safety preclinical pharmacology evaluation of the original NDA submission did not include an adequate work up of ziprasidone's effect on the QT interval.

In a meeting package (2/13/98), the sponsor concluded that ziprasidone and the metabolite ziprasidonesulfoxide did not demonstrate significant effects on the action potentials of Purkinje fibers in dogs; FDA cardiology consultants (HFD-110) expressed concern that the sponsor did not test a high enough concentration of ziprasidone in this study to fully characterize the effects of ziprasidone in the therapeutic dosage range (note: the sponsor has not submitted a study report for review). In a meeting with FDA and the sponsor (3/27/98), the sponsor presented a brief summary of data which suggested that ziprasidone may inhibit the IKr channel, an ion channel implicated in the process of QTc prolongation (no data was submitted for review).

5.0 Description of Clinical Data Sources

5.1 Primary Source Data (Development Program)

5.1.1 Study Type and Design/Patient Enumeration

The sponsor submitted three different calculations to characterize the subject exposure history of ziprasidone, and there was some discrepancy amongst the tables presented by the sponsor; this review will attempt to clarify these submissions. The following table (adapted from the sponsor's submission of 3/20/98) gives a summary of person time in the ziprasidone safety data base:

ORIGINAL NDA	ZIPRASIDONE	PLACEBO	HALOPERIDOL	RISPERIDONE
N=	2163*	366	407	206
Subject-years exposure*	626*	51	86	84
SAFETY UPDATE				
N=	2588	382	585	295
Subject-years exposure*	772	52	131	105
REPORT of 12/31/98				
N=	2993	424	653	298
Subject-years exposure*	1189	82	228	155

Subject-years exposure in ziprasidone safety data base*

* Includes Studies 105 and 120 which were not in the integrated safety data base.(see below for details)

Discrepancies in the total number of subjects exposed is explained by the fact that the sponsor did not consistently exclude or include studies 105 (11 elderly demented subjects) and study 120 (12 psychotic subject taking the IM formulation). Studies 105 and 120 were not included in the integrated safety data base. Therefore, in the original submission (submitted: 3/18/97;cut off date: 10/31/96), there were actually a total of **2140** subjects enrolled in phase II/III oral ziprasidone studies in the integrated safety data base. This is the number of subjects upon which the demographic characteristics presented below (section 5.1.2) is based.

Appendix Table 5.1.1.1 lists the cumulative number of subjects in the original integrated safety data base and the safety update with a cut-off date of May 15, 1997. The safety data base (submission of 8/29/97) for Phase I, II, and III of ziprasidone trials included a total of 3318 subjects exposed to ziprasidone. There were 742 subjects in Phase I studies and 2565 subjects in the Phases II/III studies; this doesn't include study 105 in which 11 elderly subjects with dementia were exposed to oral ziprasidone. One other trial (study 120) for the IM formulation with 12 subjects was included in the enumeration of subjects in the original submission, but the sponsor did not include this data in the integrated safety data base for the oral ziprasidone NDA.

Please refer to Appendix 5.1.1.2 for a listing of all studies. The integrated safety data base encompasses 29 Phase II/III studies. The majority of subjects were enrolled in controlled studies.

5.1.2 Demographics

Please refer to Appendix 5.1.2.1 for a demographic profile of all Phase I studies. The sponsor did not recalculate demographics based on the additional information in the safety update for phase I trials.

All demographic information for the cumulative Phase II/III safety data base as of 8/29/97 can be found in Appendix 5.1.2.2.

These tables show that the majority of subjects in Phases I and II/III were Caucasian males between the ages of 18-64 years old.

5.1.3 Extent of Exposure (dose/duration)

The modal daily dose and duration for Phase I studies are shown in Appendix 5.1.3.1 (note: these figures are based on the original submission; the sponsor did not provide additional information in the safety update for phase I trials). This table reflects that the majority of subjects were exposed to low doses (< 100 mg daily) for less than 30 days which is not unusual for Phase I studies.

Appendix 5.1.3.2 is a table of the mean daily dose and duration during all oral dosing in Phase II/III studies (including data from the safety update of 9/29/97). There have been 1686 subjects (65.7%) within this pool who have been exposed to ziprasidone in the dosage range of 80 to 160 mg daily which is the recommended dose in the proposed labeling. There were 533 subjects (20.8%) exposed to ziprasidone for six months or longer. The following table (adapted from the sponsor's safety update, and presented above) gives a summary of person time in the ziprasidone safety data base:

ORIGINAL NDA	ZIPRASIDONE	PLACEBO	HALOPERIDOL	RISPERIDONE
N=	2163	366	407	206
Subject-years exposure*	626	51	86	84
SAFETY UPDATE				
N=	2588	382	585	295
Subject-years exposure*	772	52	131	105

Subject-years exposure in ziprasidone safety data base

Med - Page 34 of 223

REPORT of 12/31/98				
N=	2993	424	653	298
Subject-years exposure*	1189	82	228	155

• Includes Studies 105 and 120 which were not in the integrated safety data base. (see Section 5.1.1 for details)

The sponsor recorded the cut off date as 10/31/98 for the original submission. The cut-off date for the safety update calculations (submitted 8/29/97) was not made clear by the sponsor despite requests to do so. The sponsor reported that the subject-years exposure calculations for the REPORT of 12/31/98 was 12/31/98.

Please note that the sponsor included subjects from blinded groups to calculate the subject-years exposure; they did not explain their procedure despite requests to do so.

5.2 Secondary Source Data

5.2.1 Other Studies

The studies conducted in Japan were not included in the integrated summary data base, but safety data including discontinuations, adverse events, and laboratory test abnormalities was included in this submission. The cut-off date, including the safety update, is May 15, 1997. The Japanese studies include the following:

Phase I Studies (n= ziprasidone exposure)

93-501 (n=13) 93-502 (n=8) 93-503 (n=6) 93-504 (n=10) 94-501 (n=6) 96-501 (n=25)

Phase II Studies 94-601 (n=49) 95-601 (n=84)

The eight Phase I and three Phase II Japanese trials included 201 subjects as of the safety update of May 15, 1997 who were exposed to ziprasidone; the sponsor calculated that this represents 14.8 subject-years exposure:

5.2.2 Postmarketing Experience

As of September 4, 1997, Ziprasidone is not marketed in any country as per a teleconference with Dr. Ritrovato at Pfizer.

5.2.3 Literature

According to a teleconference of September 4, 1997 with Dr. Ritrovato from Pfizer, all clinical studies have been done under Pfizer's sponsorship and are included in the current NDA submission.

The sponsor has submitted nineteen published papers and abstracts (NDA Vol. 153 and 154) that either focused on or contained new information about ziprasidone. Publications which make only a reference to ziprasidone were not submitted; the cut-off for the bibliography was listed as October 31, 1996. The literature search was conducted by David L. Larson, Ph.D. who has been employed at Pfizer since 1971.

Of note in this literature review, there were two studies (Bench, 1996; Bench, 1993) which reported large elevations of prolactin levels when comparing baseline and peak plasma levels of ziprasidone in twelve normal volunteers taking between 5 and 60 mg of a single dose of ziprasidone.

Otherwise, my review of the sponsor's literature search did not reveal any unexpected safety findings.

6.0 Human Pharmacokinetic Considerations

For complete details, please refer to the biopharmaceutics review.

Oral ziprasidone is absorbed up to 100 % when administered in the fed state. In the fed state, the mean half-life is 6.6 hours (variability ranging from 3 to 18 hours) and a steady state is achieved within 1 to 3 days. In addition to increasing the AUC and Cmax, food was found to delay the Cmax and decrease the half-life by approximately 4 hours. Ziprasidone is highly protein bound with an absolute bioavailability of 60 % in the fed state. The mean volume of distribution is approximately 1.5 L/kg with a mean systemic clearance of approximately 7.5 ml/min/kg. Ziprasidone demonstrates linear kinetics in the fasting state. In an interim report of study 044, a single dose pharmacokinetic study in 15 pediatric subjects with Tourette's Syndrome (ages 7-16; n=15), preliminary finding showed that the half-life range was 3.3-4.7 hours (note: an oral suspension of 40 mg/ml was used in study 044).

The major metabolites identified are ziprasidone-sulfoxide and ziprasidone-sulfone; both demonstrate a low affinity to D_2 and $5HT_{2A}$ receptors. In vitro studies of human liver microsomes suggest that ziprasidone is a cytochrome P450 3A4 substrate mainly for the metabolic processes of sulfur oxidation and N-dealkylation. Excretion was determined to be 20% in the urine and 66% the in feces.

Study 028 showed that there was a longer half-life in the elderly ($t_{x}=5.5$) than in the younger adults ($t_{x}=3.5$). The differences in Cmax and AUC were less than 15 % between the elderly and younger adults; elderly men had similar values to the young adult males and females, but elderly women displayed a higher AUC (23%) and Cmax(44%). In the proposed labeling, the sponsor concluded that a dose adjustment was not necessary for age or gender.

Study 026 examined the pharmacokinetic differences in subjects with renal impairment compared to subjects with normal renal functioning. The sponsor found that ziprasidone levels were not affected by hemodialysis and that there was no statistical significance seen in the AUC and Cmax when comparing normal subjects with renally impaired subjects. The sponsor concluded that renal impairment does not alter the pharmacokinetic properties of ziprasidone; however, the biopharmaceutics review (HFD-860: 3/3/98) makes note that the sponsor used only a dosage of 20 mg bid of ziprasidone in this studies when the proposed labeling recommends 40 mg bid as the minimum dosage.

Study 030 compared pharmacokinetic properties in subjects with hepatic cirrhosis and subjects with normal hepatic functioning. There was a higher mean half-life in the subjects with cirrhosis $(t_x=7.1)$ compared to the normal controls $(t_x=4.8)$. Otherwise, the sponsor did not find statistical significance in the Cmax and Tmax between the groups by day 5. The sponsor's proposed labeling states that impaired liver functioning does not appreciably affect ziprasidone's pharmacokinetic properties; however, as with the renal impairment study above, the sponsor used the dose of 20 mg bid ziprasidone in this study and did not test the recommended minimum dose of 40 mg bid.

Theoretically, inducers of CYP3A4 (e.g. carbamazepine) may decrease ziprasidone exposure while inhibitors (e.g. cimetidine, ketoconazole) may increase ziprasidone levels. The sponsor's studies showed that concomitant use of carbamazepine resulted in a < 40 % decrease of ziprasidone AUC and Cmax; however, it was noted in the Biopharmaceuticals Review (3/3/98) that the sponsor used the dose of 200 mg bid (for 21 days) instead of a dose in the recommended dosage range (800 to 1200 mg qd carbamazepine); this suggests that ziprasidone levels may be even further reduced when administered concomitantly with a
therapeutic maintenance dose of 800 to 1200 mg qd carbamazepine. In vivo study 050 demonstrated that the concomitant use of ketoconazole, a potent CYP3A4 inhibitor, resulted in approximately a 30 percent increase of both AUC and Cmax of ziprasidone over placebo, suggesting that ziprasidone may have some potential to inhibit the CYP3A4 isozyme.

no-

Concomitant use of the cimetidine and aluminum/magnesium antacids did not show clinically significant _ interactions with ziprasidone. Ziprasidone was shown to have no statistically significant change in the pharmacokinetics of dextromethorphan (a CYP2D6 substrate), ethinyl estradiol (a CYP3A4 substrate), and did not affect the steady state or renal clearance of lithium.

7.0 Review of Efficacy

7.1 Background

Pfizer reports they have six well controlled studies testing the effectiveness of ziprasidone in treating the psychotic symptoms of schizophrenia and schizoaffective disorders. They consider that four of these studies (106, 114, 115, and 303) are adequate to support the efficacy of ziprasidone, and that the other two studies (104 and 106) provide relevant data to their claims of effectiveness. This review will discuss the following studies which are all randomized, double blind, placebo controlled multicentered trials in subjects diagnosed with schizophrenia or schizoaffective disorder:

- Study 106, n=139 total, comparing ziprasidone 20 mg bid, 60 mg bid, and placebo, 4 weeks
- Study 114, n=302 total, comparing ziprasidone 40 mg bid, 80 mg bid, and placebo, 6 weeks
- Study 115, n=419 total, comparing ziprasidone 20 mg bid, 60 mg bid, 100 mg bid, and placebo, 6 weeks
- Study 303, n=294 total, comparing ziprasidone 20 mg bid, 40 mg bid, 80 mg bid, and placebo, 52 weeks

Study 104, n=200 total, comparing ziprasidone 5 mg bid, 20 mg bid, 40 mg bid, and placebo, 4 weeks.

Study 101 utilized the comparator control of haloperidol and will be briefly summarized, because the sponsor considered this study to be relevant to their labeling claims. There were three other controlled studies in this submission: two haloperidol controlled (studies 109 and 111) and one placebo controlled utilizing subjects with dementia (study 105); these studies were of a small size with thirty-five or less subjects.

A brief summary will be presented of the pilot study 122, an eight week, double-blind, placebo-controlled trial in the pediatric population with Tourette's Syndrome.

7.2 Review of individual studies

7.2.1 Study 106

Investigators/Location

This study was conducted in twelve centers in the United States. Please refer to Appendix 7.2.1.1 for a list of investigators and sites. The sponsor did not provide reasons why f were terminated prior to randomization of any subjects.

Study Plan

Objective(s)/Rationale

The primary objective of this study was to evaluate the safety and efficacy of ziprasidone in treating subjects with an acute exacerbation of schizophrenia or schizoaffective disorder.

Population

Subjects chosen for this study were physically healthy males and females aged 18-64 y.o. with a DSM IIIR diagnosis of chronic or subchronic (less than one year) schizophrenia with acute exacerbation, or schizoaffective disorder (for at least a year). Females of childbearing potential were required to use effective contraception during the study. Baseline scores needed to be at least 37 on the total Brief Psychiatric Rating Scale (BPRS) and at least 4 (moderate) on 2 or more of the BPRS core items (Core items include scorings for conceptual disorganization, suspiciousness, hallucinatory behavior, and unusual thought content.). Also required was a score of > 2 on the Clinical Global Impressions Improvement scale (CGI-I) assessed at baseline in comparison to the Clinical Global Impressions-Severity (CGI-S) score previously obtained during the screening period. The protocol allowed investigator discretion for a positive benzodiazepine or cannabinoids result in the urine drug screen; otherwise, it was required to be negative. Excluded from this study were patients with residual schizophrenia, mental retardation, organic mental syndromes, organic mental disorder, brief reactive psychosis, resistance to neuroleptic treatment (i.e. adequate trial of two or more marketed antipsychotics within two years prior to study), comorbid substance abuse/dependence within 6 months, use of depot neuroleptic within 8 weeks of beginning the study, and a high risk for suicide. Concurrent medications allowed during the double-blind trial period included lorazepam, benztropine, and beta-blockers; prohibited medications included other psychotropic drugs, antianginal agents, antiarrhythmics, antinauseants, anticoagulants (except aspirin), steroids, tryptophan, and insulin. If used chronically, antihypertensives, diuretics, hormones (except insulin), oral contraceptives, hypoglycemic agents and Zantac were allowed.

Design

This was a randomized, double blind, placebo controlled 28 day study. After spending four to seven inpatient days in a single-blind placebo washout period, subjects were required to be inpatients for the following 21 days of the double blind placebo controlled study, and were permitted to be inpatient or outpatient for the final 7 days of the study. Psychotropic drugs other than lorazepam and a low dose betablocker were to be discontinued during the washout phase. A history and physical was to be performed during screening; thyroid function tests were performed at screening only. Baseline data, taken at the end of the washout period, would include vital signs (including supine and standing blood pressures), routine laboratory tests, ECG, assessment of abnormal movements (Simpson-Angus Rating Scale, Barnes Scale and AIMS), efficacy instruments (BPRS, CGI, Nurse Global Impression Scale [NGI]), and the Scale for the Assessment of Negative Symptoms (SANS).

Subjects were to be randomly assigned to one of three treatment groups: ziprasidone 20 mg bid, ziprasidone 60 mg bid, or placebo; the group taking ziprasidone 60 mg would be titrated to the target dose within five days (20 mg bid x 2 days, then 40 mg bid x 2 days followed by 60 mg bid x 24 days). Dosing was to occur with mealtime. Serum samples for pharmacokinetic analysis were to be drawn in the mornings of day 1, 7, 14, 21, and 2 hours after the morning dose on days 14 and 21. Repeat laboratory tests were to be drawn prior to dosing on the morning of days 1, 7, 14, 21, and 28. Vital signs were assessed prior to the morning dose on days 1, 2, 7, 14, 21, and 28. ECG and body weight were again recorded prior to the morning dose on days 1, 14, 28. Repeat physical exams were conducted at completion of the study.

The BPRS, CGI, and NGI were scheduled 3 to 7 hours after the morning dose on days 7, 14, 21, and 28; subjects who discontinued prior to completing were to be interviewed within 24 hours of the last dose. The SANS was also repeated on days 14 and 28 or at discontinuation.

Analysis Plan

ć

The primary efficacy variables were defined in the protocol as the mean change in score from baseline to last visit in the following instruments: 1) the BPRS total score, 2) the BPRS core items (suspiciousness, conceptual disorganization, hallucinatory behavior, unusual thought content), and 3) the CGI-S and CGI-I scores.

The protocol states that the BPRS total score was to be analyzed with an analysis of variance with treatment, center, and their interaction. CGI was to be analyzed nonparametrically as a discrete variable.

Study Conduct/Efficacy Outcome

Discontinuations from Study Ziprasidone Protocol 106

Patient Disposition

Of the 203 subjects screened to enter the study, 139 subjects were randomized to one of the three treatment groups and had at least one dosage of double blind treatment. Reasons for not being chosen to participate in this trial were not provided in this submission.

The following table from the sponsor's study report itemizes reasons for discontinuations in the three treatment groups:

	Ziprasidone 20 mg BID	Ziprasidone 60 mg BID	Placebo
Amber of Subjects Randomized	44	47	48
umber of Subjects Discontinued			•••••••••••••••••
eleted to Study Drug	11	10	12
Insufficient clinical response	n	8	12
Adverse event	0	1	0
Laboratory test abnormality it related to Study Drug	<u>د</u> ۷	13	12
Adverse event	°,		
Protocol violation	2	2	,
Last to follow-up		ĩ	à
Withdrawn consent	i	6	ě
Other	i	ž	í
	.	· · · · · · · · · · · · · · · · · · ·	
UTAL	16	23	24

Appendix 7.2.1.2 shows the number of subjects per week who completed each treatment group; the reviewing statistician, Dr. Sue-Jane Wang, explained that the sample size under the "Treated" column and the "Week 1" column differ because there were subjects who did not complete the first seven days of the study. The sponsor defines their intent-to-treat as any subject who had baseline measurements and at least one post-baseline measurement; according to Dr. Wang, the sample size used in the efficacy analysis varied depending on the data available for each efficacy endpoint.

The rate of dropout for the treatment group taking ziprasidone 60 mg bid is almost identical to the profile of dropouts for placebo showing approximately a fifty percent withdraw by the end of the study. However, the group taking ziprasidone 20 mg bid appeared to have slightly fewer dropouts, with a sixty-four percent completer rate.

Demographics /Group Comparability

The majority of patients in this study were Caucasian males with mean ages of approximately 40 years old. The mean ages of the female subjects were higher than the male subjects for all treatment groups. There did not appear to be imbalances in the treatment groups. Appendix 7.2.1.3 shows the breakdown of demographics by treatment group.

The mean baseline values for the BPRS Total Score, BPRS Core Items, CGI -S, and CGI-I were very close, if not identical when comparing placebo with the two treatment groups (please refer to table below). The sponsor did not provide any statistical comparisons of baseline values.

MEAN SCORE			PLACEBO
	20 MG BID	60 MG BID	
BPRS Total	36.5	36.6	37
BPRS Core	13.4	13.6	13.9
CGI-S	4.7	4.7	4.7

Mean Baseline Values of Primary Efficacy Variables

Concomitant Medications

Lorazepam was taken by the majority of subjects in all treatment groups. Please refer to the table below for select concomitant medication use.

Selected	l concomitant med	lication used	in Study 106

	Ziprasidone				
	20 mg bid (n=44)	60 mg bid (n=47)	Placebo (n=48)		
 Lorazepam	36	40	43		
Benztropine	3	9	4		
Beta-Blocker	3	3	2		
Antidepressant	0	1 .	0		
Antipsychotic	2	- 1	2		
Antihistamine	1	6	3		

During the double-blind trial, lorazepam was used with a mean total cumulative dosage of 53 mg for the 36 subjects in the 20 mg bid group; 43 mg of lorazepam was the mean usage in the 60 mg bid group, while the placebo group had a mean usage of 30 mg. Benztropine was taken by 3 subjects in the 20 mg bid group with a mean dose of 15 mg during the trial; 14 mg was the mean dose for the 60 mg bid group, and the placebo group used a mean dosage of 5.5 mg.

Efficacy Results

Please refer to Appendix Tables for results of the important outcome measures (BPRS Total, BPRS Core, CGI-S, and SANS). The ziprasidone 60 mg bid group was the only group that showed statistical significance in week 4 for the primary efficacy variables of the BPRS Total and the CGI-S for both OC and LOCF when compared with placebo. However, neither of the treatment groups provided statistical

significance with the BPRS Core or the SANS with a 95% confidence interval when compared with placebo.

Miscellaneous Issues

Serum plasma levels of ziprasidone were obtained during this study. On day 21, the mean trough level for the ziprasidone 20 mg bid group was 18.6 ng/ml, and for the ziprasidone 60 mg bid group, the mean level was 56.4 ng/ml.

The sponsor performed an interim analysis which they state did not modify the design of the study.

Conclusions

Because this study showed statistical significance in only two of the three primary efficacy variable in week four only, it merely provides fair evidence for the antipsychotic properties of ziprasidone at a dose of 60 mg bid.

7.2.2 Study 114

Investigators/Location

This study was conducted in thirty-four centers in the United States and Canada. Please refer to Appendix Table 7.2.2.1 for a list of investigators and sites.

• •

Study Plan

Objective(s)/Rationale

The primary objective of this study was to evaluate the safety and efficacy of ziprasidone in treating subjects with an acute exacerbation of schizophrenia or schizoaffective disorder.

Population

Entrance criteria were similar to Study 106 with the exception that this study required a baseline score of > 59 for the total score of the Positive and Negative Syndrome Scale (PANSS). Please refer to Study 106 for a list of concurrent medications permitted during the double-blind trial period.

Design

This was a randomized six week, double blind, placebo controlled study. Subjects who met entry criteria were required to undergo an inpatient single-blind placebo washout period for 3-7 days followed by 14 days of inpatient double-blind treatment. During the remaining 28 days of the study, subjects could be inpatient or outpatient during which time they would be evaluated at weekly visits. A history and physical was to be performed during screening; thyroid function tests were performed at screening only. Baseline data, taken at the end of the washout period, would include vital signs (including sitting and standing), routine laboratory tests, ECG, assessment of abnormal movements (Simpson-Angus Rating Scale, Barnes Scale and AIMS), and efficacy instruments (PANSS, CGI, and the Montgomery-Asberg Depression Rating Scale [MADRS]).

Once chosen for the study, subjects were to be randomized to one of three treatment groups: 1) ziprasidone 40 mg bid, 2) ziprasidone 80 mg bid 3) placebo. The double-blind medication was to be administered orally two times a day (dosing spaced about 12 hours apart) with food. Titration occurred over a period of

3 days for the 80 mg bid group (40 mg x 2 days, then 80 mg x 40 days). Serum samples for pharmacokinetic analysis were to be drawn pre-dosing of days 7,14, 42, or at early termination. Repeat laboratory tests were to be drawn on days 7, 21, 42, or at early termination. Vital signs were assessed on days 7, 14, 21, 28, 35 and 42. ECG were again recorded on days 14 and 42 or at early termination. Repeat physical exams were conducted at completion of the study.

The PANSS and the CGI-S were to be given weekly, and the MADRS would be repeated on days 7, 14, 21, and 42. Subjects who discontinued prior to completing were to be interviewed within 24 hours of the last dose.

Analysis Plan

The primary efficacy variables specified in the protocol were the total score of the BPRS derived from the PANSS (BPRSd), BPRSd core items (conceptual disorganization, suspiciousness, hallucinatory behavior, and unusual thought content) and the CGI-S. Secondary efficacy variables included the PANSS-total, PANSS negative sub-scale, the CGI-I, and the MADRS.

The protocol stated that linear models will be fitted to the primary efficacy variables analyzing baseline values and treatment centers. The protocol also states that an analysis of discrete or categorical data may be used as an alternative method.

Study Conduct/Efficacy Outcome

Patient Disposition

Of the 440 subjects screened to enter the study, 302 subjects were randomized to one of the three treatment groups and had at least one dosage of double blind treatment. Reasons for not being chosen to participate in this trial were not provided in this submission.

The following table from the sponsor's study report itemizes reasons for discontinuations in the three treatment groups:

Discontinuations from Study Ziprasidone Protocol 114

Ziprasi	done 40 mg BID	Ziprasi	done 80 mg BID	•••••	Placebo
	•••••		· · · · · · · · · · · · · · · · · · ·		•• • • • • • • • • • • • • • • • • • • •
	106		104		92
	•••••••••••••••	• • • • • • • • • • •	•••••	•••••	••••••••••
			/mm	••	
				JZ	(34.8)
26	(24.5)	16	(15.4)	32	(34.8)
1	(0.9)	7	(6.7)	0	(0,0)
25	(23.6)	14	(13.5)	15	(16.3)
1	(0,9)	1	(1.0)	1	0.1)
3	(2.8)	ī	(1.0)	ĩ	<u>a.n</u>
6		ž		š	(3.3)
ō	(0.0)	3		ī	an
15	(14.2)	6		Ā	(8.7)
Ū.	(0.0)	i	(1.0)	ī	alii
	(48 1)	••••		47	(51.1)
	27 26 25 1 3 6 0 15 0	106 27 (25.5) 26 (24.5) 1 (0.9) 25 (23.6) 1 (0.9) 3 (2.8) 6 (5.7) 0 (0.0) 15 (14.2) 0 (0.0)	106 27 (25.5) 23 26 (24.5) 16 1 (0.9) 7 25 (23.6) 14 1 (0.9) 1 3 (2.8) 1 6 (5.7) 2 0 (0.0) 3 15 (14.2) 6 0 (0.0) 1	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

Appendix 7.2.2.2 shows the number of subjects per week who completed each treatment group; the reviewing statistician, Dr. Sue-Jane Wang, explained that the sample size under the "Treated" column and the "Week 1" column differ because there were subjects who did not complete the first seven days of the study. The sponsor defines their intent-to-treat as any subject who had baseline measurements and at least one post-baseline measurement; according to Dr. Wang, the sample size used in the efficacy analysis varied depending on the data available for each efficacy endpoint.

The dropout rate was lowest for the treatment group taking ziprasidone 80 mg bid with a sixty-four percent completer rate. The withdrawal rate for both the placebo and ziprasidone 40 mg bid treatment group was approximately fifty percent by the end of the study.

Demographics /Group Comparability

The majority of patients in this study were Caucasian males with mean ages of approximately 36 years old in all groups. The mean ages of the female subjects were higher than the male subjects for all treatment groups. There did not appear to be imbalances in the treatment groups. Appendix 7.2.2.3 shows the breakdown of demographics by treatment group.

The mean baseline values for the BPRSd Total Score, BPRSd Core Items, CGI-S, and PANSS were comparable (please refer to table below). The sponsor did not provide any statistical comparisons of baseline values.

MEAN SCORE	ZI	PRASIDONE	PLACEBO
	40 MG BID	80 MG BID	
BPRSd Total	56.5	55.0	55.1
BPRSd Core	16.9	16.6	16.4
CGI-S	4.8	4.8	4.8
PANSS total	98.2	95.8	97.3
PANSS neg.	25.4	24.3	24.9

Mean Baseline Values of Primary Efficacy Variables

Concomitant Medications

Lorazepam was taken by the majority of subjects in all treatment groups. Please refer to the table below for select concomitant medication use.

Selected concomitant medication used in Study 114

	Zip	rasidone	
	40 mg bid	80 mg bid	Placebo
	(n=106)	(n=104)	(n=92)
Lorazepam	90	96	85
Benztropine	21	26	12
Beta-Blocker	5	10	3
Antihistamine	3	3	1
Rx for alcoholism and	2	8	1
drug addiction			

The sponsor did not provide dosages of lorazepam and benztropine use for this-study.

Efficacy Results

Please refer to Appendix Tables for results of the important outcome measures (BPRSd Total, BPRSd core items, CGI-Severity, and PANSS). When compared with placebo at a 95% confidence interval, both ziprasidone treatment groups showed statistical significance by week six for efficacy variables of the BPRSd Total, BPRSd core items, CGI-S, and PANSS total for LOCF. For OC, the only statistically significant result at week six was the CGI-S for the ziprasidone 80 mg bid group.

Miscellaneous Issues

This study included data that was generated from a study site () with the investigators Drs. .

sponsor recalculated the efficacy data of Study 114 excluding data submitted from study site they concluded that the efficacy data was not significantly affected by the exclusion of Dr. data.

This efficacy review of study 114 includes data from site The only noteworthy observation from the analysis of the data without study site is that there was no statistically significant findings at week 6 for the ziprasidone 40 mg bid treatment group.

Serum plasma levels of ziprasidone were obtained during this study. On day 42, the mean trough level for the ziprasidone 40 mg bid group was 47 ng/ml, and for the ziprasidone 80 mg bid group, the mean level was 109 ng/ml.

Conclusions

This study demonstrated statistical significance when comparing both the treatment groups and placebo in the three primary efficacy variables by week six. These results provide evidence that ziprasidone is effective in treating the acute symptoms of psychosis associated with schizophrenia or schizoaffective disorder. Results for the 80 mg bid group were generally superior to the 40 mg bid group results.

7.2.3 Study 115

Investigator(s)/Location

This study was conducted in 54 sites in the United States. Please refer to the sponsor's list of investigators and sites in Appendix 7.2.3.1. The sponsor did not provide reasons for terminating prior to randomization of any subjects.

Study Plan

Objective(s)/Rationale

The primary objective of this study was to assess the safety and efficacy of ziprasidone in treating subjects with an acute exacerbation of schizophrenia or schizoaffective disorder as compared to placebo and haloperidol.

Population

Please refer to Study 114 which had the same entrance criteria. Concurrent medications were similar to those used in previous studies.

Design

This was a randomized, double-blind, placebo controlled, six week study. The details of this study's design were similar to Study 114 (please refer to Study 114 for more information).

Once chosen for the study, subjects were to be randomized to one of five treatment groups: 1) ziprasidone 20 mg bid, 2) ziprasidone 60 mg bid 3) ziprasidone 100 mg bid, 4) haloperidol 15 mg qd, and 5) placebo. The double-blind medication was to be administered orally two times a day (dosing spaced about 12 hours apart) with food. Titration occurred over a period of 3 days for the 60 mg bid group (40 mg x 2 days, then 60 mg x 40 days) and over a period of 5 days for the 100 mg bid group (40 mg bid x 2 days, then 80 mg

Med - Page 44 of 223

The

bid x 2 days, followed by 100 mg bid x 38 days). Serum samples for pharmacokinetic analysis were to be drawn pre-dosing on days 7,14, 42, or at early termination, and, also, at specified time intervals after dosing (i.e. 1-4, 4-7, or 7-10 hours) on days 14 and 21. Repeat laboratory tests were to be drawn on days 7,14, 42, or at early termination. Vital signs were assessed on days 7, 14, 21, 28, 35 and 42. ECGs were repeated on days 14 and 42 or at early termination. Another physical exam was to be conducted at completion of the study.

The PANSS, CGI-S, and CGI-I were to be given weekly; the MADRS would be repeated on days 7, 14, 21, and 42. Subjects who discontinued prior to completing were to be interviewed within 24 hours of the last dose.

Analysis Plan

The primary efficacy variables were the BPRSd total score (derived from the PANSS), BPRSd core items and the CGI-S. Secondary efficacy variables included the PANSS-total, PANSS negative sub-scale, the CGI-I, and discontinuation status due to lack of efficacy.

The protocol stated that a linear model investigating the dose-response relationship across the ziprasidone and placebo treatment groups would be used. This analysis would use the baseline values of the primary efficacy variables as a covariate and also look at interaction effects between centers and treatment. It is also noted that there were several amendments made to the analysis section of the protocol. Study Conduct/Outcome

Patient Disposition

Of the 567 subjects screened to enter the study, 419 were randomized to one of the three treatment groups and had at least one dosage of double blind treatment. Reasons for not being chosen to participate in this trial were not provided in this submission.

The following table from the sponsor's study report itemizes reasons for discontinuations in the five treatment groups:

Discontinuations from Study Ziprasidone Protocol 115

	Zipresido	ne 20 mg BiD	Zipresid	lone 60 mg BID	Ziprasid	one 100 ag BID
Number of Subjects Randomized		87		78	*******	86
Number of Subjects (%) Discontinued Related to Study Drug Insufficient clinical response Adverse event Laboratory test abnormality Not Related to Study Drug Adverse event Protocol violation Lost to follow-up Doos not meet randomization criteria Withdrawn consent	22 0 15 1 0 3	(25.3) (25.3) (0.0) (0.0) (17.2) (1.1) (0.0) (3.4) (1.1) (11.5) (0.0)	26 22 2 13 1 2 0 1 8 1 8	(33.3) (26.2) (2.6) (2.6) (16.7) (1.3) (2.6) (0.0) (1.3) (10.3) (1.3)	22 18 4 0 16 2 0 1 2 10	(25.6) (20.9) (4.7) (0.0) (18.6) (2.3) (0.0) (1.2) (2.3) (11.6) (11.2)

Discontinuations from Study Zipresidone Protocol 115

	Ha	loperidol	Pi acebo	
Number of Subjects Randomized		85		63
Number of Subjects (%) Discontinued Related to Study Drug Insufficient clinical response Adverse event Laboratory test abnormality Not Related to Study Drug Adverse event Protocool violation Lost to follow-up Does not meet randomization criteria Withdrawn consent Other	19 13 6 0 18 1 0 2 1 13 13	(22.4) (15.3) (7.1) (0.0) (21.2) (1.2) (0.0) (2.4) (1.2) (15.3) (1.2)	35 35 0 21 3 2 1 1 14 0	(42.2) (42.2) (0.0) (25.3) (3.6) (2.4) (1.2) (1.2) (1.6) (0.0)
TOTAL	37	(43.5)	56	(67.5)

Appendix 7.2.3.2 shows the number of subjects per week who completed each treatment group; the reviewing statistician, Dr. Sue-Jane Wang, explained that the sample size under the "Treated" column and the "Week 1" column differ because there were subjects who did not complete the first seven days of the study. The sponsor defines their intent-to-treat as any subject who had baseline measurements and at least one post-baseline measurement; according to Dr. Wang, the sample size used in the efficacy analysis varies depending on the data available for each efficacy endpoint.

The completer rate was highest for the ziprasidone 60 mg bid group (64%), and, as expected, lowest in the placebo group (32%). The other three groups had approximately a 45% withdraw by the end of the study.

Demographics/Group Comparability

The majority of patients in this study were Caucasian males with mean age of approximately 40 years old. There did not appear to be any imbalances in the treatment groups. Appendix 7.2.3.3 shows the breakdown of demographics by treatment group.

The mean baseline values for the BPRSd total score, BPRSd core items, CGI -S, and PANSS were comparable (please refer to table below). The sponsor did not provide any statistical comparisons of baseline values.

MEAN SCORE		ZIPRASIDO	NE	PBO	HALDOPERIDAL
	20 MG BID	60 MG BID	100 MG BID		15 MG
BPRSd Total	53.8	51.8	51.8	54.3	53.9
BPRSd Core	16.1	16.0	15.9	16.6	16.2
CGI-S	4.9	4.9	4.7	4.9	5.0
PANSS total	93.2	90.4	89.5	93.3	94.1
PANSS neg.	22.9	23.4	22.5	22.4	24.1

Mean Baseline Values of Primary Efficacy Variables

Concomitant Medications

Lorazepam was taken by the majority of subjects in all treatment groups. Please refer to the table below for select concomitant medication use.

NDA 20-825: Ziprasidone po Clinical Review - 17

Selected	i conc	omitant	medication	used in	Study 115	

	·	Ziprasidone			
	20 mg bid (n=87)	60 mg bid (n=78)	100 mg bid (n=86)	15 mg (n=85)	Placebo (n=83)
Lorazepam	76	69	75	76	75
Benztropine	24	15	36	43	26
Beta-Blocker	7	7	7	16	6
Antidepressant	0	0	0	0	0
Antipsychotic	2	4	1	5	9
Antihistamine	0	2	· 0	1	0

The sponsor did not provide dosages of lorazepam and benztropine use for this study.

Efficacy Results

Please refer to Appendix Tables for results of the important outcome measures (BPRSd Total, BPRSd Core, CGI-S, and PANSS). All treatment groups showed statistical significance by week 6 for efficacy variables of the BPRSd Total, BPRSd core items, CGI-S, and PANSS Negative total for LOCF, but not for OC when compared to placebo at a 5% significance level. In week six, the PANSS negative showed statistical significance for LOCF in the ziprasidone 100 mg bid and haloperidol groups only.

Miscellaneous Issues

Serum plasma levels of ziprasidone were obtained during this study. For the ziprasidone 20 mg bid treatment group, the mean trough serum concentrations were 28 ng/ml; the mean trough levels for the ziprasidone 60 mg bid group was 62 ng/ml, and the ziprasidone 100 mg bid treatment group had a mean trough level of 111 ng/ml.

The following table from the sponsor's submission lists the pharmacokinetic parameters established as a result of 825 samples from 237 subjects:

Pharmacolonetic Parameter		Estimated Value	Standard error	% CV
Systemic clearance	CUF (Lhr)	53.5	1.25	2.34
Volume of central compartment	V _c F (L)	381	65.2	17.1
Absorption coefficient	K (1/h)	0.661	0.147	22.2
Volume of peripheral compartment	V.F.(L)	1430	951	66.5
Intercompartmental dearance	Q(1A)	76.5	10.8	14.1
Absorption lag time	LAG (m)	1.41	0.012	0.850

The sponsor claims that a dose response relationship among the three ziprasidone treatment groups and placebo were found to be statistically significant for all primary efficacy variables. This dose response was found when the ziprasidone treatment groups were compared to placebo. However, when comparing the ziprasidone treatment groups with each other, it does not appear that increasing doses yielded higher scores on efficacy variables.

Also of note is that the number of centers was increased to 54 where as the protocol estimated 30-40 centers.

Conclusions

This study demonstrated statistical significance when comparing all the treatment groups with placebo in the three primary efficacy variables at week six. These results provide evidence that ziprasidone is effective in treating the acute symptoms of psychosis associated with schizophrenia or schizoaffective

Med - Page 47 of 223

disorder. Results were generally similar between ziprasidone dose groups. The haloperidol arm results (including negative symptoms) were numerically superior to the ziprasidone groups.

This pharmacokinetic data supports the sponsor's claim that ziprasidone follows linear kinetics.

7.2.4 Study 303

Investigator(s)/Location

Study Plan

Objective(s)/Rationale

The objectives of this fifty-two week study were to assess relapse prevention of psychotic episodes in hospitalized subjects with schizophrenia and to assess the safety and efficacy of ziprasidone compared to placebo.

Population

Subjects chosen for this study included physically healthy males and females aged 18 years or older who had been hospitalized at least two months prior to entry into the study with a primary diagnosis of chronic or subchronic schizophrenia. Females of childbearing potential were required to use methods of birth control that include an IUD, implanted or oral contraceptive methods. A baseline score of five or less on the CGI-S was required for inclusion in this trial. The protocol allowed investigator discretion for positive benzodiazepine or cannabinoids result in the urine drug screen; otherwise, it was required to be negative. Excluded from this study were patients scoring \geq five (moderate severe) on the hostility or uncooperativeness items of the PANSS, with a history of psychosurgery, mental retardation, organic mental syndromes, organic mental disorder, brief reactive psychosis, epilepsy, resistance to neuroleptic treatment (i.e. adequate trial of two or more marketed antipsychotics within two years prior to study), comorbid substance abuse/dependence within 3 months, and a high risk for suicide or homicide. Concurrent medications allowed during the double-blind trial period included lorazepam, temazepam, anticholinergic medication, and beta-blockers. Other psychotropic drugs were prohibited.

Design

This study was a randomized, double blind, placebo controlled, fifty-two week inpatient study. Subjects who met entry criteria were required to undergo an inpatient single-blind placebo lead-in for 3 days followed by 52 weeks of inpatient double-blind treatment. Screening included a history and physical (including sitting and standing vital signs), routine laboratory tests, ECG, and an ophthalmology assessment (including a funduscopy and a slit-lamp examination). Baseline data, taken at the end of the lead-in period included assessment of abnormal movements (Simpson-Angus Rating Scale, Barnes Scale and AIMS) and efficacy instruments (PANSS, CGI, and Global Assessment of Functioning [GAF])

Once chosen for the study, subjects were to be randomized to one of four treatment groups: 1) ziprasidone 20 mg bid, 2) ziprasidone 40 mg bid 3) ziprasidone 80 mg bid, and 4) placebo. The double-blind medication was to be administered orally twice daily postprandially. Titration occurred over a period of 3 days for the 80 mg bid group (40 mg x 2 days, then 80 mg for the remainder of the study). Serum samples for pharmacokinetic analysis and repeat laboratory tests were to be drawn at weeks 4, 12, 28, and 52 or at early termination; thyroid function tests were done at screening and at week 52 only. Vital signs were

assessed at weeks 2, 4, 12, 28, 40, and 52. ECGs were again recorded at weeks 12, 28, 52 or at early termination. Repeat physical exams and ophthalmology assessments were repeated at completion of the study.

The PANSS, CGI, and assessments for abnormal movements were administered at weeks 3, 6, 16, 28, 40, and 52; the GAF was evaluated at weeks 28 and 52.

Analysis Plan

The protocol states that the primary efficacy variable is the measurement of time to impending psychotic relapse. The sponsor defines "impending relapse" as CGI-I score of 6 (much worse) or greater and/or a score of 6 (severe) or greater on either of the PANSS items P7 (hostility) or G8 (uncooperativeness) on two successive days. A score of 5 (minimally worse) on the CGI-I would require ratings to be done on the following three days. If the CGI score remained at 5, then CGI ratings would be performed at weekly intervals until the score improved to 4 or less and then ratings could be performed according to the study schedule.

Other efficacy variables included the PANSS, CGI, and GAF.

The Kaplan-Meier analysis and the Cox proportional hazards model were to be used to analyze the time to discontinuation. ANCOVA models were to be used for continuous and most of the categorical efficacy variables. The analysis plans would look at the interaction effects of treatment groups. Study Conduct/Outcome

Patient Disposition

Of the 351 subjects screened to enter the study, 294 were randomized to one of the four treatment groups and had at least one dosage of double blind treatment. Reasons for not being chosen to participate in this trial were not provided in this submission.

The following tables from the sponsor's study report itemizes reasons for discontinuations in the four treatment groups:

Discontinuations from Study Liprasidone Protocol 303

mber of Subjects Randomized	· · · · · · · · · · · · · · · · · · ·				
moet of seplects weresented	76	72	71	75	
mber of Subjects (%) plscontinued lated to Study prug Insufficient clinical response Adverse event Labora tory test abnormality at Ealated to Study Drug Adverse event Labora tory test abnormality Lost to follow-up Withdrawn consent Special safety test Other	33 (43.4) 27 (35.5) 6 (7.9) 0 (6.0) 9 (11.8) 1 (1.3) 0 (6.0) 0 (6.0) 4 (6.3) 0 (6.0) 4 (6.3)	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	50 (66.7) 43 (57.3) 6 (8.0) 1 (1.3) 11 (14.7) 5 (6.7) 0 (70.0) 1 (1.3) 4 (5.3)	

Appendix 7.2.4.2 shows the number of subjects who completed each treatment group at specified intervals; the reviewing statistician, Dr. Sue-Jane Wang, explained that the sample size under the "Treated" column and the "Week 1" column differ because there were subjects who did not complete the first seven days of the study. The sponsor defines their intent-to-treat as any subject who had baseline measurements and at least one post-baseline measurement; according to Dr. Wang, the sample size used in the efficacy analysis varies depending on the data available for each efficacy endpoint.

The three ziprasidone groups had similar discontinuation rates with 46% of subjects completing both the 20 mg bid and 40 mg bid group, and 48% of subjects completing the 60 mg bid group. As expected, the completer rate was lowest for the placebo group (19%).

Demographics/Group Comparability

All patients were Caucasian; the mean ages was approximately 37 years old. The mean ages of the female subjects were higher than the male subjects for all treatment groups. There did not appear to be imbalances in the treatment groups. Appendix 7.2.4.3 shows the breakdown of demographics by treatment group.

The mean baseline values for the BPRSd total score, BPRSd core items, CGI -S, and PANSS were comparable (please refer to table below). The sponsor did not provide any statistical comparisons of baseline values.

MEAN SCORE		,	PLACEBO	
	20 MG BID	40 MG BID	80 MG BID	
BPRSd Total	46.1	47.1	45.9	48.0
BPRSd Core	11.2	11.7	11.2	11.7
CGI-S	4.0	4.0	4.0	4.1
PANSS total	85.1	86.6	85.2	88.9
PANSS neg.	24.9	24.7	25.0	25.7

Mean Baseline Values of Primary Efficacy Variables

Concomitant Medications

Lorazepam was taken by the majority of subjects in all treatment groups. Please refer to the table below for select concomitant medication use.

			· · ·	
	20 mg bid (n=76)	40 mg bid (n=72)	80 mg bid (n=71)	Placebo (n=75)
Lorazepam	46	39	41	50
Antimuscarinic Drug for	12	. 9	14	10
Parkinsonism				
Beta-Blocker	4	1	2	1
Antidepressant	O	1	0	0
Antipsychotic	0	1	0	1
Antihistamine	0	0	0	1

Selected concomitant medication used in Study 303

The sponsor did not provide dosages of lorazepam and benztropine use for this study.

Efficacy Results

The rate of relapse (as defined in the analysis section above) was lower in the ziprasidone groups (mean of 33%) than in the placebo group (57%). Compared to placebo, the three group showed a statistically significant less risk of relapse in the ziprasidone groups; however, there was no significant difference in the risk when comparing the three different ziprasidone dosage groups. (please refer to Appendix Table for details).

Please refer to Appendix Tables for results of the other important outcome measures (BPRSd total, BPRSd core, CGI-S, PANSS total and PANSS neg.). By week 52, all ziprasidone treatment groups showed

1

statistically significant improvement in these score for LOCF, but not for OC when compared to placebo at a 5% significance level.

Miscellaneous Issues

Serum plasma levels of ziprasidone were obtained during this study. For the ziprasidone 20 mg bid treatment group, the mean trough serum concentrations were 26 ng/ml; the mean trough levels for the ziprasidone 40 mg bid group was 49 ng/ml, and the ziprasidone 80 mg bid treatment group had a mean trough level of 82 ng/ml.

Conclusions

It is debatable if the sponsor has proven that ziprasidone is superior to placebo in the prevention of relapse. Their definition of "relapse" and this study design do not follow traditional methods of proving relapse prevention. There is no presumption that subjects were stable at the beginning of the study nor were subjects selected as responders in an open label study. However, the fact that subjects were hospitalized for two months prior to the start of the study may suggest that subjects had been somewhat stable prior to initiation of the study, and Dr. Wang (HFD 710:11/24/97) concluded that, using the definitions of the sponsor, the ziprasidone treatment group showed a longer time to relapse than placebo that was statistically significant.

The <u>results</u> from the other efficacy variables in this study provide support to the efficacy of ziprasidone in the treatment of psychosis in schizophrenia.

7.2.5 Study 104

Investigators/Location

This study was conducted in seventeen centers in the United States. Please refer to Appendix 7.2.5.1 for a list of investigators and sites. closed prior to randomization of subjects.

Study Plan

Objective(s)/Rationale

The primary objective of this twenty-eight day, double blind, placebo controlled study was to evaluate the safety and efficacy of ziprasidone in treating subjects with an acute exacerbation of schizophrenia or schizoaffective disorder.

Population

Subjects chosen for this study were physically healthy males and females aged 18-64 y.o. Please refer to Study 106 for details of the entrance criteria and a list of concurrent medications allowed during the double-blind trial period.

Design

This is a randomized double-blind placebo controlled 28 day study. For details of the general study design and concomitant medications, please refer to Study 106.

After the lead-in phase, subjects were to be randomly assigned to one of four treatment groups: ziprasidone (CP-88,059-1) ziprasidone 5 mg bid, 2) ziprasidone 20 mg bid, 3) ziprasidone 40 mg bid, or 4) placebo;

Junder

the group taking ziprasidone 40 mg would be titrated to the target dose within four days (20 mg bid x 3 days, then 40 mg bid for the remainder of the study).

Analysis Plan

The primary efficacy variables were defined as the mean change in score from baseline to last visit in the following instruments: 1) the BPRS total score, 2) psychotic core items, and 3) the CGI-S.

The protocol states that the BPRS total score was to be analyzed with an analysis of variance with treatment, center, and their interaction. CGI was to be analyzed nonparametrically, as a discrete variable.

Other efficacy variables included the Nurses Global Impression (NGI).

Study Conduct/Efficacy Outcome

Patient Disposition

Of the 303 subjects screened to enter the study, 200 subjects were randomized to one of the four treatment groups and had at least one dosage of double blind treatment. Reasons for not being chosen to participate in this trial were not provided in this submission.

The following table from the sponsor's study report itemizes reasons for discontinuations in the three treatment groups:

Discontinuations from Study Zipresidone Protocol 104

	Zipresidone 5eg	BID Zipresidone 20eg BID	Ziprasidone 40eg BID	Placebo
weber of Subjects Rendomized	47	55	48	50
under of Subjects (1) Discontinued lated to Study Drug Insufficient clinical response	11 (23.4) 11 (23.4)	16 (29.1) 16 (29.1)	17 (35.4) 15 (31.3)	17 (34.0) 16 (32.0)
Adverse evont ot related to Study Drug Adverse evont	0 (0.0) 6 (12.8) 2 (4.3)	0.0) 13 (23.6) 0 (0.0)	2 (4:2) 11 (22.9) 0 (0.0)	1 (2.0) 6 (12.0) 1 (2.0)
Laboratory test abnormality Protecol violation Withdrawn consent Other	0 (0.0) 3 (6.4) 6 (0.0) 1 (2.1)	1 (1.8) 5 (9.1) 7 (12.7) 0 (0.0)	0 (0.0) 5 (10.4) 5 (10.4) 1 (2.1)	0 (0.0) 0 (0.0) 4 (8.0) 1 (2.0)

Appendix 7.2.5.2 shows the number of subjects per week who completed each treatment group; the reviewing statistician, Dr. Sue-Jane Wang, explained that the sample size under the "Treated" column and the "Week 1" column differ because there were subjects who did not complete the first seven days of the study. The sponsor defines their intent-to-treat as any subject who had baseline measurements and at least one post-baseline measurement; according to Dr. Wang, the sample size used in the efficacy analysis varies depending on the data available for each efficacy endpoint.

The dropout rate was lowest for the treatment group taking ziprasidone 5 mg bid (36%). The remaining treatment groups demonstrated a dropout rate of approximately fifty percent by the end of the study.

Demographics /Group Comparability

The majority of patients in this study were Caucasian males with mean ages of approximately 40 years old. The mean ages of the female subjects were higher than the male subjects for all treatment groups. There did not appear to be imbalances in the treatment groups. Appendix 7.2.5.3 shows the breakdown of demographics by treatment group. The mean baseline values for the BPRS total score, BPRS core items, CGI -S, and CGI-I had similar scores across all treatment groups (please refer to table below). The sponsor did not provide any statistical comparisons of baseline values.

MEAN SCORE		PLACEBO		
	5 MG BID	20 MG BID	40 MG BID	
BPRS Total	34.1	34.5	36.2	33.4
BPRS Core	12.8	13.0	12.8	13.7
CGI-S	4.9	4.8	4.9	5.0

Mean Baseline Values of Primary Efficacy Variables

Concomitant Medications

Lorazepam was taken by the majority of subjects in all treatment groups. Please refer to the table below for select concomitant medication use.

Selected concomitant medication used in Study 104

	Ziprasidone						
	5 mg bid (n=47)	20 mg bid (n=55)	40 mg bid (n=48)	Placebo (n=50)			
Lorazepam	34	42	39	41			
Benztropine	6	11	6	8			
Beta-Blocker	4	2	1	3			
Antidepressant	0	1	0	0			
Antipsychotic	5	3	3	4			
Antihistamine	1	1	0	3			

During the double-blind trial, lorazepam was used with an mean cumulative dosage of 60 mg for the 34 subjects in the 5 mg bid group; 40.5 mg of lorazepam was the mean usage in the 20 mg bid group, while the 40 mg bid group had a mean dosage of 46 mg of lorazepam. The placebo group had a mean dose of 43 mg of lorazepam. Benztropine was taken by the 6 subjects in the 5 mg bid group with a mean dose of 20.5 mg during the trial; 15 mg was the mean dose for the 20 mg bid group, and the 40 mg bid group took a mean dose of 41.5 mg. The placebo group used a mean dosage of 24 mg of benztropine.

Efficacy Results

Please refer to Appendix Tables for results of the important outcome measures (BPRS total, BPRS core, CGI-S). Statistical significance when compared with placebo was seen only in the ziprasidone 20 mg bid treatment group in the BPRS total and core for OC. Otherwise, no other parameters proved to be statistically significant in this trial.

Miscellaneous Issues

Serum plasma levels of ziprasidone were obtained during this study. At week four, the mean concentration of ziprasidone found in the 5 mg bid group was 3.7 ng/ml; the 20 mg bid group had a mean ziprasidone concentration of 29.7 ng/ml, while the ziprasidone 40 mg bid group had a mean plasma concentration of 34.8 ng/ml.

N

Conclusions

This study did not provide evidence for the efficacy of ziprasidone in the treatment of an acute exacerbation of schizophrenia or schizoaffective disorder. One possible reason for these negative results is that the baseline scores in this study were lower (i.e. subjects may have been less ill) than baseline scores in studies with more positive results.

7.2.6 Other Studies

Study 101 was a four week, double-blind, haloperidol controlled trial conducted in six US centers. Ninety subjects diagnosed with schizophrenia and schizoaffective disorder were randomized to one of 5 groups: 1) ziprasidone 2 mg bid, 2) 5 mg bid, 3) 20 mg bid, 4) 80 mg bid, or 5) haloperidol 15 mg. Efficacy variables included were the BPRS, CGI, NGI, and Nurses Observation Scale for Inpatient Evaluation (NOSIE). In the protocol, a primary efficacy variable was only identified as the changes from baseline to last observation. Results from this study do not support the efficacy of ziprasidone, because no differences were found when comparing each treatment with the lowest dosage group.

7.2.7 Pediatric Studies

Study 122 was a double blind, placebo controlled, 8 week flexible dose (maximum: 20 mg bid ziprasidone) pilot study in 16 pediatric subject (ages 7-16) with Tourette's Syndrome to test the effectiveness of ziprasidone in treating their symptoms of Tourette's Syndrome. Results showed that there was statistical significance seen in one primary efficacy variable (Yale Global Tic Severity Scale), but statistical significance was not demonstrated in the other primary efficacy variable (Clinical Global Impression Severity Scale for TS) when comparing improvement of subjects in the ziprasidone treatment group with placebo.

7.3 Summary of Data Pertinent to Important Clinical Issues

7.3.1 Predictors of Response

When exploring how demographic characteristics may have affected the efficacy data, the sponsor claims to have found no significant effect on treatment based on age, gender, or race. The p-values are not significant for the interaction effects of age (<55 years or \geq 55 years), gender, race (Caucasian or African American) and any of the efficacy variables tested (PANSS total, PANSS negative, BPRS [BPRSd] total, BPRS [BPRSd] core items, CGI-S, CGI-I).

7.3.2 Choice of Dose

In the fifty-two week study 303, all doses (ziprasidone 20 mg bid, 40 mg bid, and 80 mg bid) tested were shown to be efficacious when compared to placebo. However, in study 106, statistical significance was seen in this four week study in the ziprasidone 60 mg bid group, and not in the ziprasidone 20 mg bid group. Study 114 was able to show statistically significant improvement at doses of ziprasidone 40 mg bid and 80 mg bid. The results of study 115 were supportive of efficacy in all doses tested (ziprasidone 20 mg bid, 60 mg bid, and 100 mg bid) at week six. Utilizing results from study 115, the sponsor was able to statistically show a dose response relationship when comparing all doses with placebo; however, the actual numerical differences seen between doses in the efficacy variable scores resembled more of a plateau phenomenon. Therefore, it can not be definitively concluded that increasing the dose clinically would increase the efficacy of this drug.

The drug was administered post-prandial in all studies.

The sponsor has recommended a daily dose of ziprasidone 40 mg bid administered with food. The proposed labeling includes suggestions to increase the dosage to 80 mg bid for some patients and that doses

above 80 mg bid were not shown to be more efficacious than 100 mg bid. This information accurately reflects the findings from these efficacy studies, but the sponsor does not mention that for some patients ziprasidone 20 mg bid proved to be efficacious.

7.3.3. Duration of Treatment

Both six week studies reviewed (114 and 115) demonstrated better efficacy results than the four week studies (106 and 104). It appears that the longer trials of six weeks had a better treatment outcome for psychotic symptoms. The proposed labeling does not address the issue of what time period constitutes a sufficient trial.

Study 303 shows that, after 52 weeks, this drug continues to be effective in the treatment of psychosis. As discussed previously, this trial is not useful with respect to evaluating relapse prevention.

7.4 Conclusions Regarding Efficacy Data

Ziprasidone has been proven to be efficacious in the treatment of acute exacerbation of schizophrenia and schizoaffective disorder in more than one well controlled study. Results of the a year long study have shown that responders of ziprasidone in this trial had fewer psychotic symptoms compared to placebo, but it is unclear if the sponsor's definition of prevention is accurate enough to make a claim of relapse prevention.

8.0 Integrated Review of Safety

8.1 Methods and Findings for Safety Review

The sponsor submitted the integrated safety data base of Phase II/III studies and a safety update for review. The electronic submission did not include the ability to manipulate safety data and the review was dependent on the sponsor's compilation of data into tables. The safety update and Integrated Safety Summary (ISS) did not always include needed summary information and a careful review of the text was required. There was some additional information from Japanese studies which was not a part of the integrated safety data base, but is referred to when a serious event emerged. The main focus of the review was on the Phase II/III integrated safety data base to identify significant adverse events. To examine for common adverse events, more emphasis was placed on safety data pooled from the placebo controlled studies which were of similar duration (Studies 104, 106, 114, 115) to allow for comparator control. The only other placebo controlled study was the 52 week study 303 which was examined within the integrated safety data on events occurring over a longer period of time.

Amongst the data submitted, there were several subjects listed as being in blinded groups in the safety update; the sponsor was able to reveal the specific treatment group when it was requested. Subjects in blinded groups were included in the calculations of subject years exposure.

There were 2140 subjects exposed to ziprasidone that were included in the original submission (cut-off date 10/31/96); data on an additional 425 subjects taking ziprasidone was in the safety update (cut off date 5/15/97). Therefore, the total integrated safety data base recounted the experience of 2565 psychotic patients who were medicated with ziprasidone. When calculating subject-years exposure, the sponsor chose to include the additional studies Study 120 (for dementia: n=12) and Study 105 (IM ziprasidone: n=11) despite their not being part of the integrated safety data base. Including these additional studies, the exposure time calculated by the sponsor is 772 subject-years based on a total of 2588 subjects. In the 3/20/98 submission, the sponsor submitted a recalculated patient exposure time with a cut-off date of December 31, 1997 based on 2993 subjects which totaled 1189 subject years of exposure to ziprasidone. For the purposes of this review, the calculation of subjects years exposure will be based on the figure of 772 subject-years (from the safety update of 8/29/97), because the report of December 31, 1997

did not include safety data for review. Please refer to Section 5.0 for further details of subject exposure information.

Individual subjects will be referred to by their study subject number. This review will omit the first three numbers (128) which denote that the study was conducted in the Pfizer Central Research. Also omitted will be the first zero in each subject number. In the subject number presented, the first three digits indicate the number of the study in which the subject participated. Subjects were reassigned numbers when they entered an extension study; whenever relevant, both numbers will be listed.

8.1.1 Deaths

Please refer to Appendix 8.1.1.1 for a full list of all deaths known to have occurred in subjects ever exposed to ziprasidone. As of the sponsor's cut off date of May 15, 1997 in the safety update, there have been a total of 32 subjects taking ziprasidone who have died; this includes one death (Subject JP-95-601162-02) which occurred in a Japanese study which is not considered part of the integrated safety data base. As of the sponsor's submission of 3/20/98, the sponsor reports that there was a total of 35 deaths as of 12/31/97); however, this submission was not accompanied by any supporting data. Therefore, this review can only discuss deaths as of 5/15/97.

Eighteen deaths occurred within 30 days or less of the subjects' discontinuation of ziprasidone, eight of them could be considered as sudden unexpected deaths in which the subjects were either found dead or died within 24 hours of onset of their symptoms associated with death. Of these eight deaths, it seems that only one case (115-6940394) may be determined to be temporally unrelated to the use of ziprasidone; this subject died 29 days after discontinuing ziprasidone and was being treated with risperidone at the time of death. [Note: the sponsor included an additional sudden unexpected death after the May 15, 1997 cut-off (in 12/31/97 report: submitted 3/20/98), but did not submit any data regarding this death; therefore, this additional sudden death will not be discussed in this review]. There were no sudden unexpected deaths in the placebo group in which the subjects were either found dead or died within 24 hours of onset of their symptoms associated with death (the sponsor refers to one sudden unexpected death in the placebo group in the submission of 2/13/98; however, this was unable to be confirmed by review of the safety data base.).

Of the other seven subjects who died suddenly, two subjects died of undetermined causes: 1) Subject 304E-1930379, a 52 year old male with schizophrenia died while taking a nap; 2) Subject 108-5920750, a 39 year old female with schizophrenia was found dead with a cause not clearly determined; the sponsor offered speculation regarding her diabetic and alcohol use history to explain this death. The sponsor's CRFs did not include autopsy reports for these subjects.

Three of the subjects who experienced a sudden death were reported on autopsy to have symptoms related to cardiopulmonary systems: 1) Subject 108-6070305 was a 46 year old male with schizophrenia whose death was attributed to acute and chronic asthmatic bronchitis and granulomatous myocarditis, 2) Subject 105-5340021 was a 70 year old female with dementia who died by cardiopulmonary arrest attributed to arteriosclerotic cardiovascular disease, and 3) Subject 116B-5080001 was a 54 year old male with schizophrenia who was found dead in his hospital bed with an autopsy showing COPD, cardiac hypertrophy, and diffuse atherosclerosis.

One subject (302E-3190375), a 48 year old male with schizophrenia was found dead; he had a history of seizures and polydipsia, and exhibited hypertension and tachycardia during his treatment with ziprasidone. An autopsy was not included in the NDA submission.

Subject 308-0350003, a 63 year old male with schizophrenia collapsed in a "lunch club." The cause of death was determined to be a rupture of an abdominal aortic aneurysm and gross diffuse atherosclerosis.

Another subject that deserves mention, although not meeting the definition of SUD, is subject 301-3110977, a 28 year old female with schizophrenia whose death occurred two days after discontinuing ziprasidone. Upon discharge from the study, this subject had ECG changes consistent with subendocardial ischemia with substernal pinching and was transferred to an internal medicine unit. Her death occurred two days later of unknown causes; her treatment regimen at the time of death included thioridazine and nitrazepam. The CRF did not provide more information than the patient summary and did not include any notes regarding this subjects' transfer to the medicine unit.

There were nine subjects whose death occurred within thirty days of discontinuing ziprasidone in which it appeared to be unlikely that their death was related to exposure to ziprasidone. Two subjects (116B-6590001 and 303-1970299) had been on another antipsychotic for more than three weeks after discontinuing ziprasidone. The remaining seven subjects (108-6090381, 116B-6940004, 117-6870317, 117-7060529, 302-2600156, 302E-1590029, JP-95-6011622) were victims of accidents or suicide.

In the ziprasidone treatment group there were a total of four probable suicides occurring within 30 days or less of discontinuing treatment. There was one subject (106-5520126) in the placebo group who committed suicide eight days after completing the study. There were two reports of suicide in the haloperidol group: 1) Subject 108-05820040 who committed suicide six days after discontinuing haloperidol, and 2) Subject 108-5940564 who overdosed while being treated with haloperidol. There were no subjects known to commit suicide during treatment with risperidone. The rate of suicide in 1000 subject-years for subjects in the ziprasidone group was 5.2. In the haloperidol group, the rate was 15 suicides per 1000 subject years; the rate of suicide for the placebo group was 19 suicides per 1000, and placebo group in this data base had a rate of zero. It is recognized that suicide may be a manifestation of the psychiatric disease under study, and this data does not suggest that ziprasidone increases the risk of suicide.

Appendix 8.1.1.2 gives the mortality rates for subjects in Phase II/III trials of the integrated safety data base who have died during the study or within thirty days of discontinuing the studies. In determining subject-years exposure, the sponsor used 5/15/97 as the cut-off date for deaths, while the cut-off date for the denominator determining the subjects years was sometime prior to 5/15/98 (despite requests, the sponsor did not provide the cut-off date for the denominator of the subject years calculation in the Safety Update of 8/29/97). In this data base, the placebo group demonstrates the highest mortality rate.

In Appendix 8.1.1.3, the rate of sudden unexpected death (SUD) is calculated for this NDA data base. The SUDs rate for the ziprasidone group (n=2565) is 9.1 SUD per 1000 subject years, while placebo (n=382) and haloperidol (n=585) groups had zero and the risperidone group (n=295) had a 9.5 SUD per 1000 rate. It would appear that ziprasidone and risperidone have a similar SUD rate within this data base; it is noted, however, that the risperidone rate is based on a sample size roughly one-tenth the size of the ziprasidone data base and that the SUD rate is based upon one death (due to acute aspiration).

Appendix 8.1.1.4 provides a comparison of the SUD rate of the most recent antipsychotic NDAs **submitted** to FDA; for ziprasidone, the information is based on the data base discussed in this review, while the other information has been obtained from the document: Review of Clinical Data: General Characteristics of the Deaths in the NDAs for Olanzapine, Risperidone, Quetiapine and Sertindole by Greg Burkhart, M.D. (HFD-120: 3/3/98). From this table, it appears that both ziprasidone and sertindole surpass the SUD rate of olanzapine, risperidone, and quetiapine.

In his report Review of Data Quality, Coding, All Cause Mortality and Sudden Deaths (HFD-120: 2/3/98), Dr. Boehm concludes that the SUD rate found in the ziprasidone NDA data base is 6 times higher than the SUD rate from a pool of combined data of recently approved antipsychotic NDAs.

8.1.2 Other Serious Adverse Events

The sponsor did not define a serious adverse event in the Integrated Summary of Safety (ISS). A review of the pivotal study protocols revealed that Pfizer applied the same definition for a serious adverse event that is used by FDA (i.e. any drug experience that is fatal or life-threatening, is permanently disabling, requires hospitalization, or is a congenital anomaly, cancer, or overdose). These protocols included requests that any serious adverse event be immediately reported to Pfizer. Serious adverse events were submitted as listings itemized by subjects and COSTART body system/preferred term.

There were no serious adverse events reported in the Phase I ziprasidone studies. The original submission stated that within the primary safety data base, 261 subjects (12.2%) of the 2140 subjects treated with ziprasidone experienced serious adverse events in Phase II/III studies. Based on a count of the line listings submitted in the update (the sponsor did not provide a summary table of adverse events in the safety update), 115 additional subjects appeared to experience serious adverse events; therefore, 376 subjects (14.7%).of the total 2,565 subjects exposed to ziprasidone in Phase II/III studies of the integrated safety data base experienced a serious adverse event . The sponsor may have presented a single serious adverse event under two or three COSTART terms. It is noted that some individual subjects experienced more than one serious adverse event during these trials.

Appendix 8.1.2 lists serious adverse events considered to be common events in the patient population studied or for which there is not sufficient evidence to state that they were drug related. Any event which occurred greater than 30 days after the study drug was discontinued will not be discussed in this review; in the NDA submission, the sponsor included these events which occurred greater than 30 days after the study drug was discontinued greater than 30 days after the study drug was discontinued. There were several instances in the sponsor's tables in which individual events may have been listed under two or three COSTART terms; in most cases, this review will present them as a single event in only one body system below. Also, there were some instances in which this reviewer felt that alternative COSTART terms reflected the adverse event more accurately than the sponsor's choice of category; the cases which have been recategorized in this review are discussed in section 8.1.5.2. (Note: fatal cases will not be repeated in this section.).

8.1.2.1 Syncope/Hypotension

Because of ziprasidone's alpha adrenergic properties, it not unexpected that orthostatic hypotension and syncope would be present as an adverse event. There were 5 syncopal events and one hypotensive event that were considered to be serious adverse events. It is unclear what criteria the sponsor used to report a syncopal event as a serious event, as syncope was observed in 0.7 % (15/2140) of the subjects in the phase II/III safety data base (cutoff 10/31/96), while hypotension (combining postural hypotension and hypotension) occurred in 2.5 % (53/2140) of subjects in the phase II/III safety data base. For the sake of completeness, in the phase I studies, there were 14 subjects who experienced a syncopal episode of which four experienced a second episode upon rechallenge of ziprasidone. There were at least thirteen episodes of syncope [102-5130005, 102-5140005, , 114-6170217, 116B-5980001, 116B-6940002, 117-6870313, 303-0570124, 303-0710101, 303-2650321, 303-2120222 (source: line listing in Appendix VI table 1b of sponsor's submission) and 108-6050002, 108E-5550096, 108E-5780052 (source: safety narratives)] in the Phase II/III data base that were not reported by the sponsor as a serious adverse event. The following table summarizes the subjects who where determined to have syncope/hypotension as a serious adverse events:

Syncope and Hypotension listed as a serious event in subjects taking ziprasidone

Syncope and Hypoten SUBJECT #	AGE/	MODAL	DAYS	SERIOUS ADVERSE EVENT/ COMMENTS
	SEX	DOSE	OF	
		(MG/D)	TREAT-	
			MENT	
102-5130005	62/M	40	20	Subject fell, breaking left lateral malleolus and
102-5150005	02/141	40	20	ziprasidone was discontinued. 24 hours later,
				•
				subject had two episodes (10 sec.) of loss of
1160 600000				consciousness. Etiology was not determined.
116B-5900002	31/M	160	82	Subject was on ziprasidone when he fell
				backwards with loss of consciousness while
				smoking a cigarette. Subject reported headache the
				night before and dizziness prior to fall. Etiology
•				of syncope was unclear
118-7090001	44/M	40	4	Subject fell backwards and was found to be
				hypotensive (70/50); incident attributed to
				orthostatic hypotension. One day after his last
				ziprasidone dose, subject was found on the floor
	1			unresponsive. He was found to have a sodium
				level of 120 (but his baseline was 127) and he was
				placed on fluid restriction. EEG and CAT scan
		l		were benign. No cardiac work up was done.
116B-5510003	32/M	80	43	On the same day that subject experienced syncopal
	1			episode, he developed a rash and edema in the
				regions of the lips, left hand and ankles. ECG
				showed sinus bradycardia of 52 bpm otherwise it
				was normal. Case also included in rashes.
116B-5230001	41/M	80		Case report form suggests that the syncopal event
1100-5250001	41/141			occurred because the subject had not eaten for a
				couple of days, but emergency work up was not
				located in the CRF.
SUBJECT #	AGE/	MODAL	DAYS	SERIOUS ADVERSE EVENT/ COMMENTS
SUBJECI #				SERIOUS ADVERSE EVENT/ COMMENTS
	SEX	DOSE	OF	
	1	(MG/D)	TREAT-	
T	<u> </u>		MENT	
Hypotension		·~	· · · · ·	
303-0710098	59/M	40	4	Hypotension (95/70), vomiting, hypotension,
				sweating, pallor, dypsnea (h/o diabetes)
				ECG at baseline: QRS:80, PR:160, QT:346, QTc:
				447, rate:100 bpm
	1			ECG at incident (according to CRF): QRS:90,
				QT:326, QTc:473, rate: 123 bpm. with flat t-wave

8.1.2.2 Rash

There were 21 subjects who discontinued for rashes in the originally submitted integrated safety data base; the haloperidol group only had 2 withdrawals, while the placebo group had 1 withdrawal. It was left to the investigator's discretion as to whether or not an event was reported as serious, and there did not appear to be any consistent feature that merited reporting a rash as a serious event. The following table lists the subjects whose rashes were considered serious adverse events.

SUBJECT #	AGE/ SEX	MODAL DOSE	DAYS OF	SERIOUS ADVERSE EVENT/ COMMENTS
		(MG/D)	TREAT- MENT	
106-5520124	41/M	40	15	"Facial rash" developed 9 days after starting ziprasidone. Discontinued due to rash at day 15: diagnosis of sebaceous dermatitis on face arms, thighs, and legs. Subject also had hypertension with blood pressure of 148/112.
106-5520047	46/F	120	29	Raised rash on lips, arms, back and buttocks with pruritis. Biopsy showed superficial perivasculitis with extensive edema and eosinophilia. Treated with antihistamines.
115-6560036	48/M	200	40	Severe generalized rash with itching mostly on the chest and upper extremities. Hospitalized to rule-out Stevens-Johnson Syndrome. Treated with benedryl and calamine lotion. Improvement noted within 2 days of discontinuing ziprasidone.
115-6890088	55/M	120	37	Generalized urticaria. Resolution within 5 days. Treated with prednisone, hydroxizine, and diphenhydramine.
301- 0720148*	54/F	200	33	Urticaria and itching
116B- -5510003	32/M	80	43	Syncope accompanied by a rash and edema in the regions of the lips, left hand and ankles. ECG showed sinus bradycardia of 52 bpm otherwise it was normal. Possible angioedema.

Subjects with rash as a serious adverse ever

*Safety Update

As noted above, there were three subjects who discontinued from the study whose hospitalization was extended because of the symptoms of rash. Rash was the most common adverse event resulting in withdrawal from the ziprasidone treatment groups (see 8.1.3.2). A review of the NDA safety data base revealed that there were several subjects whose rash was accompanied by an elevated white blood count, and at least two subjects with rash whose eosinophil count was elevated. Most cases of rash resolved within one week of discontinuing ziprasidone; one subject (117-5130506) experienced "bullous drug eruptions/ pruritic blisters with post-excoriated papules" on the hands, wrist, scalp, and neck which resolved 24 days after discontinuing ziprasidone. Medications used to treat rashes included steriods (oral and topical) and antihistamines. Three subjects (115-6560036, 106-5520047, and 115-6890088) required hospitalization for observation of their rashes.

8.1.2.3 Elevated Transaminase

It is unclear how the sponsor determined whether or not a laboratory value was considered a serious adverse event; therefore, the reader is referred to section 8.1.6.3 for a listing of subjects who withdrew from studies because of an abnormal laboratory values.

Abnormal transaminase values listed as serious adverse events .

SUBJECT #	AGE/ SEX	MODAL DOSE (MG/D)	DAYS OF TREAT- MENT	SERIOUS ADVERSE EVENT/ COMMENTS
104-5310185	38/M	40	7	Increased ALT and AST at first lab test after starting ziprasidone with peak five weeks (normal at baseline.). This subject was diagnosed with Hepatitis C diagnosed six months after discontinuing ziprasidone. There was a temporal relationship of the elevated LFTs and the initiation of ziprasidone.
106-5390092	35/M	120	9	ALT:165, AST:89U/L. After stopping ziprasidone, ALT returned to normal in 5 days and AST returned to normal in 19 days.
104-5360293	56/F	80	28	Elevated LDH (peak 419)
117-6390290	21/M	80	72	ALT: 133 U/L; AST: 43 U/L LFTs returned to normalized after discontinuing ziprasidone.

8.1.2.4 Neuroleptic Malignant Syndrome

The sponsor did not identify any episodes of neuroleptic malignant syndrome (NMS) in this submission. However, upon review of the case histories, it appeared that there was one 51 year old male subject (109-5650041) who presented classic symptoms of NMS, and a 40 year old female who had symptoms which may have been also been a manifestation of NMS. The following table summarizes these two cases:

Serious events manifesting symptoms of NMS

SUBJECT #	AGE/ SEX	MODAL DOSE (MG/D)	DAYS OF TREAT- MENT	SERIOUS ADVERSE EVENT/ COMMENTS
109-5650041	51/M	80	3	Became confused with temperature of 99.9 ° F, creatinine kinase of 955, and sinus tachycardia. Within two weeks of discontinuing ziprasidone, the CPK levels normalized and the tachycardia resolved within 3 days of ziprasidone discontinuation.
116B- 5530001	40/F	120	830	Episode of odd and confused behavior, a temperature of 101° F, diaphoresis, bradycardia, urinary incontinence and CPK level of 944 U/L. Her husband reported that she had similar episodes in the past. Urine culture showed E. coli infection. The sponsor chose to categorize this serious episode as UTI.

8.1.2.5 Extrapyramidal Symptoms

The following table delineates episodes of extrapyramidal symptoms which were considered serious adverse events.

Serious adverse events of extrapyramidal symptoms

SUBJECT #	AGE/ SEX	MODAL DOSE (MG/D)	DAYS OF TREAT- MENT	SERIOUS ADVERSE EVENT/ COMMENTS
109-5670019	39/M	160	23	Discontinued with ziprasidone due to severe akathisia.
115-6940394	43/M	40	16	Developed tardive dyskinesia 3 days after stopping ziprasidone and was on risperidone at the time of the incident.
116B- 06820005	52/M	120	81	Parkinsonism, restlessness and insomnia Subject took 200 mg qd x 2 weeks and was considered as an overdose.
303-1970265	68/M	40	7	Case report form listed him as having "worsening of vital functions" and was later clarified to be severe extrapyramidal symptoms (asthenia), general weakness, dehydration, hypotension, hypersalivaion. This subject was categorized by the sponsor as having hypotension as a serious adverse event when in fact his blood pressure was consistently low (approximately 90/50) for the duration of the study as well as at baseline.
303-2000113	35/F	40	13	Acute dystonia.
In combination	with halop	eridol	<u>.</u>	
116B- 05530002	62/F	200	137	Severe akathisia in combination with haloperidol.
301E- 12106661	35/M	80	43	Severe generalized dystonia 1 day after stopping ziprasidone. Subject was given one dose of haloperidol prior to episode

8.1.2.6 Aspiration Pneumonia

Because aspiration pneumonia has been associated with neuroleptic use, the following cases are listed as possibly related to ziprasidone use:

Cases of pneumo	onia			· · · ·
SUBJECT #	AGE/ SEX	MODAL DOSE (MG/D)	LENGTH OF TREAT- MENT (DAYS)	SERIOUS ADVERSE EVENT/ COMMENTS
105-5340003	98/F	6	59	Subject had cough, rales, episode of vomiting and temperature of 100.8 ° F and was admitted to the hospital for aspiration pneumonia
116B-7010012	47/M-	160	321	Hospitalized x 2 days for pneumonia.

8.1.2.7 Seizures

In the safety update (submitted 8/29/97), the sponsor reported the seizure rate to be 1.8 subjects per 100 subject years (12/772) or 0.54 % (12/2588) of the subjects in the NDA data base experienced a seizure while taking ziprasidone. The original NDA submission (3/18/97) includes six subjects who discontinued as a result of their seizure or possible seizure activity. The safety update did not include specific subject information.

The occurrences of seizure reported in the original submission is summarized in the tables below; seizures occurring beyond 6 days of treatment were not included. The first table is a summary of the subjects who experienced seizures within the original NDA submission of 3/18/97. The second table summarizes only subjects who discontinued because of an episode of seizure activity as of the safety update of 8/29/97. Information was obtained from the sponsor's ISS, patient narratives in the original NDA submission and the safety update of 8/29/97.

SUBJECT #	AGE/	MODAL	LENGTH	ADVERSE EVENT/ COMMENTS
	SEX	DOSE	OF	
		(MG/D)	TREAT-	
			MENT	
			(DAYS)	
.108-5880461	45/M	80-120	22	Complex partial seizure. Treated with phenytoin
				and lorazepam. No h/o seizure.
108E-6060082/	44/M	Unclear	9 months	Listed as possible seizure. Unresponsive when
108-6060053				brought to the hospital. MRI was normal except
				for sinusitis; blood gas was pH 7.35, $pCO_2=59$,
				$pO_2=50, HCO_3=33, Na=133, and$
			-	WBC=20D/micro. Subject became alert within
				first hour and according to the patient summary
Į	[was discharged with phenytoin and abuterol
				nebulizer. He was hospitalized two days later for
				further evaluation as he was febrile and had
				labored breathing requiring intensive care unit and
				corticosteroids.
116B-5900002	31/M	160	82	Listed as possible seizure. Subject was on
(also listed as				ziprasidone when he fell backwards with loss of
syncope)]	•		consciousness while smoking a cigarette. Subject
				reported headache the night before and dizziness
				prior to fall. Some limb movement occurred
				during episode.
301-2170651	31/M	120	45	Convulsive crisis on day 3.
303-0570124	35/M	40	364	Listed as possible seizure. Subject was found
]			unconscious in the hospital ward; no abnormalities
-				found in EEG and ECG showed "some
				tachycardia" up to nine days following the
	ł			episode. (Note: there was no CRF available and
				no specific information regarding the work up of
				this episode). Subject completed the study for
[[about two more weeks with metoprolol added to
	1			his medication regimen. One month after
				stopping ziprasidone, subject had a seizure. No
				prior h/o seizure.
303-1780041	45/M	40	2	Grand mal seizure. Treated with lorazepam.
	l			
	L			

Summary of seizures with ziprasidone treatment in the original NDA data base (cut-off 10/31/96)

.

SUBJECT #	AGE/ SEX	MODAL DOSE (MG/D)	DESCRIPTION	ADVERSE EVENT/ COMMENTS
303-2380193	49/M	40	183	Grand mal seizure on day 122. Treated with diazepam.
303-2650327	60/F	40	365	Tonic clonic seizure on day 153.

Discontinuations due to seizures in the safety update of 8/29/98 (cut-off 5/15/97)

SUBJECT #	AGE/ SEX	MODAL DOSE (MG/D)	LENGTH OF TREAT- MENT (DAYS)	ADVERSE EVENT/ COMMENTS
302E-3570475*	41/F	Unclear	7	Generalized seizure. No h/o seizure.
306E-3740017*	26/M	80	42	Tonic-clonic seizure. Treated with diazepam. H/o febrile seizure in childhood. No h/o seizure in adolescents and adulthood.
307-3770251*	36/M	40	14	Grand mal seizure. No h/o seizure.
JP-95-601 23-01* ⁴	61/M		9	Epileptic seizure. Treated with diazepam and carbamazepine. H/o one prior seizure 2° to polydipsia.

*From Safety Update

^ANot in the integrated safety data base

8.1.3 Dropouts and "Other Significant Adverse Events"

8.1.3.1 Overall Profile of Dropouts

According to the original submission, the primary integrated database included 1263 subjects (59%) of the total 2140 who prematurely discontinued treatment in Phase II/III trials. The sponsor did not provide summary data of the discontinuations in the safety update, but a count of the line listings showed that there were 72 new discontinuations. The sponsor's table below provides reasons for discontinuations for the original submission. Please note that this table does not include data from the safety update.

Insufficient clinical response was the reason that the majority of discontinuations occurred for the open label and placebo controlled trials. The sponsor's table below compiles the reason for discontinuations in four of the pivotal short term studies where a placebo comparison can be observed:

Med - Page 64 of 223

Overview of Phase II/III Study Discontinuations All Oral Dosing Phase II/III Studies

226	• • • • •	• • • • • •	• • • • • • • • • • • • •		 .					
776						,		•••••	••••••	•••••
	(10	.6)	64	(15.7)	23	(11.2)	0		18	(5.1
609	(28	. 6)	60	(14.7)	29	(14.1)	1	(5.3)	143	(40.4
16	10	75	2	(0.5)	3	(1.5)	ó			(0.3
2			ā		ă		ò		; ;	(0.3
ī			ĩ	(0.2)	i	(0.5)	ò			
46 Ž			R.	(20.1)	37		ž	(10.5)	67	(16.1
1308	(61.	 1)	209	(51.4)	. 53	(45.1)		(15.8)	229	(62,1)
	16 2 3 462	16 (0 2 (0 3 (0 462 (2)	16 (0.7) 2 (0.1) 3 (0.1)	16 (0,7) 2 2 (0,1) 0 3 (0,1) 1 462 (21,1) 82	16 (0,7) 2 (0.5) 2 (0,1) 0 3 (0,1) 1 (0,2) 452 (21,1) 82 (20,1) 1 (0,2)	16 (0,7) 2 (0.5) 3 2 (0.1) 0 0 0 3 (0.1) 1 (0.2) 1 452 (21,1) 82 (20,1) 37	16 (0,7) 2 (0,5) 3 (1,5) 2 (0,1) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 1 (0,5) 3 (1,5) 1 0 0 2 (2,1) 3 7 (10,5) 1 0 0 0 1 1 0 1 1 0 1	16 (0,7) 2 (0.5) 3 (1.5) 0 2 (0.1) 0 0 0 3 (0.1) 1 (0.2) 1 (0.5) 0 452 (21.1) g2 (20.1) 37 (18.0) 2	16 (0,7) 2 (0,5) 3 (1,5) 0 2 (0,1) 0 0 0 3 (0,1) 1 (0,2) 1 (0,5) 0 452 (21,1) 42 (20,1) 37 (10,0) 2 (10,5)	16 (0,7) 2 (0.5) 3 (1.5) 0 1 2 (0.1) 0 0 0 1 3 (0.1) 1 (0.2) 1 (0.5) 0 0 452 (21.1) 52 (20.1) 37 (18.0) 2 (10.5) 57

Includes dupiloate counting of 46 21presidone supjects and discontinued both the parent and extension study. Protocols: 015,101,102,104,104E,106,106E,104,108E,109,109E,110,111,114,115,1169,117,114,122,301,302,303,304,305

Insufficient clinical response was the reason that the majority of discontinuations occurred for the open label and placebo controlled trials. The sponsor's table below compiles the reason for discontinuations in four of the pivotal short term studies where a placebo comparison can be observed:

Overview of Phase II/III Study Discontinuations Short-Term Fixed-Dose Placebo-Controlled Oral Dosing Phase II/III Studies

Number (X) of Subjects	Ziprasidone 702		Haloperidol 85		Placebo 273	
Discontinuations Adverse event	29	(4.1)	7	(8.2)	د	(2.2)
Insufficient clinical response	165	(23.5)	18	(15.3)	96	(34.8)
Laboratory findings	4	(0.6)	ŏ		ő	
Other	118	(16.8)	17	(20.0)	49	(17.9)
fotal	316	(45.0)	37	(43.5)	150	(54.9)

Other reasons for discontinuation may include failure to meet randomization criteria, lost to follow-up, protocol violation, withdrawn consent, etc. Protocols: 104,106,114,115

As seen above, there was a higher withdrawal rate for insufficient efficacy in the placebo group than in the ziprasidone group; however, the drop out rate for adverse events was higher for the groups receiving ziprasidone.

8.1.3.2 Adverse Events Associated with Dropout

In the integrated safety data base (excluding data from the safety update), 221 subjects (10.3 %) of the 2,140 subjects exposed to ziprasidone discontinued the study due to an adverse event.

In order to establish a comparator control, it is most helpful to focus on data collected from placebocontrolled studies. The following table lists the adverse events which resulted in 2 or more withdrawals from the placebo-controlled pivotal studies (including studies 104, 106, 114, and 115):

Adverse events leading to discontinuations in short term placebo-controlled studies 104, 106, 114, 115

	Ziprasidone n=702	Placebo n=273	Haloperidol n=85
Total # of subjects discontinuing due to adverse events (%)	29 (4.1)	6 (2.2)	7 (8.2)
ADVERSE EVENT			
Rash	7 (1.0)	0	0
Nausea	3 (0.4)	1 (0.4)	0

	Ziprasidone	Placebo	Haloperidol
	<u>n=702</u>	<u>n=273</u>	<u>n=85</u>
Hypertension	2 (0.3)	0	0
Tachycardia	2 (0.3)	0	0
Vomiting	2 (0.3)	0	0
Akathisia	2 (0.3)	0	2 (2.4)
Hostility	2 (0.3)	0	0
Insomnia	2 (0.3)	· 0	0
Somnolence	2 (0.3)	0	1
LFT abnormality	2 (0.3)	0	0

NOTE: It appears from the sponsor's data that individual subjects may have experienced more than one adverse event.

8.1.4 Other Search Strategies

None.

Common Adverse Events 8.1.5

8.1.5.1 Approach to Eliciting Adverse Events in the Development Program

Pfizer did not provide their working definition of an adverse event in the Integrated Summary of Safety. All adverse events presented were classified by organ system using COSTART terminology. The sponsor stated that adverse events were collected by either direct observation by the investigator or by patients volunteering this information. This method may result in an under representation of adverse events, because the schizophrenic population may not be able to spontaneously volunteer and articulate their discomfort.

The sponsor states in the Integrated Summary of Safety (ISS) that investigators categorized adverse events as to whether or not there was a probable relationship to the study medication. The ISS presented tables which addressed events that were judged by the investigators to be related or of unknown relationship to the treatment medication; however, the sponsor stated that all causality tables included all adverse events independent of the investigator's judgment (submitted 4/9/98).

8.1.5.2 Establishing Appropriateness of Adverse Event Categorization and Preferred Terms

The sponsor reports using the COSTART dictionary, coding adverse events by body system and preferred terms. In most cases, the sponsor chose appropriate categories; however, there were several cases in which alternative categories were chosen by this reviewer to better reflect the clinical presentation as described in the patient profiles and case report forms. Serious adverse events and discontinuations which have been recategorized from the sponsor's original COSTART TERMS are as follows:

1. Subject 116B-553001: a 40 year old female who was taking 60 mg of ziprasidone daily for 830 days began to have an episode of odd and confused behavior, a temperature of 101° F, diaphoresis, bradycardia, urinary incontinence and a CPK level of 944 U/L. Her husband reported that she had had similar episodes in the past. Her urine culture showed an E. coli infection. The sponsor chose to categorize this serious episode as UTI; to be more accurate, it needs to be considered that this case may also have been a manifestation of neuroleptic malignant syndrome.

2. Subject 118-7090001: a 44 year old male who was taking 40 mg ziprasidone daily for 4 days, fell flat backwards 3 ½ hours after his last dose and was found to be hypotensive (70/50). One day after the last

ziprasidone dose, this subject was found on the floor unresponsive; he was found to have a sodium level of 120 (but his baseline was 127) and was re-hospitalized and placed on fluid restriction. EEG and CAT scan were benign. No cardiac work up was done. The sponsor chose to categorize this as a seizure despite that there was no seizure witnessed; therefore, this was recategorized as syncope.

3. Subject 101-5060084: a 52 year old male who was taking 20 mg ziprasidone daily for 4 days was found at home confused and fluid depleted secondary to hyponatremia with sodium level of 111. He was admitted to the hospital and with fluid restriction, and his sodium returned to normal. This was originally cataloged as confusion and this reviewer reclassified it as hyponatremia.

4. Subject 303-0212105: a 56 year old male who was taking 160 mg ziprasidone daily for 280 days who reportedly had a history of essential hypertension (but was normo-tensive throughout the study), fell down the steps, lost consciousness and was noted to have a blood pressure of 220/140. This case was listed as intracranial hemorrhage which actually occurred as a result of this fall; it was recategorized as a hypertensive episode because this event appeared to have preceded the fall.

The following adverse events were listed under reasons for discontinuations, but have been considered to be serious events by this reviewer:

1. Subject 109-5650041: a 51(age was listed as 48 in the summary, but CRF stated 51) year old male who, within the first few days of treatment with ziprasidone 20 mg qid, became confused with a temperature of 99.9 ° F, a creatinine kinase level of 955, and sinus tachycardia. Within two weeks of discontinuing ziprasidone, the creatinine kinase levels normalized and the tachycardia resolved within 3 days of ziprasidone discontinuation. The sponsor did not list this as a serious adverse event, but as a case of discontinuation due to the event of tachycardia. The presented data more accurately suggests that this was a serious adverse event of probable neuroleptic malignant syndrome.

2. Subject 116B-5230001: a 41 year old male who was taking 80 mg ziprasidone for 80 days experienced mild nausea and an episode of syncope and was taken to the emergency room. The sponsor categorized this as a discontinuation and did not list this as a serious adverse event; the case report form had a notation from the investigator that the emergency room physician thought that this event was due to not eating for a couple of days; however, the work up was not included in this submission. It was felt by this reviewer that this syncopal event should be classified as a serious adverse event.

3. Subject 104-5220146: a 45 year old female (taking 80 mg ziprasidone x 12 days) with a history of hypertension stabilized with nifedipine was hospitalized for a possible hypertensive crisis with a diastolic pressure up to 120. This adverse event was listed in the discontinuations by the sponsor. It is the opinion of this reviewer that this adverse event should be reclassified as a serious adverse event as this event resulted in the subject's being hospitalized.

8.1.5.3 Selecting the Best Adverse Event Tables for Characterizing the Adverse Event Profile

A helpful perspective in the attempt to determine the occurrence of events related to the study medication is to look at incidents of adverse events in placebo-controlled trials. Appendix 8.1.5.3 consists of the sponsor's table of adverse events occurring in 1 % of the subjects taking ziprasidone in the data collected from the placebo-controlled pivotal studies (104, 106, 114, and 115). The proposed labeling include a 1 % table based on this data. Events occurring in this pooled data that are not listed in the proposed labeling are anxiety, tremor, conjunctivitis, and urinary incontinence; these events were seen with equal frequency in the placebo group.

NDA 20-825: Ziprasidone po Clinical Review - 38

Med - Page 67 of 223

Appendix 8.1.5.4 is extracted from the sponsor's proposed labeling and lists all adverse events occurring in the primary safety data base in the original submission; data from the safety update is not included in this summary.

8.1.5.4 Identifying Common and Drug-Related Adverse Events

Common events were determined by identifying events which occurred in at least 5% of the ziprasidone group and occurred more than twice as frequently in the ziprasidone than in the placebo group. The drug related adverse events fulfilling this criteria were extrapyramidal syndrome and somnolence as seen in the 1% table. Weight gain was observed in 10 % (61/622) of subjects taking ziprasidone in the short term placebo controlled phase II/III studies versus 4 % (9/227) in placebo (see Appendix 8.1.7.3.2); it is noted that weight gain was omitted from the 1% table.

8.1.5.5 Additional Analyses and Explorations

Dose Response

A dose relationship for several adverse events was established when the sponsor applied the Mantel-Haenszel test to the pooled data from pivotal short term studies (104, 106, 114, 115) for doses of <40 mg bid, 40 mg bid, 80 mg bid, and \leq 100 mg bid. The sponsor's analysis showed a statistically significant dose relationship ($p \leq 0.05$) with the following adverse events: asthenia, postural hypotension, anorexia, diarrhea, dry mouth, increased salivation, arthralgia, anxiety, dizziness, dystonia, hypertonia, somnolence, tremor, respiratory tract infection, rhinitis, rash (p=0.051), abnormal vision.

Demographic Analyses

The interaction effect of gender, age, or race was determined in the common adverse events that occurred at least 5% in the ziprasidone group and more than twice as frequently than in the placebo group. The sponsor states that using the Breslow-Day Odds Ratio, there was no differences noted in the incidence of somnolence (the only adverse event analyzed) when examining gender (p=0.89), age (p=0.62) or race (p=0.89).

The demographics for this analysis is as follows:

Gender

Women: n=186 ziprasidone, n=71 placebo Men: n=516 ziprasidone, n=12 placebo

Age

18-64 y.o.: n=693 ziprasidone, n=271 placebo 65-74 y.o.: n=9 ziprasidone, n=2 placebo

Race

Caucasian: n=462 ziprasidone, n=171 placebo African American: n=175 ziprasidone, n=68 placebo Asian: n=21 ziprasidone, n=2 placebo Other: n-44 ziprasidone, n=26 placebo

8.1.6 Laboratory Findings

8.1.6.1 Extent of Laboratory Testing in the Development Program

The review of laboratory findings encompass data from the Phase II/III trials in the integrated safety data base of the original NDA submission. The Integrated Summary of Safety states that routine laboratory tests for all studies included complete blood count, electrolytes, serum hepatic and renal function. The final samples were collected up to six days after the last day of study medication. The frequency of laboratory testing varied amongst the studies; the ISS merely states that routine laboratory tests were collected at baseline and repeated during and/or at the end of the study.

8.1.6.2 Selection of Studies and Analyses for Overall Drug-Control Comparisons

This section of the review will concentrate on the pooled clinical laboratory values from the short term placebo-controlled studies 104, 106, 114, 115. This allows for comparison to be made to the placebo group and also may eliminate any confounding variable of time period as the duration of these studies is comparable (studies 104 and 106: 4 weeks; studies 114 and 115: 6 weeks). Laboratory values were collected at different intervals for these studies: 106 and 104 had weekly monitoring, while 114 and 115 had laboratory monitoring about every other week.

8.1.6.3 Standard Analyses and Explorations of Laboratory Data

8.1.6.3.1 Analyses focused on Measures of Central Tendency

The mean change from baseline of clinical laboratory values for ziprasidone and placebo can be found in Appendix 8.1.6.3.1. The sponsor performed the Kruskal-Wallis test using the RANK and ANOVA procedures with an alpha level of 0.05 to analyze the data; it is inferred from the sponsor's table in Appendix 8.1.6.3.1 that there were no statistically significant findings when comparing the ziprasidone group with the placebo group. Inspection showed that the mean changes of note in the ziprasidone group compared to placebo were values for blood urea nitrogen ($\downarrow 8$ %), bicarbonate ($\uparrow 4.5$ %), and LDH ($\uparrow 3$ %). It is important to keep in mind that the last observation value could be obtained up to six days after discontinuation from the study.

8.1.6.3.2 Analyses focused on Outliers

The sponsor used an elaborate system to determine when a laboratory value had clinical significance; this was not clearly explained in the original NDA submission and a request for further details was made (please refer to submissions of 11/18/97 and 12/17/97). Appendix 8.1.6.3.2a contains the sponsor's laboratory reference ranges used to determine whether the baseline value was normal or abnormal; baseline values were then compared to the worst laboratory value found during the study. The sponsor applied different criteria for subjects who began the study with abnormal laboratory values. Clinical significance was determined using the values in column "A" and "B" in Appendix 8.1.6.3.2b; for subjects with normal baseline values, the worst value was required to be outside the range specified by column "A". Meanwhile, for subjects with an abnormal baseline value it was required that their worst lab value be described by both column "A" and "B" in order to be considered of clinical significance and be included in the number of subjects with laboratory abnormalities.

For reasons explained above, the focus of this section will be the short term placebo controlled trials. Please refer to Appendix 8.1.6.3.2c for the incidence of clinically significant laboratory tests in the short term placebo controlled fixed dose studies 104, 106, 114, 115. Using a Fisher Exact 2-tailed test at the 0.05 alpha level, the sponsor found that statistical significance was seen in the following laboratory values: cholesterol (ziprasidone: 2% versus placebo: 0%) and triglycerides (ziprasidone: 12% versus placebo: 7%).

8.1.6.3.3 Dropouts for Laboratory Abnormalities

Of the 2140 ziprasidone subjects in the phase II/III trials in the original integrated safety data base, the following laboratory abnormalities resulted in early termination in 0.1% or greater of the subjects:

ALT elevation	0.5%
AST elevation	0.4%
Alkaline Phosphatase elevation	0.1%

A criteria to determine when a subject should withdraw from the study was not located in this submission. The following table is composed of subjects who did drop out because of laboratory abnormalities. Information for this table was generated from both the original submission and the safety update which encompasses a total of 2565 subjects exposed to ziprasidone in the Phase II/III data base.

Subject ID #	Age/Sex	Dose/duration	Most extreme levels	Outcome/comments
Abnormal liver	enzymes	l		
117-6550327	58/M	120 mg 37 days	ALT:156,AST:68 U/L AP:250, GGT:1372 Bilirubin: 2.3mg/dl	Jaundice which resolved two weeks after d/c. Five days after d/c labs began to normalized, but remained elevated.
301-2790615*	43/M	120 mg 42 days	ALT:104,AST:217 U/L	Jaundice, mild, with fever and fatigue. Follow up labs reported to be normal.
#117-622- 0033	36/M	160 mg 219 days	ALT:58,AST:158, LD:354 IU/L	LFTs reported to be resolved one day after d/c
117-622-0033	36/M	160 mg 219 days	ALT:58,AST:158, LD:354 IU/L	LFTs reported to be resolved one day after d/c
115-6380055	40/M	120 mg 10 days	ALT:116,AST:188 IU/L	LFTs resolved within two days after d/c
116B- 6220001	27/M	200 mg 117 days	ALT:77,AST:144 IU/L	LFTs normalized within one month after d/c
106-5390092	35/M	120 mg 9 days	ALT:165,AST:89U/L	ALT returned to normal in 5 days and AST returned to normal in 19 days. LFTs normalized within five days after d/c
109-5720028	34/F	80 mg 9 days	ALT:128,AST:55 U/L	LFTs normalized within eleven days after d/c
101-5090050	37/M	10 mg 7 days	ALT:56,GGT:137 U/L	Within two weeks, ALT normalized and GGT was returning to normal
101-5050065	39/M	40 mg 8 days	ALT:94,AST:197, GGT:128 U/L	LFTs were lowering within 5 days; no other f/u provided
115-6530148	42/M	120 mg 26 days	ALT:194,AST:80 U/L	LFTs were lowering within 8 days; no other f/u provided
116B- 5810013*	43/M	200 mg 171 days	AP: 257 U/L	Subject remained in study and elevated AP did not resolve by the end of the study.
114- 5290079**	40/M	80 mg 15 days	AST:377 IU/L ALT:582 LDH:784	Labs returned to normal within one month after stopping ziprasidone
304-1890367	27/M	15 days dose not provided	AST: 111 U/L ALT: 290 U/L	Subject was listed in the safety update as participating in the blinded group with laboratory test abnormalities and possible hepatitis . Sponsor revealed that subject was in ziprasidone group. Doses unclear.

Dropouts for laboratory abnormalities

Hematologic ab	normalities			
303-1800061	65/M	80 mg 89 days	Platelets:12x 10 ⁹ /L (nl:160-350)	Value at baseline was 93 x 1. Thrombocytopenia resolved · days of d/c.
104-5370243	37/M	40 mg 20 days	Neutrophils: 27 % (1242/UL)	Mild neutropenia resolved wit day of d/c
303-265-0357	36/M	160 mg 41 days	Eosin: 35.1%	Within two weeks after d/c, cosmophilia resolved.
Dropouts for lal	poratory abn	ormalities (con't)		
Subject ID #	Age/Sex	Dose/duration	Most extreme levels	Outcome/comments
Abnormal liver	enzymes/hei	natologic abnorm	ality	
303-02710228	54/M	160mg 87 days	ALT:70,AST:85 U/L Prolactin: 0.53 IU/L Hb/Hct:7.1/0.35	Accompanying symptoms of heartburn and epigastric pain were resolved within four weeks of discontinuation and treatment. Subject diagnosed with chronic gastritis by endoscopy and Helicobacter pylori infection. Follow up labs not included in submission.
Glucose abnorm	nality			
117-05130512	35/M	120 mg 194 days	Glucose: 883 mg/DL	Diabetic ketoacidosis successfully treated with insulin. Subject had one previous episode.

*From Safety Update

**Not included in sponsor's calculations (Sponsor listed this subject as discontinuation due to insufficient clinical response

As can been seen from the table above, there were several subjects whose liver function studies elevated while on ziprasidone and the pattern of resolution suggests that these episodes were drug related. Two subjects were reported to exhibit symptoms of jaundice with elevated liver functions tests: one subject (117-6550327) had an extremely high level of GGT (1372 IU/L); the other subject (301-2790615) also had symptoms of fever, and fatigue.

Also of note is the one subject (303-01800061) who withdrew because of thrombocytopenia which resolved upon withdraw from ziprasidone. It is possible that this adverse event may have been drug related.

8.1.6.4 Additional Analyses and Explorations

Prolactin studies were monitored in only two subjects in the short-term placebo controlled studies, one of which was found to have abnormal values. Therefore, it is necessary to look at the total pool of oral dosing phase II/III studies; clinically significant abnormalities were identified in 20% of the (148/741) ziprasidone subjects whose prolactin levels were monitored (see Appendix 8.1.6.4). Using the Fisher two tailed statistical test with an alpha level of 0.05, the sponsor determined that this rate of prolactin abnormality was statistically significant when compared with the placebo subjects monitored.

Thyroid function studies were not tested at all in the short-term placebo controlled studies. In referring to Appendix 8.1.6.4, it can be seen that there were only 224 subjects tested for thyroid functions and when compared to the rate of abnormal findings in the 56 placebo group monitored, there was no statistical significance identified.

8.1.7 Vital Signs

8.1.7.1 Extent of Vital Sign Testing in the Development Program

The ISS does not specify which vital signs were included in each study, nor does it state the frequency of vital sign monitoring; the final vital sign monitoring could occur up to six days after the last dose of study medication was given. The sponsor analyzed changes in standing or sitting systolic or diastolic blood pressure and sitting or standing heart rate, and weight gain or loss. Please refer to Appendix 8.1.7.3.2 for vital sign parameters used to determine clinical significance. There is no data comparing changes of supine and standing vital signs; therefore, orthostatic changes could not be adequately assessed.

8.1.7.2 Selection of Studies and Analyses for Overall Drug-Control Comparisons

The focus of this section will be on the short-term placebo controlled trials to allow for placebo comparison.

8.1.7.3 Standard Analyses and Explorations of Vital Sign Data

8.1.7.3.1 Analyses Focused on Measures of Central Tendency

Appendix 8.1.7.3.1 contains the sponsor's summary of the median change from baseline to last observation of vital signs. Using the Kruskal-Wallis test with an alpha level of 0.05, the sponsor found a statistically significant change for weight increase (61 of 622 subjects) when comparing the ziprasidone and placebo groups. From these findings, it does not appear that there are significant changes in heart rate or blood pressure reading when comparing placebo and ziprasidone; however, it must be kept in mind that the final reading could have occurred up to six days after the last dose of study medication was administered.

8.1.7.3.2 Analyses focused on Outliers

Appendix 8.7.3.2 includes the sponsor's criterion of clinically significant changes as well as the incidence of these events in the placebo-controlled studies. It can be inferred from this table that statistical significance using the Fisher exact two-tailed test was seen for weight increase only. Again, it must be considered that the baseline values may have been compared to a vital sign measurement that was taken up to six days after the last day of study treatment.

8.1.7.3.3 Dropouts for Vital Sign Abnormalities

The table below presents the number of subjects who withdrew because of vital sign abnormalities in the original NDA submission database. A similar table was not submitted in the sponsor's safety update data.

Subject withdraws for abnormal vital signs or weight measurements in P.	hase II/III trials (adapted from
sponsor's electronic submission)	

	Ziprasidone	Haloperidol	21speridone	Placebo
Number of Subjects Randomized:	2140	407	206	354
HYPERTENSION	4 (0,2) ⁻	•	1 (0.5)	• • • • • • • • • • • • • • • • • • • •
HYPOTENSION Postural hypotension	1 (0.0) 1 (0.0)	0	0	0
TACHYCARDIA NEIGHT GAIN NEIGHT LOSS	4 (0.2) 1 (0.0) 3 (0.1)	1 (0.2) 0	0 0	0 0

Med - Page 72 of 223
The following table itemizes the case reports of subjects who discontinued because of vital sign abnormalities. There is some discrepancy for the number of subjects who discontinued because of weight changes, because there were only two case listed in the sponsor's line listing of subjects for this item; however there were two cases of weight loss listed as serious adverse events (please refer to Appendix 8.1.2 under metabolic).

Subject ID #	Age/Sex	Modal Dose /duration	Reason for d/c	Outcome/comments
116B- 0617001	26/F	160 mg 139 days	weight gain	Subject reported weight gain. Investigator did not record weights.
116B- 05720001	37/M	200 mg 127 days	weight loss	Subject's weight fluctuated (baseline: 123 lb.; at discharge: 118.5; overall loss: 4.5 lb.). There was 10 lb. weight gain in first 6 weeks and then 14.5 lb. weight loss within the next two month before d/c.
117-6840608	43/M	80 mg 7 days	hypertension	Termination sitting blood pressure was 160/120 with heart rate 124 bpm. Baseline values: 145/100; 84bpm. No orthostasis.
303-2120222	58/M	160 mg 349 days	hypertension	Syncopal episode: loss of consciousness for 2-3 minutes. Upon recovery, blood pressure was 180/90 with 130 bpm. No significant ECG change. Returned to baseline the next day. CT-scan showed cerebral neoplasm with edema I temporo-parietal region of right hemisphere.
104-5220146	45/F	80 mg 12 days	hypertension	Subject had history of hypertension and diabetes with concurrent medications of nifedipine and glyburide. She awoke in the middle of the night feeling weak and tremulous and had a blood pressure of 195/105. Ziprasidone was d/c and nifedipine was increased. Over the next three days blood pressure fluctuated from 170/90 to 142/84
106-5520124	41/M	40 mg 15 days	hypertension	Blood pressure: baseline: 100/70 (sitting) at d/c: 148/112 (sitting)
Subjects who	discontinue	d due to vital sig	n abnormalities (co	on't)
Subject ID #	Age/Sex	Modal Dose /duration	Reason for d/c	Outcome/comments
303-1970265	68/M	40 mg 7 days	hypotension	Subject was hypotensive at baseline.
106-5420149	31/M	120 mg 14 days	tachycardia	Pulse at baseline: 122 bpm at d/c: 138 bpm
117-7060380	30/M	80 mg 12 days	tachycardia -	Recorded as severe tachycardia, but heart rate not available in patient profile or case report.
115-6470383	40/M	80 mg 3 days	tachycardia	Baseline: ECG: possible lateral/inferior infarct with pulse: 74 bpm, but cardiologist determined this was not an infarct. QTc:408.7 At discharge: pulse: 119 bpm, QTc:445.0

Subjects who	discontinued due t	to vital sign a	abnormalities

8.1.7.4 Additional Analyses and Explorations

There were no additional studies or analyses of vital signs done by the sponsor.

8.1.8 ECGs

8.1.8.1 Extent of ECG testing in the development program

ECGs were recorded in all Phase II/III trials. The tracings were read on site for the short term placebo controlled and most other studies. Study 303, the 52 week placebo controlled study, had ECG tracing initially read on site and then they were sent to Premier Research Worldwide of Cambridge England for blinded reading. In the original NDA submission, the ECG results of study 303 were kept separate as a different data set within the NDA data base. However, a resubmission of the data (11/13/97) included an analysis and consult by Joel Morganroth, M.D. in which the data sets from study 303 were integrated into the entire data base of the NDA, and ECGs suspected of being read by an automated system were reread by a more accurate system of analysis (Please refer to Section 8.1.8.4 for details). The tables of data submitted in Dr. Morganroth's reanalysis (submission of 11/13/97) will be used for the purposes of this review.

As part of the review process, Charles J. Ganley, M.D., cardiology consultant at FDA (review of 11/18/97) stated that ECGs were performed at trough times in some studies and the timing in other studies was unclear. He also addressed concerns regarding the dose dependent response for QT and QTc prolongation observed in some studies within this submission (11/18/97 & 1/6/98).

8.1.8.2 Selection of Studies and Analyses for Overall Drug-Control Comparisons

As done in the review of clinical laboratory and vital sign analyses, the main focus of this section of the review will be on pooled ECG data from the short term placebo controlled studies: 104, 106, 114, 115. This allows for comparison to be made to the placebo group and may eliminate any confounding variable of time period since the duration of treatment between the groups in these studies is comparable. Within this pool of data, there were 656 ziprasidone subjects and 250 placebo subjects for comparison. The tables used will reflect the most recent data reanalysis submitted by the sponsor (11/13/97).

8.1.8.3 Standard Analyses and Explorations of ECG Data

8.1.8.3.1 Analyses Focused on Measures of Central Tendency

Appendix 8.1.8.3.1 presents the mean change from baseline to the most abnormal value of ECGs from the short term placebo controlled studies. It is important to compare baseline to the most abnormal value, because the last observed value was collected up to six days after the last dose which may reflect less drug effect due to the half life. The sponsor did not provide a statistical analysis comparing placebo with the treatment groups with this reanalyzed data. From observation, it appears that subjects in the ziprasidone groups manifest more frequent ECG changes of QTc increase and heart rate than is seen in the placebo and haloperidol group. Below is a table summarizing these two parameters.

		Ziprasido	Placebo	Haloperidol			
	<80 (n=230)	80 (n=138)	120 (n=111)	160 (n=100)	≥200 (n=77)	(n=250)	(n=76)
QTc	8.6	12.6	15.2	19.8	15.0	4.3	4.1
Heart rate	7.8	6.6	5.4	6.7	4.4	4.5	4.4

As observed by Dr. Ganley's review (11/18/97 & 1/6/98), it can be seen that there is a dose response for the mean QTc change for doses less than 200 mg of ziprasidone.

8.1.8.3.2 Analyses focused on Outliers

Appendix 8.1.8.3.2 presents the sponsor's reanalysis of the incidence by percent of QTc changes collected from ECGs in the short term placebo controlled studies. The sponsor only submitted data on the incidences of QTc changes and did not submit information regarding the incidence of change in other ECG parameters.

In the original submission, the sponsor included a consult from on the original data base was that the most frequent ECG changes were from normal ST-T to abnormal ST-T with instances of isolated prolongation of the P-R, QRS, appearance of LVH, LAFB and rare PVC; Dr. Fisch did not feel that these finding were significant and he did not observe them to be dose related.

8.1.8.3.3 Dropouts for ECG Abnormalities

The sponsor did not provide information specifically itemizing subjects who withdrew because of ECG changes. The following table of subjects who discontinued due to ECG abnormalities was generated from the pool of all oral dosing phase II/III subjects.

SUBJECT #	AGE/ SEX	DOSE (MG/D)	LENGTH OF TREATMENT	SERIOUS ADVERSE EVENT/ COMMENTS
116B-6360004	69/F	80	approx. 8 months	Arrhythmia; Holter monitor showed frequent ventricular ectopy with ventricular couplets and bigeminy and supraventricular ectopy.
106-5420149	31/M	120	14 days	sinus tachycardia (138 bpm)
109-5650041	51/M	80	3 days	tachycardia-discussed as Possible NMS (See section 8.1.2.1)
117-7060380	30/M	80	12 days	sinus tachycardia (from CRF: 132 bpm) QT: minimal change from baseline
115-6470383	40/M	80-200	3 days	sinus tachycardia (119 bpm) QTc:screening:396 baseline: 408.7 termination 445.0 post study:423.8 post study: 396.2

ECG-Related Withdrawals

8.1.8.4 Additional Analyses and Explorations

The sponsor's submission dated 11/13/97 included a copy of a cardiology consult prepared by Within this submission, ! reported his additional analysis of the ECG data from the integrated safety data base; this included a blinded rereading of 3,883 ECGs using a central laboratory for purportedly more accurate QTc duration measurement to correct inaccuracies caused by automated readings of the ECG data. Using this method, concluded that of the 34 subjects originally identified as demonstrating QTc interval of ≥ 500 or having an increase of ≥ 75 msec, only one subject (116B-6220003) actually fulfilled the criteria of a QTc interval increase of ≥ 75 msec. No subjects were found to have a QTc measurement greater than 450 msec.

Dr. C. Ganley of HFD-110 also reviewed the ECG data in the integrated safety data base. Please refer to his reviews dates 11/18/97 & 1/6/98.

Because the sponsor allowed for the final ECGs to be recorded up to six days after the administration of the last dose, Dr. Boehm (HFD-120: 1/23/98) conducted an analysis that excluded any QTc value of subjects whose final ECG was recorded at one day or more after discontinuing ziprasidone in study 101 (the protocol required that the time of the ECGs be recorded); a comparison of baseline-to-final QTc was then recalculated. Dr. J. Boehm's results suggest that the mean baseline-to-final QTc changes recorded less than 1 day after discontinuing the drug were similar to the mean QTc measurements when the final ECG reading was taken up to 6 days after discontinuing the drug.

In the submission of 3/20/98, the sponsor expressed concern that the methodology of recording the initial QTc affects the determination of QTc prolongation. The sponsor explained that they recorded both screening and baseline values (with a wash out period in between) and claim that there was a dose dependent relationship of QTc prolongation observed with the baseline-to-final QTc measurements, but not with the screening-to-final. They expressed their concern that NDA data bases from recently approved anti-psychotic agents utilized measurements of screening value to determine if QTc prolongation was present. It is important to note that according to _________ eport (sponsor's submission of 11/13/97), there was a dose dependent QTc prolongation observed in both screening-to-maximum QTc value and the baseline-to-maximum value in ziprasidone treatment groups (although, the screening-to-maximum is of a lower magnitude).

Both Dr. Ganley and report present data suggesting a dose dependent increase in QTc for doses up to and including 160 mg daily of ziprasidone in the short term placebo controlled studies. Both reports also discuss that a QTc increase is not demonstrated in the analysis of the 52 week study 303.

Based on the evidence that ziprasidone has been shown to cause an increase in the QTc interval as a function of dose within the proposed therapeutic range (80-160 mg daily), Dr. Ganley concludes that the risks of arrhythmia, syncope, and sudden death may exist for ziprasidone, because these risks have been observed with other drugs which also prolong the QTc interval. He stated also that the labeling should clearly reflect this risk and that it may be necessary to consider this drug as a second line therapy if approved.

8.1.9 Special Studies

8.1.9.1 Ophthalmology

The sponsor did ophthalmology examinations including slit lamp exam in some of the long term studies (Studies 104E, 106E, 108, 108E, 109E, 116B, 117, 303). The sponsor provided the following superficial analysis of this data in the ISS:

Ophthalmology Data: Incidence of Significant Changes from Baseline All Eveluable Gral Dosing Phase II/III Studies

Number of Subjects	Ziprasidone 1168		Haloperidol 194			tisperidone 147				Placebo 84					
Changes From Baseline	Yes 1		No x	Ye N	• 1	Ħ	No x	н 1	/es 1		to T		Yes S	U.	No x
• • • • • • • • • • • • • • • • • • • •	43 (8.0) 49	1 (91.9)						(5.4)				(6.1)	6 1	(93.8)

* Visit day relative to start of open label treatment. Protocols: 104E,106E,108,108E,109E,1169,117,303

A review of the incidence of all adverse events in the original integrated safety data base (not including the update) revealed that there were no prominent patterns of eye dysfunction that would merit a more detailed analysis at this time. Of note, there was one subject reported with cataract and 87 subjects (4% of 2140 subjects) who reported abnormal vision, not otherwise specified.

8.1.9.2 Extrapyramidal Symptoms

Extrapyramidal symptoms (EPS) were assessed through the use of the Simpson Angus Rating Scale (SARS) and the amount of use of benztropine in subjects participating in studies 114 and 115, two of the pivotal six week placebo controlled studies. The following were the results observed:

1. When considering only subjects not treated with benztropine in study 114, a higher percentage of subjects in the 80 mg bid dose group had SARS scores (indicating more symptomatology) than in the 40 mg bid dose group. Both ziprasidone treatment groups had consistently higher scores than placebo. (see Appendix 8.1.9.2a). In study 114, twice as many subjects in the ziprasidone groups (40 and 80 mg bid) required the use of benztropine than in the placebo group (see Section 7.2.2 for details).

2. Results of study 115 showed a higher percentage of subjects with an increased SARS score in all ziprasidone groups compared to placebo. In study 115, benztropine was required in the 100 mg bid group approximately 1.5 times more often than in placebo group subjects; the 20 and 60 mg bid ziprasidone group's use was comparable to placebo use (see Section 7.2.3 for details).

Akathisia, a symptom of EPS, was measured using the Barnes Akathisia Scale (BAS) in both studies 114 and 115; propranolol was the medication identified to treat akathisia in the study 114 protocol and was listed as the concomitant medication to be used for EPS in study 115. The following were the results observed:

1) In study 114, the percentage of subjects who had an increase in the BAS (including only subjects who had not taken beta-blockers) showed higher scores (indicating more symptomatology) in both ziprasidone treatment groups compared to the placebo group at the last visit (see Appendix 8.1.9.2b). In study 114, beta blocker use in the ziprasidone treatment groups increased with higher dosing and was utilized more frequently than in the placebo group (see section 7.2.2).

2) In study 115, fewer subjects in the ziprasidone treatment group had an increase in BAS scores in the compared to placebo (see Appendix 8.1.9.2b), while blocker use in the ziprasidone groups was similar in slightly lower than in the placebo group (see section 7.2.3).

In conclusion, results from study 114 reveal that subjects in the ziprasidone groups consistently experienced EPS and akathisia more than subjects in the placebo group. Results from study 115 show a higher experience of EPS in ziprasidone groups versus placebo, but not akathisia.

8.1.10 Withdrawal Phenomena/Abuse Potential

The sponsor did not study the abuse potential nor the effects of sudden or gradual discontinuation of ziprasidone treatment. There were no reported case of withdrawal reaction.

8.1.11 Human Reproduction Data

The sponsor did not address this topic in the ISS and a request for information was required. The following table summarizes the subjects known to become pregnant while taking ziprasidone.

Subject #	age	ziprasidone exposure	Outcome and comments
304(E)-00390345	33	Subject had taking ziprasidone 80 mg for eight months; undetermined when pregnancy occurred.	Uterine bleeding: 1 day after stopping ziprasidone Spontaneous abortion:2 days after stopping ziprasidone.
115-07350541	26	1 st trimester (single 40 mg dose)	Term Infant with Tetrology of Fallot
116B-0523003*	31	1 st trimester (160 mg x 8 days)	Abortion; no medical complications reported
301-02730246	28	1 st trimester (20 mg x 7 days)	Healthy baby girl

This represents a limited number of exposures during pregnancy; therefore, no definitive conclusions can be drawn from this data.

8.1.12 Overdose Experience

The sponsor reports in the Integrated Summary of Safety that there were three subjects taking ziprasidone who experienced an overdose; the sponsor does not offer a specific definition of overdose. It is possible that one subject (116B6220002) experienced the sequela of ataxia; otherwise, there is no apparent sequela in the subjects who overdosed with ziprasidone. The following table summarizes the overdose cases:

SUBJECT #	AGE/ SEX	OVERDOSE MG	CONCOMITANT MEDICATIONS	COMMENTS
116B5870007	22/M	640 mg	lorazepam	Hospitalized for nausea, vomiting, shakiness, sweats, headache. Event resolved 4 days later, but treatment not recorded in submission.
116B6220002	28/M	480 mg	lorazepam, ranitidine, aluminum hydroxide/ magnesium hydroxide	Leukocytosis (also observed one month prior to overdose), slowed speech and unsteady gait. Subject was hospitalized for observation and discharged 3 days later with ataxia. Follow-up information was not located in this submission.
116B5950022	29/M	1880 mg	lorazepam, acetaminophen, topical starch suppository	Reported to also take alcohol and paroxetine. No signs or symptoms present. ECG reported to be normal. Gastric lavage revealed no pills.

Subjects with overdose of ziprasidone

Subject who was over accidentally overmedicated with ziprasidone

. .

SUBJECT #	AGE	OVERDOSE MG	CONCOMITANT MEDICATIONS	COMMENTS
_116B06820005	52/M	200 mg qd x 2 weeks		Dosage was supposed to be 80 mg qd. Subject hospitalized for \uparrow insomnia, restlessness, and Parkinsonism.

8.1.13 Pediatric Studies

There were two studies (044 and 122) described in the user fee extension submission (1/23/98) in which pediatric subjects diagnosed with Tourette's Syndrome were exposed to ziprasidone (note: there was no data submitted for a trial in pediatric subjects suffering with psychosis).

In study 044, an open label single dose (up to 20 mg ziprasidone of a liquid suspension) pharmacokinetic study in 15 children ages 7-16 y.o. with Tourette's Syndrome, adverse events were: 1) a 15 year old with a

syncopal event 3 hours and 41 minutes after a single oral dose of 20 mg (which corresponds to the t_{N} range of 3.3-4.7 hours determined in this study), 2) postural hypotension observed in 2 subjects, 3) somnolence seen in 10 of the 15 subjects, 4) an increase in prolactin levels observed in all subjects with a peak elevation at 2-4 hours post dosing, and 5) other events including nervousness, dizziness, nausea and abdominal pain. Of note, one subject (07440011) demonstrated pharmacokinetic data that reflected an exposure of up to 10 times greater than the exposure of subjects from the same treatment group of 10 mg ziprasidone; this subjects data was not incorporated into the preliminary report's calculations of mean pharmacokinetic data.

In study 122, a double blind, placebo controlled, 8 week flexible dose (maximum: 20 mg bid ziprasidone) trial, 16 pediatric subject (ages 7-16) with Tourette's Syndrome were exposed to ziprasidone. All of the subjects treated with ziprasidone experienced adverse events during the study; the most commonly reported was somnolence (12 of 16 subjects). Of note, one subject (07440020) developed a new onset of abnormal involuntary tongue movements (suggestive of a dyskinesia) on day 59 of ziprasidone treatment which continued until day 80 when he was treated with risperidone and buspirone and was discontinued from the study. Another subject (07430014) a 15 y.o. male developed gynecomastia. Other adverse events included: akathisia, insomnia, depression, dizziness, headache, arthralgia, urinary incontinence, and dysuria.

8.2 Adequacy of Patient Exposure and Safety Assessments

8.2.1 Adequacy of Clinical Experience

The clinical data of this NDA appears to be based on an adequate subject exposure of the adult population. The duration of exposure and the total number of subjects is comparable to other recently submitted NDAs for the indication of psychosis. The sponsor submitted more than one adequate and well controlled study to support the efficacy claims of ziprasidone.

Data from two pediatric studies (Studies 044 and 050) exposing ziprasidone to children with Tourette's Syndrome were included as part of the major amendment to the NDA which extended the User Fee Date by 3 months. The material submitted for this new molecular entity entailed an exposure of ziprasidone in 15 children with a single dose and 18 children exposed to ziprasidone for less than 60 days. Also, this amendment did not include any studies in children diagnosed with psychotic disorders, the indication for which this NDA has been submitted. Therefore, the pediatric data submitted thus far is not adequate to provide appropriate safety labeling for children and adolescents at this time.

8.2.2 Adequacy of Animal and/or In Vitro Testing

With respect to QTc prolongation, the preclinical cardiovascular testing in the original NDA submission did not include in vitro studies to assess ziprasidone's effect on potassium channels or on duration of action potential in Purkinje fibers. The sponsor was requested in a letter on October 3, 1997 to conduct these in vitro studies and to characterize the action potential duration in Purkinje fibers for ziprasidone, the major metabolites of ziprasidone and an active comparison group such as sotalol or terfenadine. The request suggested using multiple concentrations of each agent to generate dose response curves.

In a meeting package (2/13/98), the sponsor concluded that ziprasidone and the metabolite ziprasidonesulfoxide did not demonstrate significant effects on the action potentials of Purkinje fibers in dogs; FDA cardiology consultants (HFD-110) expressed concern that the sponsor did not test a high enough concentration of ziprasidone in this study to fully characterize the effects of ziprasidone in the therapeutic dosage range (note: the sponsor has not submitted a study report for review). In a meeting with FDA and the sponsor (3/27/98), the sponsor presented a brief summary of data which suggested that ziprasidone may inhibit the I_{tr} channel, an ion channel implicated in the process of QTc prolongation (no data was submitted for review).

8.2.3 Adequacy of Routine Clinical Testing

This submission was of adequate quality to be submitted for review. There were some concerns of what categories the sponsor chose for some serious adverse events; this is discussed in detail in section 8.1.2.1. Originally the sponsor submitted the data with the inclusion of clinical trials conducted by Drs.

sites. The sponsor concluded that the safety profile from Dr. site was consistent with the overall NDA data base. They also reported that the efficacy data was not significantly affected by the exclusion of Dr. data.

Most of the ECG recordings obtained in this data base were performed without regard for timing. There was no study which observed ECG recordings/QTc measurements at times of peak concentrations of ziprasidone. Also of concern is that no subject in these studies wore a Holter monitor. There is a possibility that QTc changes may have been more pronounced or perhaps evidence to the contrary could have been collected if ECGs had been collected just after peak doses were administered. A Holter monitor might have also provided insight into the multiple episodes of syncope observed if those patients had been monitored.

There were also a methodological flaw in the collection of the vital signs. Most of the vital signs recorded were done with sitting blood pressure rather than blood pressure recorded in the supine position; this does not allow for the most accurate assessment of orthostatic effects of ziprasidone. Also, in looking at the median changes from baseline of vital signs, the sponsor used observations that could have been recorded up to six days after the last day of study treatment; this may provide less accurate comparisons than could have been made if these measurements were recorded sooner given the half-life of this drug (t $_{15}$ =6.6 hours).

The elaborate system used by the sponsor for reporting clinical significance of laboratory values set up many restrictions that may not have captured laboratory abnormalities of interest. The criteria for a change from baseline for a baseline-abnormal subject appears extreme, and changes that may be concerning would not be picked up using this system. It would perhaps be more helpful to identify changes from baseline and use that as the criteria. It is curious that there were a significant number of subjects who had an abnormal baseline to merit different criterion; however, their laboratory values were not so abnormal that they were excluded from enrolling in the study. Also of note is that the last laboratory value was performed up to 6 days after the end of the study; some subjects may no longer have had appreciable plasma concentrations when the tests were performed, and the maximum effect of ziprasidone may not have been appreciated.

In the placebo controlled studies, there were only two subjects evaluated for prolactin studies; no thyroid studies were conducted in the placebo controlled studies. It would have provided more accurate information to assess these changes with placebo control studies; instead inference had to be made from a pool of data that included studies of different designs and duration.

8.2.4 Adequacy of Metabolic Workup

The sponsor conducted phase I studies in healthy adults testing the concomitant use of ziprasidone and carbamazepine, cimetidine, or Maalox (B) in healthy adults. However, conclusions regarding concomitant use of carbamazepine was based on results from a study which used a dose which was lower than the recommended dosage range (please refer to Section 6.0 for detail). It would be most useful for the sponsor to test concomitant use of therapeutic doses of carbamazepine to make a more accurate conclusion of its effect.

8.2.5 Adequacy of Evaluation for Potential Adverse Events for Any New Drug and Particularly for Drugs in the Class Represented by New Drug; Recommendations for Further Study

For reasons discussed above, it would be helpful to assess ECG monitoring more closely with a Holter monitor to assess QTc changes during concentration peaks. One suggestion is to challenge subjects with higher doses to prove or disprove the existence of a dose effect changes of the QTc.

To further investigate the sponsor's claim that the QTc was not adequately tested in recently approved antipsychotic medications, it would be helpful to have a study characterizing the QTc of ziprasidone and other marketed drugs in which all ECGs were evaluated at a baseline after a wash out period. If all subjects in such a study were using a Holter monitor, it might provide clarity regarding each of these drugs effect on the QTc interval.

8.2.6 • Assessment of Quality and Completeness of Data

No electronic data sets were made available as part of the electronic submission, and all data was assessed from the grouping done by the sponsor's tables. Therefore, the laboratory abnormalities were determined by the sponsor's fixed criterion.

There were some items that did not offer consistency in the NDA. As an example: the listing of cataracts in a summary analysis: in Table H.5.2a, there are no incidents of cataract reported in all phase I/II/III studies. However, in Table H.5.8a there is one listing for cataract in this pool of all phase II/III studies. The ISS gave only a cursory summary of rashes experienced in this data base, and more detail would be necessary to help characterize this adverse event.

Another example of inconsistency was the reporting of syncopal episodes. There were 3 subjects experiencing syncope found by review of the patient narratives (108-6050002, 108E-5550096, 108E-5780052) who were not listed in the sponsor's listing of subjects with adverse events (Appendix VI table 1b of sponsor's submission of 3/18/97). This leaves some question as to whether all syncopal episodes were considered when the calculation the total of subjects with syncope listed in their calculations of incidence (Table H.5.8a of sponsor's submission 3/18/97).

Given that there is so much concern for the effects of ziprasidone on QTc measurements, it would be most helpful for the sponsor to analyze and submit results of all ECGs that have been performed when a subject has been taking ziprasidone whether they have been obtained by scheduled or unscheduled visits. It would also be helpful to have subjects on Holter monitors especially if they are subjects who have already shown that they have ECG changes or symptoms such as syncope associated with the use of ziprasidone.

8.3 Summary of Selected Drug-Related Adverse Events

8.3.1 Sudden Death

The sudden unexpected death rate for the ziprasidone safety update is 9.1 SUD per 1000 subject years (7/2565) [note: the cut-off date for the deaths was 5/15/97; the sponsor did not respond to requests to specify the cut-off date for calculating the subject years (i.e. the denominator)] (see section 8.1.1, p.28).

There are a variety of classification schemes to determine sudden unexpected deaths. However, under the scrutiny of different classification schemes, ziprasidone's rate of SUD rate continues to present a signal of risk. As discussed in Section 8.1.1 (p.28), using the scheme of classification by Dr. J Boehm (HFD-120: 2/3/98), ziprasidone was found to have a SUD rate that was 6 times higher than the SUD rate from a pool of combined data of recently approved antipsychotic NDAs. The SUD rate calculated in section 8.1.1 (p.27) of this review utilized a less inclusive classification, and resulted in a similar SUD count as that provided by the sponsor (submission of 3/20/98: using 5/15/98 cut-off). Using this less inclusive SUD

Med - Page 81 of 223

count, ziprasidone continues to surpass the SUD rate of the recently approved antipsychotics olanzapine, risperidone, and quetiapine. The SUD rate of ziprasidone is comparable to sertindole, an antipsychotic NDA withdrawn by the sponsor because of safety concerns regarding QTc prolongation, high sudden death rates in clinical trials and post marketing data from the U.K. showing a high SUD reporting rate (see section 2.2, p. 2). Since ziprasidone is not currently marketed in any country, the SUD rate of ziprasidone used in the less monitored and less restricted environment of a marketed drug is not available at this time.

8.3.2 QTc prolongation

Clinically, Ziprasidone has been shown to prolong the QTc interval in a pool of the short term placebo controlled studies (see section 8.1.8.3.1, p. 45) compared to both placebo and haloperidol in a dose dependent manner. It must also be mentioned that there was no regard for the timing of the ECG in those studies and the effect of ziprasidone on the QTc at peak concentration of ziprasidone remains unknown. The sponsor expressed concern that NDA data bases from recently approved anti-psychotic agents utilized a different methodology to ascertain QTc prolongation--using screening ECGs rather than baseline readings which are recorded after a wash out period (see section 8.1.8.4, p. 46). However, when either screening or baseline QTc values are compared to the maximum QTc, ziprasidone increases the QTc values in a dose related fashion (although, the screening-to-maximum is of a lower magnitude).

Drug induced QTc prolongation may be correlated with the development of ventricular arrhythmia, syncope, and sudden death (Morganroth, 1993). Though the ziprasidone safety data base is limited, there is already a signal of a higher SUD rate when compared to other recently approved antipsychotic NDAs. No subject--not even subjects who experienced syncope--underwent Holter monitoring. Holter monitoring may have provided insight into the multiple episodes of syncope observed in this data base (see section 8.2.3 p.51).

Based on the evidence that ziprasidone has been shown to cause an increase in the QTc interval as a function of dose within the proposed therapeutic range (80-160 mg daily), Dr. C. Ganley, FDA cardiology consultant (HFD-110) concluded that there may be the usual risks observed with drugs which prolong the QT interval. He stated also that the labeling should clearly reflect this risk and that it may be necessary to consider this drug as a second line therapy if approved.

8.3.3 Hypotension/Syncope

Because ziprasidone demonstrates alpha adrenergic properties, it is not unexpected that orthostatic hypotension and syncope were observed as adverse events in this data base (see section 8.1.2.1, p. 29). It should be noted that most of the vital signs recorded in this data base were done with sitting blood pressure rather than blood pressure recorded in the supine position; therefore, orthostatic changes may not be completely appreciated (see section 8.2.3, p.51). Despite this methodological flaw, postural hypotension was seen to have a statistically significant dose response relationship (see section 8.1.5.5, p. 39).

Syncope was reported to occur in 0.7% (15/2140) of subjects in the phase II/III safety data base (cutoff 10/31/96), while hypotension (combining postural hypotension and hypotension) occurred in 2.5 % (53/2140) of subjects in the phase II/III safety data base (as per sponsor's submission of 3/18/97: Table H.5.8). Postural hypotension occurred at a higher frequency in ziprasidone groups (1.3%) compared to placebo (0.4%) in the short term placebo controlled studies (Appendix 8.1.5.3).

There were 5 syncopal events and one hypotensive event considered to be serious adverse events (see section 8.1.2.1: p.29 and section 8.1.5.2 p. 37); however, it is unclear what criteria the sponsor used to report a syncopal event as a serious event. There were at least thirteen episodes of syncope in the Phase II/III data base that were not reported by the sponsor as a serious adverse event (see section 8.1.2.1: p.29). It is possible that the actual incidence of syncope may be higher than 0.7 % because of an inconsistency found in the presentation of the safety data base (see section 8.2.6 p. 52).

It is also noted that syncope is associated with other drugs which prolong the QTc interval. Unfortunately, subjects who experienced syncope were not monitored more closely with a Holter to determine possible etiology of their syncopal events.

8.3.4 Rash

Rash was the most common adverse event resulting in withdrawal from the ziprasidone treatment groups (see 8.1.3.2, p.36). When viewing the sponsor's table of incidence of treatment emergent adverse events (submission 3/18/97: Table H.5.8a), events related to skin and appendages occurred in 10.2 % (218/2140) of subjects. The 1 % table states that rashes occurred in 4.1% (29/702) of ziprasidone subjects in the short term placebo controlled studies, compared to the placebo rate at 3.3% (see Appendix 8.1.5.3). There also was a dose response relationship seen with the occurrence of rashes (see section 8.1.5.5 p. 39).

There were 21 subjects who discontinued for rashes in the originally submitted integrated safety data base; the haloperidol group only had 2 withdrawals, while the placebo group had 1 withdrawal. It was left to the investigator's discretion as to whether or not an event was reported as serious, and there did not appear to be any consistent feature that merited reporting a rash as a serious event (see section 8.1.2.2, p.30).

It was necessary for this reviewer to go through the line listings of subjects with adverse events (sponsor's submission of 3/18/98: Appendix VI Table 1b) and then follow this up with relevant patient profiles in order to gain a better understanding of the rashes in this NDA data base. This review revealed that there were several subjects whose rash was accompanied by an elevated white blood count, and at least two subjects with rash whose eosinophil count was elevated. Most cases of rash resolved within one week of discontinuing ziprasidone; one subject experienced "bullous drug eruptions/ pruritic blisters with post-excoriated papules" on the hands, wrist, scalp, and neck which resolved 24 days after discontinuing ziprasidone. Medications used to treat rashes included steriods (oral and topical) and antihistamines. Three subjects required prolonged hospitalization to observe their rashes. (See section 8.1.2.2, p.31)

Since the sponsor's summary of the rashes in the ISS was found to be inadequate, it is recommended that the sponsor compile a detailed and thorough summary of the description, duration, related hospitalizations, severity, accompanying symptoms, treatment and resolution history of all cases of rashes observed in this NDA safety data base.

8.3.5 Seizure

In the safety update (submitted 8/29/97), the sponsor reported that 1.8 subjects per 100 subject years (12/772) or 0.54 % (12/2588) of the subjects in the NDA data base experienced a seizure while taking ziprasidone. The original NDA submission (3/18/97) includes six subjects who discontinued as a result of their seizure or possible seizure activity. The safety update did not include specific subject information (see section 8.1.2.7, p. 34). If using ziprasidone, caution would be required for patients with a history of seizure disorder.

8.3.6 Cholesterol/triglyceride elevation

In the short term placebo controlled trials, the ziprasidone groups were shown to have statistically significant increases in both cholesterol and triglyceride when compared to placebo with respect to numbers of patients exceeding threshold values. Cholesterol levels were observed in 2 % (16/685) whereas placebo had 0%. Triglycerides increased in 12 % (85/684) of the ziprasidone subjects in these trials compared to an increase in 7 % observed in the placebo group (see section 8.1.6.3.2, p. 40 and Appendix 8.1.6.3.2c). The increase in cholesterol and triglyceride levels is not listed in the sponsor's 1 % table (Appendix 8.1.5.3) which should be corrected. Increased cholesterol and triglyceride levels may be considered risks for the development of atherosclerosis.

8.3.7 Hyperprolactinemia

Prolactin studies were monitored in only two subjects in the short-term placebo controlled studies, one of which was found to have abnormal values. Therefore, it is necessary to look at the total pool of oral dosing phase II/III studies; clinically significant abnormalities were identified in 20% (148/741) of ziprasidone subjects whose prolactin levels were monitored (see section 8.1.6.4, p. 42 and Appendix 8.1.6.4).

Amongst the sponsor's literature review were two studies (Bench, 1996; Bench, 1993) which reported large elevations of prolactin levels when comparing baseline and peak plasma levels of ziprasidone in twelve normal volunteers taking between 5 and 60 mg of a single dose of ziprasidone (see section 5.2.3 p.6).

An increase in prolactin levels was observed in all subjects with a peak elevation at 2-4 hours post dosing in a single dose pediatric study of 15 children ages 7-16 y.o (Study 044). Gynecomastia was observed in one 15 y.o. male with Tourette's Syndrome in Study 122. (see section 8.1.13, p. 49)

As many neuroleptics are associated with hyperprolactinemia, it is not surprising to observe this effect with use of ziprasidone.

8.3.8 Transaminases elevation

An elevated SGOT (AST) levels were observed in 0.3 % (6/1780) of subjects in the phase II/III study data base. Elevated levels of SGPT (ALT), a more specific enzyme indicative of hepatic cell activity, was observed in 2% (17/1776) of the subject in the phase II/III data base (sponsor's submission 3/18/97: Table H.5.18a.2).

Section 8.1.6.3.3 (p.41) provides a listing of subjects who dropped out because of abnormal liver enzymes; there were several subjects whose liver function studies elevated while taking ziprasidone and normalized one to twenty days after discontinuing ziprasidone, suggesting a positive dechallenge and drug relatedness. There were two subjects in this listing who were noted to have jaundice accompanying elevated LFTs; there was inadequate follow up reported for subject 301-2790615 (safety update) which would be required for complete assessment. Cases which the sponsor considered to manifest serious adverse event of elevated transaminase is found in section 8.1.2.4 (p.32). It is unclear how the sponsor determined whether or not a laboratory value was considered a serious adverse event.

Also of note is the case of a 49 y.o. female (subject 307-2690047) who eventually died of hepatic coma, cholestatic jaundice and malignant neoplasm 95 days after stopping ziprasidone. Her initial symptoms of jaundice and elevated AST (244 U/L) and ALT (375 U/L) first appeared after 196 days of taking a daily dose of 100 mg ziprasidone. It may be possible that the ziprasidone aggravated her already compromised liver.

8.3.9 Weight Gain

Weight gain of ≥ 7 % was observed in 10 % (61/622) of subjects taking ziprasidone in the short term placebo controlled phase II/III studies (see section 8.1.7.3.1, p. 43). There was a statistically significant increase in weight gain found in subjects taking ziprasidone compared to placebo. It is noted that weight gain in not recorded in the sponsor's 1 % table (Appendix 8.1.5.3) and fits the criteria to be considered a common and drug-related adverse event (see 8.1.5.4, p.39).

8.3.10 Extrapyramidal Symptoms (EPS)

EPS was observed often enough in the ziprasidone safety data base to be considered a common and drugrelated adverse event (see 8.1.5.4, p.39). Its incidence was found to be 5 % (33/702) of the ziprasidone subjects in the short term placebo controlled phase II/III studies. As can be seen in the 1 % table, associated symptoms of akathisia, dystonia, and hypertonia were observed at higher rates in the ziprasidonegroup compared to the placebo group (see Appendix 8.1.5.3).

EPS and akathisia, a symptom of EPS, were assessed in studies 114 and 115 using rating scales and use of concomitant medications (see section 8.1.9.2, p.48). It was found in studies 114 and 115 that subjects in the ziprasidone groups consistently experienced EPS to a greater magnitude than in the placebo groups. Study 114 results revealed that akathisia was experienced more often in the ziprasidone treatment groups, whereas study 115 did not support this conclusion.

A listing of subjects whose EPS was considered a serious adverse event can be found in Section 8.1.2.6 (p.33).

8.3.11 Neuroleptic Malignant Syndrome (NMS)

Although the ISS did not identify any cases of NMS, a review of the patient narratives revealed two subjects whose adverse event description could be categorized as NMS. These cases are also summarized in section 8.1.2.5 (p.31).

8.3.12 Somnolence

Somnolence was found to be a common and drug-related adverse event (see 8.1.5.4, p.39). It was also found to have a dose response relationship (see Section 8.1.5.5, p. 39). Somnolence was observed in 14.4% (101/702) of subjects in the ziprasidone group in the short term placebo controlled phase II/III studies (see Appendix 8.1.5.3)

8.3.13 Tardive Dyskinesia

Tardive Dyskinesia is associated with the use of most neuroleptics and the symptoms may be masked by the use of antipsychotics. It is difficult to determine ziprasidone's potential to cause tardive dyskinesia as most of the controlled safety data base is of short duration of exposure.

Of note, one subject (07440020) in pediatric study 122 developed a new onset of abnormal involuntary tongue movements (suggestive of a dyskinesia) on day 59 of ziprasidone treatment which continued until day 80 when he was treated with risperidone and buspirone and was discontinued from the study (see Section 8.1.13, p. 49). More information is needed from the sponsor to characterize this episode more clearly.

8.3.14 Aspiration Pneumonia

Aspiration pneumonia has been associated with neuroleptic use and should be considered as a possible adverse event. There were two cases of pneumonia seen in the ziprasidone safety data base (see section 8.1.2.6, p. 33). As a related symptom, cough was seen to increase in 2.6 % (18/702) of the ziprasidone subjects (compared to 0.7 % placebo subjects) in the short term placebo controlled phase II/III studies (see 1 % table: Appendix 8.1.5.3).

9.0 Labeling

If approved, the sponsor's labeling will need considerable revision. Please see section 8.3 for important concerns that need to be addressed in labeling.

10.0 Conclusions

In the wake of the uncertainty and disagreement in the cardiology community of the effects of a dose dependent QTc prolongation caused by a drug, it is difficult to determine whether this quality of ziprasidone presents a major health hazard. Clearly, more understanding and research (using consistent methodology) are needed to clarify this issue for ziprasidone as well as any antipsychotic which may have this potential. What is striking about ziprasidone is that an effect of QTc prolongation was observed in the short term placebo controlled studies irrespective of methodology (i.e. the QTc interval is prolonged when both screening-to-maximum and baseline-to-maximum measurements are made). Another relevant detail regarding ziprasidone's ability to prolong the QTc is that it has been found to be dose-dependent within the therapeutic dosing range in short term placebo controlled studies.

An important factor that must be considered in reviewing the safety of this drug is the sudden unexpected death (SUD) rate. It is concerning and alarming to view the ziprasidone safety data base against other recently reviewed antipsychotic NDA safety data bases. Even under different methods of SUD analysis, the rate of ziprasidone's SUDs clearly surpass similar drugs that have been recently approved.

In clinical practice, it is always important to weigh the balance of risks and benefits for each medication prescribed to a patient. There are currently 15 antipsychotic medications marketed in the USA, and physicians continue to struggle with finding adequate treatment for many schizophrenic patients who do not respond or cannot be treated by the available armamentarium of medications. It may be that ziprasidone would offer a unique treatment in individual cases. However, thus far, the sponsor has not shown that ziprasidone is of benefit to subjects who are refractory towards treatment with other available antipsychotic medications.

Although the sponsor may be claiming that the mean weight gain observed in the ziprasidone NDA safety data base reflects less of an increase than other marketed antipsychotics, ziprasidone is associated with weight gain. Weight gain of \geq 7 % was observed in 10 % of subjects taking ziprasidone in the short term placebo controlled phase II/III studies, and this was shown to be statistically significant when compared to placebo.

Also, ziprasidone possesses the ability to induce extrapyramidal symptoms.

Ziprasidone has been shown to be effective in the treatment of schizophrenia in two placebo controlled studies. However, the risks appear numerous. Of most concern are the qualities it shares with sertindole, an antipsychotic NDA which was withdrawn by the sponsor because of safety concerns regarding QTc prolongation and high sudden death rates in clinical trials and post marketing data from the U.K. Since ziprasidone is not currently marketed in any country, we do not have the insight as to how this drug would affect a large population in an environment less monitored, less restricted and without a mechanism for informed consent.

11.0 Recommendations

According to Section 505 [355] of the Federal Food, Drug, and Cosmetic Act (July, 1993), approval of an application may be denied if the sponsor has not employed "adequate tests by all methods reasonably applicable" to show that the drug is "safe for use under the conditions prescribed, recommended or suggested in the proposed labeling," or if there is "insufficient information to determine whether such drug

is safe for use under such conditions." In the ziprasidone safety data base, there are still many uncertainties regarding the QTc prolongation, the etiologies of syncopal episodes (no Holter monitoring was used in the current safety data base), and the signal presented by the high sudden unexpected death rate.

Ziprasidone has been shown to be effective in schizophrenic patients yet it presents safety risks of unknown magnitude. Given that there are many antipsychotic drugs available whose NDA data base did not possess the qualities of both a high SUD rate and the ability to cause a dose dependent QTc prolongation, the risk benefit ratio does not support ziprasidone as a first line drug. To overcome the risk/benefit ratio, the sponsor would need to show that it can effectively treat patients who fail on other drugs. Even if approved as a second line drug, the risks of syncope, ventricular arrythmias, or sudden unexpected death need to be clearly stated in the labeling so as to alert physicians, thus enabling them to monitor patients appropriately. In light of the uncertainty of its safety profile and the unknown effectiveness in refractory patients, it is recommended that ziprasidone not be approved at this time.

151 ~ -4/30/18

Roberta L. Glass, M.D. Medical Officer, Division of Neuropharmacological Drug Products

NDA 20-825 Div File HFD-120:Laugren/Hardeman/Mosholder/Burkhart/Boehm/Glasss

5-8-98

151 Team Landy, FDP

While I am also concerned about the potential for cardiovascular risk associated with the use of ziprasidone, I do not believe this potential risk precludes entirely the possibility of approving this product. In my memo to the file, I have provided an alternative discussion of the data and issues pertinent to cardiovascular risk for ziprasidone, and I have provided a draft of labeling that I believe adequately describes the potential risks with ziprasidone and restricts its use in a way that makes it possible for it to be used in a reasonably safe manner for patients who fail on other products.

LIST OF APPENDICES

Appendices for Section 5.0 Description of Clinical Data Sources	60
Table of all studies	61

Appendices for Section 7.0 Efficacy

Study 106
Study 114
Study 115
Study 303
Study 104

Appendices for Section 8.0 Safety

Deaths (8.1.1.1)	128
Adverse Events (Section 8.1.2)	135
Sponsor's 1 % Table (8.1.5.3)	138
Clinical Laboratories (Section 8.1.6)	141
Vital Signs (Section 8.1.7)	149
ECG (Section 8.1.8)	150
EPS (8.1.9.2)	

)

Med - Page 88 of

	Ziprasidone		Haloperidol		Risperidone		Amisulpride	Other	Placebo	
Phase I Studies (Clinical Pharmacology)		******								
Single Dose Studies	409		4				•	38	. 71	
Multiple Dose Studies	333							11	78	
Subtotal: Phase I Studies	742		4					49	149	
Thase II/II Studies										
Oral Studies										
Placebo Controlled Studies										
Short-term Pixed-Dose Studies (1)	702								273	
Long-term Phred-Dose Studies (2)	246	[27]							84	Ľ
Short-term Flexible-Dose Studies (3)	16								12	-
Long-term Flexible-Dose Studies (4)	17								10	
Active Controlled Studies										
Short-term Fixed-Dose Studies (5)	683		340							
Short-term Fiexible-Dose Studies (6)	199		6		147		49			
Long-term Flexible-Dose Studies (7)	1252	[609]	267	[28]	156	[8]				
Uncontrolled Studies						L-3	•			
Short-term Studies (8)	86									
Long-term Studies (9)	6	[6]								
Subtotal: Oral Studies	2565*		585		295		49		• 370	
Dementia Patient Studies			•							
Short-term Flexible-Dose Studies (10)	11								12	
Single Dose Total	409		4					38	71	
Multiple Does Total	2909		585		295		49	11		
Grand Total	3318		589		295		49	49	531 602	
Numbers is brackets represent the number of subjects includ the parent study and are also counted in the parent study cate Protocol 115 subjects taking ziprasidence are counted in categ	gory for that tree	tment.	These subjects	are count	ed only once i	n totals	a study under the sar and subtotals.	në trëstment as		
(1) Includes studies 104, 106, 114, 115			·							
 (2) Includes studies 104E, 106E, 303 (3) Includes studies 122 										
(3) Includes studies 122 (4) Includes studies 307										
(7) Includes Studies 30/ (7) Includes studies 101, 116, 201										
(5) Includes studies 101, 115, 301										
(6) Includes studies 111, 302, 305	-									
(7) Includes studies 116B, 117, 108, 108B, 304, 302E, 30	4B						b -			
(8) Includes studies 015, 102, 109, 110, 118										

1

Appendix 5.1.1.1 Summary of all trials (adapted from the sponsor's submission of 10/27/97; cut off date is 5/15/97).

060-

1

ţ

BEST POSSIBLE COPY

Appendix 5.1.1.2 Table of all studies (adapted from sponsor's submission of 10/24/97)

061-

	Pfizer Central Research Sponsored Phase Clinical Phermacology Studies
28-001-US	Double-blind, placebo-controlled, parallel, single rising oral dose trial; healthy men (Ziprasidone: n=40, pbo: n=20); ziprasidone doses (0.5, 1,2,5,10,20,40,80,80,100 m oral suspension).
28-002-US	Double-bind, pbo-controlled, multiple oral doss triat, healthy men (Ziprasidone; n=18, pbo n=5); ziprasidone does range (20 mg qd to 40 mg bid) 18 days,
28-004-US	Open, randomized, 3-way crossover, single oral does that; healthy men (n=9); ziprasidone dose (suspension-fasting; 20 mg, capsule fasting; 2- mg, capsule-fast
128-005-US	Double-bilind, pbo-controlled, multiple oral does, parallel trial; healthy men (ziprasidone n=16, pbo; n=4); Ziprasidone does range (20 mg od to 80 mg bid). 18 days.
28-008-US	Open, non-randomized, 6-way crossover, single oral dose trial; healthy men (n=45): Ziprasidone dose (capsule-fect: 20mg: capsules-tasting: 40mg: capsules-tasting: 80mg; capsules-fact and mg).
128-007-US	Open, three way, crossover, and single does trist: healthy man (n=29); Zipreskione does (fasting: 20mg; fed: 20mg; 2 hr after meet; 20mg).
128-008-US	Open, non-randomized, crossover oral single dose trial comparative to Haloperidal: healthy men (n=5); Doses (Ziprasidone; 40mg; Halopendal; 75 mg),
128-009-US	Double-blind, pbo-controlled, randomized, single oral does, parallel trial comparative to Diazepam; healthy subjects (Ziprasidone: n=30, Diazepam; n=30; Pbo n=30); doese (Ziprasidone 20mg; Diazepam; 10mg).
28-010-US	Open, randomized, two-way crossover, single oral/IV does: healthy men (Oral: n=12, IV: n=12); does (cepsules 20mg; IV: Sing)
128-011-US	Open, randomized three-way crossover, oral single does that; healthy men (n=90): Ziprasidone does (405 mg tablet; 1x20 mg tablet; 1x20 mg capaule).
128-019-US	Double-blind, pbo-controlled, multiple oral does trial; healting men (Zipraskione m=29; pbo; n=10); Ziprasidone dose range (5 mg qd to 60 mg bid).
128-014-US	Open, randomized, two-way prossover, single oral does trial, healthy men (n=12): Ziprasidone does (2x20 mg capeule, 1x40 mg capeule),
128-015-US	Double-blind, pbo-controlled, multiple oral does trial; subjects with achizophrenia or achizoaffective disorder (n=33); Ziprasidone does (titrated to 160 mg/day), 24 days.
128-018-US	Open, randomized, sb-way crossover single oral/IV/Nescenteric infusion trial: Healthy men (n=28); Ziprasidone does (20 mg),
128-017-US	Open, persiel, oral single does, PET Scan study; healthy men (n=10); Ziprasidone dose (2x40mg capaule); PET Scan performed at 4,8, 12, and 18 hours post-dosing.
128-018-US	Open, randomized, two-way procesover, oral single dose, healthy men (n=12): Ziprasidone dose (3x20 mg capsules and 1x 60 mg capsule).
128-019-US	Open, randomized, two-way crossover oral single dose, healthy men (n=11): Ziprasidone dose (4x20 mg capeules and 1x80 mg capsule).
128-020-US	Open, non-randomized, single oral does triet; healthy men (n=4): "C/#H-ziprasidone dose (20 mg oral suspension),
128-021-US	Open, randomized, two-way crossover, single oral does fasting/fed trial: healthy men (n=12): Zipresidone does (40 mg tablet).
128-022-US	Open, rendomized, two-way crossover, single oral dose tasting / ted trial; healthy men (n=12); ziprasidone dose (40 mg tablet).
128-023-US	Open, randomized, two-way crossover, single oral does tasting / fed that; heatiny men (n=12); ziprasidoné does (40 mg tablet).
128-024-US	Open, randomized, two-way crossover, single oral does fasting / fed that; healthy man (n=12); ziprasidone does (40 mg lablet),
128-025-US	Open, randomized, placebo-controlled, parallel, lithium interaction trial; healtiny men (Zprasidone + Lithium: n=12, pbo + Lithium: n=13); ziprasidone dose (2 x 20 mg capeules bid).
128-025-US	Open, parallel, multiple dose steady-state PK trial; healthy and renally impaired subjects (Group 1 CLor >70: n=10, Group 2 CLor 30-60: n=9, Group 3 CLor 10-29; n=1 Group 4 hemodialysis: n=9; ziprasidone dose (20 mg capsules bid).
128-027-US	Open, non-rendomized, single oral dose triel; healthy men (n=4); 14C/3H-ziprasidone dose (20 mg oral suspension),
128-028-US	Open, parallel, multiple dose PK trial; healthy elderly (n=16) and young (n=19) subjects; ziprasidone dose (20 mg capaules bid),
128-029-US	Open, rendomized, two-way crossover, single oral dose tasting / led trial; healthy men (n=12); ziprasidone dose (40 mg oral suspension).
128-030-US	Open, parallel, multiple dose PK trial; healthy (n=14) and cirrhotic (n=16) subjects; ziprasidone dose (20 mg capsules bid).
128-031-US	Open, randomized, two-way crossover, single oral dose trial; healthy subjects (N=23); ziprasidone dose (capeulos; 20 mg research ve, 20 mg commercial).
128-032-US	Open, non-randomized, cross-over, single intravenous escalating dose trial; subjects with schizophrenia or schizoeffective disorder (n=8); ziprasidone doses (2.6, 6, 10 20, 40 mg).
128-039-UK	Investigator-blind, placebo-controlled, single escalating intramuscular dose trial; healthy males (Ziprasidone: n=10, pbo: n=5); ziprasidone doses (5, 10, 20 mg).
128-034-US	Open, randomized, two-way crossover, single oral dose trial; healthy subjects (n=12); ziprasidone doses (2 x 20 mg capsules vs 40 mg oral suspension).
128-035-US	Open, randomized, two-way crossover, multiple oral dose trial; healthy subjects (n=12); ziprasidone dose (20 mg research capsules bid vs. 20 mg commercial capsules bid).

t

Ongoing Japanese Trials Contributing Safety Data in Original NDA Submission (Japanese data reported separately and not incorporated into the NDA project database)



Additional Trials Reflected in Four-Month Salety Update ("In" reflects number of subjects in project database as of May 15, 1997 data cutoff)

	Pfizer Central Research Sponsored Phase I Clinical Pharmacology Studies
128-045-US	Open, randomized, three-way crossover, multiple dose trial; healthy subjects (n=13); ziprasidone doses (cross-linked capsules; 20 mg, uncross-linked capsules; 20 mg),
128-050-US	Open, randomized, two-way crossover, multiple dose ketoconazole interaction trial; healthy subjects (n=14); ziprasidone dose (2 x 20mg capsules).
	Pfizer Central Research Sponsored Phase II/III Studies
128-307-EU	Fifty-two week, double-blind, randomized, flexible-dose, placebo-controlled trial; subjects with chronic or subchronic schizophrenia (Ziprasidone: n=17, pbc; n= 10); ziprasidone doses (40-60, or 80-100 mg qd).
128-308-EU	Open, flexible-dose, extension trial; subjects with non-organic psychosis (n=0); ziprasidone dose (20-100 mg bid).
	Pfizer Japan Sponsored Phase II/II Clinical Studies
JP-06-602	Fifty-two week, open, flexible-dose, extension trial (n=0); ziprasidone dose (20-60mg bid).

Trials Contributing Safety Data to Original NDA but not to Four-Month Safety Update (Data from these trials are now contained in a separate intramuscular database)

062

128-120-South : Africa	Five-day, open, non-randomized intramuscular to oral dose trial; subjects with psychosis (n=12); ziprasidone doses (intramuscular; 2.5 to 20 mg bid to qid, oral; 20 to 60 mg bid).	7
128-121-US	Open, randomized, haloperidol-controlled, multiple dose intramuscular and oral trial; subjects with psychotic disorder (n=0); ziprasidone doses (20 to 80 mg IM up to qid, 20 to 100 mg oral capsules bid).	$\left \right $

•

Med - Page 91 of 223

Redacted)

pages of trade

secret and/or

confidential

commercial

information

Med - Page 92 of 223

·	f	· · · · · · · · · · · · · · · · · · ·	· ·····	
	t			
	<u>}</u>			
<u>i</u>				
				-
F				-
L	<u> </u>	·		·····

• •	Pfizer Central Research Sponsored Phase II/III Studies
128-101-US	Four-week, double-blind, randomized, parallel, fixed-dose, haloperidol controlled trial; subjects with schizophrenia or schizoaffective disorder (Ziprasidone: n=73, haloperidol: n=17); ziprasidone doses (2, 5, 20, 80 mg bid).
128-102-US	Four-week, open, ascending dose trial; neuroleptic resistant, chronic schizophrenia (n=13); ziprasidone doses (20, 40, 60, 80 mg bid).
128-104-US	Four-week, double-blind, randomized, fixed-dose, placebo-controlled trial; subjects with schizophrenia or schizoaffective disorder (Ziprasidone: n=150, placebo: n=50); ziprasidone doses (5, 20, 40 mg bid).
128-104E-US	One hundred four-week, double-blind, fixed-dose, placebo-controlled extension trial; subjects with schizophrenia or schizoaffective disorder (Ziprasidone: n=6, placebo: n=1); ziprasidone doses (5, 20, 40 mg bld).
128-105-US	Four to Stx week, third party-bilnd, parallel, placebo-controlled trial; subjects with dementia (Ziprasidone: n=11, pbo; n=12); ziprasidone doses (2-6 mg).
128-108-US	Four-week, double-blind, randomized, fixed-dose, placebo-controlled trial; subjects with schizophrenia or schizoaffective disorder (Ziprasidone: n=91, placebo: n=48); ziprasidone doses (20, 60 mg bid).
128-106E-US	Seventy-six week, double-blind, fixed-dose, placebo-controlled extension trial; subjects with schizophrenia or schizoaffective disorder (Ziprasidone: n=21, placebo: n=8); ziprasidone doses (20 mg bid, 20 mg gd, 60 mg bid, 60 mg gd).
128-109-US	Sb-week, double-bland, randomized, fixed-dose, parallel trial; subjects with schizophrenia or schizoaffective disorder (n=35); ziprasidone doses (20 mg qid, 80 mg bid).
128-109E-US	Fifty-two week, open, fixed-dose extension trial; subjects with schizophrenia or schizoaffective disorder (n=6); ziprasidone dose (80 mg qd).
128-110-US	Five-day, open, flexible-dose escalation trial; subjects with psychosis and acute agitation (n=10); ziprasidone dose (20 mg bid to 80 mg bid).
128-111-US	Eight-week, double-blind, randomized, haloperidol-controlled trial; subjects with schizophrenia or schizoaffective disorder and concurrent alcohol/cannable abuse/dependence (Ziprasidone: n=3, haloperidol: n=6); ziprasidone dose (40 mg bid).
128-114-US	Sbc-week, double-blind, randomized, fixed-dose, parallel, placebo-controlled trial; subjects with schizophrenia or schizoaffective disorder (Ziprasidone; n=210, placebo: n=92); ziprasidone doses (40, 80 mg bid).
128-115-US	Sb-week, double-blind, randomized, fixed-dose, parallel, haloperidol and placebo-controlled trial; subjects with schizophrenia or schizoaffective disorder (Ziprasidone: n=338, haloperidol: n=85, placebo: n=83); ziprasidone doses (20, 60, 100 mg bid).
128-118-US	Four-day, open, flexible-dose trial; subjects with psychosis and acute agitation (n=14); ziprasidone dose (20 mg bid to 120 mg bid).
128-120-South Africa	Five-day, open, non-randomized intramuscular to oral dose trial; subjects with psychosis (n=12); ziprasidone doses (intramuscular: 2.5 to 20 mg bid to qid, oral: 20 to 60 mg bid).
128-303-EU	Fifty-two week, double-blind, randomized, fixed-dose, parallel, placebo-controlled trial; hospitalized subjects with chronic or subchronic schizophrenia (Ziprasidone: n=219, pbo: n=75); ziprasidone doses (20, 40, 80 mg bid).

•

ì.

064

; |

ł

1

ţ

Appendix 5.1.2.1 Demographics of	subjects exposed to a	ziprasidone in Phase I clinical trial	(adapted from spot	onsor's electronic submission)
----------------------------------	-----------------------	---------------------------------------	--------------------	--------------------------------

÷

÷

065

			Ziprasid	one		Hali	operidol			Other	
		Hale	Femal	e ti	otal	Hale	Female	Total	Hale	Fenale	Total
Nu	mber of Subjects	59		19	715	4	0	4	44	5	49
Ag	je (years): <18 years 18-64 years 65-74 years >=75 years	57 1		2 04 12 1	6 680 28 1	0 4 0 0	0 0 0 0	0 4 0	0 44 0 0	0 5 0 0	0 49 0 1- 0
	ean age (years) ge range			36.0 1-76	30.6 11-76	30.0 25-40		30.0 25·40	27.2 18-40	27.0 20-37	27 18-
Ra	ace: Asian Black Caucasian Other	6 45 6	ē 1	1 5 02 11	6 74 560 75	0 0 4 0	0 0 0	0 0	2 4 33	1 2 2 0) 6 35 5
					15	U U	Ŷ	v	5	v	3
Ve (CO The sin	ean weight (kg) eight range DMTINUED) e numbers in each trea nce some subjects are otocols: 001.002.004.(032.03	7 49- stment group may counted in more	6.0 115 3 not match t than one of 009.010.011	64.6 4-92 he sum of these gr(013.014.0)	the IN, IV oups. 16,017,018.	79.5 68.93 / and ORAL group .019.020.021.023	os in the 1		78.0 \$7-98	59,7 48-69 1e,	
Ve (CO The sin	eight ränge DWTINUED) e numbers in each trea nce some subjects are ncoss; 001,002,004,0	7 49- stment group may counted in more 105,006,007,008,	6.0 115 115 than one of 009,010,011 37,038,039,0	64.6 4-92 he sum of these gr(013.014.0)	the IN, IV oups. 16,017,018.	79.5 68.93 / and ORAL group .019.020.021.023	os in the 1		78.0 \$7-98	59,7 48-69 1e,	
We (CO The sin Pro	eight ränge DWTINUED) e numbers in each trea nce some subjects are ncoss; 001,002,004,0	7 49- atment group may counted in more 105,006,007,008, 13,034.035,036.0	6.0 115 3 not match ti than one of 009.010.011. 37.038.039.0 Placebo	64.6 4-92 he sum of these gro 013.014.0 40.041.04	the IH, 1\ oups. 16.017.018 3.044.047.(79.5 68.93 / and ORAL group .019.020.021.023	os in the 1		78.0 \$7-98	59,7 48-69 1e,	
We (CO The sin Pro weber je (ye c 1 6	eight range DHTINUED) e numbers in each trea nce some subjects are btocols: 001.002.004.0 032.03	7 49- atment group may counted in more 305,006,007,008, 33,034,035,036,0 Male	6.0 115 3 not match ti than one of 009,010,011 37,038,039,0 Placebo Female	64.6 4-92 he sum of these gr 013.014.0 40.041.04 10.041.04 13 13	the IH, IV oups. 16.017.018 3.044.047.0 5. 0	79.5 68.93 / and ORAL group .019.020.021.023	os in the 1		78.0 \$7-98	59,7 48-69 1e,	
We (CO The sin Pro yeber je (ye 1 6 2	eight range DATINUED) e numbers in each trea nce some subjects are otocols: 001.002.004.0 032.03 of Subjects ears): (18 years 18-64 years 65-74 years 55-74 years 2-75 years pe (years)	7 49- stment group may counted in more 105,006,007,008, 13,034,035,036,0 Hale 111 0	6.0 115 3 115 3 115 3 115 3 115 3 105 3 105 3 115 3 105 3 115 3 105 3 107	64.6 4-92 he sum of these gro 013.014.0 40.041.041 1040 13 13 13 13 13 13 13 13	the IH, IV oups. 16.017.018. 3.044.047.0 5. 5. 5. 0 5. 0 0 0. 8.1 	79.5 68.93 / and ORAL group .019.020.021.023	os in the 1		78.0 \$7-98	59,7 48-69 1e,	
We (CO The sin Pro mber je (ye ci 1 6 5 2 2 2 2 3 2 2 2 3 2 2 3 2 2 3 2 3 2 3	eight range DATINUED) e numbers in each trea nce some subjects are otocols: 001.002.004.0 032.03 of Subjects ears): (18 years 18-64 years 65-74 years 55-74 years 2-75 years pe (years)	7 49- atment group may counted in more 305,006,007,008, 33,034,035,036,0 Male 111 0 111 0 0 28.0	6.0 115 3 115 3 115 3 115 3 115 3 115 3 107 018 018 019 0 Placebo Female 24 0 0 24 0 0 28.5	64.6 4.92 he sum of these gr 013.014.04 40.041.04 10.041.04 13 13 13 13 13 14 14 10 10 10 10 10	the IM, 11 oups. 16,017,018, 3,044.047.0 5 0 5 0 0 8.1 •45 1 6	79.5 68.93 / and ORAL group .019.020.021.023	os in the 1		78.0 \$7-98	59,7 48-69 1e,	

į,

1

Appendix 5.1.2.2 Demographic profile for Phase II/III trials (adapted from sponsor's submission of 8/29/97)

.

Demographic Characteristics

All Oral Dosing Phase II/I	2	iprasidone		H	Haloperidol			* Risperidone		
	Hale	Female	Total	Hale	Female	Total	Male	Female	Total	
Number of Subjects	1824	741	2565	391	194	585	203	92	295	
Age (years): <18 years 18-64 years 65-74 years >=75 years	15 1774 31 4	2 711 24 4	17 2485 55 8	0 387 3 1	0 189 4 1	576 7 2	0 203 0 0	0 87 4 1	290 4 1	
Mean age (years) Age range	38.1 / 7·78	41.9 11·82	39.2 7·82	37.2 18-75	41.0 18-82	30.5 18·82	35.7 18•64	39.6 18·75	36. 10-7	
Race: Asian Black Caucasian Other	32 267 1403 122	10 110 598 23	42 377 2001 145	9 48 323 11	4 24 163 3	13 72 486 14	4 11 169 19	4 11 76 1	8 22 245 20	
Hean weight (kg) Weight range	79.2 27·163	72.8 35-145		80.0 38·149	73.0 40-151		82.2 48-159	72.6 40-123		

(CONTINUED) Protocols: 015.101.102.104.104E.106.106E.108.108E.109.109E.110.111.114.115.1168.117.118.122.301.302.302E.303.304.304E.305.307 Date of Table Generation: 27JUN97

Four month safety update - cumulative Demographic Characteristics

All Oral Dosing Phase II/I		misulpride		Placebo				
	. Hale	Female	Total	Male	Female	Total		
Number of Subjects	29	20	49	281	89	370		
Age (years); <18 years 18-64 years 65-74 years >=75 years	1 28 0 0	0 20 0 0	1 48 0 0	8 266 6 1	4 82 3 0	12 348 9 1		
Hean age (years) Age range	38.0 8-55	34.8 24-48	36.7 8-55		41.8 10-70	39.7 8-76		
Race: Asian Black Caucasian Other	0 0 29 0	0 0 20 0	0 0 49 0	6 50 203 22	2 18 64 5	8 68 267 27		
Mean weight (kg) Weight range	74.8 50-103	67.4 55-86	••••••	78.3 25-133				

1168,117,118,122,301,302,302E,303,304,304E,305,307 Date of Table Generation: 27JUN97

1

Appendix 5.1.3.1 Number of all subjects in phase I trials taking ziprasidone (adapted from sponsor's submission)

	`	40mg	4(mg	8	Omg	H 12	odal Tota Omg	i Daily 16	Oose Per iomg		t IOmg	24	40mg	32	0mg	Tot	:al (\$)
Number of Subjects with Treatment Duration <- 1 day 2-7 days 8-14 days 15-28 days 29-60 days 61-90 days 91-180 days 181-360 days >- 361 days	620 600 000 000	,	2 70 122 14 0 0 0 0		0 1 47 15 0 0 0 0		0 0 13 0 0 0 0		0 0 0 8 0 0 0 0 0 0		0 0 0 0 0 0 0 0 0 0		0 0 0 0 0 0 0 0 0 0 0	·····.	0 0 0 0 0 0 0 0 0 0 0		8 73 169 56 0 0 0 0	(2.6 (23.9 (55.2 (18.3 (0.0 (0.0 (0.0 (0.0 (0.0) (0.0)
Number of Subjects (\$) Nean Duration Range	14 8	(4.6)	208 8	(68.0)	63 13	(20.6)	13 1R	(4,2)	8 18	(2.6)	0	(0.0)	0 n	(0.0)	0 a	(0.0)	306 10	(100.0)

Protocols: 002.005.013.025.026.028.030.035.040.041.043.047.049.203 Date of Table Generation: 10JAN97

Appendix 5.1.3.2 Number of all subjects in phase II/III trial taking ziprasidone (adapted from sponsor's submission)

2-7 days 43 84 77 16 5 1 0 0 226 8-14 days 18 59 86 32 20 13 0 0 228 15-28 days 47 103 125 51 64 26 0 0 417 (н	odal Total Daily	Dose Per Subje	ct				
Treatment Duration $\langle -1 \ day$ 5 7 2 0 0 0 0 14 2-7 \ days 43 84 77 16 5 1 0 0 226 8-14 \ days 18 59 86 32 20 13 0 0 228 15-28 \ days 47 103 125 53 64 26 0 0 417		<40mg	40	ng 80m	9 12	Omg 16	Omg 2	00mg 2	240mg	320#g	Tot	al (\$)
>- 361 days 1 40 89 28 78 16 0 0 252	Treatment Duration <- 1 days 2-7 days 8-14 days 15-28 days 29-60 days 61-90 days 91-180 days 181-360 days	5 43 18 47 15 0 2 1	59 103 59 91 21 24	126 182 62 101 126	32 51 105 84 30 47	116 52 45 62	0 1 3 74 88 21 20 16	0 0 1 0 0 0 0 0 0 0 0 0 0 0	Ň	0 0 0 0 0 0 0 0 0 0	226 228 417 552 377 218 281	(0. (8. (16. (21. (14. (1. (11. (9.

٠

90

Subjects with a modal daily dose not represented above are included in the next lowest dose category. Protocols: 015,101,102,104.104E,106E,108E,108E,109E,109E,110,111,114,115,116E,117.118.122,301.302E,303.304,304E,305.307 Date of Table Generation: 27JUN97

Redacted 3

pages of trade

secret and/or

confidential

Commercial

information

Med - Page 97 of 223

Appendix 7.2.1.2

(from Sponsor's Submission)

Subject Disposition Ziprasidone Protocol 106

	2 Number of S	ubjects	Number of Subj	of Study*	idy*		
Treatment Group	Randomized	Treated	Week 1	Week 2	Week 3	Week 4	
Ziprasidone, 20 mg BID	44	44	42	39	33	28	
Ziprasidone. 60 mg BlD	47	47	42	38	28	24	
lacebo	48	. 48	44	38	28	24	
Total:	139	139	128	115	89	76	
*Based on planned prim							

who have at least one primary efficacy measurements. Weeks are determined by visit designators. Week 1 counts subjects who have at least one primary efficacy measurement at visit 7: Week 2 similarly counts those with visit 14; Week 3 similarly counts those with visit 21: Week 4 similarly counts those with visit 28. Source Data: Appendix V Tables 6, 15, 16. Date of Data Extraction: IBSEP95. Date of Table Generation: 15JAN96.

Appendix 7.2.1.3 (from Sponsor's Submission)

- ---

Demographic Characteristics Ziprasidone Protocol 106

	Zipra	sidone 20	mg B1D	Zipra	Ziprasidone 60 mg B1D			Placebo		
	Male	Female	Total	Male	Female	Total	Male	Female	Total	
Number of Subjects Randomized	30	14	44	39	8	47	41	· 7	48	
Age (years): 18-44 44-64 >=65	. 24 5 1	7 7 0	31 12 1	29 10 0	2 6 0	31 16 0	32 8 1	4 3 0	36 11 1	
Mean age (years) Age range	38.6 25-65	44.6 21-60	40.5 21-65	37.0 19-59	47.5 23-57	38.8 19-59	38.2 21-67	43.7 29-57	39.0 21-67	
Race: Caucasian Black Oriental Other	. 22 5 0 3	9 5 0 0	31 10 0 3	26 9 2 2	8 0 0 0	34 9 2 2	29 7 3 2	6 1 9 0	35 B 3 2	
Mean weight (kg) Weight range	79.9 59·138	67.8 41- 91	••••	79.0 56-126	74.0 44-108	• • • • • • • • • • • • •	80.3 52-108	64.5 54-79	••••••••••••••••••••••••••••••••••••••	

071

..

BPRS Total Score Study 106 (from Sponsor's Submission)

BPRS Total Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 106

				•	freatm	ent Week				
Treatment Groups		eline Hean		ek 1 Mean		ek 2 Mean		ek 3 Nean		ek 4 Nean
Ziprasidone 20 mg BID 60 mg BID	43 41	36.5 36.6	42 41	·3.8 ·5.6	39 38	-5.4 -7.3	33 27	-9.0 -12.4	28 24	-8.9 -16.0
Placebo	47	37.0	44	-4.0	37	-6.1	28	·8.0	Z4	-9.4

Ziprasidone 20 mg B1D	0.838	0.923	0.767	0.395	0.747
vs placebo Zíprasidone 60 mg BlD vs placebo	0.874	0.423	0.373	0.098	0.018
Source Data: Appendix V T Date of Table Generation:	able 15. Date				

BPRS Total Score - Mean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 106

					Treato	ent Week				
Treatment Groups	Bas n	eline Mean	· We n	ek 1 Nean		ek 2 Mean	We n	ek 3 Mean	We n	ek 4 Mean
Ziprasidone 20 mg BID 60 mg BID	43 41	36.5 36.6	43 41	·3.7 ·5.6	43 41	-4.9 -7.2	43 41	-5.7 -8.2	43 41	-5.2 -10.1
Placebo	47	37.0-	47	-3.8	47	-4.3	47	·4.0	47	-4.1

				2-Sided P-Val	ues for Pairwi	se Comparisons	
Ziprasidone vs placebo	20 mg	BID	0.838	0.990	0.773	0.468	0.657
Ciprasidone : vs placebo	60 mg	810	0.874	0.380	0.208	0.108	0.022

Med - Page 99 0 27 2

BPRS Core Items Study 106

:

1

(from Sponsor's Submission)

۰.

BPRS Core Items Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 106

					Treati	ent Veci				
Treatment Groups	Bas n	seline Nean	Ve N	ek 1 Nean	We N	rek 2 Nean	Ne D	ek 3 Nean		tek 4
Ziprasidone					•••••		•••••••	·····.	n • • • • • •	Hean
20 mg BID 60 mg BID	43 41	13.4 13.6	42 41	-2.1 -2.4	39 38	-2.8 -2.9	33 28	-4.1	28 24	•3. •5.
Placedo	47	13.9	44	-2.0	38	·2.0	28	-3.5	24 24	-5. -3.
			2-Side	d P-Valu	ies foi	- Pairwis				
Cipresidone 20 mg BID		••••••	•••••	•••••			e Comp	arisons		
va placedo	Q.	526	Q.	645	0.	178	•			••••
Liprasidone 60 mg 810	Q.	661	٥	555		-	v.	368	0.	619
Source Data: Appendix y 1 Date of Table Generation:						209	0.	210	۵.	096

BPRS Core Items Score - Mean Change From Baseline and P-Values by Week-All Subjects, LDCF Ziprasidone Protocol 106

					Treate	ent"Wee)	(
Treatment Groups	Bas n	e)ine Mean	We n	ek-) Mean	We n	ek 2 Nean	. We	ek 3 Nean	We	ek 4
21prasidone 20 mg BID 60 mg BID Placebo	43 41 47	13.4 13.6 13.9	43 41 47	-2.0 -2.4 -2.0	43 41 47	-2.7 -3.0 -1.9	43 41 47	-3.1 -3.5 -2.2	••••••	-2.6 -2.3

	·····	2-Sided P-Yal	ues for Pairwi	se Comparisons	
Ziprasidone 20 mg 810 vs placebo	0.526	0.812		se Comparisons	•••••
Ziprasidone 60 mg BID		0.012	0.271	0.322	0.677
*> pracebo	0.661	0.587	0.213	0.186	0.059
Source Data: Appendix ¥ 1 Date of Table Generation:	able 15. Date Z3MAY96.	of Data Extra	tion: 28JUL95.	• • • • • • • • • • • • • • • • • • • •	•••••

CGI Severity Score Study 106 (from Sponsor's Submission)

CG1 Severity Score - Mean Change From Baseline and P-Values by Week-All Subjects. Observed Cases Ziprasidone Protocol 106

		ι.		Treatme	ent Week				
Treatment Groups	Baselin n Mea	e W	eek 1 Nean		ek 2 Mean		ek 3 Mean		ek 4 Mean
Ziprasidone 20 mg BID 60 mg BID	43 4 42 4	.7 42 .7 42	-0.1 -0.3	39 38	-0.5 -0.5	33 28	-0.8 -0.8	28 24	•0.7 •1.0
Placebo	47 4	.7 44	•0.1	38	•0.3	28	-0.3	24	-0.5
	•	2·Si	ded P-Val	ues fo	r Pairwi	se Com	parisons		
Ziprasidone 20 mg BlD vs placebo	0.978		0.690	0	. 105	0	.012	C	. 330
Ziprasidone 60 mg B1D vs placebo	0.824		0.215	-	. 233		.024	C	.033

Date of Table Generation: 23MAY96.

.....

. ...

CGI Severity Score - Mean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 106

			Treatment Week		
Treatment Groups	Baseline n Mean	Week 1 n Mean	Week 2 n Mean	Week 3 n Mean	Week 4 n Mean
Ziprasidone 20 mg BID 60 mg BID	- 43 4.7 42 4.7	43 ·0.1 42 ·0.3	43 -0.4- 42 -0.5	43 -0.5 42 -0.5	43 -0.4 42 -0.6
Placebo	47 4.7	47 -0.1	47 -0.2	47 -0.1	47 -0.2
	••••••	2-Sided P-Valu	ies for Pairwis	e Comparisons	
Ziprasidone 20 mg BJD vs placebo	0.978	0.805	0.169	0.034	0.209
Ziprasidone 60 mg BID vs placebo	0.824	0.167	0.238	0.040	0.039

Source Data: Appendix V Table 16. Date of Data Extraction: 28JUL95. Date of Table Generation: 23MAY96.

~ ---

SANS Total Score Study 106

(from Sponsor's Submission)

SANS Total Score - Mean Change from Baseline and P-Values by Week-All Subjects, Observed <u>Cases</u> Ziprasidone Protocol 106

		١	lreatm	ent Veek		
Treatment Groups	Bas(n	e)ine Mean		ek 2 Hean	We n	ek 4 Mean
Ziprasidone 20 mg BlD 60 mg BlD	42 40	52.2 50.9	38 37	-7.7 -3.8	26 22	-11.5 -14.0
Placebo	42	49.1	34	-2.1	22	7.6

2-Sided P-Values for Pairwise Comparisons

ì

Ziprasidone	20	mg	B1D	0.503	0.166	0.462
vs placebo Ziprasidone vs placebo	60	æg	BID	0.706	0.642	0.204
		• • •				

Source Data: Appendix V Table 17. Date of Data Extraction: 30MAY96. Date of Table Generation: 31MAY96.

SANS Total Score - Mean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 106

			Treatm	ent Week		
Treatment Groups	Basi n	eline Mean	We n	ek 2 Mean	We n	ek 4 Nean
Ziprasidone 20 mg BID 60 mg BID	42 40	52.2 50.9	42 40	·6.5 ·3.6	42 40	-8.6 -7.4
Placebo	42	49.1	41	·1.3	42	-2.4

	2-Sided P-Val	ues for Pairwi	se Comparisons
Ziprasidone 20 mg BID vs placebo	0.503	0.192	0.165
Ziprasidone 60 mg BID vs placebo	0.706	0.534	0.197
Source Data: Appendix ¥ Date of Table Generation	Table 17. Date n: 31MAY96.	of Data Extra	ction: 30MAY96.

.

Redacted 9

pages of trade

secret and/or

confidential

commercial

information

Med - Page 103 of 223

Appendix 7.2.2.2

(from Sponsor's Submission)

Subject Disposition Zipresidone Protocol 114

_	Number of S	ubjects	Nu	mber of Subj	ects Completin	ng Each Perio	d of Studyt	••••••
Treatment Group	Randomized	Treated	Week 1	Veek 2	Week 3	Veek 4		
Ziprasidone,		••••••	•••••	•••••			Week 5	Week (
40 mg BID	106	106	103	94	78			
Iprasidone.					/8	69	57	5
30 mg 81D	104	104	103	97				
lacebo			107	9/	92	80	73	62
	92	92	88	77	68	58	54	
						30	50	4
otal:	302	302	294	•••••••••••••		••••••••		
Based on planned primary who have at least one pri	****			268	238	207	180	166

who have at least one primary efficacy measurement at visit 7: Week 2 similarly counts those with visit 14; Week 3 similarly counts those with visit 21: Week 4 similarly counts those with visit 26; Week 5 similarly counts those with visit 35: Week 6 similarly counts those with visit 42. Source Data: Appendix V Tables 6, 15, 16. Date of Data Extraction: 29MAR96. Date of Table Generation: DIAPR96.

Appendix 7.2.2.3 (from Sponsor's Submission)

Demographic Characteristics Ziprasidone Protocol 114

	Zipr	Ziprasidone 40 mg BID			sidone E	10 mg BID	Placebo			
	Hale	Female	Total	Hale	Female	Total		Female	• • • • • • • •	
Number of Subjects Randomized	75	31	106		27	104	63	•••••	Total	
Age (years): 18-44 45-64	60 14	23	83				••••••	29 -	92	
>=65	ł		21 2	66 11 0	7	18 1	51 12 0	20 9 0	71 21 0	
Mean age (years) Age range	35.6 19-65	39.6 24-67	36.8 19-67	34.6 18-58	39.1 24-65	35.8 18-65	35.7 18-63	40.4	37.2	
Race: White Black Asian Other	52 15 1 7	25 4 1 1	77 19 2 8	55 11 3 8	18 6 0 3	73 17 3 11	39 15 2	18-64 17 9 0	18-64 56 24 2	
Hean weight (kg) Height range Source Data: APPENDIX		68.6 50-111		78.6 49-127	72.0 44-101		7 82.4 51-122	3 73.1 49-118	10	

urce Data: APPENDIX V - TABLE 2 Date of Data Extraction: 26MAR96 Date of Table Generation: 27MAR96



BPRSd Total Score Study 114

(from Sponsor's Submission)

BPRSd Total Score - Hean Change From Baseline and P-Values by Week - All Subjects, Observed Cases Ziprasidone Protocol 114

		t	reatment Week*			
Baseline n Mean	Week 1 n Mean	Week 2 n Mean	Veek 3 n Mean	Veek 4 n Mean	Week 5 n Mean	Week 6 n Nean
104 56.5 103 55.0	103 -4.1 103 -6.3	94 -4.0 96 -8.4	78 -9.1 92 -11.2	69 -11.6 80 -12.1	57 -15.8 73 -13.3	54 -15.5 67 -13.9
91 55.1	87 -1.1	77 -4.9	68 -6.0	58 -8.5	50 .10.7	45 -12.2
	2	Sided P-Value	s for Pairwise	Comparisons**		
0.4521	0.0381	0.6611	0,1020	0.1574	0.0147	0.1747
0.9250	0.0001	0.0127	0.0021	0.0315	0.1566	0.4410
	n Hean 104 56.5 103 55.0 91 55.1 0.4521	n Hean n Hean 104 56.5 103 -4.1 103 55.0 103 -6.3 91 55.1 87 -1.1 2 0.4521 0.0381	Baseline n Week 1 n Week 1 Hean Week 2 n Meek 2 n 104 56.5 103 -4.1 94 -4.0 103 55.0 103 -6.3 96 -8.4 91 55.1 87 -1.1 77 -4.9 2-Sided P-Yalue 0.4521 0.0381 0.6611	n Mean n Mean n Mean n Mean 104 56.5 103 -4.1 94 -4.0 78 -9.1 103 55.0 103 -6.3 96 -8.4 92 -11.2 91 55.1 87 -1.1 77 -4.9 68 -6.0 2-Sided P-Values for Pairwise 0.4521 0.0381 0.6611 0.1020	Baseline n Week 1 n Week 2 n Week 3 n Week 3 n Week 4 n Week 3 n Week 4 n Week 3 n Week 4 n Meen n Week 4 n Nean n Nea n Ne	Baseline n Week j Nean Week 2 n Week 3 n Week 3 Nean Week 4 Nean Week 5 n Week 5 n 104 56.5 103 -4.1 94 -4.0 78 -9.1 69 -11.6 57 -15.8 103 55.0 103 -6.3 96 -8.4 92 -11.2 80 -12.1 73 -13.3 91 55.1 87 -1.1 77 -4.9 68 -6.0 58 -8.5 50 -10.7 2-Sided P-Values for Pairwise Comparisons** 0.4521 0.0381 0.6611 0.1020 0.1574 0.0147

*Baseline - last visit prior to double-blind treatment; Week] = visit 7; Week 2 = visit 14; Week 3 = visit 21; Week 4 = visit 28 Week 5 = visit 35; Week 6 = visit 42. Missing values are imputed using value of previous non-missing visit, planned or unplanned. **Estimates of treatment effects are based on least squares means (LSMEANS) derived from an ANCOVA model with baseline response As covariate and (fixed effect) terms for center and treatment. The p-values are derived from the respective t-tests. Source Data: Appendix V Table 15. Date of Data Extraction: Z9NAR96. Date of Table Generation: 17APR96.

BPRSd Total Score - Mean Change From Baseline and P-Values by Week - All Subjects, LOCF Ziprasidone Protocol 114

							Treatme	nt Week	•					
Treatment Groups		eline Mean		ek) Mean		ek 2 Mean		ek 3 Mean	¥e n	ek 4 Mean		ek 5 Mean		ek 6 Mean
Ziprasidone 40 mg BID 80 mg BID	104 103	56.5 55.0	104 103	-4.1 -6.3	104 - 103	-3.4 -8.1	104 103	·6.2 ·10.6	104 103	-7.2 -10.5	104 103		104 103	-7.7 -10.3
Placebo	91	55.1	90	-0.8	91	·2.6	91	-2.3	91	-2.9	91	-3.2	91	-3.4
					2-Side	d P-Valu	es for	Pairwis	e Compar	isons**				•••••
Ziprasidone 40 mg BID	0.	4521	0.	0236	0.	6489	0.	0419	0.	.0439	0.	0240	0.	0472
vs placebo Ziprasidone 80 mg BID vs placebo	0.	9250	0.	0001	0.	0003	0.	0001	0.	0001		0001	0.	.0003
*Baseline - last visit p Week 5 - visit 35; Weel *Estimates of treatment As covariate and (fixed Source Data: Appendix Y	t 6 - vi t effect t effect	sit 42. Is are ba	Missing sed on for cen	values least sq	are imp juares m treatme	uted usi means (LS int. The	ng valu (MEANS) (D-value	derived	evious (from a) lerived (non-missi h ANCOVA from the	ng visi model v respect	it, plann with base	ied or ()ine re	Jubienue

Med - Page 105 of 223

. . . .

BPRSd Core Items Study 114 (from Sponsor's Submission)

.

÷

-

۰.

/BPRSd Core Items Score - Mean Change From Baseline and P-Values by Week - All Subjects, Observed Cases Ziprasidone Protocol 114

.

:

<u>·</u>··

.

						-	Treatae	nt Veek*						
Treatment Groups	Bas R	eline Mean	Ve n	ek 1 Nean	We N	ek 2 Mean	Ve n	ek 3 Nean	Ne n	ek 4 Nean		ek 5 Nean	We n	ek 6 Nean
Ziprasidone" 40 mg BID 80 mg BID	104 103	16.9 16.6	103 103	-1.9 -2.5	94 96	-2.0 -3.7	78 92	·3.6 ·4.5	69 80	-4.5 -5.1	· 57 73	-5.6 -5.8	54 67	-5.9
Placebo	91	16.4	87	-1.0	77	-2.2	68	-2.6	58	·3.2	50	-4.0	67 45	-5.8 -4.5
· ·	• • • • • •	•••••••			2-Side	d P-Value	s for f	Pairwise	Compar	isons**				
liprasidone 40 mg 810 vs placebo	0.:	3573	0.	0380	0.0	5995	0.1	487	0.1	1123	0.0			 1714
Ziprasidone 80 mg 810 vs placebo	0.1	136	0.0	0003	0.0	0045	0.0	025	0.0	0071		088		1533

Baseline - last visit prior to double-blind treatment: Week 1 - visit 7; Week 2 - visit 14; Week 3 - visit 21; Week 4 - visit 28 Week 5 - visit 35; Week 6 - visit 42. Missing values are imputed using value of previous non-missing visit, planned or unplanned. *Estimates of treatment effects are based on least squares means (LSMEANS) derived from an ANCOVA model with baseline response As covariate and (fixed effect) terms for center and treatment. The p-values are derived from the respective t-tests. Source Data: Appendix V Table 15. Date of Data Extraction: 29MAR96. Date of Table Generation: DZAPR96.

...

BPRSd Core Items Score - Mean Change From Baseline and P-Values by Week - All Subjects, LOCF Ziprasidone Protocol 114

	•													
Treatment Groups	B93 N	eline Mean		ek 1 Mean		ek 2 Mean		ek 3 Kean		ek 4 Nean		ek 5 Mean		k 6 Mean
Ziprasidone 40 mg BID 80 mg BID	104 103	16.9 16.6	104 103		104 103	-1.8 -3.5	104 103	-2.7 -4.2	104 103	-3.1 -4.3	104 103	-3.3 -4.5	104 103	-3.4 -4.4
Placebo	91	16.4	90.	-0.9	91	•1.6	91	-1.6	91	-1.7	91	-1.9	91	-2.0

		************					• • • • • • • • • • • • • • • • • • • •
Ziprasidone 40 mg BID vs placebo	0.3573	0.0256	0.5929	0.0553	0.0260	0.0207	0.0396
Ziprasidone 80 mg B1D vs placebo	0.7136	0.0002	0.0002	0.0003	0.0001	0.0001	0.0002

**Baseline - last visit prior to double-blind treatment: Week 1 - visit 7: Week 2 - visit 14: Week 3 - visit 21: Week 4 - visit 28 Week 5 - visit 35: Week 6 - visit 42. Nissing values are imputed using value of previous non-missing visit, planned or unplanned. *Estimates of treatment effects are based on least squares means (LSNEANS) derived from an ANCOVA model with baseline response As covariate and (fixed effect) terms for center and treatment. The p-values are derived from the respective t-tests. Source Data: Appendix V Table 15. Date of Data Extraction: 29MAR96. Date of Table Generation: DIAPR96.

CGI Severity Score Study 114

(from Sponsor's Submission)

CGI Severity Score - Mean Change From Baseline and P-Values by Week - All Subjects, Observed Cases Ziprasidone Protocol 114

						1	reatme	nt Week*						
Treatment Groups		line Nean		ek 1 Mean		ek 2 Mean		ek 3 Nean	Ve n	ek 4 Nean		ek 5 Mean		k 6 Nean
Ziprasidone 40 mg BID 80 mg BID	104 103	4.8 4.8	103 103	-0.3	94 97	-0.2 -0.6	78 92	-0.5 -0.7	69 80	•0.6 •1.0	57 73	-0.9 -1.0	54 67	-1.0 -1.1
Placebo	92	4.8	88	•0.1	77	-0.3	68	-0.3	58	•0.5	50	-0.6	45	-0.8
					2-Side	d P-Valu	es for I	Pairwise	Compar	isons**				
Ziprasidone 40 mg BID	0.9	9477	0.	1244	0,	9983	0.0	0677	0.	2355	0.	0162	0.2	2157
vs placebo Ziprasidone 80 mg BID vs placebo	0.9	6024	0.	0166	. O.	0111	0.	0101	0.	0010	0.	0040	0.0	0281

Week 5 - visit 35; Week 6 - visit 42. Missing values are imputed using value of previous non-missing visit, planned or unplanned. *Estimates of treatment effects are based on least squares means (LSMEANS) derived from an ANCOVA model with baseline response As covariate and (fixed effect) terms for center and treatment. The p-values are derived from the respective t-tests. Source Data: Appendix V Table 16. Date of Data Extraction: 29MAR96. Date of Table Generation: 02APR96.

CGI Severity Score - Mean Change From Baseline and P-Values by Week - All Subjects, LOCF Ziprasidone Protocol 114

.

							Treatme	nt Week*						
Treatment Groups		elfne Mean		ek] Nean	We n	ek 2 Mean	We n	ek 3 Mean		ek 4 Mean		ek 5 Mean		ek 6 Mean
Ziprasidone 40 mg BID 80 mg BID	104 103	4.8 4.8	104 103	-0.3 -0.3	104 103	·0.2 ·0.5	10 4 103	-0.4 -0.6	104 103	-0.4 -0.8	104 103	-0.5 -0.8	104 103	-0.5 -0.8
Placebo	92	4.8	91	-0.1	92	•0.1	92	•0.1	92	·0.2	92	•0.2	92	·0.2
					2-Side	d P-Valu	es for I	Pairwise	Compart	sons**				
Ciprasidone 40 mg BID vs placebo	0.9	9477	0.0	0864	0.	6033		0324	0.0)649 ·	0.0	0096	0.0	299
ziprasidone 80 mg BlD vs placebo	0.9	5024	0.0	0115	0.	0007	0.0	0002	0.0	0001	0.	0001	0.0	001

*Baseline = last visit prior to double-blind treatment; Week 1 = visit 7; Week 2 = visit 14; Week 3 = visit 21; Week 4 = visit 28 Week 5 = visit 35; Week 6 = visit 42. Hissing values are imputed using value of previous non-missing visit, planned or unplanned. *Estimates of treatment effects are based on least squares means (LSMEAMS) derived from an ANCOVA model with baseline response As covariate and (fixed effect) terms for center and treatment. The p-values are derived from the respective t-tests. Source Data: Appendix V Table 16. Date of Data Extraction: 29NAR96. Date of Table Generation: 17APR96.

088

_ _

PANSS Total Score Study 114

(from Sponsor's Submission)

PAMSS Total Score - Mean Change From Baseline and P-Values by Week - All Subjects, Observed Cases Ziprasidone Protocol 114

			۱	Freatment Week*			
Treatment Groups	Baseline n Mean	Week 1 n Mean	Week 2 n Mean	Week 3 n Mean	Week 4 n Mean	Week 5 n Mean	Week 6 n Mean
Ziprasidone 40 mg BID 80 mg BID	104 98.2 103. 95.8	103 -7.0 103 -9.9	94 -6.5 96 -14.0	78 -14.3 92 -18.6	69 -18.4 80 -20.2	57 ·25.3 73 ·22.2	54 -25.6 67 -23.5
Placebo	91 97.3	86 -1.6	77 · 7.7	68 -9.3	58 -13.8	50 -19.0	44 -21.0
			2-Sided P-Value	es for Pairwise	Comparisons**	•••••	
Ziprasidone 40 mg BID vs placebo	0.7859	0.0169	0.8971	0.0743	0.1126	0.0265	0.1217
Ziprasidone 80 mg B1D vs placebo	0.6324	0.0001	0.0058	0.0005	0.0120	0.1721	0.3418

*Baseline - last visit prior to double-blind treatment; Week 1 - visit 7; Week 2 - visit 14; Week 3 - visit 21; Week 4 - visit 28 Week 5 - visit 35; Week 6 - visit 42. Missing values are imputed using value of previous non-missing visit, planned or unplanned. **Estimates of treatment effects are based on least squares means (LSMEANS) derived from an ANCOVA model with baseline response as covariate and (fixed effect) terms for center and treatment. The p-values are derived from the respective t-tests. Source Data: Appendix V Table 15. Date of Data Extraction: 29MAR96. Date of Table Generation: 03APR96.

PANSS Total Score - Mean Change From Baseline and P-Values by Week - All Subjects, LOCF Ziprasidone Protocol 114

Treatment Groups	Treatment Week*													
	Baseline n Mean		Week] n Mean		Week 2 n Mean		Week 3 n Mean		Week 4 n Mean		Week 5 n Nean		Week 6 n Mean	
- Ziprasidone 40 mg BID 80 mg BID	104 103	98.2 95.8	104 103	-6.9 -9.9	104 103	-5.5 -13.4	104 103	-9.7 -17.4	104 103	-11.3 -17.2	104 103	-12.6 -17.0	104 103	-12.4 -17.1
Placebo	91	97.3	89	-1.2	91	-3.8	91	-3.2	91	-4.3	91	-5.0	91	-5.4
					2-Side	d P-Valu	es for	Pairwise	Compar	isons**				
Ziprasidone 40 mg BID	0.7859		0.0100		0.5010		0.0274		0.0307		0.0250		0.0478	

vs placebo Ziprasidone 80 mg 810 0.6324 0.0001 0.0002 0.0001 0.0001 0.0001 0.0002 vs placebo

*Baseline = last visit prior to double-blind treatment: Week 1 = visit 7; Week 2 = visit 14; Week 3 = visit 21; Week 4 = visit 28 Week 5 = visit 35; Week 6 = visit 42. Missing values are imputed using value of previous non-missing visit, planned or unplanned. **Estimates of treatment effects are based on least squares means (LSMEANS) derived from an ANCOVA model with baseline response as covariate and (fixed effect) terms for center and treatment. The p-values are derived from the respective t-tests. Source Data: Appendix V Table 15. Date of Data Extraction: 29MAR96. Date of Table Generation: 01APR96.

089
PANSS Negative Score Study 114 (from Sponsor's Submission)

PANSS Negative Subscale Score - Mean Change From Baseline and P-Values by Week - All Subjects. Observed Cases Ziprasidone Protocol 114

Treatment Groups	8.	eline.					Treatm	ent Week*						
Treatment Groups	n 	Nean	We 	ek 1 Mean	Ne N	ek 2 Mean	Л	ek 3 Nean		ek 4	Week 5			
Zipresidone					••••••	••••••	••••••••		n	Nean	n	Nean	n we	ek 6 Mean
40 mg BID 80 mg BID Placebo	104 103 91	25.4 24.3 24.9	103 103 86	-2.1 -2.4 0.3	94 96 77	-1.9 -3.1 -1.3	78 92 68	-3.3 -4.1 -1.7	69 80 58	-3.9 -4.7 -2.6	57 73 50	-5.7 -5.1 -5.0	54 67 45	-5.7 -5.2 -5.4
	•••••		•••••	••••••	2-Sided	[P-Value	s for P	airvise (Comnari	*****				
Ziprasidone 40 mg BJD V3 placebo	0.6	687	0.0		0.4		0.1			•••••	• • • • • • • •	••••••	• • • • • • •	
Ziprasidone 80 mg BID Vs placebo	0.5	793	0.0	001	0.0	-	0.0	-	0.0		0.1		0.2	144

*Baseline = last visit prior to double-blind treatment: Week] = visit 7: Week 2 = visit 14: Week 3 = visit 21: Week 4 = visit 28 Week 5 = visit 35: Week 6 = visit 42. Missing values are imputed using value of previous non-missing visit. Planned or unplanned. As covariate and (fixed effect) terms for center and treatment. The p-values are derived from the respective t-tests. Source Data: Appendix V Table 15. Date of Data Extraction: 29MAR96. Date of Table Generation: DZAPR96.

PANSS Negative Subscale Score - Mean Change From Baseline and P-Values by Week - All Subjects, LOCF Ziprasidone Protocol 114

						Treatme	nt Week*	•					
Bas n	seline Mean	We n	ek 1 Mean			We n	ek 3" Mean			We			ek 6 Mean
104 103 91	25.4 24.3 24.9	104 103 B9	-2.1 -2.4 0.3	104 103 91	-1.6 -3.0 -0.3	104 103 91	-2.4 -4.0 -0.4	104 103 91	-2,7 -4.0 -0.4	104 103 91	-3.2 -3.8 -1.0	104 103 91	·3.2 ·3.9 ·0.9
•••••	•••••	•••••		2-Sider	d P-Valu	es for l	Pairwise	Compar	isons*				
0.6	5687	0.0					•••••••	•••••	••••••	•••••	•••••	••••••	
0.5	5793	0.0	001	0.0	007								236
	n 104 103 91 	104 25.4 103 24.3	n Mean n 104 25.4 104 103 24.3 103 91 24.9 B9 0.6687 0.0	n Hean n Hean 104 25.4 104 -2.1 103 24.3 103 -2.4 91 24.9 B9 0.3 0.6687 0.0004	n Mean n Hean n 104 25.4 104 -2.1 104 103 24.3 103 -2.4 103 91 24.9 B9 0.3 91 2-Sider 0.6687 0.0004 0.1	Baseline n Week 1 n Week 2 n Nean 104 25.4 104 -2.1 104 -1.6 103 24.3 103 -2.4 103 -3.0 91 24.9 B9 0.3 91 -0.3 Z-Sided P-Value 0.6687 0.0004 0.1566	Baseline n Week 1 n Week 2 n Mean n Mean n Mean n Mean n Mean n Mean n n 104 25.4 104 -2.1 104 -1.6 104 103 24.3 103 -2.4 103 -3.0 103 91 24.9 B9 0.3 91 -0.3 91 2-Sided P-Values for 1 2-Sided P-Values for 1 0.6687 0.0004 0.1566 0.0	Baseline n Week 1 n Week 2 n Week 3 n Week 3 n Week 3 n Mean n Mean n Mean n Mean n Mean n	n Mean n Mean n Mean n Mean n Mean n 104 25.4 104 -2.1 104 -1.6 104 -2.4 104 103 24.3 103 -2.4 103 -3.0 103 -4.0 103 91 24.9 B9 0.3 91 -0.3 91 -0.4 91 2-Sided P-Values for Pairwise Compar 0.6687 0.0004 0.1566 0.0267 0.1	Baseline n Week 1 n Week 2 n Week 3 n Week 3 n Week 4 n Week 4 n Week 3 n Week 4 n Week 4 n	Baseline Week 1 Week 2 Week 3" Week 4 Week 4 n Mean Mea Mea Mea	Baseline Week 1 Week 2 Week 3'' Week 4 Week 5 n Mean n n n n n n n n n n n n n n	Baseline n Week 1 n Week 2 n Week 3 n Week 3 n Week 4 n Week 5 n Week 7 n Week 7 n

**Baseline - last visit prior to double-blind treatment; Week 1 - visit 7; Week 2 - visit 14; Week 3 - visit 21; Week 4 - visit 28 Week 5 - visit 35; Week 6 - visit 42. Missing values are imputed using value of previous non-missing visit, planned or unplanned. *Estimates of treatment effects are based on least squares means (LSMEANS) derived from an ANCOVA model with baseline response as covariate and (fixed effect) terms for center and treatment. The p-values are derived from the respective t-tests. Source Data: Appendix V Table 15. Date of Data Extraction: Z9MAR96. Date of Table Generation: DIAPR96.

Redacted 10

pages of trade

secret and/or

confidential

Commercial

information

Appendix 7.2.3.2

(from Sponsor's Submission)

Subject Disposition Ziprasidone Protocol 115

.

	Number of S	iubjects	Number of Subjects Completing Each Period of Study*								
Treatment Group	Randomized	Trested	Week 1	Week 2	Week 3	Week 4	Week 5	Veek			
Ziprasidone. 20 mg BID	87	87	84	77	62	57 -	51	5			
Ziprasidone. 60 mg BID	. 78	78	75	69	62	57	50	3			
Ziprasidone. 100 mg BID	86	86	82	77	69	60	57	4			
Haloperidol	85	85	77	72	67	58	50	4			
Placebo	83	83	77	65	48	35	32	2			
Total;	419	419	395	360	308	267	Z40				

*Based on planned primary efficacy measurements. Weeks are determined by visit designators. Meek 1 counts subjects who have at least one primary efficacy measurement at visit 7: Meek 2 similarly counts those with visit 14; Meek 3 similarly counts those with visit 21; Meek 4 similarly counts those with visit 28; Meek 5 similarly counts those with visit 35; Meek 6 similarly counts those with visit 42. Source Data: Appendix V Tables 6, 15, 16. Date of Data Extraction: 220CT96. Date of Table Generation: 220CT96.

APPEARS THIS WAY ON ORIGINAL

Med - Page 111 of 228 1 -

Appendix 7.2.3.3 (from Sponsor's Submission)

Demographic Characteristics Ziprasidone Protocol 115

	Zipras	Idone 2	D mg 81D	Zipras	idone 60	mg BID	Zipras	idone 100) mg BID
•	Nale	Female	Total	Nale	Female	Total	Nale	Female	Total
Number of Subjects Randomized	53	34	87	55	23	78	55	31	86
Age (years); 18-44 45-64 > ~6 5	35 18 0	25 7 2	60 25 2	42 11 2	15 8 0	57 19 2	43 12 0	20 10 1	63 22 1
Mean age (years) Age range	41.1 22-60	39.1 21-68	40.3 21-68	39.7 21-72	40.7 20-58	40.0 20-72	36.9 19-60	40.5 21-71	38.2 19·71
Race: White Black Asian Other	32 14 2 5	23 9 2 0	55 23 4 5	37 16 2 0	17 5 1 0	54 21 3 0	37 13 0 5	21 8 1 1	58 21 1 6
Nean weight (kg) Weight range	78.5 54-117	79.9 40·125	•••••	78.7 51-109	78.5 45-110	• • • • • • • • • • •	B1.9 50∙130	72.8 46:104	

Demographic Characteristics Ziprasidone Protocol 115

		Haloperi	fob		Placeb	0
· -·	Male	Female	Total	Male	Female	Total
Number of Subjects Randomized	60	25	85	54	29	83
Age (years): 18-44 45-64 >=65	44 16 0	18 5 2	62 21- 2	40 13 1	20 9 0	60 22 1
Mean age (years) Age range	38.1 18-64	40.6 21-69	38.8 18-69	38.3 18-65	40.0 18-56	38.9 18-65
Race: White Black Asian Other	37 18 2 3	18 6 1 0	55 24 3 3	30 13 1 10	20 6 1 2	50 19 2 12
Nean weight (kg) Weight range	84,4 51-141	73.6 49-110	•••••	79.7 51-133	75.9 50-112	

Source Data: APPENDIX V - TABLE 2 Date of Data Extraction: 220CT96 Date of Table Generation: 220CT96

BPRSd Total Score Study 115

(from Sponsor's Submission)

BPRSd Total Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 115

						ILGOUN	ent week						
Baseline n Mean		Week 1 n Mean		Veek 2 n Nean		Ne n	ek 3 Mean	Week 4 n Mean		Week 5 n Mean		We n	ek 6 Mean
86 76 82	53.8 51.8 51.8	84 75 82	-3.5 -4.1 -3.5	76 68 77	-5.0 -5.0 -5.9	61 62 69	-8.4 -6.5 -7.1	57 56 60	-9.3 -7.7 -7.8	51 50 57	-9.0 -7.8 -7.4	50 - 39 47	-11.0 10.5 -8.3
82	53.9	77	-5.3	72	-7.5	67	·8.2	58	-11.8	50	-11.3	47	-12.5
80	54.3	77	-0.7	64	·2.7	48	-4.5	35	-8.7	32	-8.1	27	-9.6
	n 86 76 82 82	n Mean 86 53.8 76 51.8 82 51.8 82 53.9	n Mean n 86 53.8 84 76 51.8 75 82 51.8 82 82 53.9 77	n Mean n Mean 86 53.8 84 -3.5 76 51.8 75 -4.1 82 51.8 82 -3.5 82 53.9 77 -5.3	n Mean n Mean n 86 53.8 84 -3.5 76 76 51.8 75 -4.1 68 82 51.8 82 -3.5 77 82 53.9 77 -5.3 72	Beseline n Week 1 Neen Week 2 n Week 2 n 86 53.8 84 -3.5 76 -5.0 76 51.8 75 -4.1 68 -5.0 82 51.8 82 -3.5 77 -5.9 82 53.9 77 -5.3 72 -7.5	Baseline n Week 1 n Week 2 n Week 2 n	n Mean n Mean n Mean n Mean 86 53.8 84 -3.5 76 -5.0 61 -8.4 76 51.8 75 -4.1 68 -5.0 62 -6.5 82 51.8 82 -3.5 77 -5.9 69 -7.1 82 53.9 77 -5.3 72 -7.5 67 -8.2	Baseline n Week 1 n Week 2 n Week 2 n Week 3 n Week 3 n	Baseline n Week 1 n Week 2 n Week 3 n Week 3 n Week 4 n Week 4 n 86 53.8 84 -3.5 76 -5.0 61 -8.4 57 -9.3 76 51.8 75 -4.1 68 -5.0 62 -6.5 56 -7.7 82 51.8 82 -3.5 77 -5.9 69 -7.1 60 -7.8 82 53.9 77 -5.3 72 -7.5 67 -8.2 58 -11.8	Baseline n Week 1 n Week 2 n Week 3 n Week 4 n Week 4 n Week 3 n Week 4 n Week 3 n Week 4 n Week 1 n Week 3 n Week 4 n Week 1 n Week 3 n Week 4 n Week 1 n Week 3 n Week 4 n Week n Meen n Week 1 n Week 1 n Week 3 n Week 4 n Week n Meen n Week 1 n Week 3 n Week 4 n Ween n Ween n Ween n Ween n Ween n W	Baseline n Week 1 n Week 2 n Week 3 n Week 3 n Week 4 n Week 5 n Week 5 n	Baseline n Week 1 n Week 2 n Week 3 n Week 3 n Week 4 n Week 5 n Week 5 n

2-Sided P-Values for Pairwise Comparisons

Treatment Heat

749	D.015	0.111	0.057	0.598	0.824	0.717
096	0.004	0.096	0.241	0.758	0.929	0.757
095	0.011	0.024	0.187	0.987	0.907	0.741
.754	0.000	0.004	0.228	0.275	0.254	0.363
,	.096 .095	.096 0.004 .095 0.011	096 0.004 0.096 095 0.011 0.024	.096 0.004 0.096 0.241 .095 0.011 0.024 0.187	096 0.004 0.096 0.241 0.758 095 0.011 0.024 0.187 0.987	096 0.004 0.096 0.241 0.758 0.929 095 0.011 0.024 0.187 0.987 0.907

Source Data: Appendix V Table 15. Date of Data Extraction: 205EP96. Date of Table Generation: 305EP96.

...

/

BPRSd Total Score - Mean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 115 -. - ---- ·

				Treatment Week			
Treatment Groups	Baseline n Mean	Week) n Mean	Week 2 n Mean	Week 3 n Nean	Week 4 n Mean	Week 5 n Mean	Week 6 n Mean
Ziprasidone 20 mg BID 60 mg BID 100 mg BID	86 53.8 76 51.8 82 51.8	86 -3.3 76 -4.0 82 -3.5	86 -3.6 76 -4.4 82 -5.1	86 -5.5 76 -4.8 82 -5.4	86 -5.2 76 -4.8 82 -5.5	86 -4.2 76 -4.8 82 -5.0	86 -4.9 76 -5.2 82 -5.2
Haloperido]	82 _ 53.9	. 82 -4.9	82 -6.7	82 -7.2	82 ·8.9	82 -8.5	62 ·8.8
Placebo	80 54.3	78 -0.9	80 -1.6	80 -1.6	80 -1.9	80 -1.4	80 -1.2

			2-Sided P-Values for Pairwise Comparisons										
Ziprasidone 20 mg BID	0.749	0.038	0.166	0.026	0.070	0.129	0.049						
vs placebo (1prasidone 60 mg BID	0.096	0.004	0.030	0.035	0.073	0.048	0.020						
vs placebo Liprasidone 100 mg BID	0.095	0.020	0.013	0.020	0.040	0.041	0.023						
vs placebo Haloperidol vs placebo	0.754	0.001	0.001	0.003	0.000	0.000	0.000						

Source Data: Appendix Y Table 15. Date of Data Extraction: 20SEP96. Date of Table Generation: 27SEP96.

.......

BPRSd Core Items Study 115. (from Sponsor's Submission)

 $\mathbf{r}_{\mathbf{s}}$

BPRSd Core Items Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 115

÷ ł

							Treate	ent Yeel	k i					
Treatment Groups	Bas n	eline Nean	Ne n	ek 1 Mean	Ve n	ek 2 Nean	Ne n	ek 3 Nean	We n	ek 4 Nean		ek 5 Nean	We	ek 6 Mean
Ziprasidone 20 mg BID 60 mg BID 100 mg BID Naloperidol Placebo	86 76 82 82 80	16.1 16.0 15.9 16.2 16.6	75 82 77 77	-1.5 -1.5 -1.6 -2.3 -0.9	76 68 77 72 64	-2.0 -1.9 -2.5 -3.2 -1.9	61 62 69 67 48	-2.9 -2.7 -3.3 -3.7 -2.2	57 56 60 58 35	-3.1 -3.1 -3.5 -4.8 -3.7	51 50 57 50 32	-3.3 -2.8 -3.6 -4.6 -4.3	50 39 47 47 27	-3.9 -3.9 -4.1 -5.2 -3.9

	· · · · · · · · · · · · · · · ·		2-Sided P-Va	lues for Pairw	ise Comparison	s .	
Ziprasidone 20 mg BID vs placebo	0.326	0.133	0.767	0.389	0.534	• • • • • •	•••••
Ziprasidone 60 mg BiD	0.225	0.130	0.848			0.269	0.985
vs placebo Ziprasidone 100 mg BID	0.146		0.040	0.447	0.512	0.127	0.819
vs placebo Haloperidol	0.140	0.069	0.208	0.132	0.990	0.594	0 7 30
vs placebo	0.459	0.002	0.017	0.078	0 003		0.578
Courses 0				0.070	0.227	0.595	0,119

Source Data: Appendix V Table 15. Date of Data Extraction: 205EP96. Date of Table Generation: 305EP96.

BFRSd Core Items Score - Mean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 115

							Treats	ent Week						
Treatment Groups		eline Mean	We n	ek 1- Nean	We n	ek 2 Mean	We n	ek 3 Mean	We n	ek 4 Nean	We N	ek 5 Mean	We n	ek 6 Nean
Ziprasidone 20 mg BID 60 mg BID 100 mg BID	86 76 82	16.1 16.0 15.9	86 76 82	-1.4 -1.5 -1.6	86 76 82	-1.5 -1.8 -2.2	86 76 82	-2.2 -2.1 -2.5	86 76 82	-2.1 -2.2 -2.4	86 76 82	-2.1 -2.0 -2.3	86 76 - 82	-2.3 -2.2 -2.6
Haloperidol Placebo	82 80	16.2 16.6	82 78	-2.2 -0.9	82 80	·3.0	82	-3.4	82	-4.0	82	-3.9	82	-4.1
				-0.5	80	-1.4	80	·1.0	80	-1.2	80	-1.3	80	-0.9

			2-Sided P-Va	lues for Pairw	ise Comparison:	5	
Ziprasidone 20 mg BID vs placebo	0.326	0.196	0.636	0.039	0.103	0.206	0.034
Ziprasidone 60 mg BID Vs placebo	0.225	0.167	0.339	0.051	0.083	0.252	0.046
Ziprasidone 100 mg B1D vs placebo	0.146	0.101	0.099	0.010	- 0.044	0.113	0.009
Haloperido] vs placebo	0.459	0.005	0.004	0.000	0.000	0.000	0.000

Source Data: Appendix V Table 15. Date of Data Extraction: 205EP96. Date of Table Generation: 305EP96.

.- .

CGI Severity Score Study 115 (from Sponsor's Submission)

CGI Severity Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 115

		- • •					Treatm	ient Veel	k 🛛					
Treatment Groups	Bas n	eline Mean	We n	ek 1 Nean	We N	ek 2 Nean	We R	ek 3 Nean	We n	ek 4 Nean	. Ve	ek 5 Nean		ek 6
Ziprasidone 20 mg Bl0 60 mg Bl0 100 mg BlD	86 76 83	4.9 4.9 4.7	84 75 82	-0.1 -0.2 -0.1	77 69 77	-0.3 -0.4 -0.4	62 62 69	•0.5 •0.5 •0.5	57 57 60	-0.7 -0.6 -0.6	51 50 57	-0.8 -0.6 -0.6	n 50 39 47	Mean -0.8 -0.8 -0.7
Haloperidol Placebo	83	5.0	11	-0.4	72	•0.7	67	-0.B	58	-0.9	50	-1.0	47	-0,7 -1.1
	80	4.9	77	-0.1	65	-0.2	48	-0.4	35	-0.5	32	·0.5	27	-0.7

-

			2-Sided P-Va	lues for Pairw	ise Comparison	s	
Ziprasidone 20 mg BlD vs placebo	0.897	0.598	0.574	0.430	0.224	••••••	•••••
Lipresidone 60 mg Rin	0.746	0.069	0.100		0.224	0.354	0.703
vs placebo Ziprasidone 100 mg 81D			0.103	0.197	0.298	0.677	0.582
s placebo	0.155	0.154	0.012	0.277	0.205	0.407	
laloperidol	0.700	0.001	0.000			0.407	0.508
s placebo			0.000	0,054	0.032	0.069	0.066

Date of Table Generation: 305EP96.

CGI Severity Score - Hean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 115

• • • •

							Treats	ent Week						
Treatment Groups		eline Mean	We n	ek] Mean	We n	ek 2 Nean	Ve n	ek 3 Hean	. We n	ek 4 Nean	We	ek 5 Mean		ek 6 Mean
liprasidone 20 mg BlD 60 mg BlD 100 mg BlD	86 76 83	4.9 4,9 4,7	86 76 83	•0.1 •0.2 •0.1	86 76 83	-0.2 -0.3 -0.4	86 76 83	-0.3 -0.4 -0.3	86 76 83	-0.4 -0.4 -0.4	86 76 83	-0.4 -0.4 -0.4	86 76 83	-0.4 -0.4 -0.4
Haloperidol	83	5.0	83	-0.4	83	-0.6	83	-0.7	83	·0.8	83	-0.8	83	-0.8
Placebo	80	4.9	78	-0.1	80	-0.1	80	-0.1	80	-0.0	80	-0.1	80	-0.1

	•••••		2-Sided P-Va	lues for Pairw	ise Comparison	5	
liprasidone 20 mg BID vs placebo	0.897	0.730	0.295	0.043	0.010	0.014	 0.030
iprasidone 60 mg BID s placebo	0.746	0.123	0.020	0.011	0.007	0.030	0.035
iprasidone 100 mg BID s placebo	0.155	0.354	0.004	0.008	0.007	0.015	0.006
aloperidol s placebo	0.700	0.004	0.000	0.000	0.000	0.000	0.000

Source Data: Appendix V Table 16. Date of Data Extraction: 205EP96. Date of Table Generation: 305EP96.

PANSS Total Score Study 115 (from Sponsor's Submission)

į

PÁRSS Total Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 115

							Treats	ent Week						
Treatment Groups	Bas: n	eline Mean	We n	ek 1 Nean	Ne n	ek 2 Nean	We n	ek 3 Nean	Ve n	ek 4 Nean	Ve n	ek 5 Mean	Ne D	ek 6 Nean
Ziprasidone 20 mg BID 60 mg BID 100 mg BID	86 76 82	93.2 90.4 89.5	84 75 82	-5.1 -6.3 -5.3	76 68 77	-8.0 -8.5 -9.1	61 62 69	-13.3 -11.2 -11.4	57 56 60	-14.3 -13.4 -12.2	51 50 57	-14.2 -12.4 -11.6	50 39 47	-17.3 -18.0 -14.0
Haloperido]	82	94.1	77	-8.6	72	-12.4	67	-13.7	58	·20.4	50	-18.9	47	-21.0
Placebo	80	93.3	77	-0.3	64	4,4	48	-6.1	35	-13.2	32	-12.8	27	-13.9

	·····		2-Sided P-Va	lues for Pairw	ise Comparison	\$	
Ziprasidone 20 mg BID vs placebo	0.973	0.019	0.152	0.049	0.648	0.943	0.585
Ziprasidone 60 mg B10 vs placebo	0.293	0.003	0.120	0.140	0.982	0.738	0.473
Ziprasidone 100 mg BID vs placebo	0.154	0.009	0.055	0.121	0.920	0.806	0.850
Haloperidol vs placebo	0.762	0.000	0.007	0.130	0.129	0.229	0.187

Source Data: Appendix V Table 15. Date of Data Extraction: 205EP96. Date of Table Generation: 305EP96.

• •

PANSS Total Score - Mean Change From Baseline and P-Values by Week-2

All Subjects, LOCF	
Ziprasidone Protocol	115

	Treatment Week													
Treatment Groups		eline Mean	Week 1 n Mean		Week 2 n Mean		Week 3 n Mean		Week 4 n Mean		Week 5 n Mean			ek 6 Mean
Ziprasidone 20 mg BID 60 mg BID 100 mg BID	86 76 82	93.2 90.4 89.5	86 76 82	-4.8 -6.3 -5.3	86 76 82	·5.6 -7.5 ·7.8	86 76 82	-8,6 -8,3 -8,5	86 76 82	-8.0 -8.3 -8.4	86 76 82	-6.6 -7.5 -7.8	86 76 82	-7.5 -8.6 -8.3
Haloperido]	82	94.1	82	-8.2	82	-10.9	82	-12.2	82	-15.6	82	-14.5	82	-15.2
Placebo	80	93.3	78	-0,4	80	-1.9	80	-1.1	80	-1.8	80	-1.2	80	-0.4
					2-51	ided P-Va	lues f	for Pairw	ise Co	mparison	s			

Ziprasidone 60 mg BlD vs placebo Ziprasidone 100 mg BlD	0.293 0.154	0.004 0.017	0.02 <u>2</u> 0.021	0.012 0.012	0.032 0.032	0.051 0.041	0.011 0.012
vs placebo Haloperidol vs placebo	0.762	0.001	0.001	0.001	0.000	0.000	0.000

Source Data: Appendix V Table 15. Date of Data Extraction: 205EP96. Date of Table Generation: 305EP96.

.

.....

-.

PANSS Negative Score Study 115 (from Sponsor's Submission)

PANSS Negative Subscale Score - Nean Change From Baseline and P-Values by Week-All Subjects. Observed Cases Ziprasidone Protocol 115

1

	Treatment Week													
Treatment Groups		eline Nean		ek 1 Mean		ek 2 Mean		ek 3 Mean	Ve n	ek 4 Nean		ek 5 Nean	Ve n	ek 6 Mean
Ziprasidone 20 mg BID 60 mg BID 100 mg BID	86 76 82	22.9 23.4 22.5	84 75 82	-0.9 -1.2 -1.3	76 68 77	-1.6 -1.8 -1.7	61 62 69	-2.5 -2.4 -2.6	57 56 60	-2.6 -3.5 -2.5	51 50 57	-2.9 -2.6 -2.4	50 39 47	•3.7 •4.9 •3.3
Haloperido]	82	24.1	77	-1.5	72	-2.0	67	-2.6	58	-4.1	50	•3.7	47	-4.2
Placebo	80	22.4	77	0.9	64	-0.2	48	-0.4	35	·2.5	32	-2.0	2 7	·2.3

	2-Sided P-Values for Pairwise Comparisons												
Ziprasidone 20 mg B1D	0.627	0.016	0.114	0.050	0.794	0.780	0.553						
vs placebo Ziprasidone 60 mg BID	0.334	0.012	0.193	0.128	0.746	0.844	0.244						
vs placebo Ziprasidone 100 mg BID	0.950	0.005	0.089	0.043	0.866	0.905	0.591						
vs placebo Haloperidol vs placebo	0.089	0.007	0.165	0.196	0.488	0.555	0.562						

.

Source Data: Appendix V Table 15. Date of Data Extraction: 205EP96. Date of Table Generation: 305EP96.

Treatment Groups	Bas n	eline Nean	We n	ek 1 Mean	We n	ek 2 Mean		ent Weel ek 3 Mean		ek 4 Nean	We	ek 5 Nean		ek 6
Ziprasidone 20 mg BID 60 mg BID 100 mg BID Haloperidol Placebo	86 76 82 82 80	22.9 23.4 22.5 24.1 22.4	86 76 82 82 78	-0.8 -1.2 -1.3 -1.5 0.8	86 76 82 82 80	-T.1 -1.6 -1.7 -1.9 0.3	86 76 82 82 82	-1.6 -1.8 -2.1 -2.5 0.5	86 76 82 82 82 80	-1.3 -2.1 -2.0 -3.1 0.0	86 76 82 82 82 80	-1.1 -1.5 -2.0 -2.8 0.2	n 86 76 82 82 82 80	Hea -1 -2 -2 -3. 0.
	· · · · ·	·····		••••••	2-51d	ed P-Val	lues fo	r Palrwi	ise Com	parisons				
Ziprasidone 20 mg BID vs placebo Ziprasidone 60 mg BID		627		032		168		026		206	•••••	····· 261		 121
vs placebo Ziprasidone 100 mg BID vs placebo		334 950		016 006		077		025	0.	064		204		069
		089	0.0		0.()50	0.1	005	0.	041	0.4	039	0.	020

Date of Table Generation: 305EP96.

10 7 . . - .

Redacted 4

pages of trade

secret and/or

confidential

Commercial

information

Med - Page 118 of 223

Appendix 7.2.4.2 (from Sponsor's Submission)

Subject Disposition Ziprasidone Protocol 303

_	Number of S	ubjects	ł	umber of Sub	jects Complet	ing Each Perio	d of Studys	•••••
reatment Group	Randomized	Treated	Week 3	Week 6	Week 16	Week 28	Week 40	••••••
Ziprasidone,				•••••	•••••••••		HCEK 40	Week S
10 mg 810	76	76	67	51	48	40		
Iprasidone,						40	38	3
0 mg BID .	- 72	72	66	58	47			
tprasidone.				70	4/	36	33	3.
0 mg 810	71	71	68	66	••	-		
				60	50	42	36	34
lacebo	75	75	70	••				
			70	56	31	23	16	14
otal:	294							•
Based on planned schedu Burce Data: Appendix V		294	273	241	176	141	123	

urce Data: Appendix V Tables 6, 15, 16. Date of Data Extraction: O6JAN97. Date of Table Generation: O6JAN97.

Appendix 7.2.4.3 (from Sponsor's Submission)

	- Zipra	sidone 20	3 mg 810	Zipra	sidone 40	Bg 810	Zipra	sidone B	0 mg 810		Placeb	0
	Male	Female	Total	Hale	female	Total	Hale	Female	Total	Ma)e	Female	Total
Number of Subjects Randomized	56	20	76	51	21	72	46	25	71	61	14	75
Age (years): 18-44 45-64 >=65	23 24 9	6 8 6	29 32 15	29 20 2	3 10 8	32 30 10 -	20 21 5	7 14 4	27 35 9	27 29 5	3 8 3	30 37 8
ean age (years) ge range	49.1 18-75	55.5 31-82	50.8 18-82	45.4 24-75	59.4 36-78	49.5 24-78	48.4 22-72	52.0 23-73	49.6 22-73	47.7 20-76	53.7 29.70	48.8 20-76
Race: White	56	20	76	51	21	72.	46	25	71	61	14	75
Nean weight (kg) Weight range	73.0 45-124	72.3 57-106		72.3 49-112	66.7 45- 95		73.8 51-155	67.3 44- 96		74.7 54-115	65.1 48- 90	

112

Anal ysi s of Time-to-Rel apse - Al I Subjects Ziprasi done Protocol 303

Treatment Group	N	Cumul ati ve 1 of Rei <=28 woeks		Rei aj	lity of . see	Relative Risk	95% Confidence Lower	Units Upper	P-Value***
Ziprasidone 20 mg BiD 40 mg BiD 80 mg BiD	75 72 71	23 (30.7) 21 (29.2) 22 (31.0)	27 (36.0) 22 (30.6) 24 (33.8)	0. 339 0. 326 0. 324	0. 405 0. 346 0. 358	0. 481 0. 414 0. 411	0. 296 0. 247 0. 249	0. 781 0. 693 0. 680	0. 003 0. 001 0. 001
Pi ecebo	75	35 (46.7)	43 (57.3)	0. 545	0. 708				
Overal I									<0. 001
Dose Response Zip. vs Placebo Linear Axong Zip.									0. 002 <0. 001 0. 595

.

Percent to number of patients at baseline
 ** Estimates of probability of relapse at <-28 weeks or <-52 weeks are based on the Kaplan-Neier product-limit method.
 *** The p-values for comparing each treatment with placebo and for overall are derived from a Cox regression model that includes a contrast variable for each treatment group versus placebo. The p-value for dose response is based on Cox regression model using the actual dosage levels (0 mg for placebo). The dose response is further tested for Ziprasidone groups combined versus placebo using model contrasts (-3, 1, 1, 1) and for linear effect among the Ziprasidone groups using contrasts (0, -1, 0, 1).

Source Data: Appendix III Table 27. Date of Data Extraction: O6JAN97. Date of Table Generation: O6FE897.

APPEARS THIS WAY

BPRSd Total Score Study 303

:

(from Sponsor's Submission)

BPRSd Total Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 303

	Treatment Week													
Treatment Groups	Baseline n Kean	Week 3 n Mean	Week 6 n Nean	Week 16 n Mean	Week 28 n Nean	Week 40 n Mean	Week 52 n Mean							
Ziprasidone 20 mg BID 40 mg BID 80 mg BID	75 46.1 - 72 47.1 71 45.9	67 -1.7 66 -2.1 68 -2.2	60 -4.3 58 -2.8 66 -3.6	48 -5.4 47 -5.4 50 -4.7	40 -5.8 36 -6.4 42 -7.7	38 -7.5 33 -7.1 36 -8.6	35 -9.3 33 -8.0 34 -9.2							
Placebo	75 48.0) 70 -0.2	56 -2.9	31 •5.5	23 -4.1	16 -3.0	14 -3.4							

	2-Sided P-Yalues for Pairwise Comparisons												
Ziprasidone 20 mg BID	0.302	0.120	0.068	0.265	0.056	0.127	0.543						
vs placebo Ziprasidone 40 mg BID	0.615	0.073	0.869	0.261	0.053	0.444	0.768						
vs placebo Ziprasidone 80 mg BID vs placebo	0.272	0.053	0.425	0.723	0.002	0.034	0.487						

Source Data: Appendix V Table 15. Date of Data Extraction: D6JAN97. Date of Table Generation: 07JAN97.

BPRSd Total Score - Hean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 303

							Treatm	ent Veel					
Treatment Groups	Bas n	eline Mean	#e 	ek 3 Nean	We n	ek 6 Mean	, ¥e n	ek 16 Mean	We n	ek 28 Mean	We n	ek 40 Mean	ek 52 Mean
Ziprasidone 20 mg BID 40 mg BID 80 mg BID Placebo	75 72 71 75	46.1 47.1 45.9 48.0	75 72 71 75	0.2 •0.1 •1.5 1.0	75 72 71 75	-0.1 0.2 -2.0 1.7	75 72 71 75	1.6 0.7 -0.1 4.4	75 72 71 75	3.1 2.2 0.0 7.1	75 72 71 75	2.6 2.3 0.7 9.3	 2.5 1.9 0.5 9.6

	2-Sided P-Values for Pairwise Comparisons											
Ziprasidone 20 mg BID vs placebo	0.302	0.546	0.265	0.158	0.041	0.001	• • • • • • •					
Ziprasidone 40 mg BID vs placebo Ziprasidone 80 mg BID	0.615	0.544	0.556	0.139	0.035		0.001					
	0.272	A 443			0.035	0.004	0.002					
vs placebo	0.272	0.083	0.042	0.043	0.001	<0.00 1	<0.003					

...

BPRSd Core Items Study 303

(from Sponsor's Submission)

BPRSd Core Items Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Elprasidone Protocol 303

į

Treatment Groups	Baseline n Nean	Veek 3 n Nean	Heek 6	Treatment Wei Week 16			
Ziprasidone 20 wg BID 40 wg BID			n Hean	n Nean	Week 28 a Nean	Neek 40 n Nean	Week 52 5 News
80 ag 810 Placebo	75 11.2 72 11.7 71 11.2 75 11.7	67 -0,3 66 -0,6 68 -0,6 70 0,1	60 -1.0 58 -0.8 66 -0.6 56 -0.6	$\begin{array}{rrrr} 4B & -1.1 \\ 47 & -1.3 \\ 50 & -1.3 \\ 31 & -1.5 \end{array}$	40 -1.3 36 -1.6 42 -1.8 23 -0.7	38 -1.6 33 -2.0 36 -2.2 16 -0.6	35 -2.3 33 -2.0 34 -2.1 14 -0.2
Ziprasidan- no	**********	•••••	2-Sided P-Ya	ues for Pairw	se Comparisons		
Ziprasidone 20 mg BID vs placebo Zipratidana in	D. 378	0.259			se comparisons		
Ziprasidone 40 mg BID vs placebo Liprasidone	0.980	0.052	0.127	0.844	0.098	D. 458	
iprasidone 80 mg 810 's placebo	0.446	0.071	0.567	0.534	0.070	0.468	0.602
Source Data: Appendix V T. Date of Table Generation:	able 15. Date of 07JAN97	Data Extracti	0.742	0.832	0.003	0.074	0.900 0.430
	··· •		evony/.			•••••	

BPRSd Core Items Score - Nean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 303

,

Treatment Groups	Baseline n Mean	Week 3 n Mean	Heek 6	Treatment He	ek		
Ziprasidone 20 mg BiD 40 mg BiD 80 mg BiD	75 11.2 72 11.7 71 11.7		n Nean	Week 16 n Mean	Veek 28 n Nean	Week 40 n Mean	Neek 5; A Neer
Placebo	75 11.2 72 11.7 71 11.2 75 11.7	75 0.3 72 0.1 71 -0.4 75 0.3	75 0.3 72 0.0 71 0.3 75 0.5	75 0.8 72 0.1 71 0.2 75 1.2	75 1.2 72 0.5 71 0.3 75 2.1	75- 1.3 72 0.5 71 0.4 75 2.5	75 1. 72 0. 71 0.5 75 2.6
21prasidone 20 mg 81D vs placebo 21prasidone 40 mg 81D vs placebo 40 mg 81D 11prasidone 80 mg 81D vs placebo ource Data: Appendir v 2	0.378	0.919	2-Sided P-Yel	ues for Pairwi	se Comparisons		
	0.980 0.438 0.446		0.629 0.498	0.472 0.132 0.135	0.168 0.035	0.043 0.008	0.043 0.007
ource Data; Appendix y To ate of Table Generation:	07JAN97, Date of	Data Extracti	on: 06JAN97.		9.010	0.002	0.003

179

,

CGI Severity Score Study 303 (from Sponsor's Submission)

CG) Severity Score – Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 303

	D • •						Treats	ent Veel	k					
Treatment Groups	Baseline n Mean		Week 3 n Mean		Week 6 n Nean		Veek 16 n Nean		Veek 28		Week 40		Week 52	
Ziprasidone	-								n 	Nean	n	Mean		Hean
20 mg BID 40 mg BID 80 mg BID Placebo	75 72 71	4.0 4.0 4.0		-0.1 -0.0 -0.1	60 58 66	-0.2 -0.2 -0.2	48 47 50	-0.4 -0.4 -0.4	40 36 42	-0.4 -0.6 -0.5	38 33 36	-0.5 -0.6 -0.6	35 33 34	-0.7 -0.8 -0.7
	75	4.1	70	0.1	56	-0,2	31	-0.4	23	-0.3	16	-0.3	34 14	-0.7 -0.4
	····	•••••			2-510	led P-Val	ues fo	r Pairwi	se Con	parisons				
Ziprasidone 20 mg BID vs placebo Ziprasidone 40 mg BID	0.3	724	0.	050		360		608	•••••	116	•••••	•••••• 503	•••••	••••

Date of Table Generation;	07JAN97	or vata Extrac	tion: D6JAN97.		••••••••••••••	•••••••••••••	• • • • • • • • • • • •
Source Data: Appendix V T Date of Table Generation;	ahla 16 n.a.						0.493
			0.363	0.569	0.004	0.183	
A2 blacepo	0.733	0.055	0 3/3		0.013	0.521	0.683
vs placebo Ziprasidone 80 mg 810		0.123	0.581	0.278	0.019		
Ziprasidone 40 mg BID	0.557	0.123		0.000	0.116	0.503	0.853
vs placebo	V./24	0.050	0.360	0.608			

CGI Severity Score - Mean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 3D3

	_						Treatm	ent Weel	k					
Treatment Groups		eline Mean	We n	ek 3 Mean	We n	ek 6 Mean	We N	ek 16 Mean	We	ek 28 Mean		ek 40	We	ek 52
Ziprasidone 20 mg BID 40 mg BID 80 mg BID	75 72	4.0 4.0	75 72	0.1 0.1 -0.0	75 72 71	0.1 0.1 -0.1	75 72 71	0.2			 75	Hean	 75	Hean
Placebo	71 75	4.0 4.1	7 <u>1</u> 75	0.0 0.2	71 75	-0.1 0.3	71 75	0.2 0.1 0.1 0.5	75 72 71 75	0.4 0.2 0. <u>1</u> 0.B	75 72 71 75	0.3 0.2 0.2 0.9	75 72 71 75	0.4 0.2 0.2 0.9

-	ise Comparison	
3 0.001	0.013	0.001
•	0.000	0.001
2 (0.001	0.002	<0.001
i (0,001	<0.001	<0.001
	<0.0	(0.001

-

Date of Table Generation: 07JAN97.

. .

Med - Page 123 f 236

CGI Improvement Score Study 303

(from Sponsor's Submission)

CGI Improvement Score - Means and P-Values by week-All Subjects, Observed Cases Ziprasidone Protocol 303

	Treatment Week											
Treatment Groups	Ne: n	ek 3 Mean	Ve: n	ek 6 Mean	We n	ek 16 Mean	Ve: n	ek 28 Mean		ek 40 Mean		ek 52 Mean
Ziprasidone 20 mg BID 40 mg BID 80 mg BID	67 66 68	3.5 3.6 3.6	60 58 66	3.4 3.6 3.5	48 47 50	3.1 3.2 3.3	40 36 42	3.2 3.0 2.9	38 33 36	3.1 2.8 2.8	35 33 34	2.7 2.7 2.6
Placebó	70	3.9	56	3.5	31	3.3	23	3.3	16	3.3	14	3.1

	•••••••••••		2-Sided P-Va	lues for Pairw	ise Comparison	\$
Ziprasidone 20 mg BID vs placebo	0.019	0.303	0.303	0.118	0.403	0.268
Ziprasidone 40 mg BID vs placebo	0.040	0.865	0.189	0.079	0.462	0.819
Ziprasidone 80 mg 81D vs placebo	0.059	0.902	0.553	0.007	0.069	0.204

~ ---- ..

Med - Page 124 o 22

Source Data: Appendix V Table 16. Date of Data Extraction: OGJAN97. Date of Table Generation: O7JAN97.

CGI Improvement Score - Means and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 303

.

							Treatme	nt Week	ι.				
-	Treatment Groups		k 3 Mean	Vei n	ek 6 Mean	Wee n	ek 16 Mean		k 28 Mean	Wee	k 40 Mean		ek 52 Mean
	Ziprasidone 20 mg BID 40 mg BID 80 mg BID	75 72 71	3.7 3.8 3.7	75 72 71	3.8 3.9 3.7	75 72 71	4.0 3.9 3.9	75 72 71	4.2 4.1 3.9	75 72 71	4.3 4.1 4.0	75 72 71	4.2 4.0 3.9
	Placebo	75	4,0	75	4.1	75	4.6	75	4.8	75	5.0	75	5.0

- Ziprasidone 20 mg BID vs Diacebo		2-Sided P-Values for Pairwise Comparisons										
	0.097	0.182	0.015	0.025	0.004	0.001						
vs placebo Ziprasidone 40 mg BlD - vs placebo	0.231	0.488	0.014	0.009	0.001	0.001						
ziprasidone 80 mg BlD vs placebo	0.109	0.089	0.012	0.001	<0.001	<0.001						

Date of Table Generation: 07JAN97.

PANSS Total Score Study 303 (from Sponsor's Submission)

Г

PÁNSS Total Score - Mean Change From Baseline and P-Values by Week-All Subjects. Observed Cases Ziprasidone Protocol 303

				Treatment Weel	K		
Treatment Groups	Baseline n Nean	Veek 3 n Mean	Week 6 n Hean	Week 16 B Mean	Week 28 n Mean	Week 40 n Mean	Veek 52 n Kean
Ziprasidone 20 mg BID 40 mg BID 80 mg BID	75 85.1 72 86.6 71 85.2	67 -3.2 66 -4.0 68 -3.5	60 -7.7 58 -4.9 66 -5.7	48 -10.1 47 -9.1 50 -9.1	40 -11.1 36 -12.0 42 -14.2	38 -14.2 33 -13.7 36 -15.4	35 -17.8 33 -14.7 34 -17.0
Placebo	75 88.9	70 -1.2	56 -5.9	31 -10.3	23 -8.6	16 -6.5	34 -17.0 14 -6.9

•••••		2-21060 P-Va	lues for Palrw	ise Comparison	\$	
0.210	0.150	0.127	0.311			••••••
0.449	0.083	0.664	0.513			0.277
0.227	0.135	0.962	0.849			0.821 0.350
	0.210 0.449	0.210 0.150 0.449 0.083	0.210 0.150 0.127 0.449 0.083 0.664	0.210 0.150 0.127 0.311 0.449 0.083 0.664 0.513	0.210 0.150 0.127 0.311 0.057 0.449 0.083 0.664 0.513 0.066 0.227 0.125 0.027 0.127 0.127	0.210 0.150 0.127 0.311 0.057 0.134 0.449 0.083 0.664 0.513 0.066 0.483 0.227 0.135 0.962 0.012 0.127

Date of Table Generation: 07JAN97.

PANSS Total Score - Mean Change From Baseline and P-Values by Week-All Subjects. LOCF Ziprasidone Protocol 303

Treatment Groups	Bas n	eline Nean	We n	ek 3 Mean	We n	ek 6 Nean	We	ent Wee ek 16 Mean		ek 28 Mean		ek 40	We	ek 52
Ziprasidone 20 mg BID 40 mg BID 80 mg BID Placebo	75 72 71 75	85.) 86.6 85.2 88.9	75 72 71 75	-0.2 -0.4 -2.7	75 72 71	-1.0 0.2 -3.6	75 72 71	1.4 0.8 -1.7	75 72 71	3.7 2.5 -1.5	n 75 72 71	Hean 2.9 2.6 -0.5	75 72 73	Mean 2.4 2.1 -1.1
		00.9	/3	1.0	75	1.7	75	6.2	75	10.2	75	14.1	75	14.6
	····.		·····	••••••	2-51d	ed P-Val	ues fo	r Patrwi	se Com	parísons				
Ziprasidone 20 mg BID vs placebo Ziprasidone 40 mg BID		210 449		591		285		143		 041	•••••	·····	•••••	
vs placebo Ziprasidone 80 mg BID vs placebo		227		621 110	0.1 0.0	/51)65	0.) 0.(187		042	0.	003		002
Source Data: Appendix V 1 Date of Table Generation:	able 1 07JANS	5. Date (of Data	Extraci			•••••	· · · · · · · · · ·	0.0		<0.(<0.(DO]

PANSS Negative Score Study 303

(from Sponsor's Submission)

٦

PÁNSS Negative Subscale Score – Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 303

				Treatment Veel	r i i i i i i i i i i i i i i i i i i i		
Treatment Groups	Baseline n Mean	Veek 3 n Mean	Week 6 n Nean	Week 16 n Mean	Week 28 • n Hean	Week 40 n Mean	Week 52 n Mean
Ziprasidone 20 mg BID 40 mg BID 80 mg BID	75 24.9 72 24.7 71 25.0	67 -1.4 66 -1.2 68 -1.3	60 -2.5 58 -1.4 66 -1.8	48 -3.4 47 -2.8 50 -3.1	40 -3.9 36 -3.7 42 -4.6	38 -4.8 33 -4.4 36 -5.1	35 -5.6 33 -4.2 34 -5.6
Plącebo	75 25.7	70 -1.4	56 -2.3	31 -3.4	23 -3.7	16 -2.8	14 - 3.0

	••••••				ise Comparison		
Ziprasidone 20 mg BID vs placebo	0.326	0.715	0.429	0.457	0.249	0.277	0.455
Ziprasidone 40 mg BID vs placebo	0.253	0.928	0.162	0.857	0.479	0.817	0.433
Ziprasidone 80 mg BID vs placebo	0.401	0.920	0.469	0.974	0.063	0.290	0.487

Source Data: Appendix V Table 15. Date of Data Extraction: O6JAN97. Date of Table Generation: O7JAN97.

PANSS Negative Subscale Score - Mean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 303

	Treatment Week													
8a Treatment Groups n		eline Nean	We n	ek 3 Mean	We n	ek 6 Nean	We n	ek 16 Mean		ek 28 Nean	We n	ek 40 Mean		ek 52 Nean
Ziprasidone 20 mg BlD 40 mg BlD 80 mg BlD	75 72 71	24.9 24.7 25.0	75 72 71	-0.9 -0.4 -1.4	75 72 71	-1.3 -0.4 -1.7	75 72 71	-1.4 -0.8 -2.0	75 72 71	-1.4 -0.9 -2.3	75 72 71	-1.8 -1.1 -2.3	75 72 71	-2.0 -1.0 -2.6
Placebo	75	25.7	75	-0.7	75	·0.6	75	0.1	75	0.5	75	1.2	75	1.3

2-Sided P-Values for Pairwise Comparisons									
Ziprasidone 20 mg BID	0.326	0.673	0.272	0.039	0.012	<0.001	<0.001		
vs placebo Ziprasidone 40 mg BID	0.253	0.750	0.767	0.246	0.082	0.009	0.012		
vs placebo Ziprasidone 80 mg BID vs placebo	0.401	0.203	0.118	0.007	0.001	<0.001	<0.001		

Date of Table Generation: 25FEB97.

. . , . .

119 Med - Page 126 of 223

Redacted 4

pages of trade

secret and/or

confidential

Commercial ·

information

Med - Page 127 of 223

Appendix 7.2.5.2 (from Sponsor's Submission)

Subject Disposition Ziprasidone Protocol 104

	Number of S	ubjects	Number of Sub	ects Completin	g Each Period	of Study*
Treatment Group	Randomized	Treated	Week 1	Week 2	Week 3	Week 4
Ziprasidone, 5 mg BID	47	47	45	37	33	3(
Ziprasidone. 20 mg BID	- 55	55	54	46	38	27
Ziprasidone. 40 mg BID	. 48	48	42	32	24	20
Placebo	50	50	45	34	29	27
Total:	200	200				

*Based on planned primary efficacy measurements. Weeks are determined by visit designators. Week 1 counts subjects who have at least one primary efficacy measurement at visit 7; Week 2 similarly counts those with visit 14; Week 3 similarly counts those with visit 21; Week 4 similarly counts those with visit 28. Source Data: Appendix V Tables 6, 16, 17. Date of Data CALraction: 17JUL95. Date of Table Generation: 090CT96.

Appendix 7.2.5.3 (from Sponsor's Submission)

	Zipra	si done	5mg BID	Zipri	asi done 2	Orang BiD	Zipra	asidone 4	Omo 810		 Di anab	
	Male	Femal e	Total	Mai e	Female	Total	Male	• • • • • • • • • •	Total		PI aceb	
Number of			•••••		•••••					Maie	Female	Total
Subjects Randomized	41	6	47	52	3	55	39	9	48	44	-	
Age (years):					•••••••					••••••••	6	50
18-44 45-64	31 10	2	33 14	39 13	1	40 15	32	3	35	22		•.
	• • • • • • • • •	••••••		13	2	15	7	ő	35 13	33 11	1	34 16
lean age (years) Ige range	38.4 22-60	44.7	39. 2	40, 4	51.7	41, 1	37.5	49.0	39, 7			
****************		24-57	22-60	25-64	44-56	25-64	20-61	39-63	20-63	38. 5 22-64	47.8 37-56	39. 6 22-64
ace: Caucasi an	~					· · · · · · · · · · · · · · · · · · ·			••••••		••••••••••	******
Bi ack	20 17		23 20.	27	3	30 19 2	22 12	5	27	27	•	20
Oriental	1	ŏ	1	19	0	19	12	4	16	15	2	30 17
Other	3	Ó	3	4	·ŏ	á	3	00	3	Ő	ī	ï
ean weight (kg)	77.5							•••••••	2	2	0	2
leight range	52-120	71.3		76.8	65.8		76. 9	69. 4		79.5	67.8	
ource Data: APPENDIX				55-115	63- 70		. 49-115	55- 93		58-113	57- 81	

Med - Page 128 of 223

BPRS Total Score Study 104

(from Sponsor's Submission)

BHRS Total Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocol 104

Treatment Groups	Treatment Week									
		eline Mean	We n	ek 1 Mean	We n	ek 2 Mean	We n	ek 3 Nean	We n	ek 4 Mean
Ziprasidone 5 wg BiD 20 wg BiD - 40 wg BiD	44 55 47	34. 1 34. 5 36. 2	44 54 42	-2. 3 -3. 9 -3. 4	37 46 32	-4.3 -4.9 -4.3	33 38 24	-7.5 -9.2 -8.9	30 27 20	-7.8 -13.0 -7.3
PI acebo	47	33. 4	45	-3.6	34	-4.4	29	-7.9	27	-7.3

2-Sided P-Values for Palinwise Comparisons

Ziprasidone 5 mg BID vsplacebo	0. 738	0. 454	0. 667	0. 845	0. 646
Ziprasidone 20 mg BiD vs placebo	0. 586	0. 894	0. 551	0. 352	0. 040
Ziprasidone 40 ang BiD vs placebo	0. 185	0. 836	0. 821	0. 446	0. 720
•••••••					

Source Data: Appendix V Table 16. Date of Data Extraction: 17JUL95. Date of Table Generation: 110CT96.

BPRS Total Score - Mean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 104

.....

	Treatment Week										
Treatment Groups	Baseline n Mean	Week 1 n Mean	Week 2 n Mean	Week 3 n Mean	Week 4 n Mean						
Ziprasidone 5 mg 8lD 20 mg 8lD 40 mg 8lD	44 34. 1 55 34. 5 47 36. 2	44 -2.3 55 -3.9 47 -3.0	44 -2.8 55 -3.6 47 -3.1	44 -4.3 55 -4.9 47 -3.7	44 -3.9 55 -5.4 47 -2.6						
PI acebo	47 33.4	46 - 3. 3	47 -2.4	47 -3.3	47 -3.5						

		2-Sided P-Values for Pairwise Comparisons						
Ziprasidone 5 ag BID vs placebo	0. 738	0. 445	0. 923	0. 692	0. 926			
Ziprasidone 20 mg BID vs placebo	0. 586	0. 828	0. 573	0. 438	0. 354			
Ziprasidone 40 mg BID vs placebo	0. 185	0. 590	0. 831	0. 989	0. 535			
***********************				••••••				

Source Data: Appendix V Table 16. Date of Data Extraction: 17JUL95. Date of Table Generation: 110CT96.

BPRS Core Items Study 104 (from Sponsor's Submission)

.

÷

BPRS Core Score - Mean Change From Baseline and P-Values by Meek-All Subjects, Observed Cases Ziprasidone Protocol 104

Treatment Groups	Treatment Neek									
		eline Mean	1 18 n	lek 1 Mean	Ste n	ek 2 Maan	Nie	ek 3 Mean	We	ek 4 Mean
Zipræsidone 5 mg BiD 20 mg BiD 40 mg BiD Piacebo	44 55 47 47	12. 8 13. 0 12. 8	44 54 42	-0. 9 -1. 6 -0. 7	37 46 32	-1, 6 -2, 8 -0, 8	33 38 24	-2.5 -4.1 -2.5	30 27 20	-2.9 -5.1 -2.9
	47	13. 7	45	-2. 2	34	-2.7	29	-3.8	27	.12

2-Sided P-Values for Painwise Companisons

Ziprasidone 5 mg 810 vs placebo	0. 201	0. 062	0. 313	0. 556	•••••••••••
Zi prasi done 20 mg Bi D	0 200			0.000	0, 491
VS DI ACEDO -	0. 299	0. 398	0. 761	0. 341	0. 022
Zi prasi done 40 mg BID VS placebo	0. 201	0.042	0.090		0.012
			0. 030	0. 978	0. 292
Source Data: Amendia V					

Source Data: Appendix V Table 16. Date of Data Extraction: 17JUL95.

BPRS Core Score - Mean Change From Baseline and P-Values by Week-All Subjects, LOCF Ziprasidone Protocol 104

					Treatur	ent Week				
Treatment Groups	Bas n	elíne Nean	We n	ek 1 Mean		ek 2 Nean		ek 3 Mean		ek 4 Mean
Ziprasidone 5 mg BID 20 mg BID 40 mg BID	44 55 47	12.8 13.0 12.8	44 55 47	-0, 9 -1, 6 -0, 7	44 55 47	-1.3 -2.0 -0.6	44 55 47	-1.6 -2.3 -1.2		-1.8 -2.3 -1.2
Pi acebo	47	13. 7	46	-2.0	47	-2.1	47	-2.3	47	-2.2

Ziprasidone 20 ag BID 0.299 0.593 0.974 0.981 0.751 Vs placebo Ziprasidone 40 ag BID 0.201 0.040 0.051 0.100	Source Data: Appendix V Date of Table Generation:	able 16. Date 1100796.	of Data Extra	ction: 17JUL95.		
Zipras/done 20 ag 81.0 0.299 0.593 0.974 0.981 0.751 V5 placebo Zipras/done 40 ag 81.0 0.299 0.593 0.974 0.981 0.751	vs placebo	0. 201	0. 040	0. 051	0. 193	0. 301
	VS DIacebo		0. 593	0. 974	0. 981	0. 751
Ziprasidone 5 mg BID 0.201 0.083 0.252 0.449 0.242	VS DI aceno				0. 449	0. 742

Med - Page 130 of 223

2.

CGI Severity Score Study 104

(from Sponsor's Submission)

CC: Severity Score - Mean Change From Baseline and P-Values by Week-All Subjects, Observed Cases Ziprasidone Protocoi 104

Treatment Week

Treatment Groups		line Mean	We n	ek 1 Mean	Wee n	ek 2 Mean	Wea n	ek 3 Mean	Wei n	ek 4 Mean
Ziprasidone - 5 ang BID 20 ang BID 40 ang BID	46 55 47	4, 9 4, 8 4, 9	45 54 42	-0. 1 -0. 2 -0. 2	37 46 32	-0. 1 -0. 3 -0. 2	33 38 24	-0.2 -0.6 -0.3	30 27 20	-0. 3 -0. 8 -0. 4
Pl acebo	47	5.0	45	-0.3	33	-0.3	29	-0. 7	27	-0.7

2-Sided P-Values for Pairwise Comparisons

Ziprasidone 5 mg BID vs placebo	0. 492	0. 143	0. 444	0. 055	0. 333
Ziprasidone 20 mg 810 vs placebo	0. 241	0.604	0. 985	0. 802	0. 293
Ziprasidone 40 mg 810 vs placebo	0. 788	0. 313	0, 453	0. 232	0. 595
			<i>.</i>		

Source Data: Appendix V Table 17. Date of Data Extraction: 17JUL95 Date of Table Generation: 110CT96.

...

CGI Severity Score - Mean Change From Baseline and P-Values by Week-All Subjects. LOCF Ziprasidone Protocol 104

	Treatment Week											
Treatment Groups		line Mean	We n	ek 1 Mean	We n	ek 2 Mean		ek 3 Mean	We n	ek 4 Mean		
Ziprasidone 5 mg BID 20 mg BID 40 mg BID	46 55 47	4.9 4.8 4.9	46 55 47	-0.1 -0.2 -0.1	46 55 47	-0.0 -0.2 -0.1	46 55 47	-0.0 -0.3 -0.1	46 55 47	-0. 1 -0. 3 -0. 2		
Pl acebo	47	5. 0	46	-0. 3	47	-0. 2	47	-0, 4	47	-0.4		

2-Sided P-Values for Pairwise Comparisons

					•••••
Ziprasidone 5 mg BID vs placebo	0. 492	0. 147	0. 283	0. 043	0. 076
Ziprasidone 20 mg BID vs placebo	0. 241	0. 723	0. 784	0. 784	0. 827
Ziprasidone 40 mg BID vs placebo	0. 788	0. 149	0. 337	0. 143	0. 233

Source Data: Appendix V Table 17. Date of Data Extraction: 17JUL95. Date of Table Generation: 110CT96.

..<u>.</u> - .

Appendix Table 8.1.1.1 Deaths occurring during or after trial treatment : Cut-off date: 5/15/97

SUBJECT #	AGE /	LAST DOSE	DAYS OF	CAUSE OF DEATH/COMMENTS
	SEX	(MG/D)	TREAT- MENT	
108-6070305*	46/M	80	61	Found dead (in heat of 100°F). Autopsy report stated cause of death as <u>acute and chromic asthmatic</u> bronchitis and granulomatous myocarditis. ECG: Screening: QTc =366 msec Baseline: QTc =393 Week 6: QTc=395 Was on ziprasidone at time of death.
108-5920750* 1	39/F	120	8	Found dead one day after her estimated date of death of unknown cause. Subject's face was burned and it was thought that she had fallen against a hot water pipe. The investigator's postmortem diagnosis was alcohol abuse/diabetic ketoacidosis, but there is no evidence for this. No coroner's report located in the CRF. Was on ziprasidone at time of death.
116B-5080001*	54/M	120	71	Found dead in his hospital bed. Autopsy showed generalized atherosclerosis, coronary artery disease, cerebral artery disease, visceral congestion (liver, spleen, and lung), COPD, and cardiac hypertrophy. ECGs during the study: Screening: QTc=391 msec baseline: QTc=383 week 2: QTc=367 week 6:QTc=391 Subject had complaint of chest pain once during the study, but ECG was normal and diagnosed as anxiety. Was on ziprasidone at time of death.
302E-3190375*	48/M	120	162	Found dead. CRF showed hypertension and tachycardia on last day of study with hypertension as adverse event during study. Narrative states that subject had history of polydipsia and seizure disorder. Details regarding the death are unclear. Died one day after discontinuing ziprasidone.

i

Ziprasidone subjects who died ≤ 30 days after treatment

÷

128 128

ı.

				Appendix Table 8.1.1.1 (con't)
304E-1930379**	52/M	80	221	Found dead while taking a nap. No autopsy performed and exact cause of death is unknown.
JU4E-1930379	52/14	80	221	ECG during the study as shown in the safety update: QTc at: Screening=374.7 msec Week 12=415.69
				Week 28=413.12 with flat T wave in lead AVL; no evidence of ischemic changes. The CRF had minimal information and the patient profile in the safety update had different ECG QTc values than the original submission. Was on ziprasidone at time of death.
105-5340021*	70/F	2	5	Subject had sudden onset of shallow respirations and diaphoresis. Death certificate stated acute <u>cardiopulmonary arrest due to arteriosclerotic cardiovascular disease</u> . Subject with history of right bundle branch block, otherwise ECG was normal. Was taking ziprasidone just prior to death.
308-0350003* 1	63/M	80	485	Sudden collapse and died. Coroner's report stated that cause was a ruptured abdominal aortic aneurysm and atherosclerosis. Was on ziprasidone at the time of death.
115-6940394	43/M	40	16	Found dead. Coroner's cause of death listed as <u>asphyxiation due to aspiration of vomit</u> . Was on risperidone, clonazepam and lorazepam at time of death. Subject had difficulty breathing three days before death, and complained of dyspnea on morning of death. Died 29 days after discontinuing ziprasidone.
301-3110977	28/F	120	62	Died suddenly of unknown causes. Upon discontinuing ziprasidone, this cachetic subject complained of substernal pinching and ECG changes showed an arrhythmia with probable subendocardial ischemia. She was treated with thioridazine and nitrazepam, and died two days later. Autopsy reportedly showed evidence of myocardosis. Coroner's report not found in CRF. Death occurred two days after stopping ziprasidone.

.

t

Med - Page 133 of 223

:

İ

Г

				Appendix Table 8.1.1.1 (con't)
116B-6590001	44/F	120	47	Subject had a UTI upon d/c and was diagnosed with gastritis with Helicobacter pylori 13 days later. She was seen in ER with diagnosis of panic attack 21 days after d/c (three days prior to death). Sponsor reports that the autopsy was not available due to legal issues in medical examiners, but Subject's attending physician reportedly got information from the medical examiners that subject had a benign cardiac neoplasm (myxoma). ECG: screening: QTc=444 msec baseline: QTc=443 week 1: QTc=433 week 2: QTc=440 week 6: QTc=407 Subject reported chest pain one day after starting ziprasidone: cardiology w/u was normal, but had elevated transaminases. Episodes of tachycardia and hypertension during the study: day 6:102 bpm day 20:120/100; 104 bpm day 27:140/100; 102 bpm day 42:164/98 It is unclear what medications she was on as the patient summary and the CRF do not list the same medications. Subject stopped ziprasidone 24 days before her death.
303-1970299	79/F	80	30	Cardiac arrest. No autopsy was performed. Subject had new diagnosis of atrial fibrillation and ischemic heart disease 27 days after d/c. At time of death was taking perphenazine, deparkin, digoxin, verapamil, and enalapril. Death occurred 30 days after d/c from ziprasidone.
Suicides and accide	ents			
108-6090381	21/F	160	54	Suicide by gunshot while on ziprasidone.
116B-6940004	24/M	160	146	Suicide by hanging while on ziprasidone. Subject had been complaining of increasing depressed mood; treatment included an increase in ziprasidone.
117-6870317	51/M	120	205	Death by defenestration. According to study profile, subject did not appear suicidal prior to death.
117-7060529	40/M	160	54	Subject stopped ziprasidone on his own and four days later he drove his car off a cliff. Subject was driving his car after a sleep deprived EEG against medical advice. Autopsy listed <u>asphyxiation due to</u> <u>drowning</u> and was classified as a probable traffic accident.
302-2600156	46/M	120	7	Subject's body found drowned in local river after being missing from the hospital for five days.
302E-1590029	22/M	120	179	Suicide by falling under a train. Was being treated with ziprasidone at time of suicide with plans to be admitted to the hospital that same day.
JP-95-6011622 1*	53/M	53	20	Suicide seventeen days after discontinuing ziprasidone (Japanese studies: not part of the integrated safety data base.)

Included in Sudden Unexpected Death (SUD) rate calculation in Appendices 8.1.1.3 and 8.1.1.4
 ¹ Submitted in the safety update
 ⁺ Not included in integrated safety data base

1

4

I

ł

130

ŧ

Appendix Table 8.1.1.1 (con't)

SUBJECT #	AGE/ SEX	DOSE (MG/D)	DURATION (DAYS)	CAUSE OF DEATH
104-5130213	40/M	40	28	Sudden death; cause unknown. Occurred 71/2 months after discontinuation from ziprasidone.
106-05550117	35/M	40	27	Unknown cause of death but possible seizure and aspiration of vomit. Was taking risperidone at time of death. Death occurred 4½ months after stopping ziprasidone.
108-5780020	37/M	80	8	Accidental drowning. Died 11/2 months after stopping ziprasidone.
117-6940542	38/M	160	15	Suicide by gun shot one year after d/c from ziprasidone.
301-1140331	30/M	200	26	Died of complication due to pancreatitis 9 months after stopping ziprasidone
301-1320771	34/M	120	53	Suicide by hanging approximately 3 months after stopping ziprasidone.
303-0640276	61/M	40	69	Died of bronchopneumonia with bronchial adenocarcinoma and metastasis. Death occurred 4 months after stopping ziprasidone.
303-1950250	68/M	40	350	Died of cranial trauma 2° to fall. Ziprasidone was stopped 45 days prior to death.
303-1950281	71/F	40	61	Sudden death due to acute purulent leptomeningitis. Death occurred 4 ½ months after stopping ziprasidone.
303-1970269	67/F	80	37	Bronchopneumonia. Stopped ziprasidone 112 days before diagnosis.
303-1990089 no CRF available	55/M	80	27	Sudden death due to acute cerebral edema. death occurred 60 days after stopping ziprasidone.
303-2120222	58/M	160	349	Died of post operative cerebral edema after tumor removal. Occurred two months after stopping ziprasidone
304-2040222	55/F	160	29	Sudden death with proposed cause of acute heart failure due to pulmonary disease. Death occurred approximately 2 months after stopping ziprasidone.
307-2690047** 1	49/F	100	196	Died of hepatic coma, cholestatic jaundice and malignant neoplasm 95 days after stopping ziprasidone. Discontinued ziprasidone because of jaundice and elevated AST (244 U/L) and ALT (375 U/L).

Ziprasidone subjects who died \geq 30 days after treatment

1

i

**Died after cutoff of 5/15/97 'Submitted in the safety update

1

t

Appendix Table 8.1.1.1 (con't)

1

From the sponsor's electronic submission:

Deaths in Risperidone group (table from sponsor's electronic submission)

Treatment Group: Risperidene

60 00 00

•••••	Proferred text /Invastigator text	Age åt enset	Sex.	tace	(ÜL)	Duration of Treatment (Days)	Dose at onsat (mg)	Cose (mn)	Maximum Dose (mg)	Onaet (Day)	Tresteent discentinued	Serious adverse event type
117 46379223	FOREIGH BODY IN LARYNX / Asphyxiation due to Aspiration of food	65	Mala	Caucéstan	33.4	201	4	4	6	201		DEATH / AELIFETH

Deaths in Haloperidol group (table from sponsor's electronic submission) :

Subject J.D.	Proferred taxt /Investigator taxt e	Age at inset	Sex	lietgh Bace (85	Duration of t Treatment) (Days)	Dose at onset (mg)	Mode Dose (mg)	Naximum Dose (mg)	Onset (Day)	Treatment discontinued	Serious adverse avent type	
108 05820944	SUICIDE/SELF-INFLICTED INJULY By oth/unsp fireAum / self Inflicted Gunshot Nound	,	23 Ma	le Caucastan	64,9	84	W/X			102	N	D DEATH
108 05944564	POISONING BY UNSPECIFIED Drug/NEDICINAL SUBSTANCE / INTENTIONAL DRUG OVERDOSE		30	Other			N/X				No	DEATH
	SUICIDE/SELF-INJURY BY UNSPECIFIED NEWS / SUICIDE		30	Other			N/A				N a	DEATH
	SUICIDE/SELF-HOISONING BY CORROSIVE/CAUSTIC SUBST / INTENTIONAL INGESTION OF CORROSIVE SUBSTANCE		30	Other			H/X				No	DEATH
108206840077	ACUTE MI, UNSPECIFIED SITE, UNSPECIF EPISOPE CARE / ACUTE MYOCARDIAL INFARCTION	72		Other		N/A					Yas DEATH	/ LENOSP

t

Appendix Table 8.1.1.1 (con't)

Deaths in Placebo Group (Table from the sponsor's electronic submission): Treatment Group: Placebo

i

Døbject I.D.	Preferred text /Investigator text	Age et onset	Sept	t+co	lie i ght (ÉG)	Duration of Treatment (Days)	Dose at onset (mg)	Mada Dasa (mg)	Haximum Done (ng)	Onset (Day)	Treatment discontinued	Serlew adverse event type
105 05340018	CONSESTIVE HEART FAILURE /	R	Female	Caucasian	76.0	21	•	•	•	76	B e	DEATH / AENOSP
	RESPIRATORY FAILURE /	62	Fenale	Caucastan	76.4	21	٠	٩	٠	76	Ho	DEATH / AEHOSP
105 05340019	PHEIMORITIS OUE TO INHALATION " OF FOOD OR YORITUS / Aspiration Pheimonia	87	Fenale	Caucastan	46.8	13	٠	0	0	17	Yes	DEATH / AELIFETH / AEXOSP
	RESPIRATORY FAILURE / RESPIRATORY FAILURE	67	Female	Caucastan	46.8	13	٠	\$	٠	17	Yes	BEATH / AELIFETH / Achosp
106 05510053	PYREXIA OF BHENOM BELGIN / COMPLICATIONS OF WYPERTNERMIA	м	Hale	Caucastan	79.6	14	•	٠	•	61		DEATH
106 05520126	POISUNING BY ANTIALLENGIC AND ANTIENETIC DAUGS / INTENTIONAL OVELOOSE - DIVIENTOBALLEE	37	Famale	Caucastan	79.4	32	•	٥	0	39	No	DEATH
	SUICIDE/SELF-INJURY BY UNSPECIFIED MEANS / SUICIDE	37	Fanate	Caucasian	79.4	32	٥	0	0	39	He	DEATH
303 01810067	MALIGUANT NEOPLASM OF PLEURA, UNSPECIFIED / LEFT SIDE PLEURAL MESOTHELIGNA	· 43	Halq	Coucesien	66.0	232	•	٥	٠	286	Yes	DEATH / AELIFETH / AENOSP
	SECTOREP RALIGA BEOPL LYNNH NODESINTBATNORACIC / METASTASES TO MEDIASTINAL LYNNH RODES	43	Male	Caucastan	66.0	232 1	•	٠	Û	286	Tes	DEATH / AELIFETH / AENOSP
	SECONDARY MALIGN BEOPL OF Liver, spec as secondary / Metastases to liver	43	Mate	Caucastan	\$6.9	- 292	0	0	0	286	Yes	DEATH / AELIFETH / AEHOSP
303 41970272	OLONCHOPHELMONIA, ONGANISM UNSPECIFIED / DECONCHOPHELMONIA	76	Hate	Coucasian	56.0	16	• •	0	٥	19	I.	DEATN
303 02120224	ACUTE PANCREATITIS / ACUTE HEMORRHAGIC PANEREATITIS		66 Fe	male Cau	cestan	70.0	10	٥	()	0 49	NO DEATH
	PULMONARY BIDOLISH AND INFARCTION / PULMONARY EMBOLISH		66 Fe	mala Caus	cesta n	70.0	10	٠	(•	0 49	No DEATH
	UNSPECIFIED CIRCULATORY SYSTEM DISORDER / CIRCULATOR INSUFFICIENCY	n r	66 Fe	male Caud	cast an	70.0	10	٥	()	0 49	No DEATH
847 01788994	EITED BENOR POST INJURY 11/0 Open IC 100, REP CONS / Eiteadural Newstona	đ				r	N/A				6	DEATE
807 02120122	HEART DESEASE, DISPECIFIED / CARDIAC DECOMPERSATION	n					N/A				Ye	DEATE / AELIFETE /
	PHE BYOINTA, OR CANTERN UNISPECIFITIO / LEFT SIDED PHE BYONIA	n					N/A				Ye	ALENDER DEATE / AELIFETE / Alender

BEST POSSIBLE COPY

t

Med - Page 137 of 223

DRUGS	Number of Subjects '	Subject-years exposure '	Total # deaths	# deaths ≤ 30 days	Crude mortality rate ²	Mortality per 100 subject- years ²
Ziprasidone	2588	772	31 3	17	0.007	2.20
Placebo	382	52	9	5	0.013	9.62
Haloperidol	585	131	3	3	0.005	2.29
Risperidone	295	105	1	1	0.003	0.95

APPENDIX 8.1.1.2 Mortality rate for Phase II/III clinical programs in ziprasidone NDA 20-825

¹Includes integrated safety data base, Study 105 (IM: ziprasidone n=11; placebo n=12) and Study 120 (dementia: ziprasidone n=12) ²Based on # of deaths \leq 30 days

³Does not include subject 307-269-0047 (died after the cut-off date of 5/15/97) and subject JP-95-6011622

APPENDIX 8.1.1.3 Rate of Sudden Unexpected Death* (SUD) in ziprasidone NDA 20-825

DRUGS	Number of Subjects ¹	Subject-years exposure	# Sudden Deaths	SUD per 1000 subject years	
Ziprasidone	2588	772	7*	9.1	
Placebo	382	52	0	0	
Haloperidol	585	131	0	0	
Risperidone	295	105	1	9.5	

*Sudden Unexpected Death (SUD) refers to subjects found dead or who died within 24 hours of symptoms

Refer to Appendix 8.1.1.1 for listing of deaths considered to be SUD.

#Does not include subject 115-6940394; please refer to the text of Section 8.1.1

¹Includes integrated safety data base, Study 105 (IM: ziprasidone n=11; placebo n=12) and Study 120 (dementia: ziprasidone n=12)

APPENDIX 8.1.1.4 Rate of SUD in most recently submitted antipsychotic NDA data bases *

DRUGS	Subject-years exposure	# Sudden Deaths	SUD per 1000 subject years	
Ziprasidone	772	7	9.1	
Sertindole	476	5	10.5	
Olanzapine	1122.2	4	3.5	
Risperidone	508	2	3.9	
Quetiapine	865.3	1	1.1	

*Sources are the current NDA 20-852 and Review of Clinical Data: General Characteristics of the Deaths in the NDAs for Olanzapine, Risperidone, Quetiapine and Sertindole by Greg Burkhart, M.D. (HFD-120: 3/3/98)

				APPENDIX 8.1.2		
SUM	MARY OF N	IONFATA		S ADVERSE EVENTS OCCURING IN SUBJECTS TAKING ZIPRASIDONE		
2		1.2.1.1		SIDERED UNLIKELY TO BE DRUG RELATED		
Subject #	Age/Sex	Modal	Duration	Adverse Event		
	l .	Dose	(days)			
	L	(mg/d)	L			
Cardiac				· · · · · · · · · · · · · · · · · · ·		
101-5050003	33/M	40	28	Syncopal event. Ziprasidone was d/ced 11 days prior to episode and subject was on multiple medications at time of incident.		
104-5220146	45/F	80	12	Subject hospitalized with hypertensive episode with diastolic pressure up to 120 mm Hg tremouslousness and weakness. Subject with history of hypertension which had been stable with nifedipine.		
106-5520124	41/M	40	15	Hypertension with peak of 152/110. No h/o of hypertension also had facial rash (see dermatology)		
116B-5510007	44/M	120	12	Chest pain, thought to be anxiety. Subject was hospitalized for observation and sponsor concluded this was a manifestation of anxiety.		
116B-6590008	72/F	80	239	Hypertension of 200/90. Profile states that subject had a history of hypertension, but it was normal for the six months of the study prior to this onset.		
118-7090004	40/M	40	4	Sinus bradycardia (48 bpm) at end of study. Elevated CPK, but this was not checked until the end of the study just as subject started risperidone.		
303-2120105	56/M	160	280	Hypertensive episode (220/140) fell down steps resulting in subdural hematoma, skull fracture, pneumothorax		
Gastrointestinal	, d,,,,,,	A	d			
106-5550119	57/F	120	7	Incarcerated left inguinal hernia		
109-5720027	49/F	80	3	Subject was hospitalized for chest pain; original work-up in CRF suggested ECG		
	1		ſ	changes of QTc prolongation. Cardiology work up was negative and Dr. Charles Ganley,		
				HFD-110 consultant, determined that ECGs did not reflect true QTc changes. Final		
	1		1	diagnosis was exacerbation of gastroesophageal reflux.		
116B-6590001	44/F	120	47	Gastritis with Helicobacter pylori diagnosed 13 days after discontinuing ziprasidone.		
				Died of <u>benign cardiac neoplasm (myxoma)</u> 24 days after stopping ziprasidone.		
303-2710228	54/M	160	87	Subject had heartburn, epigastric pain and weight loss and within three weeks of these		
			1	symptoms discontinued ziprasidone and was hospitalized. Endoscopy showed chronic		
				gastritis with Helicobacter pylori.		

.

١.

t

:

Continuer			<u> </u>	APPENDIX 8.1.2 (CON'T)			
Genitourinary			1.000				
116B-5870006	37/F	160	256	Hysterectomy and B/L oophorectomy			
116B-5900005	33/F	160	440	Uterine Fibroids			
117-6220029	53/M	40	4	benign prostatic hypertrophy			
117-6380304	57/F	80	363 \	Total abdominal hysterectomy			
117-6690008	39/F	160	315	Urinary bladder suspension surgery.			
Pulmonary							
105-5340006	86/F	6	57	Tracheobronchitis 2 ¹ / ₂ months after d/c of ziprasidone			
106-5550136	59/M	40	2	Asthma attack; subject with history of COPD.			
106E-5550133	60/F	40	92	chest pain/exacerbation of COPD/bronchitis/adenovirus			
110-5370007	33/M	60	4	Diagnosed with lung cancer five days after d/c of ziprasidone			
116B-5680007	43/F	120	240	Test positive for HIV virus and had event of pneumocystis crainii			
116B0-74004/	33/M	80	301	Spontaneous pneumothorax			
108-5740080							
Metabolic							
117-5130512	35/M	120	194	Diabetic episode with 16 lb. wt loss and thirst and ketoacidosis			
116B-6940003 26/M		200	164	Nausea, vomiting and loss of appetite. Weight loss of > 30 lb; was hospitalized for			
				exacerbation of schizophrenia. Also, loss of appetite, n/v.			
		52	Dehydration, erratic eating pattern associated with exacerbation of schizophrenia in th				
				subject. Weight loss of 13.5 lb.			
Miscellaneous							
101-05060084	52/M	4	20	Hyponatremia (Na=111), confusion. Hospitalized, resolved in 2 days with fluid			
				restriction. Ziprasidone was discontinued.			
104-05250132	64/M	40	28	Cellulitis, right lower leg occurring twenty-two days after d/c of ziprasidone.			
104-5200275	63/F	80	28.	Fall (unclear how subject fell) with fracture of left humeral head on day 15; on day 24			
				subject developed right lower leg cellulitis and lymphangitis and was treated with			
_				antibiotics, bedrest, and leg soaks.			
116B-665008/	23/M	160	365	Knee surgery.			
117-6650101							
118-709-0004	40/M	20	4	Pt had elevated CPK, but no CPK levels done until the subject was taking risperidone			
			I I	day after taking ziprasidone. At that time the CPK levels were very high (702 U/L) and			
	1	1	1	were eventually stabilized and decreased to 544 U/L before subject was discharged fro			
	1			the hospital.			

136

÷ 1

Ţ

.

i

١

				APPENDIX 8.1.2 (CON'T)			
303-01980076	44/M	80	364	Subject was mugged and had brain concussion and facial contusions.			
303-2640336	34/M	160	359	2 nd degree burns of face, neck and shoulder from fire at psychiatry clinic.			
303-2120222	58/M	160	349	Blood pressure was 180/90 after this loss of consciousness (2-3minutes). No ECG changes, but CT showed cerebral neoplasm.			
 From Japanese study which is not part of integrated safety data base 	51/F	40,80 or 100 mg/d	24	Hyponatremia (Na=110 MEq/L), hypokalemia (K=3.2Meq/L), CPK=10,760 U/L, T AST=91, Tcortisol (32 ug/ml), LDH=796 U/L, 3+ blood in urine and decreased level of consciousness (unresponsive to verbal stimulation and abnormal reflexes). Hypertension (208/110) and tachycardia (110/min)Baseline values were normal for the above tests except for K=5.3 Meq/L. Within 8 days of hospitalization, this subjects level of consciousness improved and she was able to eat without assistance. This incident occurred 8 days after discontinuation of ziprasidone (she had akathisia and worsening depression) and subject was taking biperidene and sulpiride at time of the incident.			
Overdose			1				
114-7150457	20/M	160	20	Overdose of metformin.			
116B-5510005	38/M	80	72	Overdosed with acetylsalicylic acid.			
116B-5950018	33/F	200	182	Overdosed on lorazepam and flurazepam in presence of hospital staff.			
116B-6020005	27/M	160	159	Overdosed on lorazepam 10 days after d/c of ziprasidone.			
117-05080352	54/M	80	308	Subject overdosed on chloral hydrate and lorazepam was treated with activated charcoal and developed rhabdomyolysis with nerve damage in leg. This occurred one day after d/c of ziprasidone.			
117-06310477	35/F	80	92	Overdose on lithium eight days after d/c of ziprasidone.			

÷

137

Note: Psychiatric serious adverse events are not discussed in this review, as it is not always possible to distinguish between the effects of the study drug and the symptoms of the underlying illness.

1

t

Appendix 8.1.5.3 (From Sponsor's Electronic Submission)

Treatment-Emergent Adverse Events Reported in at Lasst 1% of Zipresidone Subjects (All Causalities) Incidence Rate for Zipresidone Greater than Placebo Short-Term Fixed-Dose Placebo-Controlled Gral Dosing Phase II/III Studies

Number of Subjects: Evaluable for Adverse Events	Ziprasidone	lobireçi do l	Placebo	
lumber of Subjects.				
Eveluable for Adverse Events	702	85	273	
	*****	•••••••••		
S KITO ACTERIE LYNDIS		,		
BODY AS A HHOLE Accidental Injury	4 4		1.6	
ACCIDENTAL INJUST	7.2	2.4 0.2 2.4	1.	
	1.3	2.4	2.5	
INFELTION	1.3	••••	0.1	
CLEDIOVASCULLE			•••	
POSTURAL HYPOTENSION	1.3 1.6	2.4	0.4	
TACHYCARDIA	1.6	Ž.4	1.1	
DIGESTIVE				
ANOR EXIA CONSTIPATION	1.9	4.7	1.1	
PIARREA	7.3 4.8	4.7	8.4	
DEY MOUTH	4.1	11.6	2.2	
DYSP EPSIA	·	16.3	7.0	
THEREASED SALIVATION	1.3	1.2	0.7	
MAUS EA	9.5	9.4	7.0	
MUSEULOSKELETAL				
ATTHRALGIA	3.3	2.4	2.9	
MYALGIA	1.1		0.4	
NERYOUS	• •		-	
AKATNISIA ANXIETY	8.4 4.4	29.2 1.2	7.0	
DIZZ INESS	7.8	9.4	9,0	
DYSTONIA	4.0	11.6	5.9 2.2	
EXTRAPYRANIDAL SYNDROME	4.7	14.1	1.1	
HYPERTONIA	3.4	11.8	Í.	
SCHIFOLENCE	14.4	23.5	6.6	
TREMOR	2.4	10.6	1.6	
RESPIRATORY				
COUGH INCREASED	2.6 4.8	2.4	0.7 1.1	
RESPIRATORY DISORDER RESPIRATORY TRACT INFECTION	4.8 3.0	8.2	2.2	
RESTINATORY TRACT INFECTION RHIMITIS	3.7	Ø.1	2.2	
SKIN AND APPENDAGES			4.4	
FUNGAL DERMATITIS	1.7		1.1	
RASH	4.1	2.4	3.1	
SPECIAL SENSES				
ABNORMAL VISION	2.7	4.7	1.5	
CONJUNCTIVITIS	1.1	1.2	1.1	
UROGENITAL				
URINARY INCONTINENCE	1.3	1.2	1.1	

Subjects with multiple occurrences of the same adverse event are counted only once for that adverse event. Only adverse events occurring while on study treatment or within the six days after the last day of study treatment were included in this table. Protocols: 104,106,114,115

÷

APPEARS THIS WAY ON ORIGINAL

Appendix 8.1.5.4

(Selected from Sponsor's Proposed Labeling)

Other Adverse Events Observed During the Premarketing Evaluation of Ziprasidone -Following is a list of COSTART terms that reflect treatment- emergent adverse events as defined in the introduction to the ADVERSE REACTIONS section reported by patients treated with ziprasidone at multiple doses > 4 mg/ day within the database of 2163 patients. All reported treatment- related events are included except those already listed in Table 1 or elsewhere in labeling, those event terms which were so general as to be uninformative, and events reported only once and which did not have a substantial probability of being acutely life- threatening. It is important to emphasize that, although the events reported occurred during treatment with ziprasidone, they were not necessarily caused by it.

Events are further categorized by body system and listed in order of decreasing frequency according to the following definitions: frequent adverse events are those occurring in at least 1/ 100 patients (only those not already listed in the tabulated results from placebo- controlled trials appear in this listing); infrequent adverse events are those occurring in 1/ 100 to 1/ 1000 patients; rare events are those occurring in fewer than 1/ 1000 patients.

Body as a Whole Frequent: abdominal pain, back pain, fever, flu syndrome, headache, pain, suicidal ideation: *Infrequent:*, abscess, accidental fall, accidental overdose, allergic reaction, cellulitis, chills, bacterial infection, fungal infection, intentional overdose, lab test abnormal, malaise, photosensitivity reaction, suicide attempt: *Rare:* abdomen enlarged, hangover effect, neoplasm, pelvic pain.

Cardiovascular System Frequent: hypertension, hypotension: *Infrequent:* angina pectoris, arrhythmia, bradycardia, electrocardiogram abnormal, hemorrhage, migraine, palpitation, syncope, vasodilation. *Rare:* peripheral vascular disorder, QT interval prolonged, retinal vascular disorder.

Digestive System Frequent: vomiting: *Infrequent:* cheilitis, duodenal ulcer, dysphagia, flatulence, gastritis, gastroenteritis, gingivitis, increased appetite, liver function tests abnormal,

oral moniliasis, rectal hemorrhage, tongue edema, tooth caries: *Rare:* eructation, fecal incontinence, gum hemorrhage, stomach ulcer.

Hemic and Lymphatic System Infrequent: anemia, ecchymosis, eosinophilia, leukocytosis, leukopenia: Rare: iron deficiency anemia, thrombocytopenia.

Metabolic and Nutritional Disorders Frequent: weight gain, weight loss: *Infrequent:* albuminuria, dehydration, edema, hyperglycemia, peripheral edema, SGOT increased, SGPT increased, thirst: *Rare:* bilirubinemia, hypercholesteremia.

Musculoskeletal System Infrequent: arthrosis, bone pain, leg cramps, myasthenia, tenosynovitis.

Nervous System Frequent: agitation, delusions, depression, dyskinesia, hallucinations, hostility, insomnia, manic reaction, myoclonus, nervousness, paranoid reaction, paresthesia, personality disorder, psychosis, schizophrenic reaction, speech disorder, tardive dyskinesia, thinking abnormal, twitching: *Infrequent:* abnormal dreams, abnormal gait, akinesia, amnesia, apathy, ataxia, catatonic reaction, choreoathetosis, cogwheel rigidity, confusion, convulsion, delirium, dementia, depersonalization, drug dependence, dysarthria, emotional lability, euphoria, grand mal convulsion, hyperkinesia, hypesthesia, hypokinesia, libido decreased, libido increased, neurosis, oculogyric crisis, paralysis, sleep disorder, stupor, vertigo,

Med - Page 143 1239

withdrawal syndrome: Rare: diplopia, incoordination, neuropathy, nystagmus.

Respiratory System Frequent: bronchitis, dyspnea, pharyngitis: *Infrequent:* asthma, epistaxis, pneumonia, respiratory distress syndrome, sinusitis: *Rare:* pneumothorax, voice alteration.

Skin and Appendages Frequent: pruritus: *Infrequent:* acne, alopecia, contact dermatitis, dry skin, eczema, exfoliative dermatitis, herpes simplex, maculopapular rash, psoriasis, seborrhea, skin hypertrophy, skin ulcer, sweating, urticaria, vesticulobullous rash: *Rare:* furunculosis, lichenoid dermatitis, pustular rash.

Special Senses Infrequent: blepharitis, deafness, dry eyes, ear pain, eye pain, otitis externa, otitis media, retinal disorder, taste perversion, tinnitus: *Rare*: abnormality of accommodation, mydriasis.

Urogenital System Infrequent: abnormal ejaculation, amenorrhea, cystitis, dysmenorrhea, dysuria, gynecomastia, hematuria, impotence, leukorrhea, menorrhagia, metrorrhagia, polyuria, urinary frequency, urinary retention, vaginitis: *Rare:* anorgasmia, breast pain, kidney pain, nephritis, pyelonephritis, uterine fibroids enlarged.

APPEARS THIS WAY ON ORIGINAL
Appendix 8.1.6.3.1 Mean Change from Baseline to Last Observation for Laboratory Test Data Short-term Fixed-Dose Placebo Controlled from Phase II/III* (adapted from sponsor's electronic submission)

				Ziprasid	one	Placebo			
•			R I	BASELINE MEDIAN	CHANGE FROM BASELINE		DISELIHE MEDIAN	CHANGE FROM DASELINE	
GROUP HEMATOLOGY	PARMETER Hemoglobin (HGB)	UNITS# G/DL	690	15.3	-0.1	261	15.1		
	Hematocrit (HCT) RBC Count Platelets	g Hill/Chm Thou/Chm	··· 679 690 678	45 4.9 255	-1	261 261	45 4.9	0	
	HDC Count Eosinophils (S)	THOU/CHM X	680 679	7.4	¢.1	261	256 7.5 3	-1 0 0	
LIVER FUNCTION	Neutrophils (abs) Total Bilirubin Total Protein	THOU/CHM MG/DL G/DL	679 681	4.41	0.1 <u>3</u>	261	4, 34 0, 5	-0.06 0.1	
_	Serum Albumin Serum Globulin	G/DĽ G/DL	682 681 690	7.1 4.3 2.9	0	261 261 260	7,2 4,3 3	0	
•	SGOT(AST) SGPT(ALT) LDH	IU/L IU/L IU/L	682 681	18 20	ě.	261 261	18 19	0 -1	
RENAL FUNCTION	Alk, Phosphatase Blood Urea	IU/L MG/DL	682 682	140 73	-1	261 261	142 76	-1	
ELECTROLYTES	Nitrogen Serum Creatinine	NG/DL	682 682	1 3 1	-1 9	261 261	12 1	0	
CLECTROL 1123	Unic Acid Sodium Potassium	MG/DL MEQ/L MEQ/L	682 682 682	5.2 140	0.Z	201	5,2 140	9.Ż	
	Chloride Bicarbonate	MED/L MED/L	682	4.4 103 22	0 0 1	261 261	4,4 103	0	
	Calcium Phosphorus Glucose, Fasting	NG/DL NG/DL NG/DL	682 681	9,6 3,8	Õ Q	zi Zi	9.6 3.8	-0.1	
	Glucose, Random Magnestum	MG/DL MG/DL	3 678 1	. 86 93 2	2 0	1 259	190 93	-41 3	
LIPIDS URINE	Cholesterol Triglycerides	MG/DL MG/DL	682 691	185 126	-3	26j 260	186 130	·12 ·13	
	Specific Gravity Urine pH Protein (gual)		679 678 588	1.02 5.5 0	0	258 258	1.02	0 0 0	
	Urine Glucose Ketones (qual)		588 588	0	Ó	214 214 214	0 0 0	0	
	Bilirubin (qual)		588	0	0	214	Ó	õ	

Based on Laboratory Test Results. 1. Converted to Standard Reporting Units 2. Adjusted to a Common Set Upper and Lover Reference Limits Ø Kruskal-Wallis test yielded a significant association at the .05 level of alpha by using the RAMK and AMOVA procedures

Med - Page 145 of 223 4 1

*The sponsor clarified that these were studies 104, 106, 114, 115

Appendix 8.1.6.3.2a Sponsor's Laboratory Reference Ranges to Determine Baseline Abnormality (11/7/97 submission from sponsor)

ę.

Ziprasidone Project Labor	ratory Kelere	nce kanges	
TEST	REF_MIN	REF_MAX	STDUNIT
Hemoglobin	· .		G/DL
Hematocr1t			%
Red Blood Cells			MILL/CMM
Platelets			THOU/CMM
White Blood Cells			THOU/CMM
Eosinophils (%)			96 ·····
Erythrocyte Sedimentation Rate	•		MM/H
Prothrombin Time Quick			SEC
Total Bilirubin			MG/DL
Direct Bilirubin			MG/DL
Protein (total)			G/DL
Albumin			G/DL
Globulin			G/DL
Aspartate Aminotransferase (GOT)			IU/L
Alanine Aminotransferase (GPT)			IU/L
Lactate Dehydrogenase			IU/L
Alkaline Phosphatase	:		IU/L
Blood Urea Nitrogen			MG/DL
Creatinine			MG/DL
Jrate			MG/DL
Sodium			MEQ/L
Potassium			MEQ/L
Chloride			MEQ/L
Bicarbonate			MEQ/L
Calcium	•		MG/DL
Phosphate			MG/DL
Cholesterol			MG/DL
Iriglycerides			MG/DL
Slucose (fasting)			MG/DL
Slucose (random)			MG/DL
Jrine Specific Gravity			~
Jrine pH	-		
Jrine Protein			
Jrine Glucose			
Jrine WBC Jrine RBC			/HPF
Jrine Ketones			./HPF
Irine Granular Casts			
Irine Hyaline Casts		•	/LPF
frine Bilirubin			/LPF

142 Med - Page 146 of 223

--

Appendix 8.1.6.3.2a (con't) Sponsor's Laboratory Reference Ranges to Determine Baseline Abnormality (11/7/97 submission from sponsor)

Cholesterol (LDL) Cholesterol (HDL) MG/DL MG/DL Thyroxine (T4) Magnesium MCG/DL MG/DL Prolactin NG/ML Urine Calcium Urine Glucose (24 Hr) Quantitative MG/DAY Urine (24hr) Protein MG/DAY TSH MG/DAY MCIU/ML Urine WBC Cast Urine (24hr) Creatinine /LPF MG/DAY Urine RBC Casts /LPF Neutrophils (Abs) THOU/CMM

APPEARS THIS WAY ON ORIGINAL

5

Med - Page 147 of 22

Appendix 8.1.6.3.2b Sponsor's criterion for determining post baseline clinical significance of laboratory values (adapted from sponsor's submission of 11/7/97)

					Column "A"	Column "B"
Test Code	Lab Test	Standard Unit	Test Type	Baseline Abnormality Criterion	Criterion for BL normal/abnormal	Post-baseline Clin Sig Criterion for BL abnormal
1	Hemoglobin (HCB)	G/DL	HEMATOLOGY	>1.0± ULN	(Tier 1) >20% Decrease from baseline	(Ther 2) < 75% of buscline
				<1.0 x LLN	>20% Decrease from baseline	< 90% of baseline
2	Hematocrit (HCT)	%	HEMATOLOGY	> 1.0 x ULN	>20% Decrease from baseline	< 75% of baseline
	•			<1.0xLLN	>20% Decrease from baseline	< 90% of baseline
3	RBC Count	MILLICMM	HEMATOLOGY	> 1.0 x ULN	>25% Decrease from baseline	< 75% of bescline
				<1.0 x LLN	>25% Decrease from baseline	< 90% of baseline
5	Platelets	THOU/CMM	HEMATOLOGY	$> 1.0 \times ULN$ $< 1.0 \times ULN$	> 700	> 120% of baseline
				<1.0x11M	3</td <td>< 80% of baseline</td>	< 80% of baseline
	WBC Count	THOURDA	HEMATOLOGY	>1.0 x ULN	> 17.5	> 125% of baseline
	WDC COUN	1100/Caller		<1.0xLLN	<25	< 75% of baseline
	ESR	MM/H	HEMATOLOGY	> 1.0 x ULN (x)	>1.2 x ULN	> 120% of baseline
	Prothrombin Time	SEC	HEMATOLOGY	> 1.0 x ULN	> 1.2 x ULN	> 120% of baseline
608	Neutrophils (abs)	THOU/CMM	HEMATOLOGY	< 1.0 x ULN	<1.0	< 75% of baseline
9	Eosinophils (%)	96	HEMATOLOGY	>1.0 x ULN	>m 10%	> 150% of baseline
21	Total Bilirubin	MG/DL	LIVER FUNCTION	> 1.0 x ULN (x)	> 1.5 x ULN	> 150% of baseline
22	Direct Bilirubin	MG/DL	LIVER FUNCTION	> 1.0 x ULN (x)	> 1.5 x ULN	> 150% of baseline
24	Total Protein	G/DL	LIVER FUNCTION	> 1.0 x ULN	> 1.1 x ULN	> 110% of baseline
				< 1.0 x LLN	0.9 < x LLN	< 90% of baseline
25	Serum Albumin	G/DL	LIVER FUNCTION	> 1.0 x ULN	> 1.1 x ULN	> 120% of baseline
				< 1.0x LLN	< 0.9 x LLN	< 80% of baseline
26	Serum Globulin	G/DL	LIVER FUNCTION	> 1.0 x ULN	> 1.2 x ULN	> 150% of baseline
		1		< 1.0 x LLN	< 0.8 x LLN	< 50% of baseline

Appendix 8.1.6.3.2b (con't) Sponsor's criterion for determining post baseline clinical significance of laboratory values (adapted from sponsor's submission of 11/7/97)

					Column "A"	
Test Code	Lab Test	Standard Unit	Тезі Турс	Baseline Absormality Criterion	Post-baseline Clin Si Criterion for BL normal/abnormal (Tier 1)	Post-baseline Clin Sig Criterion for BL abnorms (Ther 2)
28	SOOT(AST)	TUL	LIVER FUNCTION	>1.0 x ULN (x)	> 3 x ULN	> 200% of baseline
30	SOPT(ALT)	TU/L	LIVER FUNCTION	> 1.0 x ULN (x)	> 3 x ULN	> 200% of baseline
32	НДН	IN.	LIVER FUNCTION	> 1.0 x ULN (x)	> 3 x ULN	> 200% of baseline
	Alicaline Phosphatase	TUAL	LIVER FUNCTION	> 1.0 x ULN (x)	> 3 x ULN	> 150% of baseline
	BUN	MO/DL	RENAL FUNCTION	> 1.0 x ULN (x)	>13xULN	> 130% of baseline
48	Creatinine	MG/DL	RENAL FUNCTION	> 1.0 x ULN (x)	> L3 x ULN	> 130% of baseline
- 54	Sodium	MEQ/L	ELECTROLYTES	> 1.0 x ULN	> 1.05 x ULN	> 105% of baseline
		+		< 1.0 x LLN	< 0.95 x LLN	< 95% of baseline
55	Potessium	MEQIL	ELECTROLYTES	> 1.0 x ULN	> 1.1 x ULN	> 110% of baseline
				< 1.0 x LLN	< 0.9 x LLN	< 90% of baseline
	h					
- 20	Chloride	MEQIL	ELECTROLYTES	> 1.0 x ULN < 1.0 x LLN	> 1.1 x ULN	> 110% of baseline
					< 0.9 x LLN	< 90% of baseline
57	Bicarbonate	MEQ/L	ELECTROLYTES	> 1.0 x ULN	> 1.1 x ULN	> 110% of baseline
				< 1.0 x LLN	< 0.9 x LLN	< 90% of bescline
58	Calcium	MG/DL	ELECTROLYTES	> 1.0 x ULN	> 1.1 x ULN	> 110% of baseline
				< 1.0 x LLN	< 0.9 x LLN	< 90% of baseline
59	Phosphorus	MG/DL	ELECTROLYTES	> 1.0 x ULN	> 1.2 x ULN	> 120% of baseline
	<u> </u>	_		< 1.0 x [L]N	< 0.8 x LLN	< 80% of baseline
50	Uric Acid	MG/DL	ELECTROLYTES	> 1.0 x ULN	> 1.2 x ULN	> 120% of baseline
199	Magnesium	MEQAL	ELECTROLYTES	> 1.0 x ULN	> 1.1 x ULN	> 110% of baseline
				< 1.0 x LLN	< 0.9 x LLN	< 90% of baseline
63	Cholesterol	MG/DL	LIPIDS	> 1.0 x ULN (x)	> 1.2 x ULN	> 150% of baseline
173	HDL Cholesterol	MG/DL	LIPIDS	<1.0 x LLN (?)	< 0.8 x LLN	< 80% of baseline
	LDL Cholesterol	MG/DL	LIPIDS	> 1.0 x ULN(x)	> 1.2 x ULN	> 120% of baseline
64	Triglycerides	MG/DL	LIPIDS	> 1.0 x ULN (x)	>12xULN	> 150% of baseline
67	Glucose, Fasting	MG/DL	1	> 1.0 x ULN	> 1.2 x ULN	> 150% of baseline
				<1.0x11N	< 0.6 x LLN	< 50% of baseline
223	Prolactin	NGML.		> 1.0 x ULN (x)	> 1.1 x ULN	> 150% of baseline

Med - Page 149 of 2234 5

Appendix 8.1.6.3.2b (con't) Sponsor's criterion for determining post baseline clinical significance of laboratory values (adapted from sponsor's submission of 11/7/97)

					Column "A"	Column "B"
Test Code	Lab Test	Standard Uzit	Test Type	Baseline Abnormality Criterion	Post-baseline Clin Si, Criterion for BL normal/abnormal (Ther 1)	Post-baseline Clin Sig Oritorion for BL absorms (Ther 2)
78	Protein (qual)	- 	URINE	>1.0xULN	>= 2+	> baseline + 2
_	Urine Obscore		URINE	>10xULN	>= 2+	> bescline + 2
80	Urine WBC	APP	URINE	>1.D x ULN	>= 6	> baseline + 6
81	Urine RBC	APF	URINE	>1.0 1 ULN	>= 6	> bescline + 6
86	Keiones (goal)		URINE	>1.Dz ULN	≻= +	> baseline + 1
88	Granular Casts	APF	URINE	>1.0 x ULN	>1	> bascline + 1
90	Hyaline Casts	1.PF	URINB	>10xULN	>1	> baseline + 1
115	Bilirubis (qual)		URINB	>1.0 x ULN	≻=}+	> baseline + 1
600	Red Cell Cast	/LPF	URINE	>1.D x ULN) = <u> </u>	> baseline + 1
442	White Cell Cast	1.PP	URINE	>10xULN	>=1	> baseline + 1
76	Specific Gravity		URINE	>10xULN	> 1.035	> 1.035
	ļ	7		< 1.0 x LLN	< 1.000	< 1.000
77	Urine pH		URINE	>1.0x ULN	> 1.1 x ULN	> 1.1 x ULN
				<1.0 x LLN	< 0.9 x LLN	<0.9 x LLN
495	Creatinine	MODAY	URINE	> 1.0 x ULN (x)	> 1.1 x ULN	> 110% of baseline
302	Calcium (quant)	MG/DAY	URINE	> 1.0 x ULN (x)	> 1.1 x ULN	> 110% of baseline
308	Protein (guant)	MGIDAY	URINE	$> 1.0 \times ULN(x)$	> 1.1 x ULN	> 110% of baseline
307	Glucose (quant)	MG/DAY	URINE	> 1.0 x ULN (x)	> 1.1 x ULN	> 110% of baseline

Column "A" Column "B"

APPEARS THIS WAY ON ORIGINAL

Med - Page 150 of 223

Appendix 8.1.6.3.2c Incidence of Clinically Significant Laboratory Abnormalities for Short term Fixeddose placebo controlled studies 104, 106, 114, 115. (adapted from sponsor's electronic submission)

Number of Subjec Evaluable fo With Clinica	r laboratory abnorm lly significant lab	alities oratory abnorm	wlities	- • -	sidone 688 (\$7%)		P1ac 86	:#D0 261 (33£)
					jects v ormalit		Sub. Abri	ects with irmalities
Group	Parameter	Units	Criteria* Ø	• .	••			
NENATOLOGY	Hemoglobin (HGP) Hematocrit (HCT) RBC Count Platelats	G/DL X MILL/CHH THOU/CHH	> 205 decrease* > 205 decrease* > 255 decrease* < 75 > 700	694 694 694 693 693	3 1 2 2 0	0 0 0	261 261 261 261	0 0 0 2
	HBC Count	THOU/CHIN	< 2.5 > 17.5	684 684	04	0 0 1	261 261 261	0
•	Ecsinophils (2) Prothrombin Time	X SEC	>= 10\$ > 1.2 x ULN	683 2 683	21 9 3	\$ 0	261 261 1	3 1 0 0
	Neutrophils (abs) ESR	MH/H	<1.0 >1.2 x ULN		-	•	261	1
LIVER FUNCTION	Total Bilirubin Direct Bilirubin Total Protein	HG/DL HG/DL G/DL	> 1.6 x ULN > 1.6 x ULN < 0.9 x LLN	685 9 685	2	11	261 4 261	
	Serum Albumin	6/DL	> 1.1 x ULH < 0.9 x LLH	685 685 685	Ì	Ó Q	261 261 261	0
	Serum Globulin	G/DL	> 1.1 x ULN > 1.2 x ULN	684	ŏ	0	261 261	0
	SGOT(AST) SGPT(ALT)	IV/L IV/L	< 0.8 x LLN > 3.0 x ULN > 3.0 x ULN	684 685 694	2	0 1	261 261 261	0 0 1
RENAL FUNCTION	LDN Alk. Phosphatase Blood Urea	IU/L IU/L MG/DL	> 3.0 x ULN > 3.0 x ULN > 1.3 x ULN	685 685	1	0	261 261	0
	Nitrogen Serum Creatinine	MG/DL	> 1.3 x ULN	685 685	1	0	261 261	0
ELECTROLYTES	Uric Acid Sodium	MG/DL MEQ/L	> 1.2 x ULN < 0.95 x LLN > 1.05 x ULN	685 685 685	34	0	261 261 261	1
	Potassium	HEQ/L	< 0.9 x LLH	685	Ó	ŏ	261	Ô,
	Chloride	MEQ/L	> 1.1 x ULN - < 0.9 x LLN	685 685	10 2	1 0	261 261	5
	Bicarbonate	HEQ/L	> 1.1 x ULN < 0.9 x LLN	685 21 21	1	ŏ	261	0
	Calcium	MG/DL	> 1.1 x ULH < 0.9 x LLH	685	1	- 0	261	0
	Pho sphor us	MG/DL	> 1.1 x ULN > 1.2 x ULN < 9.8 x LLN	685 685 685	0 7 2	0 1 0	261 261 261	26
LECTROLYTES	Glucose, Fasting	MG/DL	> 1.2 x ULN	14	0	0	3	0
	Glucose, tandom	NG/DL	< 0.6 x LLN > 1.2 x ULN	14 684	67	9 8	261 261	20
	Nagnesium	MG/DL	< 0.6 x LLN < 0.9 x LLN > 1.1 x ULN	684 4	3 0 0	0		1 10
IPIDS IRINE	Cholesterol Triglycerides Specific Gravity	MG/DL MG/DL	> 1.2 x ULH 0 > 1.2 x ULH 0 < 1.000	685 684 685	16 85 9	12	261 261 269 259	0 17 0
	Urine pH		> 1.035 Ø < 0.9 x LLN	685 685	0	0	2) 2)	0 3
	Protein (qual)		> 1.1 x ULR >= 2+	685 685	3	0	259	ī
	Urine Glucose Urine NBC	/HPF		685 566	2 53 52	11	259 228 223	21 21
	Urine BBC Ketones (gual)	/HPF	>= 6 >= 1+	561 685	n n n	6 2	23	4
	Granular Casts Nyaling Casts	/LPF /LPF		25	0 3	0 60	4	i 2
	Bilirubin (qual)	NG/DL	>="1+ > 1.2 x ULN	684	ĩ	ŝ	259	0
	LDL Cholesterol HDL Cholesterol	HG/DL	(Ö.Ö.X LLII	2	ò	ò	1	0
	Protein (quant) White Cell Cast	MG/DAY /LPF	>1.1 x UL# >⇒1	1	0 1	0 100	1	v
	Red Cell Cast	ILFF .	j ⊢ ī	ĩ	ĩ	100		

Includes protocols 104, 106, 114, 115

B = Total number of subjects with at least one observation of the given lab parameter while on study treatment or during lag time.
 n = Mumber of subjects with a clinically significant abnormality
 Change from baseling
 Fisher Exact 2-tailed test yielded a significant association at the .05 level of alpha
 Note: The criteria in this table lists only Column "A" from Appendix 8.1.6.3.2a when in fact both

Column "A" and Column "B" were used to establish criteria for abnormal values in this table.

Med - Page 151 of 25 7.

Appendix 8.1.6.4	Incident of Clinically Significant Laboratory Test in	all oral Phase II/II Studie
(adapted from sp	ponsor's electronic submission)	

Kuber of Subjec Evaluable fo With Clinica	ts: r laboratory abnorm lly significant lab	alities oratory abox	•	esi done 1794 6 (\$716)			oebo 835 (47%)	
	· · · · · · · · · · · · · · · · · · ·			bjects of normaliti			eots.w	
Group	Parameter	Units		A	*	ĸ	a	%
ENATOLOGY	Hemoglobin (HGB) Hematoorit (HGT) RBC Count Platelets	C/DL % WILL/CNN THOU/CNN	1778 1778 1778 1778 1777	6 2 4 6	0	833 833 833 833	1 1 0 8	0 0 0 1
	WBC Count	THOU/CIN	1777	2 5 16	0 0 1	833 893 833	0 0 1	0 0 0
	Eosinophiis (%) ESR	NV/H	1775 3	63 0	4 D	833 1	8	200
	Prothroubin Time	SEC	6	0	0	1	Ō	
IVER FUNCTION	Neutrophils (abs) Total Billrubin	NG/DL	1771	9	1	833	1	0
	Direct Billrubin	KG/DL	1780	9	1	334	3	1
	Total Protein	G/DL	19 1778	1 3	ŝ	\$ 334	0	0
			1778	š	ŏ	834	ó	ŏ
	Serum Albumin	6/DL	1780	3	Ó	334	1	0
	Comm Clabulis	ć m	1780	1	Ó	334	Ó	Ó
	Serum Globulin	CADE	1352	2 "	0	264	0	
	SCOT (AST)	10/2	1352 1780	6 6	0 0	264 334	0 2	0
	SCPT (ALT)	iŭ/L	1776	17	ĭ	334	2	- i
	LOH	1U/L	1361	1	0	264	ō	ó
P	Alk. Phosphatase	10/1	1781	. 0	0	334	Ó	Ò
ienal function	Blood Urea	NG/DL	4000	-	•		-	_
	Nitrogen Serum Creatinine	NG/DL	1363	3.	0	264	1	0
LECTROLYTES	Uric Acid	WG/DL	1783 1359	49	0 1	334	1	0
	Sodium	WEQ/L	1782	26	1	264 334	3	1
			1782	3	ò	334	ō	6
	Potassium	NEQ/L	1779	Ó	Q	334	Ó	0
	Al land de	NED /I	1779	17	1	334	7	2
	Chloride	¥EQ/L	1363 1363	4	0	264	0	Q
	Bicarbonate	WEQ/L	41	ò ·	ŏ	264 6	ŏ	0
			ăi	ŏ	ŏ	6	ŏ	ŏ
	Calcium	NG/DL	1362	Ž	ō	264	ĩ	ŏ
		No 101	1362	1	0	264	2	1
	Phosphorus	WG/DL	1361	16	1	264	7	3
LECTROLYTES	Oherehomut	NG/DL	1961	•••••	1	264	1	·····
	Phosphorus Glucose, Fasting	NG/DL	1361 70	8 5	7	204		0 33
	eroverst resoring		70	ă	ó	š	ó	:0:
	Glucose, Random	NG/DL	1344	178	13	264	32	. <i>S</i> Ż
			1344	7	1	264	0	0
	Nagnesium	WG/DL	10	<u> 0</u>	0	1	ç	0
IPIDS	Chol esterol	NG/DL	10 1363	0 227	0 17	7 264	1. 40	100 15
	Triglycerides	NG/DL	1359	322	24	264	48	18
RINE _	Specific Gravity		1359	- O	Ö	262	ŏ	ŏ
			f 1359	4	0	262	4	2
	Urine pH		1359	Q	0	262	<u> </u>	0
	Protein (qual)		1359 1761	29	1 2	262 332	3.	1
	Urine Glucose		1760	13	1	832	ŏ	5
	Urine NBC	THPF	1201	147	12	231	24	10
	Urine RBC	THPF	1189	91	8	226	25 11	- 11
	Ketones (qual)	A DC	. 1760	41	2	332		3
	Granular Casts	ሊ ኦ ና ለ 05	5 17	1	20 53	3	0	0
	Hyaline Casts Bilirubin (qual)	<i>N</i> .PF	1259	92	0 23	262	1	25 0
	LDL Cholesterol	NG/DL	4	ō	ŏ	6 V %	•	
	HOL Cholesterol	NG/OL	Ğ	ō	0			
	Protein (quant)	NG/DAY	2	0	Ó	1	0	0
	White Cell Cast	/LPF	1.	···· 1	100			
	Urine Creatinine	86/DAY /LPF	1	· 1	100 100			
ORNONES	Red Cell'Cast Thyroxine (T4)	NGG/DL	218	2		37	1	,
			218	5	2 0	57 57	ó	20
	Prolactin	NG/NL	# 741	148	20	75	3	4
	TSH	NCIU/NL '	224 224	4	2	56 56	0	0 2

Includes protocols 015, 101, 102, 104, 104, 105, 106, 108, 108, 109, 109, 110, 111, 114, 115, 1169, 117, 118, 122, 301, 302, 303, 304, 305

It a Total number of subjects with at least one observation of the given isb parameter shile on study treatment or during lag time. n • Number of subjects with a olinically significant abnormality • Change from baseline # Fisher Exact 2-tailed test yielded a significant association at the .05 level of signa Criteria for clinically significant laboratory abnormalities not adjusted for abnormal baseline laboratory values Mod 1

14

Med - Page 152 of 223

Appendix 8.1.7.3.1 Median changes from baseline for short term placebo controlled Phase II/III trials (from sponsor's electronic submission)

		Zipra	1 done		Halop	eridol	Flacebo				
		Hedian	Median Change From Baseline To Last Obs		Median Baseline			Hedian Baseline	Median Change From Baseline To Last Obs		
Standing	•••••			••••	•••••	•••••	••••	••••	• • • • • • • • • • • • • • • • • • • •		
Systolic SP (milg)	674	114.5			129.4	1.0	260	118.0	0.1		
Diastolic BP (maig)	674	78.0	A A	23 23	80.0	. .	200	80.0	ě.		
Heart Bate (bpm)	669		0.0	Ä	- 		266		ě.		
Sitting	,		•.•						۷.		
Systolic 87 (andig)	690	118.0	0.0		120.0	2.0	265	118.0	•		
Plastolic SP (mmHg)	690	76.0	ě.ě	- 11	78.0	0.4	26	76.0	¢. ¢.		
Heart Bate (bom)	690	80.0	0.0	- 22		1.6	26	80.4	ě.		
Supine	0 74	.v	4.4	-	ev. v	4.0	2.00	ev.v	۷.		
Systelic 97 (mmia)	,	120.0	-8.0								
Plastolic 8P (mmig)		78.0	-10.0								
Heart Bate (bpm)	1	70.0	Z.9								
esperature (C)	243	36.6	9.0					X. 6	Q .		
wight (kg)?	622	75.8	0.5	- 74	79.4	0. 0	227	75.6	٩.		

If is the total number of subjects with a baseline observation and at least one observation while on study treatment or within six days after the last day of study treatment for the given wital sign parameter. * Trusted-labils test yielded a significant association at the .46 level of alpha by using the RANK and ANOVA procedures. Protocols: 104.106.114.115

Appendix 8.1.7.3.2 Incidence of Clinically Significant Changes in Vital Signs in Short-term Placebo Controlled Studies:

	1	iprasido	ne	H	aloperid	01		Flacebo	
			Percent Changed			fercent Changed	H	Total Changed	
Studios Sustalis BD (mile)	••••••	•••••	•••••	• • • • • • •			•••••	•••••	••••••
Standing Systolic BP (mmHg) Increase (BP>180, CHG>=20)	674	1	0.1	83	0	0.0	260	1	0.4
Decrease (BPC90, CHG/=20)	674		3.3	83	š	3.6	260	ģ	3.5
Standing Diastolic BP (mmHg)	0/4	4	3.3	60	•	3.0	1.04	,	3.9
Increase (BP>105, CH6>=15)	674	14	2.1	\$3	,	1.2~	260	4	1.5
Decrease (BPC50, CHGK=-15)	674		0.4	83		3.6	260	3	1.2
Standing Heart Bate (bom)		3	v	04		2.0	1.00		4.4
Increase (BP>129, CHG>=15)	669	21	3.1	83	,	2.4	256	9	3.5
Decrease (BP(50, CHG(=-15)	669		0.4	ន័	2	1.2	256	ő	0.0
Sitting Systolic BP (mmHg)		•	•.•		-	• • •		•	•.•
Increase (BP)180, CH6>=20)	690	,	0.1		0	0.0	265	1	0.4
Decrease (BPC90, CHG(=-20)	690		1.6		š		265	6	2.3
	030		1.0		3	3.0	100		2.3
Sitting Diastolic BP (amilg) Increase (BP>105, CH6>=15)	690	12	1.7	84	,	1.2	265	,	20
Decrease (BPC50, CH6/=15)	690		0.7	- 5		0.0	265	7 2	2.6
Sitting Heart Rate (bpm)	074		V.7		•	v.v	¥ 63	2	9.8
Increase (BP>120, CH6>=15)	690	,	0.1	84	0	0.0	265	z	9.8
Dacrosse (BP/120, CH6/=-15)	690		0.3		š	3.6	265	2	0.8
	034	4	V.3		3	3.0	t 63	4	Ψ.8
Supine Systolic BP (mHg)		٥	0.0	ð	•		ó	•	
Increase (BP>180, CHG>=20)		ě		š	ŏ		ŏ	0	
Decrease (BPC90, CH6(=-20)		•	9.0	v	v		v	Ŷ	
Supine Diastolic SP (mmHg)								•	
Increase (BP>105, CHE>=15)		0	0.0	0	0			0	
Decrease (BP(50, CHG(=-15)	1	¢	0.0	Q	v		0	Q	
Supine Heart Late (bpm)					•			•	
Increase (BP>120, CHG>=16)	1	0	0.0		0		0		
Decrease (BPK50, CH6K=-15)	1	0	0.0	9	9		Q	Q	
Height (kg)			• •			• •			
Increase (CHG)=75)*	622		9.8	- 74		5.4	227	9	4.9
Decrease (CHGK#-7%)	622	- 16	2.6	- 74	1	1.4	227	7	3.1

If is the total number of subjects with a baseline observation and at least one observation while on study treatment or within six days after the last day of study treatment for the given vital sign parameter. To be a clinically significant change, a value has to both meet the criterion value and represent a change from baseline of at least the magnitude noted at any time during the study treatment or within the six days after the study treatment. If fisher exact two-tailed test yielded a significant association at the .05 level of alpha. Protocols: 104.106.114.115

Appendix 8.1.8.3.1 Mean and Mean Change values of electrocardiogram variables comparing baseline and end of treatment with ziprasidone and placebo in short term placebo controlled Phase II/III trials (from sponsor's submission dated 11/13/97).

Table H.S.23.1 (Change from Base Short-Term Fixed By BID Dose - Ce	line to Naz Doze Place strolly Ree	faun Read 20-Contro	tine Value	In ECG R Desing P	eedines.			40ma 810		••••••••••••••••••••••••••••••••••••••	dina BID	`		ge 1 of 2	!
		Base Nea	Nex. Heen**		Base Hean	Nax Nax	•	Base Nean	Max Notare		Base Nea	Kex Necn++	••••••	Bese Neen	Nex Hean**
"OIC Int QI int Weart Rate PR int QRS int	250 250 250 250 254 - 250	199.8 348.6 80.4 149.2 86.1	4.3 7.6 4.6 3.4 1.6	230 236 238 238 230	396.9 354.6 76.6 148.0 86.6	8.6 5.0 7.8 4.3 1.5	138 138 138 138 138	297.6 251.2 78.6 148.4 86.3	12.6 7.6 6.6 1.9 1.3	111 111 111 111 111 111 111	294.0 352.6 38.5 149.2 84.8	15.2 15.3 5.4 2.6 1.7	100 100 100 100 100	394.6 352.6 76.7 150.0 85.5	19.8 14.4 6.7 2.5 1.8
(CONTINUED) Protocols: 104.3 *OIC int - OI in * Maximum changed - the last day of Date of Lable ge	06,114,116 st/SORT(60/(e from base number of s study treat	Heart Bat line for ubjects (ment.	each subje	ict where	beselin	ts the la	nst (C6 t	aten befi	ore the fl	rst day e	fstady				

Table H.S.23.1 [Change from Base Short-Term Fixed By BID Dose - Ce	line to Max Dose Place	i nus Re ad De-Contri	ting talue	10 FCC P	Sadi ant		Page 2 of 2
	-<	100mg 81)	Ha	loperidol		
		Base Mean	Max Nean**		Base Mean	Nex Nean**	
*QTc int QT int Weart Rate PR int QRS int	77 71 71 71 71	402.7 354.9 78.6 148.3 87.3	15.0 17.1 4.5 1.6 1.2	76 76 76 76 76	400.2 357.3 76.8 150.9 87.1	4.1 7.1 4.4 3.1 0.9	
Protocols: 104.1 *OTc int - OT in ** Maximum chang Total Changed - the last day of Date of table ge	st/SQRT(60/(e from base number of s study treat	line for ubjects (ment.	each suble	ct where baselin	baselin e ECG val	is the last ue and post-	ECG taten before the first day of study treatment. baseline value within six days after

BEST POSSIBLE COPY

APPEARS THIS WAY

Appendix 8.1.8.3.2 ECG: percent of subjects who met sponsors criteria for clinically significant changes in short term placebo controlled studies. (from sponsor's submission dated 11/13/97)

			flacula		·	<time 40mg="" bid="" bid<="" th=""><th colspan="3">68mg \$10</th><th colspan="2">Stars BID</th><th></th></time>			68mg \$10			Stars BID				
			Total Changed	1	U	Tota) Changed	5	r	Teta) Changed	t		Total Changed	1	8	Tetal Changed	1
•Øfc lat	>-400 msec >-540 msec >-75 msec increase ** >-105 increase ** >-155 increase **	21 250 250 250 250 250	11 2	4.4 4.4 9.9	213 713 714 239 239 239	0 8 9 32 1 9	5.7 9.4	138 138 139 139 138 138	11 11	ŧ.;	112 112 111 111 111 111	1 8 8 12 1	¢.9 Q 18.8 4.9	100 100 100 100 100	9 0 16 - 3	16.C 3.C

	>>180mg BID Haleperide)								
			Total hanged	5		Teta) Changed	1		
•QTc 1nt	>-480 msec >-500 msec >-75 msec increase ** >-105 increase ** >-155 increase ** >-205 increase **	77 77 77 71 71 71	0 0 11 1	0 9 14.3 1.3 0	76 76 76 76 76 76	0 0 4 0	0 0 5.3 0		

. . .

APPEARS THIS WAY ON ORIGINAL

BEST POSSIBLE COPY

151 Med - Page 155 of 223

(adapted from sponsor's submission)

Simpson-Angus Rating Scale Score** - Percent of Subjects with a Postbaseline Change by Visit -All Subjects Not Taking Benztropine, Observed Cases Ziprasidone Protocol 114

		Day 7+	Day 21	Day 42	Last
Ziprasidonę, 40 mg BID	Increase Decrease No Change N	19.8 34.6 45.7 81	30 36.7 33.3 60	16.3 48.8 34.9 43	15.7 42.2 42.2 83
Ziprasidone. 80 mg BID	Increese Decrease No Change N	19.7 18.4 61.8 76	27.9 20.6 51.5 68	39.1 23.9 37 46	32.5 23.4 44.2 77
Placebo	Increase Decrease No Change N	21.1 23.7 55.3 76	16.4 24.6 59 61	16.7 40.5 42.9 42	21.5 25.3 53.2 79

**Simpson-Angus Rating Scale Score equals the sum of SARS items 1 through 10.

*Day 7 - visit 7; Day 21 - visit 21; Day 42 - visit 42; Last - last visit, planned or unplanned.

BEST POSSIBLE COPY

Appendix 8.1.9.2b Akathisia Assessment for Studies 114 and 115 using the Barnes Akathisia Scale

(adapted from sponsor's submission)

		Day 7+	Day 21	Day 42	Last
iprasidone, O mg BID	Increase Decrease No Change N	23.3 • 26.7 50 60	20.9 23.3 55.0 43	30.3 30.3 39.4 33	27 28.6 44.4 63
iprasidone,) mg BID	Increase Decrease No Change N	15.3 28.8 55.9 59	16.3 40.8 42.9 49	12.5 37.5 50 32	20 33.3 46.7 60
iprasidone. DO mg BID	Increase Decrease No Change N	17 25.5 57.4 47	20.5 41 38.5 39	20 36:7 43.3 30	21. 38. 40. 41
lalopertdol	Increase Decrease No Change N	27.8 8.3 63.9 36	33.3 10 56.7 30	35 15 50 20	49 19 40
lacebo	increase Decrease No Change N	15.4 21.2 63.5 52	20 22.9 57.1 35	4.8 23.8 71.4 21	13 25.9 61.1 54

Simpson-Angus Rating Scale Score** - Percent of Subjects with a Postbaseline Change by visit -

All Subjects Not Taking Benztropine, Observed Cases

**Simpson-Angus Rating Scale Score equals the sum of SARS items 1 through 10.

+Day 7 - visit 7: Day 21 - visit 21: Day 42 - visit 42: Last - last visit, planned or unplanned.

Barnes Akathisia Scale Global Score - Percent of Subjects with a Postbaseline Change by Visit -All Subjects Not Taking Beta-Blockers, Observed Cases

rnes Akathisia Scale Global Score - Percent of Subjects with a Postbaseline Change by Visit -1 Subjects Not Taking Beta-Blockers, Observed Cases prasidone Protocol 114 Day 7* Last Day 21 Day 42 iprasidone. Increase 20.6 12.5 15.7 16.3 0 mg 810 Decrease 12.4 16.7 17.6 18.4 No Change 70.8 66.7 65.3 67 97 72 51 98 ï iprasidone, Increase 12.9 13 19.5 13.8 0 mg 810 18.5 17.1 22.4 18.3 Decrease No Change 68.5 63.4 63.8 68.8 92 82 58 93 12.7 8.2 11 1 lacebo Increase 9.3 27.1 Decrease 23.5 27 25.6 hand No Change 65.4 60.3 65.1 64.7 81 63 43 85 ay 7 - AR 7; Day 21 - visit 21; Day 42 - visit 42; Last - last visit, planned or unplanned. i

 Ziprasidone Protocol 115

 Day 7*
 Day 21
 Day 42
 Last

 Ziprasidone, increase
 17.9
 10.5
 11.1
 13.8

 20 on RID
 Derrease
 14.1
 21.1
 20
 21.3

Ziprasidone. 20 mg BID	Increase Decrease No Change N		17.9 14.1 67.9 78	21.1 68.4 57	20 68.9 45	21.3 65 80	
Ziprasidone, 60 mg BID	Increase Decrease No Change N		14.7 14.7 70.6 68	16.4 10.9 72.7 55	8.3 13.9 77.8 36	14.5 10.1 75.4 69	
Ziprasidone. 100 mg BID	Increase Decrease No Change N		18.4 6.6 75 76	9.5 17.5 73 63	9.3 18.6 72.1 43	14.5 14.5 71.1 76	
Haloperidol	Increase Decrease No Change N		29 12.9 58.1 62	29.1 21.8 49.1 55	23.1 20.5 56.4 39	29.9 19.4 50.7 67	
Placebo	Increase Decrease No Change N	ţ	14.3 21.4 64.3 70	18.6 20.9 60.5 43 Med	8.3 29.2 ¹ 62.5 24 - Page 156	16.4 27.4 56.2 73 of 223	

*Dav 7 - visit 7; Day 21 - visit 21: Dav 42 - visit 42: Last - last visit, planned or unplanned

Review of Clinical Data

Review of Data Quality, Coding, All Cause Mortality and Sudden Deaths

NDA:	NDA 20-825
Sponsor:	Pfizer
Drug:	Ziprasidone
Route of Administration:	Oral
Reviewers:	Gerard Boehm, M.D., M.P.H. James F. Knudsen, M.D., Ph.D.
Author:	Gerard Boehm, M.D., M.P.H.
Review Completion Date:	2/3/98

The objectives of this review are to evaluate the methods of coding and overall quality of the data and to review all cause and cause specific mortality with emphasis on sudden deaths in ziprasidone trials. This review covers information presented in the NDA and the 4 month safety update. I used the results from Dr. Knudsen's evaluation of the data quality and coding for this report.

Methods

Accuracy of the database

To verify accuracy of the data submitted with the NDA, we cross checked the data for deaths and cardiac adverse events presented in line listings/tabulations, narrative summaries, and CRF's. Specifically, the sources were examined for inconsistencies and omissions.

To evaluate the coding process, subsumed investigator verbatim terms for adverse events were compared with the preferred terms included in appendix VI. In addition, the events that were coded with selected preferred terms were reviewed in more detail. Using narrative summaries and CRF's, the adverse events coded with the preferred terms arrhythmia, tachycardia, bradycardia, syncope, hypotension, postural hypotension, heart arrest and circulation failure were compared across sources to determine if the coding process resulted in appropriate groupings.

Deaths

The overall mortality rates were compared between ziprasidone group and the comparator groups using the number of deaths that occurred within 30 days of the last

exposure to drug for deaths presented in the NDA and safety update. The ziprasidone mortality rate was also compared to the mortality rates of other recently approved antipsychotic medications.

I reviewed and summarized the narrative summaries and the CRF data for ziprasidone patient deaths. I then classified the deaths into 4 categories. Those deaths for which there was a definite external cause (ex. suicide jumped in front of a train) were placed in the first category. Those deaths with a probable external cause, but where there remains some question about the possibility of a sudden death (ex. a car accident where the driver could have experienced sudden death) were placed in the second category. The third category contains deaths that appear to be related to an underlying process (ex. cancer or pneumonia). The last category contains the deaths that occurred suddenly, noting which deaths were explained by an underlying process (example sudden death with ruptured aneurysm discovered on autopsy). Using this scheme, I calculated an external cause (definite + probable) and sudden death rate for the NDA data alone, and for the combined NDA and safety update information. The calculated sudden death rate for ziprasidone was compared to the sudden death rate for recently approved antipsychotic medications.

Results

Approach to Safety, Data Accuracy and Specificity of the AE Coding

The sponsor defined treatment emergent adverse events as 1) events not present at baseline or during the baseline period and that occurred after treatment began 2) events that were present at baseline but increased in severity after beginning treatment. AE surveillance occurred at each study visit. Investigators recorded observed or volunteered AE's that occurred during the treatment period or within 6 days after the last day of treatment. The sponsor translated investigator verbatim terms to preferred terms using the COSTART dictionary. The AE's were presented in tabular form with information about treatment emergence, body system classification, investigator assessment of severity (mild, moderate or severe) and causality. When an event for a patient was reported with more than one severity, the summary tables reported the greatest severity recorded by the investigator. Events without investigator assessments of severity were classified as severe.

The investigator verbatim terms listed in the CRF's of the patients who died, had serious cardiovascular adverse events, or dropped out due to cardiac related events, were congruent with those listed in the tabulations/data listings and included in the narrative summaries. The narrative summaries often provided more clinical detail, particularly about past medical history. Additionally, the narrative summaries often clinical information with the exception that they contained vital sign, ECG, and laboratory data which was less complete in the narrative summaries.

In general, the COSTART coding of the investigator verbatim terms was appropriate. The sponsor provided a comprehensive line listing of adverse events aggregated across all Phase II/III studies which included the investigator verbatim and coded preferred term for each report. With few exceptions, the sponsor's coding practices were neither excessively narrow nor broad. The sponsor may have been too overly inclusive in deciding what to subsume under the preferred term postural hypotension. The verbatim terms orthostatic hypotension, dizziness on standing, and lightheadedness with standing were subsumed under the preferred term postural hypotension. This approach increases the sensitivity of the coding for detecting patients with orthostatic changes in blood pressure but probably lessens the specificity.

Orthostatic change in pulse, irregular pulse, and arrhythmia were subsumed under the preferred term arrhythmia, whereas orthostatic tachycardia and elevated pulse were subsumed under the preferred term tachycardia. This grouping probably decreased the ability to detect orthostatic changes by report of change in pulse. Since there are other parameters for detecting orthostatic changes (blood pressure change, symptoms) these coding practices probably had minimal impact on the ability to identify this event. Loss of consciousness and fainting were appropriately subsumed under the preferred term syncope and not under hypotension.

The data quality for death and cardiovascular adverse events was adequate. With few exceptions, the sponsor's coded terms accurately reflected the investigator verbatim terms. These data allow an accurate assessment of the events occurring during the ziprasidone development program.

Deaths

There were 32 deaths reported with the NDA and safety update. Eighteen deaths occurred within 30 days of the last exposure to ziprasidone. One of these deaths (suicide from the safety update) was from the Japanese database which was maintained separately and did not contribute to the person time used to calculate rates (p.5 Safety Update). That death is not included in the following discussions or the mortality rate calculations.

In the original NDA submission, there were 14 deaths that occurred within 30 days of the discontinuation of ziprasidone. The all cause mortality rate for this population was 2.2/100PY (14/626PY). The safety update included 3 additional deaths within 30 days of last exposure and almost 150 additional patient years of exposure. The all cause mortality rate including the safety update data was 2.2/100PY (17/772PY). This mortality rate is higher than the rates observed for other recently reviewed antipsychotic medications.

The sponsor also provides a calculation of crude mortality for the comparator groups through the safety update. The cumulative crude mortality rate for the placebo exposed group is 9.6/100 patient years exposure (5 deaths in 52 patient years). The crude mortality rate observed in patients exposed to haloperidol was 2.3 per 100 patient years (3 deaths in 131 patient years) and for risperidone 0.95 per 100 patient years (1 death in 105 patient years).

Cause Specific Mortality

Cardiovascular related deaths were more commonly observed in ziprasidone treated individuals compared to patients receiving placebo or active comparator medications. Ten of the 17 deaths that occurred within 30 days of the last dose of ziprasidone (including the safety update) are possibly cardiovascular related. These deaths included diagnoses such as cardiac arrest, ruptured abdominal aortic aneurysm and several unwitnessed, and sudden deaths without clear diagnoses that are suspicious for cardiovascular events. The other causes of death reported in ziprasidone trials were suicide (3), drowning (2), aspiration (1), and accident(1). The deaths in the placebo patients were from pneumonia (3), extradural hematoma, and suicide. In the haloperidol group there were 2 suicides and a post operative MI leading to death. The death in the risperidone group was due to aspiration of food.

Ziprasidone and Sudden Deaths

In the sponsor's review, a death was "sudden" if it occurred within 24 hours of the onset of symptoms directly associated with the death. Including safety update data, the sponsor classified 7 deaths as sudden, giving a sudden death rate of 0.9 per 100 PY exposure (7 sudden deaths in 772 patient years exposure). In my opinion, 6 deaths (35%) had an external cause (definite + probable) and 11 deaths (65%) were sudden. One of the sudden deaths included with the safety update was explained at autopsy (ruptured abdominal aortic aneurysm).

Considering only the NDA death information (not including the safety update), I felt that 6 deaths (43%) had an external cause (definite + probable), and 8 deaths (57%) were sudden (see appendix). The death rate due to external cause was 0.9 per 100PY (6/626) and the sudden death rate was 1.3/100PY (8/626).

A combined sudden death rate was calculated from the NDA data for several recently approved antipsychotic medications. For these 3 drugs, there were 6 sudden deaths in 2496 patient years exposure giving a sudden death rate of 2 per 1000PY. The population in the phase II/III ziprasidone trials was similar to the populations studied in other recently reviewed antipsychotic medications with respect to mean age (39.6), age range (7-82), percent males (72%) and percent caucasians (76%).

Discussion

Because the placebo and active comparator experience within the NDA was limited, the death rate information for ziprasidone was compared with the rates observed for other recently reviewed antipsychotic medications. The all cause mortality rate for ziprasidone was higher than the rate observed with the other medications. After applying the

classification scheme described above, the causes of death appeared to differ from the causes observed with the other medications. Sudden deaths were 6 times more common for the ziprasidone group compared to the rates for the other medications. The rate of deaths with external, explained causes in the ziprasidone development program was similar to the rates estimated for other recently reviewed drugs.

The classification of sudden deaths in these analyses is a function of the quality of available data and reviewer opinion. The quality of information is dependent, in large part, on the data collection methods employed by the sponsor and the investigators. One must consider that variability in the quality of data across NDA's could lead to a classification bias resulting in the observed differences in mortality rates.

Development programs are usually separated by place and time and differ with respect to populations studied, inclusion/exclusion criteria, and concomitant medications allowed. Any or all of these factors could have an influence on the observed sudden death rate. Therefore, comparisons across NDA's should be interpreted cautiously and in the context of all available safety data.

Conclusions

- 1. The data quality and approach to coding were adequate for the purpose of this review.
- 2. The all cause mortality rate was higher for ziprasidone compared to rates for other recently reviewed antipsychotic medications.
- 3. The sudden death rate for ziprasidone is 6 times higher than that for combined data from recently reviewed NDA's.

Because of potential differences in available data and in the drug development programs, these mortality rate comparisons are not conclusive evidence for determining if there is increased risk associated with exposure to ziprasidone. The signal of increased sudden deaths may become an important complementary piece of evidence in the presence of other safety data. One concern that has potential relevance to this finding is an apparent dose dependent QTc prolonging effect associated with ziprasidone in the STFDPC study population.

Gerard Boehm M.D., M.P.H.



cc:HFD-120\Boehm\Burkhart\Laughren\Leber \Glass

Appendix

Summaries for the 17 deaths that occurred within 30 days of last exposure to drug from the ziprasidone NDA and the safety update.

Category 1 (Definite external cause) 128-302E-159-0029 Suicide, falling under a train. 128-116B- 694-0004 Suicide, hanging. 128-108-609-0381 Suicide, gunshot.

Category 2 (Probable external cause)

128-117-706-0529 This 40 YO male with a past medical history significant for bronchitis, took ziprasidone for 54 days and the last recorded dose was 160mg per day. Four days after self-discontinuing ziprasidone, he underwent a sleep deprived EEG to evaluate complaints of amnesia that began following an assault. He left the testing site after 30 hours of sleep deprivation and drove himself home. Two hours after leaving the site, he drove his car off a cliff and died. The autopsy listed asphyxiation due to salt water drowning as the cause of death. Concomitant medications at the time of death included propranolol, antacids, and clonazepam. The investigator felt that this was an accidental death caused by sleep deprivation. There sponsor noted that the patient had plans for the future and there was no suicide note.

128-117-687-0317 This 51 YO male took ziprasidone for 205 days at a dose of 120mg daily. He died from injuries sustained from a fall from his 10th floor balcony. The patient was taking no other medications at the time of death. His last study visit was 9 days prior to his death and he appeared to be doing well and had plans for the future. The sponsor notes that on the day of his death he went to a church crisis center and appeared intense. The sponsor states that the coroner initially felt this was a suicide but was reconsidering that initial conclusion. The investigator felt that this was an accident.

128-303-260-0156 This 46 YO male took ziprasidone for 7 days and the last recorded dose was 120mg daily. His body was found in a river 5 days after leaving a hospital where he was an inpatient. He was taking no other medications at the time of his death. The autopsy listed drowning as the cause of death. He had no prior history of suicidal ideation or suicide attempts. The investigator felt that this was possibly a suicide.

Category 3 (Underlying disease)

Category 4 (Sudden deaths)

128-308-035-0003 *This 63 YO male was treated with ziprasidone for 75 days and the last recorded dose was 80mg daily. While at a lunch club, he collapsed and died suddenly. He was taking no other medications at the time of his death. The coroners report listed ruptured abdominal aortic aneurysm as the cause of death.

128-115-694-0394 This 43 YO male took ziprasidone for 16 days and the last recorded dose was 40mg daily. He withdrew consent and left the study. Subsequently, he had 2 psychiatric hospitalizations for severe generalized anxiety and tardive dyskinesia. He also had been seen in an ER for breathing difficulties and was discharged without prescribed treatment. Thirty days after stopping ziprasidone, he awoke with difficulty breathing and while getting ready to go to the hospital, he vomited. He was found unconscious by his sister, and did not respond to resuscitative measures. The coroner listed asphyxia due to aspiration of vomit as the cause of death. He was taking risperidone, lorazepam, and clonazepam at the time of death.

128-116B-659-0001 This 44 YO female took ziprasidone for 40 days and the last recorded dose was 80mg daily. The study drug was discontinued due to lack of efficacy. While still taking drug, she experienced chest pain. Cardiac work-up (including stress echo) was normal. A GI work-up revealed mildly elevated LFT's and non specific gastritis. Thirteen days after stopping ziprasidone she was admitted to a hospital for hematemesis and endoscopy revealed gastritis with H. pylori. She was treated with clarithromycin, omeprazole, and cisapride. Five days later she was treated in an ER for panic attack. Three days after the ER visit (24 days after stopping ziprasidone) she was found dead in bed. Medications at the time of death included buspirone, sucralfate, acetaminophen, lorazepam, omeprazole, clarithromycin, and cisapride. An autopsy report was not available, but the subject's attending physician learned of a possible atrial myxoma from the ME's office.

128-304E-0193-0379 This 52 YO male took ziprasidone for 221 days and the last recorded dose was 80mg daily. The patient was found dead in bed. At the time of death, the patient was taking acetaminophen as needed for headaches. The physician who attempted resuscitation felt that death was due to cardiac arrest with possible myocardial --infarction. Multiple ECG's recorded for the study did not show ischemic changes. The patient had 3 cardiovascular disease risk factors (male, tobacco use and sedentary lifestyle). There was no autopsy.

128-301-311-0977 This 28 YO female received ziprasidone for 62 days and the last recorded dose was 120mg daily. The drug was stopped for insufficient response. On the day that ziprasidone was stopped, the patient had an end study ECG suggesting subendocardial ischemia. Because of this finding, she was transferred to an internal medicine service in another hospital. She complained of a substernal pinching sensation. The physician did not think this symptom was due to a cardiac disorder. The admitting diagnoses were schizophrenic psychosis, somatic asthenia, chronic disturbance of food intake, and arrhythmia without clear cause. The patient was started on thioridazine, nitrazepam, and a liquid diet supplement. The next day she was noted to have orthostatic symptoms and an ECG reportedly showed sinus arrhythmia at a rate of 58 beats per minute. She died the next day (2 days after stopping ziprasidone). Concomitant medications were biperiden, temazepam, thioridazine, fresubin liquide, nitrazepam, duovit, and vitamin B complex. At the time of death, she was described as cachetic and malnourished (height 5'6" and weight 42kg). An autopsy revealed atrophic and

Med - Page 163 of 223

dystrophic heart muscle fibers which could be associated with toxins, infection, or metabolic abnormalities. A consultant pathologist was unable to determine the cause of death and did not find any changes in the myocardium suggesting a specific disease or drug reaction. A consultant cardiologist reviewed the ECG tracings and felt the bradycardia and T-wave abnormalities were consistent with abnormalities seen with malnutrition.

128-108-607-0305 This 46 YO male took ziprasidone for 61 days at a dose of 80mg daily. He had a past medical history significant for asthma, COPD, PUD, gastroenteritis, microhematuria, and gallstones. During the study, he had a hospitalization for pneumonia and gastroenteritis. Approximately 1 month later, he was found dead in a chair on his front porch presumably after mowing his lawn on a hot day. Concomitant medications at the time of death were beclamethasone, metaproterenol, and Maalox. Autopsy revealed acute and chronic asthmatic bronchitis and granulomatous myocarditis. The liver and brain each had a granulomma-like lesion. The conclusion of the autopsy was that the patient died from asthmatic bronchitis possibly exacerbated by lawn mowing. An independent pathology consultant interpreted the histologic slides as focal myocarditis of unknown etiology. The sponsor admitted that the exact cause of death in this case was uncertain and offered several possible explanations(asthmatic bronchitis, possible heat related illness, myocardial inflammation).

128-116B-508-0001 This 54 YO male took ziprasidone for 72 days, and his last recorded dose was 120mg daily. His past medical history was significant for hypertension, COPD, GERD, constipation, and peripheral vascular disease. He was found dead in his hospital bed. Concomitant medication at the time of death included Procardia XL, ipratropium, triamcinolone, albuterol, docusate lorazepam, aspirin, disulfiram, propranolol, and ranitidine. The sponsor noted that the patient had an episode of chest pain approximately 1 month prior to death. ECG done at that time was reportedly normal. An autopsy revealed severe generalized atherosclerosis, cerebral artery disease, visceral congestion, COPD, cardiac hypertrophy, and vocal cord congestion. The investigator felt that the death was due to atherosclerosis.

128-108-592-0750 *This 39 YO female received ziprasidone for a total of 7 days and the last recorded dose of the drug was 120mg daily. She had a past medical history significant for diabetes mellitus type II, tobacco abuse and alcohol abuse. She was found dead in her apartment. The diagnosis, according to the ME, was chronic alcoholism with hepatic steatosis, supported by a clinical history of alcohol abuse. The sponsor noted that decomposition of the body made a final post-mortem diagnosis difficult. The investigator attributed the death to alcohol abuse and DKA (no supporting blood glucose or pH data provided).

128-302E-319-0375* This 48 YO male took ziprasidone for 162 days and the last recorded dose was 120mg. He had a past medical history significant for hypertension. The investigator discontinued ziprasidone because of increasing psychopathology and started the patient on haloperidol. The following day he was found dead in his apartment.

He had a prior history of water intoxication and associated hyponatremia. The sponsor stated that the autopsy report was not available. A preliminary opinion from the ME attributed the death to cardiac arrest secondary to sodium depletion associated with water intoxication. Supporting data (serum sodium concentration) was not provided.

128-303-197-0299 This 79 YO female took ziprasidone for 30 days and the last recorded dose was 80mg daily. She had a past medical history significant for ischemic heart disease and atrial fibrillation. Thirty days after discontinuing ziprasidone, she experienced a presumed cardiac arrest. An autopsy was not performed.

128-105-534-0021 This 70 YO female took ziprasidone for 5 days and the last recorded dose was 2mg daily. She was treated with ziprasidone for behavioral disturbances associated with dementia. She had a past medical history significant for COPD, right bundle branch block, tobacco abuse, hypothyroidism, and hip fracture. On the 5th day of treatment, she developed shallow respirations and diaphoresis. She was taken to the hospital, arrested and died. The death certificate listed acute cardiopulmonary arrest due to arteriosclerotic cardiovascular disease as the cause of death. Concomitant medications at the time of death included atenolol, ranitidine, levothyroxine, potassium chloride, theophylline, heparin, albuterol, ipratropium bromide, diltiazem, furosemide, docusate, mycolog, and glypizide.

*Indicates death was identified in the safety update.

APPEARS THIS WAY ON ORIGINAL

APPEARS THIS WAY ON ORIGINAL

(54012)

Review of Clinical Data

 Preliminary Review of Study 054

 IND:
 []]

 Sponsor:
 Pfizer

 Drug:
 Ziprasidone

 Route of Administration:
 PO

 Reviewer:
 Greg Burkhart, M.D., M.S.

 Review Completion Date:
 January 11, 2000

Study 054 was an open-label study comparing QT interval duration between patients randomly assigned to ziprasidone, risperidone, quetiapine, olanzapine, thioridazine, or haloperidol. Pfizer initiated the study to address concerns raised in the June 17, 1998 not-approvable letter about ziprasidone's capacity to increase cardiac repolarization.

We received a summary of the findings from study 054 on December 6, 1999 and the full study report on January 3, 2000. My review is preliminary because the study report was incomplete. It relied upon the square root method of correcting the QT interval for heart rate. Dr. Laughren, prior to completion of study 054, had informed Pfizer that the square method causes a bias particularly for drugs that increase the heart rate. This bias was discovered after study 054 was in progress but is a significant concern since many antipsychotics increase the heart rate. While Pfizer provided summary data for analyses that used other more appropriate methods of correcting the QT in a separate submission, they will not provide a full presentation and discussion of these findings until they submit their response to the not-approvable letter.

Study 054 began with a 7-day out-patient period during which pre-existing antipsychotic medication was tapered. Patients then entered the treatment facility for a 7-day washout/ baseline period. Following baseline data collection, patients were titrated to the maximum marketed dose for their assigned drug (80 mg BID for ziprasidone) and observed at steady state. (Drop-outs were replaced.) Because of different kinetic properties, the duration of the dose escalation period varied by drug group with ziprasidone and thioridazine having the shortest (10 days) and risperidone having the longest (18 days). Once patients had achieved steady state and ECGs were collected, a protocol-specified metabolic inhibitor was added to each drug group and observation continued. After ECGs were collected in the presence of the metabolic inhibitor, treatment was tapered and terminated for all patients.

CLI INDL NDA 20825 HED. 120 Burkhart, Katz, Laughern, Glass, Hurdeman

Med - Page 166 pf 223

Daily ECGs were taken for the last 3 days of baseline. A single ECG was taken at day 2 of the dose escalation period and daily ECGs were taken for 3 days at steady state. Pfizer timed the 4 ECGs taken during exposure to the estimated c max for each drug. Daily ECGs were also collected for 3 days in the presence of the metabolic inhibitor. A blinded reviewer read all ECGs to manually measure the QT interval. The analysis focused on the change from baseline that occurred at steady state and any change occurring in the presence of the metabolic inhibitor.

I have attached some of the tables and figures from the study report and those provided in a separate fax. Figure 2.1 shows that at steady state (period 3) all drugs but haloperidol caused an increase in the heart rate. This certainly raises the possibility that the square root correction method would not standardize the QT, and could more importantly, cause a biased comparison with haloperidol. (I am showing the data for completers, but the findings are the same when using all randomized patients.)

As a reminder, the haloperidol group was added to the study at the insistence of the FDA because we were aware of a large amount of data across several NDAs showing that oral haloperidol had no effect on the QT when compared to placebo. (One study was a 7-arm study that evaluated three doses of haloperidol.) Thus, it was our view at the start of study 54 that haloperidol would serve as the control arm providing a formal basis for comparison. We also recommended that thioridazine be added as a treatment group since there was some data suggesting that it prolonged the QT and there were also cases of TDP reported in the post-marketing literature.

Figure 3.1 shows the change from baseline in the uncorrected QT. For drugs that cause an increase in heart rate, one would expect the QT to decrease, as occurred with risperidone, olanzapine, and quetiapine. However, for thioridazine and ziprasidone, both of which caused an increase in heart rate, the QT increased. (Haloperidol also had an increase in QT, but the heart rate decreased from baseline in this group so that the increase in QT would be expected.)

Figure 1.1.1 shows the square root corrected OTCs by treatment group whereas tables 6b and 6c show OTCs using other methods of correction that do a better job of standardizing the QT for heart rate. Focusing on tables 6b and 6c, it seems fairly clear that there are only two between group differences in study 054 when using haloperidol as the basis for comparison. Both ziprasidone and thioridazine clearly caused a prolongation in the QT with the size of the effect much more impressive for thioridazine. The effect for ziprasidone appears to be about 10 msec, which was similar to that observed in the NDA. (Pfizer did not provide any statistical testing of these findings, but by examining the 95% CIs, one can reasonably conclude that there is significant statistical evidence of a difference when comparing either ziprasidone or thioridazine with haloperidol; the evidence for thioridazine being much stronger since there is no overlap in CIs.)

After seeing the findings from study 054, I believe there are three significant conclusions. First, ziprasidone affects cardiac repolarization within its proposed dose range whereas risperidone, quetiapine, and olanzapine do not have such an effect, at least within their marketed dose range. Second, the size the effect attributable to ziprasidone is about 10 msec, which is similar to that observed in the ziprasidone NDA. Finally, thioridazine clearly has a large effect upon cardiac repolarization that is almost certainly clinically significant and life threatening. Several cases of TDP have been documented with thioridazine use in the literature and significantly lower doses of thioridazine than that studied in 054 have also been shown to prolong the QT.

I should also point out that patients in all dose groups seemed to tolerate the study fairly well. There were no deaths or serious events on drug, and no reported cases of TDP or syncope. One patient on thioridazine had an adverse event reported as "QT prolonged". The groups were also fairly comparable in the rate of discontinuation. For ziprasidone, 35 patients were randomized with 31 completing the study.

Pfizer appears to have concluded that study 054 shows that all drugs in the study prolonged the QTC. This conclusion is apparently based upon the fact that there was a positive change from baseline in QTC observed in each drug group. However, in my view, the empirical evidence does not support this conclusion. An additional control group such as placebo or even a lower dose of haloperidol that experienced significantly less change in QTC during study would be necessary to justify such a conclusion.

The sponsor also argues that the size of the effect attributable to ziprasidone is not clinically significant because of the absence of any clinical evidence of a risk and because of the terfenadine experience. In the ziprasidone NDA, there was no compelling evidence that sudden death was increased above background, no patients had QT prolongation at levels considered clinically significant and there were no cases of TDP. It is true that the experience with ziprasidone is in direct contrast to that with sertindole. With sertindole the effect on the QT was larger, patients had clinically significant prolongation, and the rate of sudden death appeared to be increased in both the NDA (compared to other NDAs) and in the European post-marketing experience. Pfizer also points out that terfenadine at its c max had a similar effect size and that no cases of TDP or QT prolongation have ever been found with terfenadine in the absence of a metabolic inhibitor.

I think there are at least 2 problems with this line of reasoning. First, while the facts with terfenadine are generally correct, patient exposure to parent terfenadine is fleeting - in the order of a few minutes. So patients do not remain at risk, if there was any, for very long until terfenadine is co-administered with a metabolic inhibitor. Second, the capacity of any development program to detect an increased rate of sudden death is limited. It would have to be a large increase compared to the rates observed in other NDAs for one to reasonably conclude there was a signal of concern. Likewise, the timing of ECGs to c max may have not been sufficient to detect prolongation. TDP can be difficult to capture clinically unless investigators are looking for such events. Thus, the absence of any clinical signal in the NDA development program is not a compelling argument that they won't necessarily occur.

In short, while I agree that the clinical significance of a relatively small effect like 10 msec is not established, I don't agree that the finding is dismissible given any historical experience. I also think an even more important finding from study 054 than the experience with ziprasidone is that we can now conclude with much more certainty than in the past that risperidone, quetiapine, and olanzapine probably do not prolong the QT in their marketed dose range. This relative difference in effect between ziprasidone compared to some of the more recently approved antipsychotics raises significant need to inform investigators and patients under the ziprasidone IND about the findings from study 054.

My recommendations are (1) to include the findings from study 054 in the investigator brochure and to inform patients of these findings, in effect, conducting informed consent again for all patients exposed under the IND; (2) to ask the sponsor to try and get ECGs in patients that remain under the IND at c max and to consider discontinuing any patient with a prolonged QT; (3) to restrict any additional studies conducted under the IND to defining the level of risk from the QT effect or establishing a comparative efficacy benefit between ziprasidone and other antipsychotics, and (4) to consider withdrawing thioridazine from the market. At a minimum, I would recommend a boxed warning and reserving thioridazine for second line use.

APPEARS THIS WAY ON ORIGINAL

APPEARS THIS WAY ON ORIGINAL

128-054

STUDY REPORT

Page 15

A diagram showing the general study design and the timing of critical measurements is given below.



During Period 1, each subject's antipsychotic medication was tapered to the lowest possible dose over about 7 days. The investigator was to contact subjects on alternating days during this one-week period.

One day prior to the start of Period 2 (day -5), subjects entered the clinical research facility and had liver function tests performed. Results of these tests had to be reviewed before subjects were randomized to treatment assignment via tele-randomization. During Period 2, subjects received placebo once daily at approximately 0800 hours for 5 days (days -4 to 0). To standardize circadian and meal effects on QTc, the exact time of morning dosing established on day -4 was to remain fixed for the remainder of the study.

On the last 3 days of Period 2 (days -2, -1, 0), baseline ECG measurements were obtained three times daily at times specified according to treatment assignment. The times were selected so that QTc would be assessed at timepoints surrounding the mean Tmax of each agent, controlling for post-prandial time. The schedule of ECG measurements, relative to the morning dose of placebo (Period 2), antipsychotic (Period 3), and antipsychotic plus metabolic inhibitor (Period 4) is listed in Section 5.4.

During Period 3, subjects received the assigned antipsychotic drug while under continuous medical supervision in the clinical research facility. Study drug was administered in open-label fashion. The duration and dosing schedule in Period 3 were unique for each agent due to differences in tolerability, pharmacokinetics, and the time required to reach steady-state exposure. The time to achieve steady-state conditions was estimated from the average t½ of each study drug or major metabolite. For the dosing regimen used, it was anticipated that the ziprasidone, risperidone, olanzapine, quetiapine, thioridazine, and haloperidol groups would reach steady-state conditions by days 8, 16, 11, 10, 8, and 10, respectively. For risperidone, olanzapine, and quetiapine, the dose escalation schedule and the Tmax were based on information provided in product labeling; for thioridazine¹² and haloperidol³, the estimate of Tmax was based on literature reports. The time to achieve steady-state conditions and the Tmax for ziprasidone were based on data from previous pharmacokinetic studies that used doses from 40 to 80 mg BID.

The duration of treatment and the maximal daily dose differed across study drugs as shown below:

Med - Page 170 of 223

Table 5.1

Summary of Mean (CV%) Concentrations (ng/ml) for Parent Drug Obtained at the Time of Expected Peak Systemic Exposure Protocol 128-054

10

Med - Page 171 of 223

Drug Group	Period 3 Day 2	Period 3 Low Dose*	Period 3 Steady-State	Period 4 With Inhibitor	Ratio Period 4/3
Ziprasidone	49 (41)	· N/A	171 (34)	224 (35)	1.39 (40)
Risperidone	14.8 (61)	24.8 (67)	58.7 (79)	124.0 (48)	2.47 (35)
Olanzapine	9.2 (54)	N/A	55.1 (39)	84.5 (27)	1.77 (45)
Quetiapine	175 (48)	N/A	1280 (61)	3740 (43)	4.03 (70)
Thioridazine	215 (43)	N/A	765 (46)	799 (50)	1.04 (20)
Haloperidol	2.1 (91)	N/A	16.1 (95)	27.1 (75)	1.94 (50)

Source: Section 13 Tables 1.2.1 - 1.2.6

• = Applies only to the risperidone treatment group; sample obtained on day 5.

N/A = Not applicable.

APPEARS THIS WAY

H:/128/Report/054/PKReport.doc 12/01/99

Figure 2.1

`>

Mean Change from Baseline Heart Rate(bpm) and 95% Confidence Intervals at Each Period by Treatment Group - Completers Ziprasidone Protocol 054



Z=Ziprasidone, R=Risperidone, O=Olanzapine, Q=Quetiapine, T=Thioridazine, H=Haloperidol.

* Bar on left is Risperidone 6-8 mg, bar on right is Risperidone 16 mg.

+ Contains only pre (3/16/99) amendment values, post-amendment values are provided in the listings. Source Data: Table 5.2.2.1.1. Date of Data Extraction: 03JUN99. Date of Figure Generation: 07JUL99.

Med - Page 172 of 223

t

Figure 3.1

7

1000

Souther Street

Mean Change from Baseline QT Interval(msec) and 95% Confidence Intervals at Each Period by Treatment Group - Completers Ziprasidone Protocol 054



Z = Ziprasidone, R = Risperidone, O = Olanzapine, Q = Quetlapine, T = Thioridazine, H = Haloperidol. * Bar on left is Risperidone 6-8 mg, bar on right is Risperidone 16 mg. + Contains only pre (3/16/99) amendment values, post - amendment values are provided in the listings.

Source Data: Table 5.2.3.1.1. Date of Data Extraction: 03JUN99. Date of Figure Generation: 07JUL99.

Med - Page 173 of 223



Mean Change from Baseline QTc Interval(msec) and 95% Confidence Intervals at Each Period by Treatment Group - Completers Ziprasidone Protocol 054



Z = Ziprasidone, R = Risperidone, O = Olanzapine, Q = Quellapine; T = Thioridazine, H = Haloperidol.

* Bar on left is Risperidone 6-8 mg, bar on right is Risperidone 18 mg.

+ Contains only pre (3/16/99) amendment values, post-amendment values are provided in the listings. Source Data: Table 5.2.1.1.1. Date of Data Extraction: 03JUN99. Date of Figure Generation: 07JUL99.

Table 6b. C	QT¢ Change (F	FDA Correction	(0.37 Power)) from	Baseline to Last	Observation; Stud	y 054 Completers
-------------	---------------	----------------	--------------------	------------------	-------------------	------------------

Ketz, Director,

NDA-20-825, ND

.

JUNIT.

]

	Ziprasidone	Risperidone	Olanzapine	Quetiapine	Thioridazine	Haloperidol
Baseline					•	•
Mean (msec)	391.1	388.9	389.0	388.3	389.2	384.8
(95% CI)	(384.4, 397.8)	(382.8, 394.9)	(381.5, 396.4)	(381.3, 395.3)	(383.1, 395.3)	(378.6, 391.0)
Period 3: Steady	-State (except of	days 5-7 for Ris	peridone)			
Mean ∆ (msec)	16.5	4.9	2.3	6.9	30.8	6.8
(95% Cl)	(11.1, 21.8)	(1.2, 8.5)	(-3.1, 7.8)	(2.9, 10.9)	(25.6, 36.1)	(1.4, 12.2)
%Δ	4.3	1.3	0.7	1.8	8.0	1.8
(95% CI)	. (2.9, 5.7)	(0.3, 2.2)	(-0.8, 2.1)	(0.7, 2.8)	(6.6, 9.3)	(0.4, 3.2)
Period 3: Steady	-State for Rispe	eridone				
Mean ∆ (msec)		4.3				
(95% CI)		(-2.3,10.9)				
%Δ		1.2				
(95% CI)		(-0.6, 2.9)				
Period 4: Inhibito	or Present					
Mean ∆ (msec)	17.0	2.7	3.3	9.5	29.3	12.8
(95% CI)	(11.0, 23.0)	(-4.6, 10.0)	(-1.7, 8.3)	(4.7, 14.3)	(23.2, 3 5.5)	(7.0, 18.6)
% ∆	4.5	0.7	0.9	2.5	7.6	3.4
(95% CI)	(2.9, 6.0)	(-1.2, 2.7)	(-0.4, 2.2)	(1.2, 3.7)	(6, 9.3)	(1.8, 4.9)

Table 6c. QTc Change (Framingham Linear Correction) from Baseline to Last Observation; Study 054 Completers

	Ziprasidone	Risperidone	Olanzapine	Quetiapine	Thioridazine	Haloperidol
Baseline						
Mean (msec)	389.4	388.2	388.4	388.1	388.5	383.7
(95% Čl)	(383.5, 395.3)	(382.6, 393.9)	(381.3, 395.6)	(381.4, 394.9)	(382.6, 394.4)	(378.0, 389.3)
Period 3: Steady	-State (except o	lays 5-7 for Ris	peridone)			
Mean ∆ (msec)	14.9	3.6	1.6	4.4	28.5	6.1
(95% CĪ)	(9.9, 19.8)	(-0.1, 7.3)	(-3.7, 6.8)	(0.8, 8.1)	(23.0, 34.0)	(1.0, 11.2)
%∆ 	3.9	0.9	0.5	1.2	7.4	1.6
(95% CI)	(2.6, 5.2)	(0.0, 1.9)	(-0.9, 1.8)	(0.2, 2.1)	(6.0, 8.8)	(0.3, 3.0)
Period 3: Steady	-State for Rispe	eridone				
Mean (msec)		3.7	•			
(95% CI)		(-2.2, 9.7)				
%Δ		1.0				
(95% CI)		(-0.6,2.6)		•		
Period 4: Inhibito	or Present					
Mean (msec)	15.5	2.5	2.8	5.9	28.6	12.8
(95% CI)	(9.7, 21.4)	(-4.5, 9.5)	(-2.0, 7.6)	(1.6, 10.2)	(22.2, 35.0)	(7.3, 18.2)
% Δ	4.1	0.7	0.8	1.6	7.4	3.4
(95% CI)	(2.6, 5.6)	(-1.2, 2.5)	(-0.4, 2)	(0.4, 2.7)	(5.7, 9.1)	(1.9, 4.8)
Source: Data on	File					

December 24, 1999

Katz, Director NDA-20-825, NDA

JINDE

7



Figure 1b. Individual Mean QTc Changes (FDA-proposed Correction) from Baseline vs. Ziprasidone Serum Concentration; Study 054



Figure 1c. Individual Mean QTc Changes (Framingham Correction) from Baseline vs. Ziprasidone Serum Concentration; Study 054

Med - Page 176 of 223

able 5.2.3.1.1 ummary of QT Inte iprasidone Protoc	erval (msec) 8 ol 054	aseli	ne and Change fro	om Baseline by Tr	eatment Group - C	omplețers	' Page	1 of 1
		••••			Treatmen	. Group		· · · · · · · · · · · · · · · · · · ·
Period			Ziprasidone	Risperidone+	Olanzapine	Quetlapine	Thioridazine	Haloperido
Baseline*	Std. D		31 362.2 21 7	25 368.9 26.6	24 365.0 23 4	27 362.4 23.2	30 371.3 1A.2	358 15
	Ka	nge ČI	(354.3, 370.2)	(358, 379.9)	(355.1, 3/4.9)	(353.2. 3/1.0)	(304.5, 3/0.1)	(332.0, 304.)
Day 2	Std. D	N ean ev. nge	31 -4.5 17.8	25 -5.9 10.4	24 -3.8 13.2	27 -0.3 12.6	30 0.4 16 4	0.
		ĩŠĩ	(-11.1, 2)	(-10.2, -1.7)	(-3.3, 1.0)	(-3.3, 4./)	(-5./, 0.5)	(-4.9. 6.6
Per1od 3 ₽	Std. De	N ean ev.	31 6.8	25 -12.1 12.9	24 - 8.9 18.7	27 -12.2 15.1	30 18.7 22.1	2 12. 16.
	nai	nge ČI	(-0.1, 13.6)	(-17.46.7)	(-10.8, -1)	(-10.2, 0.2)	(10.5, 2/)	(2.3, 13.1
Period 3^	Mi Std. Di Rai			25 • 8.0 • 9		I		
	ngr	ČĨ		(-16.1. U.Z)	1			
Period 4	Std. De		31 10.0	20 1.1	24 - 1 - 8 1 7 - 8	27 -15.8 16 9	30 33.3 23.1	22 22 19
	Rar	ČI	(2.1, 17.8)	(-7.6; 9.8)	(•9.3, 5.7)	(-22.5, -9.1)	(24./. 41.9)	. (13.1, 31.8

٠.

3.1

;

Baseline is defined as the average of the planned ECGs collected on days -2. -1, and 0.
 # Risperidone 6-8 mg
 * Risperidone 16 mg
 + Period 4 contains only pre (3/16/99) amendment values, post-amendment values are provided in the listings.
 Source Data: Section 13 Table 18.1. Date of Data Extraction: 03JUN99. Date of Table Generation: 08JUL99.

Med - Page 177 of 223

. .

t

BEST POSSIBLE COPY

Period		Ziprasidone	•••••••••••••	Treatmer	nt Group		••••••••••••
Baseline*	· · · · · · · · · · · · · · · · · · ·	31	Risperidone+	Olanzapine	Quetlapine	Thioridazine	
	Mean Std. Dev. Range	75.1	25 70.5	24 172,2	27	30	Haloperidol+
Day 2	ČI	(70.7, 79.6)	(05,9, 75)	8.8 100.0, 10.91	73.0 7.7	68.9 8.2	27 73.5 10.6
	N Mean Std. Nev	31 3.6 7.2	25	24	(/U, /U)	103.0. 11.31	(09.4, //./)
	CI CI		3.8 5.6	1.2 7 n	27 0.4 5.2	30 4.3	27
Period 30	N	(U.8, 6.2) 31	(1.1. 0.5)	(-1./, 4.2)	(-1.0, 2.5)	7 ă (1.5, /)	0.3
	Hean Std. Dev. Range	4.6 8.2	25 9.5 6.0	24 6.5	27 11.2	30	(*2.5, 3)
erlod 3n	ČI	(1.6, /./)	(/, 11.9)	(J. 1U)	A D	5.7	27 -2.9 A 3
	N Mean Std. Dev. Range		25 6.4 6.9	(3. 10)	(8, 14,5)	(2.8, 8.0)	(-0.1. 0.3)
entod 4	CI	•	(3.6, 9.3)				
	Mean Std. Dev.	31 3. g	20 9.5	:"4 3.0		30	
seline is defined	Range Ci	(0.7. 6.5)	(2.4, 3.4)	¢ ·•	1515 (11.8, 18.3)		- 20 5 7 9 9
peridone 6-8 mg) as the average of	the planned ECG	s collected on day	's ·2, ·1 and 0	•••••••••••••••	(-5.1, U.y)	(-10, -1.4)
e Data: Section	nly pre (3/16/99) 13 Table 18.1. Da	amendment values te of Data Extrac	. post-amendment v ction: 03JUN99. D	alues are provid ate of Table Gen	led in the listings Peration: OBJUL99.	•	
					'		

Review of Clinical Data

JAN 23 1998

Review of Ziprasidone ECG Data

NDA:	NDA 20-825	
Sponsor:	Pfizer	
Drug: •	Ziprasidone	
Route of Administration:	Orai	
Reviewer:	Gerard Boehm, M.D., M.P.H.	
Review Completion Date:	1/23/98	

In this document, I review the sponsor's ECG data presentation included in the NDA ISS. I follow by summarizing the reviews provided by Drs. Ganley and I then provide the results of additional analyses that I conducted using the sponsor's data from the STFDPC trials, and from study 101. I conclude with a discussion of the evidence.

NDA ISS data

Description of the collection of ECG data

There were no studies specifically designed to evaluate the effect of ziprasidone on the QTc. For the analyses presented in the NDA, the sponsor examined data recorded by 12lead ECG's and did not identify any studies that used Holter monitors. In selected studies, investigators recorded screening, baseline, on treatment, and end of study ECG's. Investigators obtained screening ECG's during recruitment and used them to identify volunteers with ECG abnormalities for the purpose of excluding them from the study. The sponsor defined the baseline ECG as the last tracing recorded before the first day of study treatment. Depending on the protocol, these tracings could have followed a washout period where investigators discontinued the medications that subjects were taking at the time of enrollment. In some cases, the baseline ECG's were the screening ECG's (ex studies 104,106). End of study or final ECG's were the last ECG's recorded while on study treatment or within six days after the last day of study treatment. Investigators may have recorded additional ECG's during the trials (depending on the individual study protocols). With few exceptions, protocols did not specify the timing of the ECG with respect to dosing. ECG machines measured the intervals for the tracings analyzed in the ISS. The one exception was study 303, where machines read the tracings on site but the investigators also forwarded ECG's to a central site for blinded reading using an accurate, digitized methodology. The sponsor entered the ECG data into a database and calculated mean values for various parameters. The sponsor compared the mean baseline QTc to the

mean final QTc to look for evidence of prolongation. Tables in the ISS presented the data from the oral dose Phase II/III studies, the short term placebo controlled fixed dose studies (104,106,114,115), and study 303. In addition to looking for mean changes in the study population, the sponsor examined individual tracings to identify outliers (criteria: QTc \geq 500 or an increase in QTc \geq 75).

NDA Presentation of ECG Data

The NDA included the initial review of the ECG data as well as the review by DrL the sponsor's cardiology consultant.

In all oral Phase II/III trials, the final mean QTc increased from baseline for ziprasidone treated subjects (3.8) and decreased for those treated with placebo (-2.5). Similarly, in the subset of short term fixed dose placebo controlled trials, the sponsor described an increase in the final mean QTc compared to baseline in ziprasidone treated individuals (6.6), and a decrease in placebo exposed subjects (-2.6). The apparent dose response relationship presented in the following table strengthened the argument for ziprasidone's effect for prolonging the QTc.

Treatment Group	N	Acebo Controlled Trials Mean Change QTc from baseline
Placebo	251	-2.6
Zip<40 BID	232	4.0
Zip 40 BID	137	4.5
Zip 60 BID	111	7.3
Zip 80 BID	100	10.5
Zip≥100BID	77	12.1
Haloperidol	76	0.2

Change from Baseline to Last Observation in QTc Short Term Fixed Dose Placebo Controlled Trials

From Sponsor's table H.5.23c p.1070 Vol. 1.1

The sponsor's analysis of mean QTc difference from baseline in Study 303, a 52 week inpatient study, did not demonstrate a dose response relationship (see appendix).

The sponsor identified few ziprasidone treated patients (n=13) who met outlier criteria. The percentage of ziprasidone exposed subjects who met outlier criteria was similar to the percentage of subjects given placebo or an active comparator who met the criteria. None of the outlier ziprasidone treated patients had adverse events associated with prolonged QTc (i.e. syncope, documented arrhythmia, sudden death) and none had a prolonged QTc on more than one occasion.

The sponsor's consultant, Dr [found that the ECG interval measurements were incorrect in several instances. He re-measured the intervals from some of the tracings from the STFDPC trials. Following re-measurement, he did not find a dose response relationship for QTc prolongation, but he did describe a statistically significant increase 7
in mean final QTc compared to baseline. Dr[QTc was small and not clinically significant. felt that ziprasidone's effect on the

An FDA cardiologist, Dr. Ganley, independently reviewed the sponsor's NDA ECG presentation as well as the ECG SAS data sets (see NDA consult dated November 18,1997). Without interval re-measurement, he found dose response relationships when analyzing all phase II/III studies and the subset of short term fixed dose studies. Dr. Ganley did not find a dose response relationship for QTc prolongation in study 303 (although he noted an increase in mean QT for the 80mg bid group accompanied by a decrease in mean heart rate which was not observed in the STFDPC trials). Dr. Ganley agreed with Dr that the intervals were misread for many of the tracings. Despite the inconsistencies, he felt that the evidence suggested a dose response relationship for QTc prolongation and compared the change observed with ziprasidone to that seen with therapeutic doses of terfenadine. He could not estimate a risk of torsades from the data and recommended, in lieu of further clinical studies, in vitro studies of the effect of ziprasidone and its metabolites on action potential duration.

The sponsor hired Dr.(to review ECG data from ziprasidone studies. He reassessed the previously identified outlier ECG tracings. Using his readings, none of these subjects had a QTc \geq 500 and only one met the criteria for a clinically significant change from baseline (QTc increased \geq 75 over baseline). Re-measurement of intervals using the accurate, digitized methodology supported Dr.(assessment.

Dr. Analyzed data from study 301, and pooled data from studies 117, 108, 108E, (ongoing studies) and 303. These studies were selected because the ECG intervals had been measured using an accurate digitized methodology. He found no evidence of a dose response relationship for QTc prolongation in these data.

The sponsor then had the ECG's from the STFDPC trials re-measured using an accurate digitized methodology. After re-measurement, Dr.() noted that the mean QTc difference from baseline increased with increasing ziprasidone dose for all dose groups except the ≥ 100 mg bid group.

Short Term Fixed	Dose Pl	laceoo Controlled Trials
Treatment Group	N	Mean Change QTc from baseline
Placebo	250	-2.6
Zip<40 BID	230	0.6
Zip 40 BID	138	5.9
Zip 60 BID	111	7.7
Zip 80 BID	100	9.7
Zip≥100 BID	77	6.4
Haloperidol	76	-1.6
From Dr.	table	H.5.23.2 Appendix IV

Change from baseline to Last Observation In QTc Short Term Fixed Dose Placebo Controlled Trials Dr I. provided an additional analysis of STFDPC trial data. He compared the ECG tracing with the largest QTc change at any time during the study with the baseline QTc. Aside from the greater magnitude of change, the results were similar to above (increasing mean QTc difference with increasing dose for all groups except the ≥ 100 mg bid group).

When looking at the evidence in aggregate, Dr 7 felt the data should be interpreted as showing no ziprasidone related dose changes. He felt that the lack of outliers argued against ziprasidone related prolongation. An additional analysis using the mean last QTc minus the mean screening QTc did not reveal a dose response relationship (contrary to what he found using the baseline QTc). He felt that if a true effect was present it should be seen regardless of the QTc used for comparison (baseline or screening). Therefore, he felt that natural variability, rather than drug effect, explained the observed changes. Additionally, the results of the analysis of study 301 did not support a dose response effect. Lastly, Dr presented the results of IM study 046. In this study, ziprasidone was dosed at 5 to 20mg intra-muscularly four times a day. Investigators obtained ECG's following the 4th dose on day 2. The mean QTc changes from baseline were: 5.3 for placebo; 3.5 for 20mg; 11 for 40mg; and 12.5 for 80mg. Prolongation did not appear to correlate with the estimated serum concentrations.

Re-analysis of ECG data from STFDPC Trials

Methods

Using the sponsor's data sets for the re-measured intervals from STFDPC trials, we were able to conduct our own analyses. I began by reviewing information about the individual trials to determine if it was reasonable to pool the results for analysis. Using the data sets provided, I attempted to replicate Drl. γ 's findings. The sponsor included variables that identified the baseline QTc and last QTc for the patients from the STFDPC trials. I was able to reproduce the mean changes from baseline that were included with Dr. γ 's results. Unfortunately, the sponsor did not include a variable that identified the QTc used in calculating the maximal QTc change from baseline. The results I obtained for this analysis are the same as Dr γ 's for all the dose groups except for <40 and 40 which differ only slightly.

During the review of the QTc prolongation issue, we became concerned about the possibility of an effect due to the timing of the last ECG tracing (up to 6 days after the last day on study medication). We hypothesized that measurement of QTc after completing treatment could reflect the return to baseline and not the effect that would be observed if the tracing were recorded while the subject was taking the drug. If this were true, the measured effect would then underestimate the actual effect of the drug. I attempted to control for this potential effect by conducting two additional analyses. After excluding any QTc value from a patient whose last ECG was recorded greater than or equal to one day after discontinuing the medication, I compared the last QTc with the

baseline QTc values. In addition, since all of these trials included a day 14 ECG, I performed an analysis comparing the day 14 QTc and the baseline QTc values.

In reviewing Dr.). Is analyses, the mean OTc change values using the remeasured intervals from the STFDPC trials appeared to depict a dose response trend. We were interested in evaluating the strength of the evidence for a dose response relationship. First, we wanted to determine if the mean QTc change values were significantly different. If they were, we would perform pair wise comparisons of the mean differences. If a dose response relationship was present, and there were adequate numbers of observations, we expected that the mean QTc differences would be significantly higher in the high dose categories compared to the lower dose categories, and placebo. The STFDPC trials data was provided with the following dose groups: placebo, <40mg bid, 40mg bid, 60mg bid, 80mg bid and ≥100mg bid. These categories were used for the independent variables and the mean QTc difference from baseline was the response variable. An analysis of variance was conducted to evaluate the difference between the mean QTc change for all dosage groups. The Tukey-Kramer means comparison test was used for pair wise comparisons of the mean OTc differences for each dosage group. Regression lines were fitted using first dose group, and then estimated dose (using 12.5mg as an estimate for the categorical group <40mg and the dose in mg for the rest of the groups) by mean QTc difference from baseline. A t-test was conducted to evaluate the slope of the resulting line.

Results

The STFDPC analyses use data pooled from 4 studies. One of the potential advantages of pooling data is to increase the precision of an estimate by increasing the number of observations. Unfortunately, pooling does not result in additional observations for the 80mg bid and 100mg bid doses because all the subjects for each of these dose groups came from one study.

The STFDPC trials include four separate studies. Studies 104 and 106 were 4 weeks long and had similar entrance criteria. As mentioned above, these protocols required only a screening ECG prior to beginning the study. They both allowed the use of low dose β blockers and lorazepam during the washout and double blind phases of the studies. Study 104 looked at the use of ziprasidone at 5mg bid, 20mg bid, and 40mg bid doses. Study 106 used 20mg bid and 60mg bid doses.

Studies 114 and 115 were similar to each other but differed slightly from the studies mentioned above. These protocols required both screening and baseline ECG's. These studies were 6 weeks in duration and both allowed the use of lorazepam, but not β -blockers, in the washout period. Both studies allowed the use of lorazepam, benztropine, or β -blockers during the double-blind phases of the studies. Study 114 used 40mg bid and 80mg bid doses while study 115 used 20mg bid, 60mg bid, and 100mg bid doses.

The following table provides the mean QTc change from baseline for the individual studies, using the mean last QTc minus the mean baseline QTc (accurate measurement methodology).

Study	Placebo	<40mg bid	40mg bid	60mg bid	80mg bid	>=100mg bid
104	-3.4	2.1	2.2		[
106	0.1	-3.4		5	· ·	
114	-3.7		7.3	1	9.7	
115	-2.3	. 1.2	[9.2		6.4

Examination of individual studies, last QTc minus baseline QTc

Looking at the individual study results, in all but one dosage group, the drug exposed groups had a greater mean QTc difference than the group exposed to placebo. In 2 of the 4 studies the trend was for increased effect with increased dose. The study designs and results were similar and pooling of data was considered appropriate.

The following table lists the results of the analyses using QTc values obtained at different times to calculate the mean QTc difference from baseline. The Last-baseline analysis provide the findings from Dr. Vs analysis. The last on treatment-baseline represents the analysis which excluded patients whose last ECG was not done on the day that treatment ended. The 14 day-baseline represents the results using the mean day 14 QTc compared to the mean baseline QTc. The QTcmax-baseline represents the results of my analysis which used the maximal mean QTc difference at any time during the study compared to the mean baseline QTc (see appendix for complete tables).

Comparison	Placebo	<40mg bid	40mg bid	60mg bid	80mg bid	100mg bid
Last-baseline	-2.6	0.6	5.9	7.7	9.7	6.4
Last on tx-baseline	-4.2	0.8	4.6	7.9	8.8	8.0
14 day- baseline	-2.6	3.4	6.1 -	8.4	13.3	8.5
QTcmax-baseline	4.3	9.5	12.8	15.2	19.8	15.0

· Ziprasidone Short term fixed dose placebo controlled trials (104,106,114,115)

Comparison of the mean QTc differences

For all of the comparisons, the mean QTc difference from baseline increased with increasing dose for all ziprasidone dose groups except for the 100mg bid group. In each of the above analyses, the results of the one way ANOVA was consistent with different means (p<.0001). Appendix 1 provides the pair wise comparisons for the Tukey-Kramer test. Using the mean last QTc on treatment minus baseline analysis, all dosage groups except the <40mg bid group were significantly different from placebo, but differences between mean changes for the different ziprasidone dosage groups did not achieve statistical significance. For the remaining 3 analyses, the QTc difference from baseline was also significantly higher than placebo for all dose groups except the <40mg. Additionally, the 80mg bid group was significantly higher than the <40mg bid group. The

Mean QTc difference (final on treatment from baseline) Study 101

Haloperidol	Zip 2mg bid	Zip 5mg bid	Zip 20mg bid	Zip 80mg bid
0.1(n=17)	-5.4 (n=17)	0.3 (n=17)	4.9 (n=17)	3.4 (n=19)

I repeated this analysis using the sponsors' data set that was included with the NDA submission and my results were comparable to those above. To assess the potential effect of ECG's obtained after ziprasidone was stopped, I conducted additional analyses using the day 7 (prior to am dose) and day 21 (3-7 hours after the am dose) ECG tracings.

Mean QTc difference (day 7 from baseline, day 21 from baseline) Study 101

Comparison	Haloperidol	Zip 2mg bid	Zip Smg bid	Zip 20mg bid	Zip 80mg bid
Day 7	-6.8 (n=13)	1.4 (n=14)	6.4 (n=14)	-1.3 (n=15)	3.0 (n=17)
Day 21	-11 (n=10)	-7.6 (n=10)	5.9 (n=7)	-5.8 (n=10)	-3.0 (n=13)

I did not find a dose response relationship for QTc prolongation when comparing the mean QTc on day 7, or day 21 to the mean baseline QTc. Separate analyses looking at QT intervals by dose group also show no evidence of prolongation.

Discussion

QTc prolongation has been associated with increased mortality in healthy individuals as well as in patients with ischemic heart disease.^{1,2,3,4,5} Arrhythmias and death have been associated with drugs causing QTc prolongation either alone or in combination with other agents that compete for degradation pathways.^{6,7,8,9} To evaluate the potential of ziprasidone to cause QTc prolongation, the sponsor examined ECG data collected during the development program and described conflicting results. Some evidence suggests that ziprasidone causes dose dependent increases in QTc while other evidence does not support such an effect. The ability of these data to accurately describe the effect of ziprasidone on QTc is limited for several reasons.

Previous research has demonstrated that the QTc is a dynamic parameter, with demonstrated variability within individuals 10,11,12,13 The protocol instructions for collection of ECG data in the ziprasidone NDA trials did not take this variability into consideration. For example, the baseline ECG used in these studies was a single measurement at an unspecified time (defined as the last ECG recorded prior to taking the study medication).

Timing with respect to dosing was not specified in many of the study protocols. As a result, for the on treatment tracings, some of the measurements could reflect trough concentrations, some could reflect peak concentrations, and others could reflect an intermediate concentration. Additionally, there are problems with the timing of what the sponsor defines as the last ECG. The sponsor acknowledged that some of the last ECG's were done as many as 6 days after the last dose of study drug. This disregard for timing could limit the ability to detect a drug effect, if present.

differences between the remaining groups did not achieve statistical significance.¹

These results demonstrate that in the STFDPC trials, the difference in mean QTc from baseline was greater in the groups exposed to ziprasidone than in the groups exposed to placebo. Within the ziprasidone dosage groups, the mean QTc difference generally increases with increasing dose, but the difference achieves statistical significance only for the 80mg bid group compared to the <40mg bid group. Use of different tracings from the database for comparison appeared to have little effect on the detected mean QTc difference. The mean QTc difference was smaller for the 100mg bid group than the 80mg bid group in each of these analyses.¹

Simple regression lines were fitted using mean last QTc and mean day 14 QTc difference from baseline by categorical dose groups and estimated dose(see appendix). The slopes of the regression lines were positive and were significantly different than zero. These findings are consistent with a dose response effect, although the models did not explain the data well with an r^2 equal to .04.¹

Analysis of ECG data from Study 101

Methods

To further explore the possible effect of ziprasidone on the QTc, I analyzed the ECG data collected from study 101. One of the previously identified potential sources of error in the development program was the lack of protocol instructions about the timing of ECG tracings with respect to dosing. I selected Study 101 because the protocol required that ECG's be done at specified times during the trial. The study design was reviewed with emphasis on ECG data collection. I compared the mean QTc from the day 7 and day 21 tracings to the mean baseline QTc to look for prolongation.

Results

This trial was a 4 week double blinded, haloperidol controlled study in acute exacerbation of schizophrenia and schizoaffective disorder. Patients were randomized to one of 5 treatment groups (ziprasidone 2mg bid, 5mg bid, 20mg bid, 80mg bid, or haloperidol 15mg daily). Twelve lead ECG's were recorded for each subject at screening and prior to the morning dose on days 1,3,7,10,and 14. ECG's were also done 3-7 hours after dosing on days 21 and 28. The intervals were reported from the study site and were not reanalyzed using the accurate, digitized methodology mentioned previously. The QTc was derived using the following formula The sponsor reported the ECG results in tabular format in the study report. The results listed in table 9.1 are summarized in the following table.

¹Methods and results discussed with D. Hoberman, PhD.

Med - Page 186 of 223

The sponsor concludes that ziprasidone does not cause QTc prolongation. Although there is evidence to support this conclusion (analyses of studies 108,108E,107,and 301; 303;101) because of the known variability of the QTc and the identified limitations in the methods of data collection, the ability to detect an effect, if present, is expected to be low. The sponsor dismisses the results of the STFDPC trials despite finding an increasing QTc prolongation for each increasing dose group except the highest. Additional analyses demonstrate that for all groups except the <40mg group, mean QTc difference from baseline is significantly longer than placebo. The mean QTc difference from baseline is also significantly longer for the 80mg group compared to the <40mg group. If we are to accept that there is no relationship between ziprasidone and QTc prolongation, then the sponsor must adequately explain the circumstances that led to the apparent dose response relationship findings in the STFDPC trial data.

Like much of the safety data in drug development, the ECG information was collected without intent to evaluate a specific effect. The were no studies designed to determine if ziprasidone causes QTc prolongation. The results from these analyses are only useful for generating hypotheses. The finding of SUDs rates in ziprasidone trials that are higher than observed for other recently approved antipsychotic medications, further stresses the need for additional testing to clarify the effect of ziprasidone on the QTc.

n or

Gerard Boehm, M.D., M.P.H. Safety Reviewer, Neuropharmacological Drug Products, HFD-120

cc:HFD-120\Boehm\Burkhart\Laughren\Leber \Glass

1-23-98 there is sisn i Firmt in white In my of inion, In my of a pose-response to Einthemal evidence of a pose to get to get of evidence of a pose to get of the to would not be set of the first in market of power ed to the first he provinitely power ed to the first to market of power ed the first he pose to get of the to how to the power ed to the first to how to the power ed to the first to how to the power ed to the first to how to the power ed to the first to how to the power ed to the first to how to the power ed to the first to how to the power ed to the first to how to the power ed to the first to how to the power ed to the first to how to the power ed to the first to how to the first of the first to how the first of the first of the first of the first to how the first of the firs

Appendix

Table H.5.23aCha	nge from baseline to last observation all phase II/III
stud	lies
Table H.5.23bCha	nge from baseline to last observation STFDPC phase
11/11	I studies
Table H.5.23mCha	nge from baseline to last, protocol 303
Sponsor's ConsultantEffe	-
-	trally Read 108,108E, 117, &303 Change from baseline
	ll post baseline values
Result of reanalysis of data from	STFDPC trials
Tukey Kramer Pairwise Compa	
Simple Regression, Newmode (categorical modal dose) by last QTc minus baseline

Simple Regression, Dose (estimated dose group) by last QTc minus baseline Simple Regression Newmode (categorical modal dose) by day 14 QTc minus baseline Simple Regression Dose (estimated dose group) by day 14 QTc minus baseline References

APPEARS THIS WAY ON ORIGINAL

Table H.5.23a Change from Baseline to Last Observation in ECG Readings

Page 1 of 1

Varlable	Treatment Group	N	Base Mean	Base Median	Base Range	Finat Nean	Final Median	Final Range	Mean Change
QTc int (msec)	Ziprasidone	1460	409.0	408.7		412.8	412.6		J.8
	Haloperidol	281	410.3	410.5		408.7	406.7		-1.7
	Risperidone	165	409.9	407.6		407.3	407.1		-2.6
	Anisulpride	12	400.4	403.5		401.6	406.5		1.2
	Placebo	251	- 411.3	409.0		408.7	407.8		-2.5
I int (msec)	Ziprasidone	1460	364.9	364.0		366.6	364.0		1.7
	Haloperidol	281	365.0	360.0		364.6	362.0		-0.4
	Risperidone	165	367.3	365.0		363.9	362.0		- 3.4
	Anisulpride	12	353.7	360.0		367.4	360.0		13.8
	Placebo	251	361.5	360.0		360.5	360.0		-1.0
eart Rote (bpm)	Ziprasidone	1460	76.9	75.0		11.1	77.0		0 8
	Haloperidol	281	77.5.	76.0		76.8	76.0		-0.7
	Risperidone	165	76.3	74.0		76 B	75.0		05
	Amisulpride	12	78 7	75 0		72:3	72.5		-6.3
	Placebo	251	79.3	77.0		78.9	78.0		-0.4
R int (msec)	Ziprasidone	1461	i 153.0	152.0		152.5	152.0		-0.5
	Hatoperidol	283	155.4	159 0		155.4	160.0		0.0
	Risperidone	165	152.4	153.0		152.1	150.0		-0.2
	Anisulpride	12	163.4	160.0		155.5	160.0		-7.9
	Placebo	251	150.6	151.0		151.5	152.0		0.8
ORS Int (msec)	Ziprasidone	1464	85.1	84.0		84.9	84.0		-0.2
	Hatoperidol	283	83.9	84.0		84.5	84.0		0.6
	Risperidane	165	84.6	83.0		85.8	85 .0		1.2
	Anisulpride	12	66.7	66.5		65 9	70.0		-0.8
	Placebo	251	86.8	87.0		87.3	86.0		0.4

÷

t

Page 1 of 1

Table H.5.23b Change from Baseline to Lost Observation in ECG Readings Short-Term Fixed-Dose Placebo-Controlled Oral Dosing Phase 11/111 Studies

Variable	Treatment Group	N	Base Nean	Base Median	Base Range	Final Mean	Final Median	Final Range	Mean Change	P-Value*
				400.0		416.8	415.0		6.6	0.001
''QTc int (msec)	Ziprasidone	657	410.2	408.0		409.9	409.0		Ŏ.Ž	0.001
	lla loper i dol	76	409.7	411.6		409.9	407.8		-2.6	
	Placebo	251	411.3	409.0		100.0	407.0		-2.0	
	2 Lorge Ldone	657	364.6	364.0		367.2	364.0		2.7	0.253
QT Int (msec)	Ziprasidone		366.1	363.5		365.7	364.0		-0.5	
	Haloperidol	76				360.2	360.0		-1.3	
:	Ptacebo	251	361.5	360.0		300.2	300.0			
Heart Data (box)	Ziprasidone	657	77.5	15.0		78.9	78.0		1.4	0.322
Heart Rate (bpm)		76	76.8	75 0		77 0	75.5		0.2	
	Hatoperidol		79.3	11.0		79.0	78.0		-0.3	
	Placebo	251	79.3	11.0		73.0	10.0		0.0	
PR Int (msec)	Ziprasidone	655	150.8	150.0		149.8	150.0		-1.0	0.171
PR INC (MSEC)	Haloperidol	76	152.0	152.0		153.3	156.0		1.3	
	Placebo	251	150.6	151.0		151.7	152 0		1.1	
	Praceou	231	130.0	131.0						
QRS Int (msec)	Ziprasidone	657	86.3	86.0		86.1	85.0		-0.1	0.619
des me (msec)	Haloperidol	76	85 1	85.5		85 7	84.0		0.6	
	Placebo	251	86.8	87.0		67.3	86.0		0.4	
	PTACEDO	40 I	00.0	01.0					••••	

¹⁰⁶⁹

paserine - last the taken derore the first day of study treatment. Final - last ECG taken while on study treatment or within six days after the last day of study treatment. Date of table generation: IOFED97.

1

Med - Page 190 of 223 · ... /

÷

Protocols: 104,106,114,115 *Twn-tailed T.test ____

Tuble H.5.23m

Change from Baseline to Last Observation in ECG Readings

Protocol	303
----------	-----

			Baae	Base	Base	Final	Final	Final	Heen
Variable	Treekment Group	N	Hean	Median	Range	Hean	Hedian	Range	Change
QTc int (meec)	20mg BID	62	409.4	407.1		411.4	409.2		2.0
	40mg 810	61	399.9	401.4		403.5	406.7		3.5
	SQuer BITD	59	405.6	403.0		404.0	403.7		-0.8
	Placebo	48	406.1	408.3		401.4	404.8		-4.7
QT int (meec)	20mg 1510	62	364.5	365.5		363.2	355.0		-1.3
	40mg 800	61	358.5	357.0		358.9	360.0		0.4
	Song BID	59	359.1	350.0		ງຄ.ງ	3හ.0		9.2
	Placebo	48	365.4	365.0		350.1	351.0		-7,3
mart Rate (bgm)	20mg 810	62	17.7	74.7		79.1	78.7		1.4
-	40mg 81D	61	76.1	74.3		77.5	75.7		1.4
	90mg 81D	59	78.5	79.7		74.5	72.7		-4.0
	Placebo	48	76.6	73.4		77.0	75.8		0.5
R int (meec)	20mg 1830	62	149.1	145.0		149.8	149.0		-0.3
	40mg 810	60	157.3	157.5		153.5	155.0		-3.0
	SONG BID	59	151.7	153.0		151.5	146.0		-0.2
	Placebo	48	149.5	150.0		145.3	145.0		-3.2
gs int (meec)	20mg BID	62	86.0	85.0		87.7	87.0		1.6
	40mg BID	61	87.4	86.0		87.2	87.0		-0.2
	SCarg BID	59	86.5	86.0		84.9	B4.0		-1.6
	Placebo	40	65.2	86.5		86.5	85.0		1.3

Protocol: 300

Processor: -----"QTC int Buselins = Last BCG taken before the first day of study treatment. Final = Last BCG taken while on study treatment or within six days after the last day of study treatment. Date of table generation: 1200097.

BEST POSSIBLE COPY

PAGE 560

h

t

EFFECT OF ZIPRASIDONE ON THE ECG

Objective: Purpose of the study was to determine the effect, if any, of varying doses of Ziprasidone on the ECG and to compare it with the effect of placebo and Haloperidol.

Material: The group included 658 individuals dosed with Ziprasidone, doses of 5, 20, 40, 60, 80 and 100 mg b.i.d. The N for Haloperidol and placebo was 79 and 245 respectively.

<u>Method</u>: The ECG variables included the heart rate, the P-R interval, QRS duration, intraventricular conduction defects, QT, QTc, ST-T, arrhythmias.

<u>Results</u>: There were no significant changes in heart rate, the P-R and intraventricular conduction after administration of Ziprasidone.

A statistically significant change at a p level of 0.05 or less was noted in the duration of the QT when compared with placebo at doses of 80 and 100 mg. The absolute prolongation was in the order of ________ nsec respectively. No statistically significant change was noted at doses of 5, 20, 40 and 60 mg. The absolute difference between baseline and final mean for the 80 and 100 mg dose was 7.8 and 8.2 msec respectively.

A statistically significant prolongation of QTc, when compared with placebo, at a level of 0.05 or less, was noted at doses of 20, 40, 60, 80 and 100 mg. However, the respective, absolute prolongation was No statistically significant change was noted at 5 mg dose. The absolute difference between the baseline and final mean for the 5 doses was 5.9, 3.6, 5.8, 10.6 and 8.4 msec respectively. In no instance did the QTc exceed 490 msec.

Conversion from normal to abnormal was observed in 37 (5.6%) of the 658 dosed individuals. Similarly, 39 (5.9%) converted from abnormal to normal during therapy. Of the 324 Haloperidol and placebo 15 (4.9%) converted from normal base to abnormal and 25 (7.7%) from abnormal to normal. The most frequent change was from normal ST-T to abnormal ST-T. Isolated prolongation of the P-R, QRS, appearance of LVH, LAFB, and rare PVCs were also noted.

Med - Page 192 of 223

<u>Comment</u>: As indicated there was no physiologically or statistically significant change in heart rate, P-R and QRS intervals. The ST-T changes most frequently noted are common, reflecting the very labile nature of repolarization (ST-T) and, thus, expected over the period of the trial. Furthermore, similar changes from abnormal ST-T to normal were observed. Similar incidence of conversion from normal to abnormal ST-T was recorded in the placebo group. Furthermore, the abnormalities did not appear dose related.

The isolated prolongation of the P-R, QRS, the LVH and the rare PVC are rarely if ever due to drugs.

Although there was some statistically significant prolongation of the QTc after dosing, and when compared with placebo and Haloperidol, the absolute prolongation was small and clinically insignificant.

Summary: There was no clinically significant effect of Ziprasidone on the ECG.

/3/

M.D.

Table H.5.23.2 [Pooled] - Centrally Read 108,108E,117, & 303 Change from Baseline to All Post Baseline Values in ECG Readings By Actual Study Drug Dose Taken the Day of the Reading Centrally Read Maintenance Oral Dosing Phase [1/11] Studies - - -

Zip <20mg BID Zip 20-(40mg BID Zip 40.<60mg BID Zip 60-(80mg B10 21p 80-<100mg 810 . Base Base Base Base . Hean Base Hean** M Hean Hean** Mean Hean** Hean · • • • • • • • • • • • • • • Nean** Hean Nean** _ _ _ _ _ _ _ _ _ _ _ _ _ *OIC int 234 234 234 405.1 364.7 75.7 309 309 309 309 306 309 0.3 172 406.8 2.8 407.0 165 165 165 165 165 3.0 n 410.0 .0.6 01 Int 368.3 74.8 152.3 408.D +2.5 1.5 173 1.7 0.6 3.5 366.3 20 311.1 366.3 76.0 2.1 Heart Rate 173 9.6 1.0 žŏ 74.9 -1.1 PR Int 233 153.9 -1.4 .3.2 -1.2 155.8 -1.9 20 154.4 QRS Int 234 151.7 86.1 -0.1 173 85.7 .0.7 -1.3 85.5 ·0.1 20 83.1 ·0.8 84.6 -. 0.1 (CONTINUED)

Protocols: 108,1086 117 303

*OTc int

1

"Old int ** Mean comparison wasering to subject baseling value using all post baseling readings. Total Changed - number of postbaseling ECG values within six days after the last day of study treatment for which that dose level of study drug was taken the day of the ECG readings. Date of table generation: 210CT97.

Page 1 of 3

.

.

Sheet1

Results of the reanalysis of ECG data from the STFDPC trials

Last minus	baseline					
	n	mean QTc baseline	mean QTc last	QTc difference	QT difference	pulse difference
Placebo	250	399	396.5	-2.6	0.3	-1.1
<40 mg	230	396.9	397.5	0.6	-4.4	2.4
40 mg	138	397.6	403.4	5.9	0.3	2.2
60 mg	111	398	405.7	7.7	7.1	-0.2
80 mg	100	394.6	404.3	9.7	7.2	0.4
>=100 mg	77	402.7	409.1	6.4	5.8	0.2

Last on treatment minus baseline

ĺ	n	mean QTc baseline	mean last QTc	QTc difference	QT difference	pulse difference
Placebo	152	399.8	395.6	-4.2	0.2	-1.4
<40 mg	147	397.1	397.8	0.8	-2.7	1.7
40 mg	78	398.6	403.1	4.6	1.8	1.1
60 mg	74	398.1	405.9	7.9	8.8	-0.9
80 mg	57	394.7	403.4	8.8	6	0.8
>=100 mg	52	403.1	411.1	8	7.9	-0.3

Day 14 minus baseline

	n	mean baseline QTc	mean d14 QTc	QTc difference	QT difference	pulse difference
Placebo	197	399.7	397.1	-2.6	0	-1.1
<40 mg	179	386.4	399.7	3.4	-5.9	4.1
40 mg	116	397.7	403.8	6.1	-1.2	3.3
60 mg	97	398.8	407.2	8.4	6.3	0.2
80 mg	92	395.7	409	13.3	4.1	3.2
>=100 mg	69	402	410.5	8.5	10.1	-0.8

QTc max minus baseline

.

	n	mean baseline QTc	mean max QTc	QTc difference	QT difference	pulse difference
Placebo	250	399	403.4	4.3	-1	2.2
<40 mg	230	396.9	406.4	9.5	-5.2	6.1
40 mg	138	397.6	410.4	12.8	0.6	4.6
60 mg	111	398	413.2	15.2	5.7	3
80 mg	100	394.6	414.4	19.8	6.2	4.6
>=100 mg	77	402.7	417.7	15	9.7	1.8

.

Tukey Kramer Pair wise comparison of the mean QTc differences

	Placebo	<40	40	60	80	100
Placebo		-	+	+	+	 +
<40	-		•	-	-	-
40	+	-		-	-	1 -
60	+	-	-			-
80	+	-	•	-		
100	+	-	•		•	

Using Last minus baseline

•	Placebo	<40	40	60	80	100
Placebo		•	+	+	+	
<40	-		-	-	+	- 1
40	+	-		-	•	- 1
60	+	-	-		-	- 1
80	+	+	-	-		1.1
100	+	-	-	-	-	
	+Signific	ant dif	ference			

Using Day 14 QTc-baseline

-	Placebo	<40	40	60	80	100
Placebo		-	+	+	+	+
<40	•	[•	-	+	-
40	+	-	T	-	- 1	-
60	+	-	-	l	-	1.
80	+	+	-	-	1	1 -
100	+		-	-	-	
	+Signific	ant diff	ference		•	

Using QTcmax minus baseline

	Placebo	<40	40	60	80	100
Placebo		- 1	+	+	+	+
<40	-		-	- 1	+	1.
40	+			-	-	-
60	+	-	-	1	-	•
80	+	+	-	- 1	1	-
100	+	-	-	- 1	1 -	
	+Simific	ant diff	erence			

+Significant difference





			·	Line	ear Fi	t
-	 	_				

Linear Fit

last minus bl = -4.0292 + 2.48476 NEWMODE

Summary of Fit

RSquare	0.038618
RSquare Adj	0.037555
Root Mean Square Error	20.17928
Mean of Response	2.909492
Observations (or Sum Wgts)	906

Analysis of Variance

Source_	DF	Sum of Squares	Mean Square	F Ratio
Model	1	14786.80	14786.8	36.3131
Error	904	368111.78	407.2	Prob>F
C Total	905	382898.58		<.0001

Parameter Estimates

Term	Estimate	Std Error	t Ratio	Prob>[t]
Intercept	-4.0292	1.332402	-3.02	0.0026
NEWMODE	2.4847649	0.412338	6.03	<.0001
			··• +	

last minus bl By dose



---- Linear Fit

Linear Fit

last minus bl = -1.1723 + 0.12024 dose

Summary of Fit

RSquare	0.038604	
RSquare Adj	0.03754	
Root Mean Square Error	20.17943	
Mean of Response	2.909492	
Observations (or Sum Wgts)	906	-

Analysis of Variance

Analysis of Venance							
Source	DF	Sum of Squares	Mean_Square	F Ratio			
Model	1	14781.35	14781.4	36.2991			
Error	904	368117.23	407.2	Prob>F			
C Total	905	382898.58		<.0001			

	Paramete	er Estimates		
Term	Estimate	Std Error	t Ratio	Prob> t
Intercept dose	-1.172281 0.1202434	0.953125 0.019958	-1.23 6.02	0.2190 <.0001

Med - Page 198 of 223

.. .

difference By NEWMODE



		 Linear	Fit
•			

Linear Fit

difference = -3.6359 + 2.83924 NEWMODE

Summary of Fit

RSquare	0.049201
RSquare Adj	0.047929
Root Mean Square Error	20.62923
Mean of Response	4.56
Observations (or Sum Wgts)	750

Analysis of Variance

Source	DF	Sum of Squares	Mean Square	F Ratio
Model	1	16472.11	16472.1	38.7064
Error	748	318322.69	425.6	Prob>F
C Total	749	334794.80		<.0001

Parameter Estimates

Term	Estimate	Std Error	t Ratio	Prob>(t)
Intercept	-3.635928	1.517522	-2.40	0.0168
NEWMODE	2.8392359	0.456362	6.22	<.0001

1

difference By dose



----- Linear Fit

Linear Fit

difference = -0.2855 + 0.13481 dose

Summary of Fit

RSquare	0.047369
RSquare Adj	0.046095
Root Mean Square Error	20.64909
Mean of Response	4.56
Observations (or Sum Wgts)	750

Analysis of Variance

Source	DF	Sum of Squares	Mean Square	F Ratio
Model	1	15858.89	15858.9	37.1938
Error	748	318935.91	426.4	Prob>F
C Total	749	334794.80		<.0001

	Parameter Estimates							
Term	Estimate	Std Error	t Ratio	Prob> t				
Intercept	-0.285523	1.095343	-0.26	0.7 944				
dose	0.13481	0.022105	6.10	<.0001				

References

¹ Karjalainen J, Reunanen A, Ristola P, Viitasalo M. QT Interval as a Cardiac Risk Factor in a Middle Age Population, Heart 1997; 77:543-548.

² Schouten EG, Dekker J, Meppelink P, Kok F, Vanderbroucke J, Pool J. QT Interval Prolongation Predicts Cardiovascular Mortality in an Apparently Healthy Population, Circulation 1991; 84: 1516-1523.

³ Alga A, Tijssen JG, Roelandt JR, Pool J, Lubsen J, QTc Prolongation Measured by Standard 12 Lead Electrocardiography is an Independent Risk Factor for Sudden Death due to Cardiac Arrest, Circulation 1991; 83:1888-1894.

⁴ Pudda PE, Bourassa MG, Preediction of Sudden Death from QTc Interval Prolongation in Patients with Chronic Ischemic Heart Disease, J Electrocardiol 1986; 19: 203-211.

⁵ Schwartz PJ, Wolf S, QT Interval Prolongation as a Predictor of Sudden Death in Patients with Myocardial Infarction, Circulation 1978; 57: 1074-1077.

⁶ Benedict E, The QT Interval and Drug Associated Torsades de Pointes, Drug Invest 1993; 5:69-79.

⁷ Schoenenberger RA, Haefeli WE, Weiss P, Ritz RF, Association of Intravenous Erythrtomycin and Potentially Fatal Ventricular Tachycardia with QT Prolongation, BMJ 1990; 300: 1375-1376.

⁸ Engav MB, Coodley I, Magnusson AR, Torsades de Pointes after Inhaled Pentamidine, Ann Emerg Med 1992; 21: 1413-1415.

⁹ Woolsey RL, Chen Y, Freiman JP, Gillis RA, Mechanism of the Cardiotoxic Actions of Terfenedine, JAMA 1993; 269: 1532-1536.

¹⁰ Morganroth J, Brown A, Critz S, Crumb W, Kunze D, Lacerda A, Lopez H, Variability of the QTc Interval: Impact on Defining Drug Effect and Low Frequency Cardiac Event, AM J Cardiol 1993; 72: 26B-31B.

¹¹ Morganroth J, Brozovich F, McDonald J, Jacobs R, Variability of the QT Measurement in Healthy Men_with Implications for Selection of an Abnormal QT value to Predict Drug Toxicity and Pro Arrhythmia, AM J Cardiol 1991; 67: 774-775.

¹² Molnar J, Zhang F, Weiss J, Ehlert F, Rosenthal J, Diurnal Pattern of QTc Interval: How Long is Prolonged? Possible Relation to Circadian Triggers of Cardiovascular Events, J AM Coll Cardiol 1996; 27: 76-83. ¹³ Pratt C, Ruberg S, Morganroth J, McNutt B, Woodward J, Harriss S, Ruskin J, Moye L, Dose Response Relation Between Terfenedine (Seldane) and QTc Interval on the Scalar Electrocardiogram, Am Heart J 1996; 131: 472-480.

١

APPEARS THIS WAY ON ORIGINAL

APPEARS THIS WAY ON ORIGINAL

MEMORANDUM

Food and Drug Administration Center for Drug Evaluation and Research Division of CardioRenal Drug Products Consultation

Date: 6/14/00

To:

Russell Katz, MD Division Director, HFD-120

From:

Through:

Shaw Chen, MD. PhD_______ Medical Team Leader, HFD-110

Maryann Gordon, MD Medical Reviewer, HFD-110

Dr. Raymond Lipicky______ Division Director, HFD-110

Subject:

Ziprasidone, NDA# 20825 Study report of Clinical Pharmacology Protocol #128-054

Conclusion

Study 128-054 has demonstrated that the antipsychotic agents ziprasidone and thioridazine adversely affect cardiac repolarization in that these drugs prolong the QTc and QT intervals in a concentration-related manner. Patients who take drugs that prolong these ECG intervals are at risk of serious cardiac arrhythmias such as torsade de points (TdP) and sudden death. The effect on cardiac repolarization of the other antipsychotic agents used in study 128-054 for comparison appears to be minimal or absent.

Taking into account ECG data from this study as well as other trials, ziprasidone increases the QTc from baseline on average about 10-20 msec, thioridazine approximately 36 msec, and sertindole, an antipsychotic removed from the UK market for causing TdP and sudden death, about 21 msec. Although the magnitude of the increase of the QTc (and QT) is thought by experts to be important, it is not predictive of the degree of risk of TdP or other serious ventricular arrhythmias.

The co-administration of a metabolic inhibitor with ziprasidone and thioridazine increased blood levels and QTc only slightly compared to the use of these drugs alone. Therefore, drug-drug interactions similar to what occurred with terfenadine (when blood levels increased dramatically when ketoconazole was taken along with terfenadine) are much less of a concern with these agents.

In summary, a certain proportion of patients taking ziprasidone or thioridazine will have an increased risk of potentially fatal ventricular arrhythmias. The Cardio-Renal Division considers it essential that any agent with an added safety risk, unless efficacy data suggest superior benefit compared to other drugs for the same indication, should either not be made available or should be reserved for second line therapy.

Finally, adverse effects such as increases in total cholesterol and large weight gains reported with some of the other antipsychotic agents are unlike sudden death in that they can be identified early and the patient at risk can be switched to another agent. Therefore, the claim that ziprasidone has less cardiovascular risk factors because the drug causes less weight gain and/or improves lipid profile cannot offset its likely

1

WICSAYPITED JUNN22 02000

propensity to cause sudden death.

Introduction

(Please refer to previous consults written by Dr. C. Ganley and dated 12/17/98, 11/18/97, and 2/21/97)

The sponsor of ziprasidone was sent a letter by the Agency on 6-17-98 stating that the drug was not approvable because of its effect on cardiac repolarization. The concern was that the "modest" effect (QTc¹ prolongation of about 10 msec with the 160 mg dose) was an underestimation because the ECGs were not obtained at maximum drug concentration. The study reviewed here was specifically designed to address this issue and also to compare the effect of ziprasidone on cardiac repolarization to the effect of other approved antipsychotic drugs.

Study Design, protocol #128-054

This was a randomized, open-label, parallel, multi-center study in subjects with normal ECGs (QTc <450 msec) and psychotic disorders. Following a screening phase, the trial consisted of five different treatment periods:

Period 1: subjects who were eligible for enrolling in the study had their current medication tapered over several days as an out-patient;

Period 2: subjects entered a clinical research facility to be completely withdrawn from current therapy. Period 3: subjects were randomized to one of six treatments (ziprasidone, risperidone, olanzapine, quetiapine, thioridazine or haloperidol) with the dose escalated over 10 to 19 days;

Period 4: after the maximum dose of randomized therapy was achieved, the metabolic inhibitor selected for each drug was administered;

Period 5: after steady state is reached with the combination of randomized therapy and a metabolic inhibitor, the subjects were withdrawn from therapy.

The study diagram is shown below.



ECGs were obtained at baseline, at start of study drug (day 2), at steady state (period 3), and with the inhibitor (period 4). ECGs were recorded at times estimated to correspond with the mean Tmax for each study drug.

Dosing and metabolic inhibitors

Subjects were to be titrated to the highest dose tolerated. The initial and maximum doses used for each

¹ Bazett's correction: QTc=QT/sqrt (60/hr rate)

treatment group and the doses of the metabolic inhibitors are shown below.

Study Drugs (Period 3)	Potency (mg)	Initial Dose (mg/day)	Maximum Dose (mg/day)	
Ziprasidone	20, 40, and 80 (capsules)	40	160	
Risperidone	1, 2, 3, and 4 (tablets)	2	16	
Olanzapine	5 and 10 (tablets)	5	20	
Quetiapine	25, 100, and 200 (tablets)	50	750	
Thioridazine	25 and 100 (tablets)	50	300	
Haloperidol	2, 5, and 10 (tablets)	2	15	
Metabolic Inhibitors (Period	<u>4)</u>	- ·		
Paroxetine	20 (tablet)	20		
Ketoconazole	200 (tablet)	400		
Fluvoxamine	50 (tablet)	50	100	

There were changes with the administration of the inhibitors during period 4: Originally,

-ketoconazole (200 mg BID) was administered with ziprasidone and quetiapine,

-paroxetine (20 mg 00) was administered with thioridazine and risperidone,

-fluvoxamine (50 mg escalating to 100 mg QD) was administered with olanzapine.

-paroxetine (20 mg QD) and ketoconazole (200 mg BlD) were administered with haloperidol.

Late in the study, ketoconazole (200 mg BID) was substituted for paroxetine as the metabolic inhibitor in the risperidone group, and the regimen for dosing ketoconazole to the haloperidol group was changed from 200 mg BID to 400 mg QD by protocol amendment.

Comments on the protocol raised by Dr. Ganley

1) the study was to enroll a sufficient number of subjects such that 150 subjects (25 per group) completed the entire study. There was no explanation in the protocol to justify the sample size.

2) The protocol was lacking in its description of how the QTc data should be interpreted. There was, however, an expectation that the change in QTc interval with ziprasidone therapy was to be different from haloperidol.

Study results

A total of 183 subjects were randomized and had evaluable ECG data. Patient demographics are shown below.

	Ziprasidone				Olanz	Olanzapine Quetia		_	Thor	Thoridazine		Haloperidol
_	Ma	FD	M	F	м	F	- M	F	м	F	M	E
Number of subjects	25	10	22	6	20	8	22	7	25	6	25	7
Mean age (yrs)	38.6	36.1	38.3	37.3	38.2	38.6	38.1	40.6	35.5	37.3	33.7	43.0
Age range (yrs)	22-58	20-47	20-55	29-47	22-58	25-53	26-47	27-57	21-48	30-44	20-47	35-4
Mean weight (kg)	85.9	79.8	84.1	88.0	86.0	86.2	83.9	87.7	90.5	87.1	77.9	75.6

aM=male; bF=female

Mean age and range, mean weight and number of subjects were reasonably similar for the different treatment groups.

There were 8 subjects (2 ziprasidone, 3 quetiapine, and 3 haloperidol who did not reach the protocolspecified maximum daily dose of study drug in Period 3. Seven of the eight were discontinued prematurely. The eighth received 600 mg of quetiapine at steady-state rather than 750 mg because of adverse events. This subject completed the study. One thioridazine subject required a dose higher than that specified in the protocol.

Heart rate and correction factors

QT interval is inversely related to heart rate (normally, the slower the heart rate the longer the QT interval). To compensate for normal variations in heart rate, the Bazett's correction, known as QTc, is used. The use of Bazett's correction factor is controversial with drugs that increase heart rate. Among the group of drugs studied here, quetiapine was the only one that consistently raised heart rate throughout the study. The mean change from baseline heart rate for the various agents are shown in the attachment.

<u>OT/OTc</u>

Baseline is defined as the average of the planned ECGs collected on days -2, -1, and 0. All ECGs were obtained at Tmax and all were read centrally. QTc intervals were provided by the central reader.

Mean changes

The tables below show the mean change QTc and QT from baseline at the start of titration (day 2) and at steady state (period 3).

Start of titration

	Zisprasidone	Risperidone	Olanzapine	Quetiapine	Thioridazin e	Haloperidol
QTc	3.0 (10.7)	4.7 (14.1)	0.3 (9.0)	-0.5 (8.6)	11.2 (13.2)	2.2 (12.4)
QT	-3.6 (17.3)	-5.3 (11.2)	-3.7 (12.8)	-0.1 (12.4)	1.1 (16.6)	-0.8 (15.6)

Mean change (SD) from baseline at Day 2: msec

Tables 5.2.2.1 and 5.2.3.2.1

At the start of dosing, only thioridazine shows a substantial prolongation of the QTc (11.2 msec). Changes from baseline in QTc/QT are similar for ziprasidone and the rest of the agents.

Steady state

Mean change (SD) from baseline at Period 3: msec

	Zisprasidone	Risperidone	Olanzapine	Quetiapine	Thioridazin e	Haloperidol
QTc	20.6 (16.4)	10.0 (11.1)	6.4 (13.6)	14.5 (12.7)	35.8 (13.5)	4.7 (16.9)
QT	7.0 (18.40)	-11.8 (12.8)	-9.3 (18.0)	-12.2 (15.1)	19.7 (22.3)	12.5 (16.7)

Tables 5.2.2.1 and 5.2.3.2.1

Thioridazine at steady state caused a 35.8 msec increase in QTc (9% increase from baseline) and a 19.7 msec increase in QT (5% increase from baseline). The next largest increase was caused by ziprasidone with a 20.6 msec increase in QTc (5% increase from baseline) and a 7 msec increase in QT (2% increase from baseline). While risperidone, olanzapine, and quetiapine were associated with an increased QTc, the QT was decreased for these agents. Haloperidol increased QTc by 4.7 msec and it is generally accepted, perhaps erroneously, that its effect on QTc is not different from placebo.

With metabolic inhibitor

	Zisprasidone	Risperidone	Olanzapine	Quetiapine	Thioridazin e	Haloperidol
Ratio [^]	1.39	2.47	1.77	4.03	1.04	1.94
QTc	20.4 (17.0)	3.2 (16.9)	5.3 (12.8)	19.7 (13.5)	28.0 (17.3)	8.9 (15.0)
QT	9.9 (21.0)	1.1 (18.6)	-1.8 (17.8)	-15.8 (16.9)	33.3 (23.1)	22.5 (19.9)

Mean change (SD) from baseline at Period 4: msec

^drug concentrations period 4: period 3

Tables 5.2.2.1, 5.2.3.2.1, and page 38 of study report

Quetiapine showed the largest increase in plasma concentration when subjects were also given a metabolic inhibitor. Compared to steady state, the concentration of ziprasidone increased slightly while the mean QT/ QTc prolongation (20.4/9.9 msec) was basically unchanged.

Outliers

The tables below shows the number and percent of subjects with QTc increases from baseline of ≥ 30 , ≥ 60 , and ≥ 75 msec for the various drugs at steady state (period 3) and with the metabolic inhibitor (period 4).

Increase from baseline	Zisprasidone N=33	Risperidone 16 mg N=28	Olanzapine N=26	Quetiapine N=27	Thioridazine N=30	Haloperidol N=20
$QTc: \ge 30$ msec	21 (64)	12 (46)	9 (35)	14 (52)	30 (97)	11 (41)
$QTc: \ge 60$ msec	7 (21)	1(4),	1 (4)	3 (11)	9 (29)	1 (4)
QTc: ≥ 75 msec	1 (3)	0	0	0	3 (10)	0

Number and (percent) of subjects Period 3

Table 5.3.3.2

Number and (percent of subjects) Period 4

Increase from baseline	Zisprasidone N=32	Risperidone N=20	Olanzapine N=24	Quetiapine N=27	Thioridazine N=30	Haloperidol N=20
$QTc: \ge 30$ msec	25 (78)	8 (40)	8 (33)	8 (67)	27 (90)	· 9 (45)
$QTc: \ge 60$ msec	4 (13)	0	0	4 (15)	6 (20)	0
QTc: ≥ 75 msec	1 (3)	0	0	0	4 (13)	0

Table 5.3.4.2

Only thioridazine and ziprasidone increased QTc by 75 msec or more in at least 1 study patient.

Relationship to drug concentration.

The attached figures² show individual QTc and QT values plotted against drug concentration on a log scale for each of the antipsychotic agents. The steepness of the slope indicates the magnitude of increase in QTc and QT for every log increase in concentration.

Thioridazine and ziprasidone showed the steepest-slope for both QTc and QT followed by haloperidol. While the effect of quetiapine on the QTc was impressive (slope of 15), the changes in QT was negative. Olanazepine had a small positive slope and the slope for risperidone was flat.

Lipid profiles

Median changes and median percent changes from baseline at last planned visit prior to discharge in fasting serum cholesterol and triglycerides are shown below by treatment group.

	Baseline							
	Ziprasidone	Risperidone	Olanzapine	Quetiapine	Thioridazine	Halopendol		
Cholesterol								
Total	197.5(-14.5 [°])	204.0(-3.0*)	201.0(4.0')	196.0(5.0')	186.0(21.0 [°] ')	193.0(-22.0 ^c)		
% Change	-7.5 [®]	-1.6	2.1'	2.4	13.7 ^{cJ}	-11.5°		
HDL	43.5(0.0)	41.0(-2.0)	44.0(-2.0)	45.0(-3.0)	41.0(1.5)	43.0(-3.0 [*])		
% Change	0.0	-4.9	-4.6	-8.6	3.0	-6.0 [*]		
LDL	122.0(-11.0)	125.0(9.0)	128.0(1.5)	117.0(-0.5°)	121.0(20.0 ^{c.(})	121.0(-14.0 ^c)		
% Change	-8.5	6.5	1.1	-0.3	18.6 ^{c.1}	-10.5 ^c		
Triglycerides	141.0(-37.0 ^c)	158.0(-17.0)	148.0(43.0 ^{c.)})	124.0(25.0 ^{c.})	120.0(9.07)	118.0(-18.0)		
% Change	-28.0°	-6.7	31.0 ^{c.i}	18.3°'	7.9	-18.0 [•]		
Total/HDL	4.31(-0.33°)	5.43 (0.31)	5.14(0.28*)	4.42(0.48 ^ª ')	4.61(0.41")	4.26(-0.22)		
Ratio	· ·				-			
% Change	-7.5 [•]	5.9 ^{**}	5.4**	10.8 ^{8,}	12.4 ^{c.i}	-7.0 *		

*p<0.05; *p<0.01; *p<0.001 versus baseline using Wilcoxon signed rank test on change from baseline values against 0; *p<0.05, *p<0.01, 'p<0.001 versus ziprasidone by two-sided Wilcoxon test.

The sponsor claims that ziprasidone has a beneficial effect on lipid profiles. In the Division's opinion, if patients need control of their lipids, treating them with a lipid lowering agent would be preferred.

СС

Orig. HFD-110/SChen HFD-120/RGlass/TLaughren

² courtesy of Dr. Gabriel Robbe, Biopharmacology Reviewer



Mean Change from Baseline Heart Rate(bpm) and 95% Confidence Intervals at Each Period by Treatment Group - All Subjects Ziprasidone Protocol 054



Med - Page 209 of 223

Z = Ziprasidone, R = Risperidone, O = Olanzapine, Q = Quetiapine, T = Thioridazine, H = Haloperidol.

* Bar on left is Risperidone 6-8 mg, bar on right is Risperidone 18 mg.

+ Contains only pre (3/16/99) amendment values, post-amendment values are provided in the listings. Source Data: Table 5.2.2.2.1. Date of Data Extraction: 03JUN99. Date of Figure Generation: 07JUL99.





Z = Ziprasidone, R = Risperidone, O = Olanzapine, Q = Quetlapine, T = Thioridazine, H = Haloperidol.

* Bar on left is Risperidone 6-8 mg, bar on right is Risperidone 16 mg.

+ Contains only pre (3/16/99) amendment values, post-amendment values are provided in the listings. Source Data: Table 5.2.1.2.2. Date of Data Extraction: 03JUN99. Date of Figure Generation: 07JUL99.



Mean Change from Baseline QT Interval(msec) and 95% Confidence Intervals at Each Period by Treatment Group – All Subjects Ziprasidone Protocol 054



Z=Ziprasidone, R=Risperidone, O=Olanzapine, Q=Quetlapine, T=Thioridazine, H=Haloperidol.

* Bar on left is Risperidone 6-8 mg, bar on right is Risperidone 18 mg.

+ Contains only pre (3/16/99) amendment values, post-amendment values are provided in the listings. Source Data: Table 5.2.3.2.1. Date of Data Extraction: 03JUN99. Date of Figure Generation: 07JUL99.

1

t

r.



Effect of Thioridazine on QTc Interval

Med - Page 212 of 223



Effect of Thioridazine on QT Interval

Intercept = 272.776 Slope = 44.17474 r²=0.2164634033



Effect of Ziprasidone on QTc Interval

Intercept = 372.711 Slope = 21.57282 r2 = 0.09080

Med - Page 214 of 223



Effect of Ziprasidone on QT Interval

Intercept = 319.724 Slope = 23.342 r²= 0.082985

Med - Page 215 of 223



Effect of Haloperidol on QTc Interval

Med - Page 216 of 223



Effect of Haloperidol on QT Interval

Intercept = 351.81535 Slope = 21.5923 r ²=0.1552740511

Med - Page 217 of 223



Effect of Quetiapine on QTc Interval

Intercept = 365.1656 Slope = 15.0783 <u>r</u> ²=0.1713726855



Intercept = 390.7492 Slope = 5.81969 r ²= 0.0177752817

Med - Page 219 of 223



Effect of Olanzepine on QT Interval

Intercept = 354.47023 Slope = 3.9619 r²= 4.262398787e-3

Med - Page 220 of 223 ·



Effect of Risperidone on QTc Interval

-Intercept = 406.506151 Slope = -1.701107 r²=1.2749627819e-3

Med - Page 221 of 223



Effect of Risperidone on QT Interval

Risperidone Concentration (ng/ml)

Intercept = 367.81725 Slope = -3.157232 r²=2.4245779374e-3

MEMORANDUM

DEPARTMENT OF HEALTH & HUMAN SERVICES Public Health Service Food and Drug Administration Center for Drug Evaluation and Research

Date: May 7, 1997 From: HFD-110 Subject: Comment on the Proposed ECG data displays for the ziprasidone NDA (#20-825) Information Provided: volumes 2.1, 2.2 To: HFD-120

Background

HFD-110 has provided comments on the proposed submission of ECG QT and QTc interval data for NDA 20-825 (IND). The NDA has been submitted. HFD-120 requests comments on the submitted data displays.

Comments

• The sponsor has provided the data tables as proposed in the previous consult regarding the display of ECG data.

• Only post-randomization ECG recordings are provided in enclosure 3. The sponsor should provide baseline and any other post-randomization EKGs for the patients listed in enclosure 3.

• The data provided is sufficient to initiate a review of QT interval data. If a formal consult is requested to interpret QT interval data, additional pre-clinical and clinical information from the NDA will be needed from HFD-120.

Charles J. Ganley, M.D.

Raymond Lipic

cc:

Division File HFD-110/ganley HFD-120/Hardemann/Laughren/

Med - Page 223 of 223